Orthogonal Separation of The Hamilton-Jacobi Equation on Spaces of Constant Curvature

by

Krishan Rajaratnam

A thesis presented to the University of Waterloo in fulfillment of the thesis requirement for the degree of Master of Mathematics of Applied Mathematics

Waterloo, Ontario, Canada, 2014

© Krishan Rajaratnam 2014

I hereby declare that I am the sole author of this thesis. This is a true copy of the thesis, including any required final revisions, as accepted by my examiners.

I understand that my thesis may be made electronically available to the public.

Abstract

What is in common between the Kepler problem, a Hydrogen atom and a rotating blackhole? These systems are described by different physical theories, but much information about them can be obtained by separating an appropriate Hamilton-Jacobi equation. The separation of variables of the Hamilton-Jacobi equation is an old but still powerful tool for obtaining exact solutions.

The goal of this thesis is to present the theory and application of a certain type of conformal Killing tensor (hereafter called concircular tensor) to the separation of variables problem. The application is to spaces of constant curvature, with special attention to spaces with Euclidean and Lorentzian signatures. The theory includes the general applicability of concircular tensors to the separation of variables problem and the application of warped products to studying Killing tensors in general and separable coordinates in particular. Our first main result shows how to use these tensors to construct a special class of separable coordinates (hereafter called Kalnins-Eisenhart-Miller (KEM) coordinates) on a given space. Conversely, the second result generalizes the Kalnins-Miller classification to show that any orthogonal separable coordinates in a space of constant curvature are KEM coordinates. A closely related recursive algorithm is defined which allows one to intrinsically (coordinate independently) search for KEM coordinates which separate a given (natural) Hamilton-Jacobi equation. This algorithm is exhaustive in spaces of constant curvature. Finally, sufficient details are worked out, so that one can apply these procedures in spaces of constant curvature using only (linear) algebraic operations. As an example, we apply the theory to study the separability of the Calogero-Moser system.

Acknowledgements

I want to thank my supervisors Dong Eui Chang and Raymond McLenaghan for their support and advice. I'm especially indebted to Raymond McLenaghan for introducing this exciting and fruitful problem to me, and for carefully reading my work and providing valuable feedback. I also want to thank Spiro Karigiannis for carefully reading this long thesis and providing valuable feedback.

Contents

Li	st of	Figur	es	x
Li	st of	Table	S	xi
A	crony	yms		xii
Li	st of	Notat	ions	xiii
	List	of New	Results	xiv
	List	of Rest	ılts	xv
	List	of Exa	mples	xix
1	Intr	oducti	ion	1
	1.1	Prefac	e	2
	1.2	Histor	ical Outline	2
	1.3	Summ	ary of Main Results	3
	1.4	Notati	ions and Conventions	4
		1.4.1	pseudo-Euclidean spaces	4
		1.4.2	pseudo-Riemannian manifolds	7
2	Ove	erview	of Chapters and Theory	9
	2.1	The In	ntrinsic Characterization of Separation	10
	2.2	Conci	rcular tensors	11
	2.3	Separa	ation of Geodesic Hamiltonians	13
		2.3.1	Benenti tensors	13
		2.3.2	Concircular tensors with Multidimensional Eigenspaces and KEM	
			webs	14
		2.3.3	Necessity of KEM webs in spaces of constant curvature $\ . \ . \ . \ .$	19
	2.4	Separa	ation of Natural Hamiltonians	19
	2.5	Conclu	usion	23

I General Theory

3	Wa	rped Products	25
	3.1	pseudo-Riemannian Submanifolds and Foliations	25
		3.1.1 Brief outline of The Theory of Pseudo-Riemannian Distributions .	25
		3.1.2 Specialization to pseudo-Riemannian submanifolds	28
	3.2	Circles and Spheres [*]	29
	3.3	Product Manifolds	34
	3.4	Nets and their Integrability	35
	3.5	Warped and Twisted Products	37
4	Kill	ing tensors	43
	4.1	Hamiltonian mechanics on the Cotangent bundle	43
	4.2	Definition in terms of Poisson and Schouten brackets and Covariant Derivative	45
	4.3	Conformal Killing tensors and special classes	49
	4.4	Orthogonal conformal Killing tensors	49
	4.5	Killing tensors in Warped Products	58
5	Har	nilton-Jacobi separation via Characteristic Killing tensors	62
	5.1	Hamilton-Jacobi Theory	62
	5.2	Separation of the Hamilton-Jacobi equation	67
	5.3	Intrinsic characterization of Separation for geodesic Hamiltonians	70
		5.3.1 Killing-Stackel spaces in Warped Products	75
	5.4	Intrinsic characterization of Separation for natural Hamiltonians \ldots .	78
		5.4.1 Separation of natural Hamiltonians in Warped Products	81
	5.5	Notes	83
6	Cor	ncircular tensors and KEM webs	85
	6.1	General Valence	85
	6.2	Torsionless Conformal Killing tensors	87
	6.3	Concircular 2-tensors	91
		6.3.1 Characterizations of OCTs	94
		6.3.2 Relation to Geodesically Equivalent Metrics	97
		6.3.3 Existence in arbitrary manifolds	99
	6.4	$Concircular vectors^* \dots \dots \dots \dots \dots \dots \dots \dots \dots $	100
	6.5	KEM webs	102
	6.6	The Killing-Stackel space of KEM webs	106
	6.7	Separation in KEM webs: The BEKM Separation Algorithm	113
	6.8	Notes	116

24

Π	\mathbf{S}	pecial	lization to Spaces of Constant Curvature	118
7	In S	Spaces	of Constant Curvature: Separable webs are KEM webs	119
	7.1	Summ	nary of Results	. 119
	7.2	Prelin	ninary results	. 123
	7.3	Classi	fication of orthogonal separable coordinates with diagonal curvature	. 125
	7.4	Spaces	s of Constant Curvature	. 135
	7.5	Conclu	usion	. 136
	7.6	Notes		. 136
8	\mathbf{Pre}	limina	ries from Linear Algebra and Geometry	138
	8.1	pseude	o-Euclidean space	. 138
	8.2	Self-ac	djoint operators in pseudo-Euclidean space	. 140
		8.2.1	Minkowski Space	. 143
	8.3	Spaces	s of Constant Curvature in pseudo-Euclidean space	. 145
	8.4	Warpe	ed products in Spaces of Constant Curvature	. 146
9	Cor	ncircula	ar tensors in Spaces of Constant Curvature	151
	9.1	Prelin	ninaries and Summary	. 152
		9.1.1	Concircular tensors	. 152
		9.1.2	Properties of OCTs	. 154
		9.1.3	Summary of Results	. 155
	9.2	Canon	nical forms for Concircular tensors in pseudo-Euclidean space	. 159
		9.2.1	Standard Model of pseudo-Euclidean space	. 159
		9.2.2	Parabolic Model of pseudo-Euclidean space	. 160
		9.2.3	Existence of Canonical forms	. 163
		9.2.4	Uniqueness of Canonical Forms	. 170
	9.3	Canon	ical forms for Concircular tensors in Spherical submanifolds of pseudo-	
		Euclid	lean space	. 172
		9.3.1	Obtaining concircular tensors in umbilical submanifolds by restriction	on 172
		9.3.2	Concircular tensors in Spherical submanifolds of pseudo-Euclidean	
			space	. 173
	9.4	Prope	rties of Concircular tensors in Spaces of Constant Curvature \ldots .	. 175
		9.4.1	Central Concircular tensors	. 178
		9.4.2	Axial Concircular tensors	. 187
		9.4.3	Concircular tensors in Spherical Submanifolds of pseudo-Euclidean	
			space	. 192
	9.5	Classi	fication of reducible concircular tensors	. 196
		9.5.1	In pseudo-Euclidean space	. 196

		9.5.2 In Spherical submanifolds of pseudo-Euclidean space \ldots .	. 206
	9.6	Applications and Examples	. 209
		9.6.1 Enumerating inequivalent separable coordinates	. 209
		9.6.2 Constructing separable coordinates	. 211
	9.7	Notes	. 214
10	Sepa	aration of Natural Hamiltonians	216
	10.1	Known Separable Potentials	. 217
	10.2	Example: Calogero-Moser system	. 221
	10.3	The BEKM separation algorithm	. 228
		10.3.1 Spherical KBD Equation	. 228
		10.3.2 In pseudo-Euclidean space $\ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots$. 230
		10.3.3 In Spherical submanifolds of pseudo-Euclidean space \ldots	. 231
11	Con	clusion	233
A	PPEI	NDICES	234
A	Cofa	actor Operators via Exterior Algebra	235
в	Nije	enhuis tensors, Haantjes tensors and Integrability of Eigenspaces	239
	B.1	Properties of Torsionless Tensors	. 243
\mathbf{C}	Self	-adjoint operators in pseudo-Euclidean space	245
	C.1	Preliminaries on Scalar product spaces	. 246
	C.2	Preliminaries from Operator Theory	. 246
	C.3	Existence of the metric-Jordan canonical form	. 248
	C.4	Uniqueness of the metric-Jordan canonical form	. 254
D	War	rped products in Spaces of Constant Curvature	259
	D.1	Spherical Submanifolds of Spaces of Constant Curvature	. 260
	D.2	Standard spherical submanifolds of pseudo-Euclidean space	. 260
	D.3	Warped product decompositions of Spaces of Constant Curvature	. 261
	D.4	Warped product decompositions of pseudo-Euclidean space	. 263
		D.4.1 Spherical submanifolds of pseudo-Euclidean space	. 263
		D.4.2 Warped product decompositions of pseudo-Euclidean space	. 265
	D.5	Isometry groups of Spherical submanifolds of pseudo-Euclidean space *	. 274
	D.6	Warped Product decompositions of Spherical submanifolds of Pseudo-	
		Euclidean space	. 277
		D.6.1 Spherical submanifolds of $\mathbb{E}^n_{\nu}(\kappa)$. 277

D.6.2	Warped Product decompositions of Spherical submanifolds of Pseudo-	
	Euclidean space	. 280
E Lexicogra	phic ordering of complex numbers	28 4
References		285
Index		29 4

List of Figures

2.1	Concircular tensor with simple eigenspaces $E_1, \ldots, E_n \ldots \ldots \ldots$	14
2.2	Concircular tensor with eigenspaces E_1, \ldots, E_m, D	15
2.3	KEM web I	16
2.4	KEM web II	18
2.5	KEM Web III	18
2.6	Concircular tensor with eigenspaces E_1, \ldots, E_m, D	20
2.7	Possible KEM web that can be constructed	20

List of Tables

6.1	$\dim \mathbf{C}^2(M) \text{ if } n = 3 \dots \dots$	99
6.2	$\dim \mathbf{C}^2(M) \text{ if } n = 4 \dots \dots$	99
9.1	Separable Coordinate Systems in \mathbb{E}^2	212

Acronyms

KV Killing vector 46
KT Killing tensor 46, 57, 80
ChKT characteristic Killing tensor 70, 87
CKT conformal Killing tensor 48, 85
CV concircular vector 98, 225
CT concircular tensor also called a C-tensor 83
OCT orthogonal concircular tensor 90, 149
ICT irreducible concircular tensor 91, 151
KS-space Killing-Stackel space 10, 72, 85
KEM Kalnins-Eisenhart-Miller 101
KBD Killing Bertrand-Darboux 112, 213
KBDT Killing Bertrand-Darboux tensor 92
WP-net warped product net 40
TP-net twisted product net 40
HJ Hamilton-Jacobi 61

List of Notations

- \mathbb{E}_{ν}^{n} pseudo-Euclidean space, an *n*-dimensional vector space equipped with a metric with signature ν xii, 136, 260
- $\mathbb{E}_{\nu}^{n}(\kappa)$ A hyperquadric of pseudo-Euclidean space. More precisely the central hyperquadric of \mathbb{E}_{ν}^{n} with curvature κ . 144
- \oplus The orthogonal direct sum. 4, 245
- sgn Given a real number a, sgn a is the sign of a if $a \neq 0$ and 0 if a = 0. 144
- $\mathcal{F}(M)$ The set of functions defined on the manifold M. 6
- $\mathfrak{X}(M)$ The set of vector fields defined on the manifold M. 6
- $\Gamma(E)$ The set of vector fields tangent to the distribution E. 6, 25
- $S^p(M)$ The set of symmetric contravariant tensors of valence p defined on the manifold M. 6
- $C^p(M)$ The vector space of concircular contravariant tensors of valence p defined on the manifold M. 83
- $C_0^p(M)$ The vector space of covariantly constant contravariant tensors of valence p defined on the manifold M. 83

 \odot The symmetric product of two tensors, i.e. if u, v are tensors then $u \odot v$ is the symmetrization of $u \otimes v$. 5, 150

List of New Results

Corollary 4.4.6		53
Corollary 4.4.8	Conformal Killing tensors induce twisted product nets	. 54
Corollary 4.4.11		. 55
Proposition 4.5.2	Killing tensors in Warped Products	. 58
Corollary 4.5.4		61
Proposition 5.3.10	Extending a Killing-Stäckel space into a warped product	. 77
Proposition 5.3.12	The Killing-Stäckel space of a reducible separable web	. 77
Proposition 5.4.3	Reduction of The dKdV equation on warped products	. 81
Theorem 5.4.4	Separation in Warped Products	. 83
Corollary 6.3.2	OCTs induce Warped Products	. 93
Proposition 6.3.7	Characterization of Reducible OCTs	. 96
Proposition 6.6.3	Extension of the L-sequence	. 109
Proposition 6.6.6	Some Geometrical properties of torsionless CKTs	. 112
Corollary 6.6.8	The KS-space of an ICT	. 112
Proposition 9.1.2		. 154
Proposition 9.3.2	Concircular tensors in $\mathbb{E}^n_{\nu}(\kappa)$	173
Proposition 9.4.3		177
Proposition 9.4.7		180
Proposition 9.4.9		182
Proposition 9.4.13		187
Proposition 9.4.14	Determinant of Axial Concircular tensors	. 188
Proposition 9.4.19	Determinant of Spherical CTs	. 192
Proposition 9.5.3	Constructing Reducible CTs in \mathbb{E}^n_{ν}	197
Proposition 9.5.9		. 205
Theorem 9.5.11	Classification of Reducible CTs in \mathbb{E}^n_{ν}	. 205
Proposition 9.5.12	Restricting Reducible CTs to $\mathbb{E}^n_{\nu}(\kappa)$	206
Theorem 9.5.15	Classification of Reducible CTs in $\mathbb{E}^n_{\nu}(\kappa)$	208
Proposition 10.1.1	Constructing separable potentials I	. 217
Corollary 10.1.3		. 218
Proposition 10.2.1		222
Proposition 10.3.1		228
Proposition 10.3.3	Spherical KBD equation	. 230

List of Results

Theorem 2.1.1	Orthogonal Separation of Geodesic Hamiltonians [Eis34]	10
Theorem 2.1.2	Orthogonal Separation of Natural Hamiltonians [Ben97]	11
Theorem 2.3.4	Separable Webs in Spaces of Constant Curvature	19
Proposition 3.1.1		26
Proposition 3.1.2	Geometric Characterization of The Second Fundamental Form	27
Theorem 3.2.3	Circles and Spheres [ANY90]	32
Theorem 3.2.4	Circles and Spheres II [ANY90]	33
Theorem 3.2.6	Spheres in spaces of constant curvature [LN71]	34
Theorem 3.4.3	Characterizations of integrable nets [RS99]	36
Corollary 3.4.4		36
Proposition 3.5.6		38
Corollary 3.5.7		39
Corollary 3.5.8	Sectional curvature of warped products	40
Proposition 3.5.9	Properties of the Twisted Product [MRS99]	40
Theorem 3.5.10	Geometric Characterization of Twisted and Warped Products	
[MRS99]		41
Theorem 3.5.14	Twisted and warped product nets [MRS99]	42
Proposition 3.5.16	Lifting isometries from Killing distributions	42
Theorem 4.2.1	Schouten bracket [Sch53]	45
Theorem 4.2.2	Schouten and Poisson brackets	46
Proposition 4.2.3	Properties of the Schouten bracket	46
Proposition 4.2.4	Schouten bracket and Connections	47
Corollary 4.2.5		48
Theorem 4.2.6	Killing tensors and constants of motion	48
Theorem 4.3.1	Characterizations of conformal Killing tensors	49
Proposition 4.4.2	[CFS06]	51
Theorem 4.4.3	Geometric Characterization of Orthogonal CKTs $[{\rm CFS06}]$	51
Proposition 4.4.12		56
Proposition 4.4.13	[KMJ80]	57
Proposition 4.4.15	Restriction of CKTs to Invariant Submanifolds	57
Corollary 4.5.3	WP-nets always admit KTs	60
Theorem 5.1.1	Jacobi Theorem	63
Theorem 5.1.4	Complete solutions via First integrals	65
Theorem 5.2.2	Levi-Civita equations [LC04]	68
Proposition 5.2.3	$[Ben 89] \ldots \ldots$	69
Theorem 5.2.4	Separable solutions via First integrals [Ben89]	70
Theorem 5.3.2	Orthogonal Separation of Geodesic Hamiltonians $[{\rm Eis}34]$	71

Corollary 5.3.3		72
Theorem 5.3.5	Orthogonal Separation of Geodesic Hamiltonians II [KMJ80] .	74
Corollary 5.3.6		74
Theorem 5.4.1	Benenti's Theorem [Ben97]	79
Corollary 5.4.2		80
Proposition 6.1.1	Properties of Concircular tensors [Cra08]	86
Theorem 6.1.2	The Vector Space of Concircular tensors [Cra08]	86
Corollary 6.1.4	Concircular tensors in spaces of constant curvature	86
Proposition 6.2.1	Characterization of torsionless orthogonal CKTs	88
Proposition 6.2.2	Properties of torsionless orthogonal CKTs [Ben05]	88
Corollary 6.2.4	Benenti tensors induce Separable Webs	89
Proposition 6.2.5	Equivalent Torsionless Orthogonal CKTs	90
Proposition 6.3.1	Properties of Concircular 2-tensors [Cra03]	92
Proposition 6.3.5	$[Ben05] \ldots \ldots$	94
Proposition 6.3.6	Characterization of orthogonal CTs	96
Theorem 6.3.9		98
Theorem 6.3.10	Lacunae in the dimension of the space of CTs [Sha00] \ldots .	99
Theorem 6.3.11	Lacunae in the dimension of the space of CVs $[Cra07]$	99
Theorem 6.3.12	[Cra03]	99
Proposition 6.4.3	Concircular vectors in \mathbb{E}^n_{ν} [Cra07]	100
Proposition 6.4.4	Concircular 2-tensors in \mathbb{E}^n_{ν}	101
Proposition 6.4.5	Properties of Concircular vectors	101
Theorem 6.5.4	KEM webs	103
Proposition 6.5.5		104
Theorem 6.6.1	Killing tensors from Concircular tensors [Cra03]	107
Theorem 6.6.2	[Ben04]	108
Theorem 7.1.1	Separable Webs in Spaces of Constant Curvature	119
Theorem 7.1.2	KEM Separation Theorem	121
Theorem 7.1.3	KEM Separation Theorem II	122
Proposition 7.3.2		126
Proposition 7.3.3		128
Proposition 7.3.4		128
Proposition 7.3.5		129
Proposition 7.3.6		134
Proposition 7.3.7		134
Proposition 8.2.1		141
Theorem 8.2.2	Complex metric-Jordan canonical form [O'N83]	141
Theorem 8.2.5	Real metric-Jordan canonical form [O'N83]	142

Theorem 8.2.7	Properties of self-adjoint operators in Minkowski space 145
Theorem 8.4.1	Spherical submanifolds of $\mathbb{E}^n_{\nu}(\kappa)$
Theorem 8.4.5	Standard Warped Products in \mathbb{E}^n_{ν} [Nol96]
Theorem 8.4.7	Restricting Warped products to $\mathbb{E}_{\nu}^{n}(\kappa)$
Corollary 9.1.4	Geometric Equivalence of OCTs
Theorem 9.1.5	The Vector Space of Concircular tensors [TCS05]
Corollary 9.1.6	Concircular tensors in spaces of constant curvature
Theorem 9.1.8	Canonical forms for CTs in \mathbb{E}^n_{ν}
Proposition 9.2.1	Concircular vectors in \mathbb{E}^n_{ν} [Cra07]
Proposition 9.2.2	Concircular tensors in \mathbb{E}^n_{ν}
Proposition 9.2.3	Isometry group of \mathbb{P}^n_{ν}
Proposition 9.2.6	
Proposition 9.2.10	
Theorem 9.2.13	Isometric Equivalence of CTs in \mathbb{E}^n_{ν}
Proposition 9.3.1	Restriction of CTs to umbilical submanifolds [Cra03] 172
Theorem 9.3.5	Isometric Equivalence of CTs in $\mathbb{E}^n_{\nu}(\kappa)$
Proposition 9.4.1	
Proposition 9.4.2	
Theorem 9.4.8	[KMR84]
Proposition 9.4.15	ICT metrics in \mathbb{E}^n_{ν}
Corollary 9.4.18	
Proposition 9.4.21	ICT metrics in $\mathbb{E}^n_{\nu}(\kappa)$
Corollary 9.5.5	
Corollary 10.1.2	
Proposition 10.1.5	Constructing separable potentials II [Bla03; Bla05]
Corollary A.0.6	
Proposition A.0.7	
Corollary A.0.8	
Proposition A.0.9	Leverrier sequence
Proposition A.0.10	Derivative of the Characteristic Polynomial
Proposition A.0.12	Projectors onto Eigenspaces
Proposition B.0.14	Equivalent Definitions of The Nijenhuis tensor
Theorem B.0.19	Haantjes [Haa55; FN56]
Theorem B.0.20	Nijenhuis [Nij51]
Corollary B.0.21	Integrability of almost product structures
Proposition B.1.1	Jacobi's formula for Torsionless Tensors
Proposition B.1.2	
Theorem C.2.2	

Theorem C.2.4	Sylvester's Law of Inertia
Theorem C.3.1	Fundamental Properties of Self-Adjoint Operators
Theorem C.3.7	Existence of the metric-Jordan canonical form $[{\rm O'N83}]$ 253
Theorem C.4.2	Uniqueness of the metric-Jordan canonical form
Theorem C.4.5	Isometric Equivalence of self-adjoint operators
Proposition D.1.1	
Proposition D.1.2	Spherical Submanifolds in Spaces of Constant Curvature 260
Proposition D.2.1	
Proposition D.2.2	
Theorem D.4.1	Spherical submanifolds of \mathbb{E}^n_{ν}
Theorem D.4.5	Warped products in \mathbb{E}^n and M^n
Theorem D.4.6	Standard Warped Products in \mathbb{E}^n_{ν} [Nol96]
Corollary D.4.10	Canonical form for Warped products of \mathbb{E}^n_{ν}
Proposition D.5.1	Lifting isometries from Killing distributions
Proposition D.5.2	
Theorem D.6.2	Spherical submanifolds of $\mathbb{E}^n_{\nu}(\kappa)$
Theorem D.6.5	Restricting Warped products to $\mathbb{E}_{\nu}^{n}(\kappa)$
Corollary D.6.7	Warped product decompositions of $\mathbb{E}^n_{\nu}(\kappa)$
Theorem D.6.8	Warped products in Spherical submanifolds of \mathbb{E}^n and M^n 283

List of Examples

Elliptic coordinates in \mathbb{E}^2 13
Cylindrical coordinates in \mathbb{E}^3
Spherical coordinates in \mathbb{E}^3
Calogero-Moser system
Geodesics in Spherical Submanifolds [Kas10]
Non-integrable nets
Product Nets
Prototypical warped product
Separability of The Schwarzschild metric
Complex Separation [DR07]
The Dilatational vector field in \mathbb{E}^n_{ν}
KEM webs
Calogero-Moser system
Separation in Static space-times
Constructing multiply warped products
Separable coordinates in hyperbolic space
Generalization of elliptic coordinates to \mathbb{E}^n_{ν}
Circular coordinates
Spherical Coordinates I
Spherical Coordinates II
Central CTs
CTs in $\mathbb{E}^n_{\nu}(\kappa)$
Spherical coordinates in \mathbb{S}^2
Oblate/Prolate spheroidal coordinates in \mathbb{E}^3
Product coordinates in \mathbb{E}^4
Constructing separable potentials I

Example 10.1.6	Constructing separable potentials II $\ldots \ldots \ldots \ldots \ldots$	220
Example C.4.4		257
Example D.4.4	Proper Circles in pseudo-Euclidean space	264

Chapter 1 Introduction

The method of separation of variables for the Hamilton-Jacobi equation is considered to be a powerful tool for obtaining exact solutions. Classically, this method was one of the only known methods for obtaining exact solutions. Until recently (last few decades), it was not known how to fully exploit this method to its maximum potential.

We say coordinates are *separable* if they are orthogonal and they separate the geodesic Hamilton-Jacobi equation. Separable coordinates can be used to integrate the geodesic equations by quadratures. Additionally, these coordinates are important to mathematical physics for other reasons. Assuming the Ricci tensor is diagonalized in them, one can show that they separate the Laplace equation and the Klein-Gordon equation from relativity.

The *fundamental problems* concerning the separation of the Hamilton-Jacobi equation are the following:

- 1. Give a (pseudo-)Riemannian manifold, what are the "inequivalent" coordinate systems that separate the geodesic Hamiltonian?
- 2. How do we determine, intrinsically (coordinate-independently), the "inequivalent" coordinate systems in which a given natural Hamiltonian separates?
- 3. If we have determined that the natural Hamiltonian is separable in coordinates (u^1, \ldots, u^n) , what is the transformation to these coordinates from the original coordinates (q^1, \ldots, q^n) in which the natural Hamiltonian is defined?

Concircular tensors can be used to obtain an elegant solution to these problems in spaces of constant curvature, as we shall show throughout this thesis. In Chapter 2 we will give an overview of this solution.

1.1 Preface

We give a textbook like account of the modern geometric theory of orthogonal separation of variables for the Hamilton-Jacobi equation, with special attention to spaces of constant curvature. We first present the general theory in the first part and then specialize it to spaces of constant curvature in the second. The second part, for the most part, can be read separately from the first. The chapters tend to provide greater details and generalizations on certain aspects of the theory. Hence it is necessary to the them together, this is done in Chapter 2. In this chapter, we also give an overview of the main results.

The thesis is organized so that it can be a useful reference on several subtopics and related topics. Although the thesis is not completely self-contained, most proofs are given, especially when they are difficult to find elsewhere, or are important for understanding the theory. Sections marked with an asterisk (*) are mostly optional. They can be skipped with little to no loss of continuity until their results are referenced elsewhere, which is usually a rare occurrence.

We assume the reader has some familiarity with differential geometry, including the notion of distributions (plane fields) and the related Frobenius theorem (e.g. see [Lee12]). It is assumed the reader has a sufficient knowledge of (pseudo-)Riemannian geometry to have a basic understanding of general relativity (e.g. see [O'N83]).

By using the notions and tools of differential geometry to solve this problem, we provide a fairly general setting in which our results are applicable. We are able to generalize previous results given for Riemannian spaces of constant curvature in [Kal86] to arbitrary signature. Furthermore we are able to present solutions to problems, in a single framework, which are usually solved separately.

Much of the content of this thesis comes from other articles. The content from [RM14b] is split up into Sections 4.4, 4.5, 5.3.1, 5.4.1, 6.5 and 6.7. The content from [RM14c] makes up Chapter 9 and Sections 10.2 and 10.3. The content from [RM14a] makes up Chapter 7 and also appears in Section 6.5. The content from [Raj14a] is summarized in Section 8.2, and it makes up Appendix C. The content from [Raj14b] is summarized in Section 8.4, and it makes up Appendix D and Sections 3.1 and 3.2.

1.2 Historical Outline

The theory presented in this thesis is a synthesis of decades of research in this area. The seminal result from which we start is the intrinsic characterization of separation for geodesic Hamiltonians given by Eisenhart in [Eis34]. This result can be deduced from the Levi-Civita equations originally given in [LC04].

This problem has a long history, for the work preceding Eisenhart see [Kal86] and

references therein. Recent interest in the subject was due to the discovery of the separability of the Hamilton-Jacobi equation for the geodesics in the Kerr solution from general relativity [Car68]. Some of this research culminated in the Kalnins-Miller classification of separable coordinates in Riemannian spaces of constant curvature. This classification, which was based on Eisenhart's results, was originally presented in the articles [KM86; KM82], and then combined in the book [Kal86].

The next result was due to Benenti in [Ben92a], where he obtained an intrinsic method to calculate Killing tensors associated with certain separable coordinates in Euclidean space such as elliptic and parabolic coordinates. This was done with the help of a certain torsionless conformal Killing tensor.

Around the same time, Benenti had come up with the intrinsic characterization of separation for natural Hamiltonians presented in [Ben93]. These results were eventually generalized to general (possibly non-orthogonal) separation in [Ben97].

The results given by Benenti in [Ben92a] were further refined by Crampin in [Cra03], and concircular tensors were first formally introduced into separation of variables. By then it was known that certain separable coordinates in Euclidean space, such as elliptic coordinates, could be intrinsically characterized using concircular tensors.

A recursive algorithm for separating natural Hamiltonians in Euclidean and spherical space was given by Waksjo and Wojciechowski in [WW03]. It turns out that this algorithm, which is based on the Kalnins-Miller classification, can be intrinsically defined using concircular tensors.

Further research on concircular tensors was done by Benenti in [Ben05] and Crampin et al. in [TCS05; Cra07].

Motivated by the desire to obtain an intrinsic algorithm to separate natural Hamiltonians in spaces of constant curvature, the author developed a more general theory on the application of concircular tensors to the separation of variables problem in [RM14b]. This theory was then applied to spaces of constant curvature in [RM14c].

1.3 Summary of Main Results

The main purpose of this thesis is to present the theory of concircular tensors (CTs) and their application to the separation of variables problem in general, and in spaces of constant curvature in particular.

The first result given in Section 6.5 shows how to use CTs to construct a special class of separable coordinates, hereafter called Kalnins-Eisenhart-Miller (KEM) coordinates, in a given pseudo-Riemannian manifold. The second result is a converse to the first result, which shows that all orthogonal separable coordinates in spaces of constant curvature are KEM coordinates (see Chapter 7). This result generalizes the corresponding result by Kalnins and Miller in [Kal86] and serves as an independent verification of it.

A recursive algorithm is given (see Section 6.7) which allows one to intrinsically (coordinate independently) search for KEM coordinates which separate a given (natural) Hamilton-Jacobi equation. This algorithm is exhaustive in spaces of constant curvature.

Results given originally by Benenti in [Ben92a] are generalized to give a recursive procedure to construct the KS-space associated with KEM coordinates in Section 6.6.

Finally, sufficient details are worked out in Chapter 9, so that one can apply these procedures in spaces of constant curvature using only (linear) algebraic operations. Further details are worked out in Chapter 10 and then applied to study the separability properties of the Calogero-Moser system.

Additionally, various generalizations are given. The most significant is the study of Killing tensors and separation in warped products. See Sections 4.5 and 5.3.1 for the former and Section 5.4.1 for the latter.

1.4 Notations and Conventions

Our notations and conventions build on those in [O'N83] and [Lee12].

1.4.1 pseudo-Euclidean spaces

Suppose V is a vector space over a field \mathbb{F} (which for us is \mathbb{R} or \mathbb{C}). A symmetric bilinear form on V is bilinear function $\langle \cdot, \cdot \rangle : V \times V \to \mathbb{F}$ such that $\langle x, y \rangle = \langle y, x \rangle$ for all $x, y \in V$. A symmetric bilinear form $\langle \cdot, \cdot \rangle$ is called *non-degenerate* if for a fixed $x \in V$, $\langle x, y \rangle = 0$ for all $y \in V$ implies x = 0. If $x \in V$ then we denote $x^2 := \langle x, x \rangle$ and $||x|| := \sqrt{|x^2|}$. A vector $x \in V$ is called a *unit vector* if ||x|| = 1.

Given a non-zero vector $x \in V$, it is classified as follows:

timelike If $\langle x, x \rangle < 0$

lightlike (null) If $\langle x, x \rangle = 0$

spacelike If $\langle x, x \rangle > 0$

We define a scalar product (metric) on a vector space V as a non-degenerate symmetric bilinear form on V. A real vector space V equipped with a scalar product is called a scalar product space (pseudo-Euclidean space). The index of a real scalar product space V, denoted ind V, is defined as the number of timelike basis vectors in an orthogonal basis for the scalar product, which is an invariant of the scalar product by Sylvester's law of inertia. For all notions related to the index, we will assume the scalar product space is real. The invariant ind V is also called the signature of the metric $\langle \cdot, \cdot \rangle$. A pseudo-Euclidean space of dimension n and signature ν is often denoted \mathbb{E}^n_{ν} .

The *Euclidean metric* given as follows is an example of a non-degenerate scalar product:

$$\langle x, y \rangle = \sum_{i=1}^{n} x_i y_i$$

An *n*-dimensional real vector space equipped with the Euclidean metric is called *Euclidean space* and denoted \mathbb{E}^n . The standard example of a non-degenerate scalar product with non-zero signature is the *Minkowski metric* given as follows:

$$\langle x, y \rangle = \sum_{i=2}^{n} x_i y_i - x_1 y_1$$

A *n*-dimensional real vector space equipped with the Minkowski metric is called Minkowski space and sometimes denoted M^n .

Given a subspace $H \subseteq V$, we denote the *orthogonal subspace* of H as H^{\perp} which is defined as follows:

$$H^{\perp} = \{ x \in V : \langle x, y \rangle = 0 \quad \text{for all } y \in H \}$$

 H^{\perp} is complementary to H (i.e. $V = H \oplus H^{\perp}$) iff the restriction of the scalar product to H is non-degenerate [O'N83, P. 49]. One can also show that for a non-degenerate subspace H, ind $V = \operatorname{ind} H + \operatorname{ind} H^{\perp}$. If U, W are subspaces of V, then $V = U \oplus W$ means that $V = U \oplus W$ and $U \perp W$.

Tensors We now discuss notation related to tensors on a vector space V. Let T be a type $\binom{a}{b}$ tensor on V. If b = 0 (resp. a = 0) we say T is a *contravariant* (resp. *covariant*) tensor of valence a (resp. b). Now suppose V is a scalar product space. Without further specification, *tensor* is short for valence 2-tensor and the type depends on the context.

Let T be an endomorphism (i.e. $\binom{1}{1}$ tensor) of V. A subspace D is called T-invariant if $TD \subseteq D$. T is said to have a simple eigenvalue λ , if λ is real and has algebraic multiplicity equal to 1. T is said to have simple eigenvalues if all its eigenvalues are simple. T is called self-adjoint if

$$\langle Tx, y \rangle = \langle x, Ty \rangle$$
 for all $x, y \in V$

The above condition is equivalent to requiring T to be metrically equivalent to a symmetric contravariant (covariant) tensor. When V is non-Euclidean a self-adjoint endomorphism is not necessarily diagonalizable. Hence, by an *orthogonal tensor*, we mean a symmetric contravariant tensor whose associated endomorphism is diagonalizable

with real eigenvalues. One can check that the eigenspaces of such an endomorphism are necessarily pair-wise orthogonal non-degenerate subspaces. Finally given a subspace $W \leq V$, the restriction of T to W is denoted $T|_W$.

Index Notation We will occasionally use index notation for calculations. Suppose V is a vector space and let T be a type $\binom{a}{b}$ tensor on V. Let v_1, \ldots, v_n be a basis for V with dual basis f^1, \ldots, f^n for the dual space V^* , which satisfy:

$$f^i(v_j) = \delta^i_{\ j}$$

where δ^{i}_{j} is the Kronecker delta. Index notation is defined by:

$$T^{i_1\cdots i_a}_{j_1\cdots j_b} = T(f^{i_1},\cdots,f^{i_a},v_{j_1},\cdots,v_{j_b})$$

We will make use of the Einstein summation convention, which is illustrated with the following example:

$$S^i T_i = \sum_{j=1}^n S^j T_j$$

Furthermore, if the abbreviation (n.s.) appears beside an equation, it means "no sum". The symmetrization of a type $\binom{a}{0}$ tensor T is defined as follows:

$$T^{(i_1,...,i_a)} = \frac{1}{a!} \sum_{\sigma \in S_a} T^{i_{\sigma(1)},...,i_{\sigma(a)}}$$

where the sum is over all elements of the symmetric group S_a on a elements. In this notation, the symmetric product takes the form:

$$(T \odot S)^{i_1, \dots, i_a, j_1, \dots, j_b} = T^{(i_1, \dots, i_a} S^{j_1, \dots, j_b)}$$

Similarly, the *anti-symmetrization* of a type $\binom{0}{a}$ tensor T is defined as follows:

$$T_{[i_1,\dots,i_a]} = \frac{1}{a!} \sum_{\sigma \in S_a} (\operatorname{sgn} \sigma) T_{i_{\sigma(1)},\dots,i_{\sigma(a)}}$$

where $\operatorname{sgn} \sigma = \pm 1$ denotes the sign of the permutation σ . In this notation, the *wedge* product takes the form:

$$(T \wedge S)_{i_1,\dots,i_a,j_1,\dots,j_b} = \frac{(a+b)!}{a!b!} T_{[i_1,\dots,i_a} S_{j_1,\dots,j_b]}$$

1.4.2 pseudo-Riemannian manifolds

All differentiable structures are assumed to be smooth (class C^{∞}). Let M be a *pseudo-Riemannian manifold* of dimension n equipped with covariant metric g. By a *Riemannian manifold*, we mean a pseudo-Riemannian manifold with a positive-definite metric. Unless specified otherwise, it is assumed that M is connected and $n \geq 2$. The contravariant metric is usually denoted by G and $\langle \cdot, \cdot \rangle$ plays the role of the covariant and contravariant metric depending on the arguments. We denote $S^p(M)$ (resp. $A^p(M)$) as the set of symmetric (resp. anti-symmetric) contravariant tensor fields of valence p on M and $S(M) = \bigcup_{p\geq 0} S^p(M)$. Furthermore $\mathcal{F}(M) = S^0(M)$ is the set of functions from M to \mathbb{R} and $\mathfrak{X}(M) = S^1(M)$ denotes the set of vector fields over M. If $f \in \mathcal{F}(M)$ then $\nabla f \in \mathfrak{X}(M)$ denotes the gradient of f, i.e. the vector field metrically equivalent to the exterior derivative df.

We assume the reader is familiar with the concept of a distribution, foliation, and the (local) Frobenius theorem (see, for example, [Lee12]). A distribution E naturally induces a subspace of $\mathfrak{X}(M)$, denoted $\Gamma(E)$. More precisely, $v \in \Gamma(E)$ if $v \in \mathfrak{X}(M)$ and for every $p \in M$ we have $v|_p \in E_p$.

We will also use the existence and uniqueness theorem for the following class of PDEs:

$$\frac{\partial y_i}{\partial x^j} = A_j^i(x^1, \dots, x^p, y^1, \dots, y^q) \quad i = 1, \dots, q, \ j = 1, \dots, p$$

It can be deduced from the Frobenius theorem (see [AMR01, Section 7.4B] or [Lee12, Proposition 19.29]) that the above system of PDEs has a complete solution y = f(x, c) where c = f(0, c) are the initial conditions iff the mixed partials commute, i.e. $\frac{\partial^2 y_i}{\partial x^k \partial x^j} = \frac{\partial^2 y_i}{\partial x^j \partial x^k}$.

All notions from pseudo-Euclidean space generalize point-wise to pseudo-Riemannian manifolds. Definitions with subspaces in pseudo-Euclidean space naturally generalize to distributions in pseudo-Riemannian manifolds. For example, given a distribution E, the orthogonal distribution E^{\perp} is defined at each point $p \in M$ by $(E^{\perp})_p = (E_p)^{\perp}$. All definitions are only required to hold locally. For example, given a self-adjoint $\binom{1}{1}$ -tensor T on M, we say it is an orthogonal tensor if it is point-wise diagonalizable on some (non-empty) open subset of M and we tacitly work on this subset. Similarly we say Tis not an orthogonal tensor on M if T is not point-wise diagonalizable on a open dense subset of M. Similar definitions apply to other notions such as constancy of functions on M.

Smoothness of Eigenvalues and Eigenvectors Suppose T is a $\binom{1}{1}$ -tensor on M. The question arises weather the eigenfunctions of T and the eigenvector (fields) are smooth.

First of all, we should make clear that we always work in an open subset of M where the Jordan form is of a fixed type. For example, if T is point-wise diagonalizable, we assume the multiplicities of the eigenvalues are constants and the eigenvectors corresponding to a given eigenfunction form a smooth distribution. Now, suppose T is a $\binom{1}{1}$ -tensor of class C^p such that for some $q \in M$, T_q has simple eigenvalues. Then one can show that there exists a neighborhood of q in which T has simple eigenfunctions of class C^p , and T admits a basis of eigenvector fields of class C^p . The proof is an application of the implicit function theorem (see, for example [Die08, Theorems 10.2.1-10.2.4]). Details can be found in [Kaz98], see also [Lax07].

If we relax the condition that the eigenfunctions of T are simple, then the problem inevitably gets more complex (see [Kaz98] for some examples). For our applications though, we will not need such general results. When stating general results involving the eigenfunctions/eigenvector fields of a $\binom{1}{1}$ -tensor, we will always assume the eigenfunctions/eigenvector fields are smooth.

Riemann curvature tensor Our sign convention for the Riemann curvature tensor, R, is the opposite of that in [O'N83]. Hence it is defined by:

$$R(X,Y)Z = \nabla_X \nabla_Y Z - \nabla_Y \nabla_X Z - \nabla_{[X,Y]} Z \quad X,Y,Z \in \mathfrak{X}(M)$$

A space of constant curvature κ is intrinsically defined by the following condition on the Riemann curvature tensor R [O'N83, Corollary 3.43]:

$$R(X, Y, V, W) = \kappa(\langle V, X \rangle \langle Y, W \rangle - \langle V, Y \rangle \langle X, W \rangle) \quad X, Y, V, W \in \mathfrak{X}(M)$$
(1.4.1)

or in index notation $R_{ijkl} = \kappa (g_{ik}g_{jl} - g_{jk}g_{il})$. We will be working with specific models of these spaces which will be introduced when needed.

Chapter 2

Overview of Chapters and Theory

In this chapter we will connect the following chapters by giving an overview of the theory. References are given to the appropriate chapters for more details, proofs and generalizations. As much of the chapters in this thesis attempt to be somewhat comprehensive and contain generalizations, this chapter is crucial for understanding the theory. This chapter motivates many of the generalizations presented in later chapters and some key notions such as concircular tensors and warped products. In this chapter we basically show, fairly abstractly, and with some examples, how one can use concircular tensors to solve the fundamental problems (given in the introduction) for spaces of constant curvature. For illustrative purposes, we will sketch how one obtains the separable coordinate systems for the Calogero-Moser system.

Some proofs are included when they are simple and allow the reader to understand the theory. However, the chapter is recommended to be read lightly at first, and in more detail after reading the subsequent chapters in the first part of the thesis. We emphasize this point for Sections 2.3.2 and 2.4 which are more technical, and so the proofs can be skipped on the first reading. We also note that some earlier review articles written by Benenti may complement our exposition given here, see [Ben04; Ben93].

In this chapter (M, g) is a pseudo-Riemannian manifold and T^*M denotes the cotangent bundle of M. If (q, p) denote the canonical (position-momenta) coordinates on T^*M , then the (natural) Hamiltonian H with potential $V \in \mathcal{F}(M)$ is defined by:

$$H(q,p) := \frac{1}{2} \langle p, p \rangle + V(q)$$

The geodesic Hamiltonian is obtained by setting $V \equiv 0$ in the above equation. The Hamilton-Jacobi equation is a PDE defined on M in terms of the Hamiltonian. Coordinates (q^i) (for M) are called *separable* if they are orthogonal and the Hamilton-Jacobi equation separates in them. This is all one needs to know about the Hamilton-Jacobi equation to understand the theory which we are about to present.

2.1 The Intrinsic Characterization of Separation

The first cruical result is due to Stäckel in 1893 [Sta93]. Stäckel showed that if the Hamilton-Jacobi equation of a natural Hamiltonian is orthogonally separable then it admits n quadratic first integrals F_1, \ldots, F_n (where $F_1 := H$) each having the following form in canonical coordinates (q^i, p_i) [Sta93] (see [Kal86, P. 9] for English readers):

$$F = \frac{1}{2}K^{ij}p_ip_j + U(q^i)$$
(2.1.1)

with:

$$\{F_i, F_j\} = 0, \quad \mathrm{d}F_1 \wedge \dots \wedge \mathrm{d}F_n \neq 0$$

where $\{\cdot, \cdot\}$ is the Poisson bracket. One can show that the condition $\{F, H\} = 0$ is equivalent to the following equations on M (see Theorem 4.2.2 and Eq. (4.2.7)):

$$\nabla_{(i}K_{jk)} = 0 \tag{2.1.2a}$$

$$\mathrm{d}U = K\mathrm{d}V \tag{2.1.2b}$$

where ∇ is the Levi-Civita connection induced by g. The first of the above equations, Eq. (2.1.2a), shows the important fact that K is a Killing tensor (KT) on (M, g). Using this fact Eisenhart was the first to obtain an intrinsic characterization of separation for geodesic Hamiltonians [Eis34]. In order to present this theorem, we first need a definition. A *characteristic Killing tensor* (*ChKT*) is a Killing tensor which has simple eigenfunctions and admits coordinates in which it is diagonalized.

Theorem 2.1.1 (Orthogonal Separation of Geodesic Hamiltonians [Eis34])

The geodesic Hamiltonian is separable in orthogonal coordinates (q^i) iff there exists a ChKT which is diagonalized in these coordinates.

Given a ChKT, K, let $\mathcal{E} = (E_1, \ldots, E_n)$ denote the collection of eigenspaces of K. The above theorem shows that any coordinates (q^i) with the property that span $\{\partial_i\} = E_i$ are separable. Hence we call the collection \mathcal{E} a *separable web*. More generally, any collection $\mathcal{E} = (E_1, \ldots, E_n)$ of pair-wise orthogonal non-degenerate 1-distributions which admit local coordinates (q^i) satisfying span $\{\partial_i\} = E_i$ is called an *(orthogonal) web*. Since separable webs are uniquely determined by ChKTs, we will often work with them instead of coordinates.

The second equation, Eq. (2.1.2b), is a compatibility condition between the KT K and potential V. The structure of Eq. (2.1.2) hints at the fact that the separation of the geodesic Hamiltonian is a necessary condition for the separation of a natural Hamiltonian. Benenti was the first to obtain the intrinsic characterization of separation for natural Hamiltonians [Ben97].

Theorem 2.1.2 (Orthogonal Separation of Natural Hamiltonians [Ben97])

A natural Hamiltonian with potential V is separable in orthogonal coordinates (q^i) iff there exists a ChKT K diagonalized in these coordinates which satisfies:

$$d(KdV) = 0$$

The above equation is called the dKdV equation associated with the KT K and potential V.

For geodesic separation, each first integral given by Eq. (2.1.1) has a corresponding KT K. It can be deduced from Stäckel's theorem that the n KTs are point-wise independent on M and span an n-dimensional vector space of KTs which are simultaneously diagonalized in the separable coordinates. This vector space of KTs is called the *Killing-Stackel space* (KS-space) associated with a separable web.

We conclude with the following observations. Firstly, Theorem 2.1.1 implies that the problem of classifying separable coordinates for a geodesic Hamiltonian is equivalent to the problem of classifying ChKTs. Secondly, Theorem 2.1.2 shows that the problem of classifying ChKTs is important for separating natural Hamiltonians as well. Killing tensors are studied in greater detail in Chapter 4. The Hamilton-Jacobi equation, its separation, and the intrinsic characterization of separation is studied in greater detail in Chapter 5.

2.2 Concircular tensors

In the previous section we have given an intrinsic characterization of separation, which allows one to, in principle, obtain all separable coordinates systems defined on a given pseudo-Riemannian manifold. There are several problems one confronts when trying to apply the theory, particularly to spaces of constant curvature. We list some of these problems, assuming n > 2.

- It is difficult to obtain an algebraic expression for the general ChKT in a space of constant curvature.
- Given a ChKT, it's hard to obtain the transformation from separable coordinates $(u^i) \rightarrow (q^i)$ to the standard coordinates.
- It's also hard to find canonical forms for ChKTs modulo the action of the isometry group.

When n = 3, one can manage with these difficulties. Indeed, in [HMS05] (resp. [HM08; HMS09]), building on results from [Eis34], canonical forms for the isometrically inequivalent ChKTs in \mathbb{E}^3 (resp. M^3) were given together with the corresponding transformations from separable coordinates. Furthermore, the authors were able to solve all the fundamental problems given in the introduction. However, generalizing their solution to higher dimensions seems intractable.

Kalnins and Miller in [Kal86] were able to devise a procedure to construct the transformation from separable coordinates for all isometrically inequivalent ChKTs on Riemannian spaces of constant curvature, thereby solving the first fundamental problem (1). Waksjo and Wojciechowski in [WW03] used this procedure to solve the last two fundamental problems (2 and 3) for Euclidean and spherical spaces. A careful study of these solutions and works by others (e.g. [Ben05; Cra03]) show that concircular tensors have a fundamental role to play in these solutions.

A concircular tensor (CT), $L \in S^2(M)$, is defined by the following equation:

$$\nabla_k L_{ij} = \alpha_{(i}g_{j)k}$$

for some covector α . One can obtain a general solution to the above equation in \mathbb{E}^n . First, define the *dilatational vector field* in \mathbb{E}^n in Cartesian coordinates (x^i) by $r := x^i \partial_i$. Then the general solution is given as follows (see Proposition 6.4.4):

$$L = A + 2w \odot r + mr \odot r$$

where A is a symmetric and constant matrix, w is a constant vector and m is a constant scalar. We denote the unit sphere in \mathbb{E}^n by \mathbb{S}^n . Then the restriction of the above tensor to \mathbb{S}^n gives the general CT (see Proposition 9.3.2). CTs solve the problems confronted with ChKTs listed above. Indeed, in this thesis, we will show that CTs can be used to solve the fundamental problems in spaces of constant curvature.

We say a CT is an orthogonal concircular tensor (OCT) if it is point-wise diagonalizable. An important property of OCTs is that they always admit local coordinates which diagonalize them. More precisely, suppose L is an OCT, then there exist local coordinates (x^i) such that L has the following form (see Propositions 6.3.1 and 6.3.6):

$$L = \sum_{a \in M} \sigma_a \partial_a \otimes \mathrm{d}x^a + \sum_{I \in P} e_I \sum_{i \in I} \partial_i \otimes \mathrm{d}x^i$$
(2.2.1)

where $\{1, \ldots, n\} = M \cup (\bigcup_{I \in P} I)$ is a partition (here P is an index set and each $I \in P$ is a subset of $\{1, \ldots, n\}$), the $\sigma_a(x^a)$ are non-constant and the e_I are constants.

Another important property of CTs is that some special KTs can be constructed using them. Indeed, if L is a CT, it can be shown that the following sequence of tensors are KTs (see Section 6.6):

$$K_0 = G,$$
 $K_a = \frac{1}{a} \operatorname{tr}(K_{a-1}L)G - K_{a-1}L$ $1 < a < n$ (2.2.2)

The KT K_1 is so special, it deserves a name. If L is a CT, then the tensor

$$K = \operatorname{tr}(L)G - L$$

is a KT, called the *Killing Bertrand-Darboux tensor (KBDT)* associated with L. This KT will be useful for connecting CTs with the general theory of separation given in the previous section. An important observation is that it has the same eigenspaces as L.

We conclude by noting that concircular tensors in general are studied in Chapter 6 while those in spaces of constant curvature are studied in Chapter 9.

2.3 Separation of Geodesic Hamiltonians

2.3.1 Benenti tensors

We say a CT L is a *Benenti tensor* if it has simple eigenfunctions. A key observation made by Benenti is that any Benenti tensor induces a separable web [Ben92a]. Indeed, since the KBDT is a KT with simple eigenfunctions and can be diagonalized in a coordinate system (see Eq. (2.2.1)), it's a ChKT, hence the result follows by Theorem 2.1.1. Furthermore it can be shown that the KTs given by Eq. (2.2.2) form a basis for the KS-space associated with this separable web.

An important class of Benenti tensors are the *irreducible concircular tensors (ICTs)*. A CT L is called irreducible if it's a Benenti tensor and its eigenfunctions are functionally independent. By Eq. (2.2.1) any Benenti tensor with non-constant eigenfunctions is irreducible. This class of CTs are of interest, because in this case, by Eq. (2.2.1) the eigenfunctions can be used as separable coordinates! We will see shortly that ICTs can be used as building blocks to construct more general classes of separable coordinates. The following is the prototypical example of an ICT:

Example 2.3.1 (Elliptic coordinates in \mathbb{E}^2)

Let $M = \mathbb{E}^2$ and fix an orthonormal basis $\{d, e\}$ for this Euclidean space. Let (x, y) be Cartesian coordinates for \mathbb{E}^2 so that $d = \partial_x$ and $e = \partial_y$. Then consider the following CT:

$$L = \lambda_1 d \odot d + \lambda_2 e \odot e + r \odot r$$

WLOG we can assume $\lambda_1 < \lambda_2$. We will show how to obtain the transformation from separable to Cartesian coordinates after showing that L is a Benenti tensor. The characteristic polynomial of L is given as follows:

$$p(z) = \det(zI - L) = (z - \lambda_1)(z - \lambda_2) - x^2(z - \lambda_2) - y^2(z - \lambda_1)$$

From the above equation, we note the following:

$$p(\lambda_1) = x^2(\lambda_2 - \lambda_1) \qquad \qquad p(\lambda_2) = y^2(\lambda_1 - \lambda_2) \qquad (2.3.1)$$

Now, assume that $x, y \neq 0$. Then we observe that $p(\lambda_1) > 0$, $p(\lambda_2) < 0$ and $\lim_{z\to\infty} p(z) = \infty$. Hence by the intermediate value theorem, at each point, p(z) has two distinct roots $u^1 < u^2$ satisfying:

$$\lambda_1 < u^1 < \lambda_2 < u^2$$

Thus L is a Benenti tensor. Since $dp \neq 0$, it follows that L cannot have constant eigenfunctions, thus from the preceding discussion we see that L is an ICT. Now observe that we can write $p(z) = (z - u^1)(z - u^2)$. Then Eq. (2.3.1) can be used to obtain the transformation from the separable coordinates (u^1, u^2) to Cartesian coordinates (x, y):

$$x^{2} = \frac{(\lambda_{1} - u^{1})(\lambda_{1} - u^{2})}{(\lambda_{2} - \lambda_{1})} \qquad \qquad y^{2} = \frac{(\lambda_{2} - u^{1})(\lambda_{2} - u^{2})}{(\lambda_{1} - \lambda_{2})} \qquad \qquad \square$$

The above example will be generalized to higher dimensions and signatures later on, see Example 9.4.11. Proceeding as in the above example and using additional results from Chapter 9, one can classify all (isometrically inequivalent) separable coordinates associated with Benenti tensors in \mathbb{E}^2 , including polar and Cartesian coordinates. The results of this classification are summarized in Table 9.1.

We conclude by introducing a diagram for Benenti tensors (see Fig. 2.1) which will be used later on. It represents the structure of the separable web associated with the Benenti tensor, which is the simplest possible.

Figure 2.1: Concircular tensor with simple eigenspaces E_1, \ldots, E_n

$$\begin{bmatrix} E_1 \end{bmatrix} \cdots \begin{bmatrix} E_n \end{bmatrix}$$

2.3.2 Concircular tensors with Multidimensional Eigenspaces and KEM webs

More generally, any orthogonal concircular tensor can (possibly) be used to construct separable webs, as we will see in this section. Suppose L is a non-trivial¹ concircular tensor with a single multidimensional eigenspace D and denote by D^{\perp} the distribution orthogonal to D. Then one can show that (see Proposition 6.3.6):

- There is a local product manifold $B \times F$ of (pseudo-)Riemannian manifolds (B, g_B) and (F, g_F) such that:
 - $\{p\} \times F$ is an integral manifold of D for any $p \in B$ and
 - $B \times \{q\}$ is an integral manifold of D^{\perp} for any $q \in F$.
- $B \times F$ equipped with the metric $\pi_B^* g_B + \rho^2 \pi_F^* g_F$ for a specific function $\rho : B \to \mathbb{R}^+$ is locally isometric to (M, g).

Such a product manifold is called a *warped product* and is denoted $B \times_{\rho} F$. The manifold B is called the *geodesic factor* and F is called the *spherical factor* of the warped product. We also say that the warped product $B \times_{\rho} F$ is *adapted* to the splitting (D^{\perp}, D) , which is often called a *warped product net (WP-net)*. When a distribution D admits an adapted warped product as above, it is called a *Killing distribution*. See Chapter 3 for more details on these matters.

We note here that warped products are rigid. For example, in Euclidean space, it can be shown that if an open connected subset U is isometric to a warped product with a single spherical factor, then the warped product must have one of the following forms:

1. $\mathbb{E}^m \times_{\rho} \mathbb{S}^r$

2. $\mathbb{E}^m \times_1 \mathbb{E}^r$

The warped products in more general spaces of constant curvature are systematically obtained in Appendix D, see Section 8.4 for a summary sufficient for the purposes of this thesis.

Now, if we enumerate the one dimensional eigenspaces of L by E_1, \ldots, E_m and denote the multidimensional eigenspace of L by D as above, then Fig. 2.2 gives a diagram for L. In this figure, the block containing the eigenspace D represents a "degeneracy" which needs to be removed to uniquely specify a separable web. We now describe how to do this.

Figure 2.2: Concircular tensor with eigenspaces
$$E_1, \ldots, E_m, D$$

 $E_1 \qquad \cdots \qquad E_m \qquad D$

A remarkable property of the warped product decomposition is the following. Let \tilde{K} be a ChKT on F, this can be canonically lifted to a tensor, $\tilde{K} \in S^2(B \times_{\rho} F)$, which is

¹A CT is called non-trivial if its not a multiple of the metric.

in fact a KT on $B \times_{\rho} F$! Hence if K' is the KBDT associated with L, then locally we can assume that $K' + \tilde{K}$ is a ChKT on $B \times_{\rho} F$. Indeed, one can show that L induces a Benenti tensor, \tilde{L} , on B by restriction. Let (x^i) be any coordinates on B which diagonalize \tilde{L} . Note that we observed in the previous section that these coordinates are separable on B. Suppose (y^j) are coordinates on F which diagonalize \tilde{K} , hence are separable (see Theorem 2.1.1). Then since the product coordinates (x^i, y^j) diagonalize $K' + \tilde{K}$ (see Eq. (2.3.2)), Theorem 2.1.1 implies that $K' + \tilde{K}$ is a ChKT² and that these coordinates are separable. Note that in these coordinates $K' + \tilde{K}$ have the following form:

$$K' + \tilde{K} = \sum_{i} (\operatorname{tr}(L) - \lambda_i) \partial_i \otimes \mathrm{d}x^i + \sum_{j} (\operatorname{tr}(L) - c + \tilde{\lambda}_j) \partial_j \otimes \mathrm{d}y^j$$
(2.3.2)

where λ_i are the eigenfunctions of \tilde{L} , c is the constant eigenfunction of L associated with D and $\tilde{\lambda}_j$ are the eigenfunctions of \tilde{K} . In conclusion, we have shown how to construct separable coordinates (x^i, y^j) using the CT L and ChKT \tilde{K} .

Now take \tilde{K} to be the KBDT associated with a Benenti tensor on F which has eigenspaces $\tilde{E}_1, \ldots, \tilde{E}_k$. Then Fig. 2.3 is a diagram for the above construction applied to \tilde{K} , which represents the tree-like structure of the constructed separable web. It should be interpreted as a tree diagram, where the one dimensional eigenspaces are the leaves. We illustrate this construction with two simple examples in \mathbb{E}^3 , both of which are depicted by Fig. 2.3 with m = 1 and k = 2.



Example 2.3.2 (Cylindrical coordinates in \mathbb{E}^3)

Fix a unit vector $d \in \mathbb{E}^3$ and consider the following CT:

$$L = d \odot d$$

The eigenspaces of L are span $\{d\}$ and d^{\perp} . Identify $\mathbb{E} = \text{span}\{d\}$ and $\mathbb{E}^2 = d^{\perp}$, then the warped product $\psi : \mathbb{E} \times_1 \mathbb{E}^2 \to \mathbb{E}^3$ given by $(q, p) \to q + p$ is adapted to the eigenspaces of L. We can construct separable coordinates in \mathbb{E}^3 by parameterizing \mathbb{E}^2 with any of the separable coordinates from Table 9.1. For example, let e, f be an orthonormal basis for d^{\perp} , let q = x d and $p = \rho \cos \theta e + \rho \sin \theta f$, then we obtain cylindrical coordinates:

²The eigenfunctions may not exactly be simple, but one can modify \tilde{K} so that they are locally simple.
$$\psi(q,p) = x \, d + \rho \cos \theta \, e + \rho \sin \theta \, f \qquad \Box$$

The following is a more interesting example of this construction.

Example 2.3.3 (Spherical coordinates in \mathbb{E}^3)

Consider the following CT in \mathbb{E}^3 :

$$L = r \odot r$$

The eigenspaces of L are span $\{r\}$ and r^{\perp} . Fix a unit vector $a \in \mathbb{E}^3$, identify $\mathbb{E} = \mathbb{R}^+ a$, let \mathbb{S}^2 be the unit sphere in \mathbb{E}^3 and $\rho_1 := \langle q, a \rangle$ for $q \in \mathbb{E}$. Then the warped product $\psi : \mathbb{E} \times_{\rho_1} \mathbb{S}^2 \to \mathbb{E}^3$ given by $(q, p) \to \rho_1 p$ is adapted to the eigenspaces of L. We can construct separable coordinates in \mathbb{E}^3 by parameterizing \mathbb{S}^2 with any of the separable coordinates defined in it.

For example, one can take spherical coordinates on \mathbb{S}^2 . Indeed, fix a unit vector $d \in \mathbb{E}^3$. Then one can show that the restriction of $d \odot d$ to \mathbb{S}^2 is a Benenti tensor diagonalized in spherical coordinates (see Example 9.6.3), which are given as follows:

$$p = \cos(\phi)d + \sin(\phi)(\cos(\theta)e + \sin(\theta)f)$$

where e, f is any orthonormal basis for d^{\perp} . Hence the above coordinates are separable in \mathbb{S}^2 . If we let $q = \rho a$ where $\rho > 0$ and take p as above, then we obtain spherical coordinates in \mathbb{E}^3 :

$$\psi(q,p) = \rho(\cos(\phi)d + \sin(\phi)(\cos(\theta)e + \sin(\theta)f))$$

More examples can be found in Section 9.6.2.

This construction procedure can be generalized in two ways. Firstly, we can recursively apply this procedure, by treating $B \times_{\rho} F$ as the spherical factor of a larger warped product and use $K + \tilde{K}$ in place of \tilde{K} . Figure 2.4 depicts such a construction where the CT L has eigenspaces E'_1 and D'. Again, this figure depicts the tree-like structure of the KEM web where the leaves are the one dimensional eigenspaces of the CTs that make it up.

Secondly, we can allow L to have multiple distinct multidimensional eigenspaces. These procedures can also be combined to create even more complex webs, as the following example will show. Figure 2.5 depicts the natural generalization of the above construction procedure to CTs with multiple multidimensional eigenspaces. In this case, the CT L has only multidimensional eigenspaces D_1, \ldots, D_r .

We emphasize here that in each case, the constructed web is separable. Any coordinates constructed using this procedure are called *Kalnins-Eisenhart-Miller (KEM) coordinates*



and the associated webs are called KEM webs. We will show that KEM webs are always separable.

We've shown how CTs can be used to construct a special class of separable webs called KEM webs. A significant advantage of KEM webs is that we can reduce the problem of classifying isometrically inequivalent KEM webs to the similar problem for CTs. We will see that the problem of classifying isometrically inequivalent CTs in spaces of constant curvature can be reduced to problems in linear algebra (see Chapter 9).

In conclusion, we mention how some of the ideas presented here are generalized. The observation that CTs (which are in fact CKTs) induce a warped product decomposition of the (pseudo-)Riemannian manifold, motivates the more systematic study of CKTs in Section 4.4. This culminates in Corollary 4.4.8 and Corollary 4.4.11. In Section 4.5 we will be able to prove that the KT \tilde{K} in the above construction is a KT on the warped product. In fact, this observation will be generalized to describe KTs which are "decomposable" in a warped product. In Section 6.5 we will prove the facts on KEM webs we presented here more rigorously. In Section 6.6 we will apply the theory developed in Section 5.3.1 on KS-spaces in warped products to show how one can obtain the KS-space associated with a KEM web.

2.3.3 Necessity of KEM webs in spaces of constant curvature

In the previous section we have shown how to construct a class of separable webs called KEM webs. These webs were originally discovered by Kalnins and Miller when classifying the separable webs in Riemannian spaces of constant curvature [Kal86]. Generalizing their results, one can prove the following.

Theorem 2.3.4 (Separable Webs in Spaces of Constant Curvature)

In a space of constant curvature, every separable web is a KEM web.

This theorem is proven in Chapter 7. It involves a long calculation in which we solve the Levi-Civita equations together with the equations satisfied by the Riemann curvature tensor in a space of constant curvature (see Eq. (1.4.1)).

The above theorem allows us to tractably solve problem (1) in spaces of constant curvature. Motivated by the above theorem, in Chapter 9 we study (orthogonal) concircular tensors in spaces of constant curvature. In that chapter, we obtain the required information to reduce problem (1) to simple problems in linear algebra.

2.4 Separation of Natural Hamiltonians

In this section we will sketch how concircular tensors can be used to separate natural Hamiltonians. We will use Theorem 2.1.2 and our knowledge of the structure of KEM webs to develop a recursive algorithm to separate natural Hamiltonians in KEM webs.

Fix some $V \in \mathcal{F}(M)$. Let *L* be the general concircular tensor on *M* and let $K := \operatorname{tr}(L)G - L$ be the KBDT associated with *L*. The Killing-Bertrand-Darboux (KBD) equation on *M* is defined as follows:

$$d(KdV) = 0$$

It can be shown that this equation defines a linear system of equations with at most $\frac{1}{2}(n+1)(n+2)$ unknowns, where the maximum is achieved iff the space has constant curvature.

Let L be a particular solution of the KBD equation which is point-wise diagonalizable with k distinct eigenfunctions. We analyze the following cases.

Case 1 (k = 1, i.e. all the eigenfunctions coincide)

L = cG for some $c \in \mathbb{R}$. This is the trivial solution which gives no information.

Case 2 (the eigenfunctions are simple)

L has simple eigenfunctions, hence it's a Benenti tensor. Then V separates in any coordinates which diagonalize L by Theorem 2.1.2.

Case 3 (at least one eigenfunction is not simple)

Assume for convenience, that L has a single multidimensional eigenspace D. If E_1, \ldots, E_m denote the one dimensional eigenspaces of L, then so far we know that V is "compatible" with the partial separable web in Fig. 2.6.

Figure 2.6: Concircular tensor with eigenspaces E_1, \ldots, E_m, D $\begin{bmatrix} E_1 \end{bmatrix} \cdots \begin{bmatrix} E_m \end{bmatrix} \begin{bmatrix} D \end{bmatrix}$

Now the goal is to fill in the degeneracy coming from D. This is done as follows. Let $B \times_{\rho} F$ be a local warped product adapted to (D^{\perp}, D) . Let $\tau : F \to B \times F$ be an embedding. Assume the natural Hamiltonian on F associated with potential $V \circ \tau$ is separable in some coordinates (y^j) . Let (x^i) be separable coordinates associated with the induced Benenti tensor on B. Then one can show that the natural Hamiltonian associated with V (on $B \times_{\rho} F$) is separable in the product coordinates (x^i, y^j) .

Indeed, this can be seen as follows. Let \tilde{K} be a ChKT on F diagonalized in (y^j) , and K' be the KBDT associated with L. In the discussion preceding Eq. (2.3.2), it was shown that we can assume $K := K' + \tilde{K}$ is locally a ChKT on $B \times_{\rho} F$ diagonalized in (x^i, y^j) . Given the assumptions, one can show that V satisfies the dKdV equation with K on $B \times_{\rho} F$, hence by Theorem 2.1.2 it's separable in the coordinates (x^i, y^j) .

In the third case, in order to obtain the separable coordinates (y^j) , the idea is to apply the same procedure again on F with the potential $V \circ \tau \in \mathcal{F}(F)$. So one has to solve the KBD equation on F with the potential $V \circ \tau$ and then go through each case. This gives us a recursive algorithm for separating natural Hamiltonians, which is called the Benenti-Eisenhart-Kalnins-Miller (BEKM) separation algorithm. Figure 2.7 gives a possible KEM web that can be constructed, assuming the solution of the KBD equation on F is a Benenti tensor with eigenspaces $\tilde{E}_1, \ldots, \tilde{E}_k$.



In principle, one can construct any KEM web using the BEKM separation algorithm. For example, just take V = 0. We now briefly illustrate the execution of this algorithm with the following example.

Example 2.4.1 (Calogero-Moser system)

The Calogero-Moser system is a natural Hamiltonian system with configuration manifold \mathbb{E}^3 given by the following potential in Cartesian coordinates (q_1, q_2, q_3) :

$$V = (q_1 - q_2)^{-2} + (q_2 - q_3)^{-2} + (q_1 - q_3)^{-2}$$

First note that the constant vector $d = \frac{1}{\sqrt{3}}(\partial_1 + \partial_2 + \partial_3)$ is a symmetry of V, i.e. $\mathcal{L}_d V = 0$. One can prove that the general solution of the KBD equation associated with V is³:

$$L = c d \odot d + 2w d \odot r + m r \odot r$$

where $c, w, m \in \mathbb{R}$. We note that given a CT L, then for any $a \in \mathbb{R} \setminus \{0\}$ and $b \in \mathbb{R}$, the CT aL + bG is a CT which is equivalent to L. After classifying the above CTs modulo this equivalence and isometric equivalence, we can obtain canonical forms. Before we present these, fix an orthonormal basis e, f for d^{\perp} . We have the following canonical forms.

Cartesian: $L = d \odot d$

From Example 2.3.2 we know that a warped product manifold adapted to L has the form $\mathbb{E} \times \mathbb{E}^2$. Let (q'_1, q'_2, q'_3) be Cartesian coordinates adapted to this product manifold, then one can show that V takes the form:

$$V = \frac{9(q_3'^2 + q_2'^2)^2}{2q_2'^2(3q_3'^2 - q_2'^2)^2}$$

In this case V naturally restricts to a potential on \mathbb{E}^2 with coordinates (q'_2, q'_3) . In \mathbb{E}^2 one can apply the BEKM separation algorithm to find that the only solution of the KBD equation (up to constant multiplies) is $L = r \odot r$. One can show that polar coordinates diagonalize this CT. Hence V is separable in cylindrical coordinates:

$$x d + \rho \cos \theta e + \rho \sin \theta f$$

Spherical: $L = r \odot r$

From Example 2.3.3 we know that a warped product manifold adapted to L has the form $\mathbb{E} \times_{\rho} \mathbb{S}^2$. One can show that the restriction of V to \mathbb{S}^2 satisfies the KBD equation associated with the CT obtained by restricting $d \odot d$ to \mathbb{S}^2 . Hence from Example 2.3.3, V is separable in spherical coordinates:

 $\rho(\cos(\phi)d + \sin(\phi)(\cos(\theta)e + \sin(\theta)f))$

³We ignore constant multiples of the metric.

Elliptic: $L = cd \odot d + r \odot r, c \neq 0$

In this case L is a Benenti tensor. If we let $a := \sqrt{|c|}$, then if c > 0, V is separable in prolate spheroidal coordinates:

$$a\cos\phi\cosh\eta d + a\sin\phi\sinh\eta(\cos\theta e + \sin\theta f)$$

If c < 0, V is separable in oblate spheroidal coordinates:

 $a\sin\phi\sinh\eta d + a\cos\phi\cosh\eta(\cos\theta e + \sin\theta f)$

Parabolic: $L = 2d \odot r$

In this case L is a Benenti tensor, and so V is separable in rotationally symmetric parabolic coordinates:

$$\frac{1}{2}(\mu^2 - \nu^2) d + \mu\nu(\cos\theta \, e + \sin\theta \, f) \qquad \Box$$

The above example will be done in much greater detail and for a more general potential in Section 10.2. The BEKM separation algorithm is presented in more detail and with proofs in Section 6.7. It motivates the study of separation of natural Hamiltonians in warped products in Section 5.4.1.

Completeness of the BEKM separation algorithm In spaces of constant curvature, the BEKM separation algorithm gives a complete test for orthogonal separation. This is a consequence of Theorem 2.3.4. We also note that the separable coordinates can be explicitly constructed by following through the algorithm, this is shown by way of example in Section 10.2. Hence the BEKM separation algorithm solves problems (2) and (3) in spaces of constant curvature.

Spaces of constant curvature In order to apply the BEKM separation algorithm (i.e. reduce it to problems in linear algebra) in spaces of constant curvature, CTs in these spaces are studied throughly in Chapter 9. In order to do this in spaces with arbitrary signature, one needs to solve some non-standard problems in linear algebra. The prerequisite theory is covered in Appendix C, and summarized in Section 8.2. The results obtained in Chapter 9 are used in Section 10.3 to concretely carry out the BEKM separation algorithm, and in Section 10.2 to study the separability properties of a well known example, the Calogero-Moser system.

Separable potentials We also mention here that in Section 10.1, we give special potentials that can proven to be separable in KEM webs.

2.5 Conclusion

We have given an overview of how the fundamental problems are solved in this thesis, and how this solution is broken down in the various different chapters. The first part of this thesis will present the theory more rigorously, with greater detail, and present some generalizations of parts of it. The second part of this thesis will (mainly) apply the theory to spaces of constant curvature.

Part I General Theory

Chapter 3

Warped Products

In this chapter we will present the theory of warped products, most of which will be used in later chapters. Warped products will not be defined until Section 3.5. We will first develop the necessary theory to characterize these products in terms of the distributions they induce. This characterization is better suited for applications later on. In fact, motivated by applications, we will study a more general product structure known as a twisted product. In this chapter we provide a fairly comprehensive introduction to the warped product and so it is recommended that it be read lightly at first and in more detail when the need arises.

3.1 pseudo-Riemannian Submanifolds and Foliations

In this section we will summarize the theory of pseudo-Riemannian submanifolds and foliations that will be useful to us. We can conveniently treat this as a special case of the theory of pseudo-Riemannian distributions, so we will present this first. For more details on pseudo-Riemannian submanifolds see (for example) [O'N83; Lee97]. Similarly for pseudo-Riemannian foliations see [Rov98; Ton88].

3.1.1 Brief outline of The Theory of Pseudo-Riemannian Distributions

The following brief exposition of the theory of pseudo-Riemannian distributions is a combination of that given in [MRS99] and [CFS06]. Suppose E is an m-dimensional non-degenerate distribution defined on a pseudo-Riemannian manifold \overline{M} . Then we use the orthogonal splitting $T\overline{M} = E \oplus E^{\perp}$, $V = V^E + V^{E^{\perp}}$, to define a tensor $s^E : T\overline{M} \times E \to E^{\perp}$ and a linear connection ∇^E for E by:

$$\nabla_X Y = \nabla_X^E Y + s^E(X, Y)$$

for all $X \in \mathfrak{X}(\bar{M})$ and $Y \in \Gamma(E)$. s^E is called the generalized second fundamental form of E and the above equation is referred to as the Gauss equation. One can also check that ∇^E is metric compatible, i.e. $X \langle Y, Z \rangle = \langle \nabla^E_X Y, Z \rangle + \langle Y, \nabla^E_X Z \rangle$ for all $X \in \mathfrak{X}(\bar{M})$ and $Y, Z \in \Gamma(E)$.

For the remainder of the discussion we set $s^E := s^E|_{(E \times E)}$. For $X, Y \in \Gamma(E)$, we can further decompose $s^E(X, Y)$ into its anti-symmetric and symmetric parts

$$s^{E}(X,Y) = (\nabla_{X}Y)^{E^{\perp}} = \frac{1}{2}(\nabla_{X}Y + \nabla_{Y}X)^{E^{\perp}} + \frac{1}{2}(\nabla_{X}Y - \nabla_{Y}X)^{E^{\perp}}$$
$$= h^{E}(X,Y) + A^{E}(X,Y)$$
$$A^{E}(X,Y) := \frac{1}{2}(\nabla_{X}Y - \nabla_{Y}X)^{E^{\perp}}$$
$$h^{E}(X,Y) := \frac{1}{2}(\nabla_{X}Y + \nabla_{Y}X)^{E^{\perp}}$$

Since ∇ is torsion-free, $A^E(X,Y) = \frac{1}{2}([X,Y])^{E^{\perp}}$, hence *E* is integrable iff $A^E \equiv 0$. h^E is called the *second fundamental form* of *E*. The second fundamental form can be decomposed in terms of its trace to get a further classification of *E* as follows:

$$h^{E}(X,Y) = \langle X,Y \rangle H_{E} + h^{E}_{T}(X,Y)$$
$$H_{E} = \frac{1}{m} \operatorname{tr}(h^{E})$$

where h_T^E is trace-less. H_E is called the mean curvature normal of E. E is called minimal, umbilical or geodesic¹ if $s^E(X,Y) = h_T^E(X,Y)$, $s^E(X,Y) = \langle X,Y \rangle H_E$ or $s^E(X,Y) = 0$ respectively for all $X, Y \in \Gamma(E)$. We add the qualification "almost" to the three definitions above by replacing s^E with h^E ; this just drops the requirement that $A^E \equiv 0$. For example E is almost umbilical iff $h_T^E = 0$. We remark that when E is one dimensional $h_T^E = 0$ trivially, hence all one dimensional non-degenerate foliations and all one dimensional pseudo-Riemannian submanifolds are trivially umbilical. If E is umbilical and $\nabla_X^{E^{\perp}} H_E = 0$ for all $X \in \Gamma(E)$ then E is called spherical. Finally if E is spherical and E^{\perp} is geodesic then E is called Killing.

We also note here that s^E and $s^{E^{\perp}}$ are not independent of each other:

Proposition 3.1.1

For $X, Y \in \Gamma(E)$ and $Z \in \Gamma(E^{\perp})$, the following holds:

$$\left\langle s^{E}(X,Y),Z\right\rangle = -\left\langle Y,s^{E^{\perp}}(X,Z)\right\rangle$$

¹Note that some authors use the name auto-parallel instead [MRS99].

Proof

$$0 = \nabla_X \langle Y, Z \rangle$$

= $\langle \nabla_X Y, Z \rangle + \langle Y, \nabla_X Z \rangle$
= $\langle s^E(X, Y), Z \rangle + \langle Y, s^{E^{\perp}}(X, Z) \rangle$

The following proposition gives an important geometric characterization of the second fundamental form. It is the key lemma used to connect twisted products with their differential counterparts as we will see in Section 3.5. It is taken from [Rov98, Proposition 2.7] (cf. [Ton88, Theorem 5.23], [Zeg11, Lemma 2.3]).

Proposition 3.1.2 (Geometric Characterization of The Second Fundamental Form) Let E be a non-degenerate distribution. Denote the covariant metric by g and by g_E the restriction of g to E. Suppose $U, V \in \Gamma(E)$ and $Z \in \Gamma(E^{\perp})$, then the second fundamental h of E is characterized by the following equation:

$$(\mathcal{L}_Z g_E)(U, V) = -2g(Z, h(U, V))$$

E is almost geodesic (almost umbilical) iff g_E is invariant (resp. conformal invariant) under flows of vector fields orthogonal to E, i.e. for $V \in \Gamma(E^{\perp})$, $\mathcal{L}_V g_E = 0$ (resp, $\mathcal{L}_V g_E = -2g(H, V)g_E$ where H is the mean curvature normal of E).

PROOF By definition, we have the following:

$$(\mathcal{L}_Z g_E)(U, V) = Z g_E(U, V) - g_E([Z, U], V) - g_E(U, [Z, V])$$

= $Z g(U, V) - g([Z, U], V) - g(U, [Z, V])$

Now since $[Z, U] = \nabla_Z U - \nabla_U Z$ for a torsion-free connection and with a similar equation holding for [Z, V], the above equation becomes

$$(\mathcal{L}_Z g_E)(U, V) = Zg(U, V) - g([Z, U], V) - g(U, [Z, V])$$
$$= g(\nabla_U Z, V) + g(U, \nabla_V Z)$$

Since $0 = Ug(Z, V) = g(\nabla_U Z, V) + g(Z, \nabla_U V)$, with a similar equation holding for Vg(U, Z), the above equation becomes:

$$(\mathcal{L}_Z g_E)(U, V) = g(\nabla_U Z, V) + g(U, \nabla_V Z)$$
$$= -g(Z, \nabla_U V) - g(Z, \nabla_V U)$$
$$= -2g(Z, h(U, V))$$

The remaining assertions follow from the definitions of almost geodesic and almost umbilical distributions.

3.1.2 Specialization to pseudo-Riemannian submanifolds

Suppose $\phi: M \to \overline{M}$ is a local embedding of (a pseudo-Riemannian submanifold) M^m inside \overline{M}^n . Then for any point $p \in M$, it is known that there exist local coordinates (x^i) on \overline{M} , such that the subset

$$\{(x^1, \dots, x^m, x^{m+1}, \dots, x^n) : x^{m+1} = c_{m+1}, \dots, x^n = c_n\}$$

for some $c_{m+1}, \ldots, c_n \in \mathbb{R}$ can be identified with $\phi(U)$ where U is an open subset with $p \in U \subseteq M$. These coordinates induce a local foliation L in a neighborhood of p, with M being a leaf given by the above equation. We will refer to such a foliation as a (local) foliation of \overline{M} associated with M. Now suppose L is an arbitrary foliation of \overline{M} associated with M, and let E be the induced distribution. Locally we can assume L is a foliation by pseudo-Riemannian submanifolds of \overline{M} , hence E is non-degenerate and the discussion in the previous section applies to it. Since E is integrable, it follows that for any $X, Y \in \Gamma(E)$, that $[X, Y] \in \Gamma(E)$. Throughout this discussion, for any $X \in \Gamma(E)$, we let $\tilde{X} \in \mathfrak{X}(M)$ denote the unique vector field such that for any $p \in M$, we have $X_{\phi(p)} = \phi_* \tilde{X}_p$. Then for any $X, Y \in \Gamma(E)$ we see that

$$[X,Y]|_{\phi(p)} = \phi_*[\tilde{X},\tilde{Y}]|_p$$

Thus $[X, Y]|_{\phi(p)}$ depends only on $[\tilde{X}, \tilde{Y}]|_p$ in M.

Now denote by ∇ (resp. ∇) the Levi-Civita connection on M (resp. \overline{M}). By the uniqueness properties of the Levi-Civita connection on M, it follows that for any $X, Y \in \Gamma(E)$ we have for any $p \in M$ that

$$(\bar{\nabla}_X^E Y)|_{\phi(p)} = \phi_*(\nabla_{\tilde{X}}\tilde{Y})|_p$$

Thus $(\bar{\nabla}_X^E Y)|_{\phi(p)}$ depends only on $(\nabla_{\tilde{X}} \tilde{Y})|_p$ in M. By also using the Gauss equation, we observe that for any $p \in M$, that $(\bar{\nabla}_X Y)|_{\phi(p)}$ depends only on \tilde{X} and \tilde{Y} .

In consequence of these observations, it follows that the theory presented for pseudo-

Riemannian distributions induces a similar one for pseudo-Riemannian manifolds. We now connect this with the standard notations [Che11]; in effect this removes the appearance of the extraneous distribution, E.

In this case $s^E \equiv h^E$ and $h := (h^E)|_M$, then the Gauss equation becomes:

$$\bar{\nabla}_X Y = \nabla_X Y + h(X, Y)$$

for all $X, Y \in \mathfrak{X}(M)$. We denote the set of normal vector fields over M, i.e. the restriction of $\Gamma(E^{\perp})$ to M by $\mathfrak{X}(M)^{\perp}$. The Gauss equation for E^{\perp} is usually called the Weingarten equation and is only defined for $X \in \mathfrak{X}(M)$ and $Y \in \mathfrak{X}(M)^{\perp}$. This is because in this case, $\overline{\nabla}_X Y$ depends only on the values that X and Y take on M^2 . Thus we can let $A_Y(X) := -s^{E^{\perp}}(X,Y), \, \nabla^{\perp}_X Y := \bar{\nabla}^{E^{\perp}}_X Y$ and the Gauss equation (for E^{\perp}) becomes:

$$\bar{\nabla}_X Y = \nabla_X^{\perp} Y - A_Y(X)$$

for all $X \in \mathfrak{X}(M)$ and $Y \in \mathfrak{X}(M)^{\perp}$. Note that the properties of $\overline{\nabla}^{E^{\perp}}$ imply that ∇^{\perp} is a connection³ on $\mathfrak{X}(M)^{\perp}$. In this notation, the relationship between s^{E} and $s^{E^{\perp}}$ given in Proposition 3.1.1 becomes:

$$\langle h(X,Y), Z \rangle = \langle A_Z(X), Y \rangle$$
 (3.1.1)

for all $X, Y \in \mathfrak{X}(M)$ and $Z \in \mathfrak{X}(M)^{\perp}$.

Finally, we note that the definitions of minimal, umbilical, or geodesic foliations induces corresponding definitions for submanifolds. For example, a submanifold is geodesic if its second fundamental form vanishes identically.

In conclusion, we should mention that even though we have given a concise presentation of the theory, it's not useful for practical calculations. For these, one will have to evaluate these quantities in terms of curves on M. See for example, Proposition 4.8 in [O'N83].

Circles and Spheres* 3.2

In this section we will briefly overview the theory of circles and spherical submanifolds (spheres) of pseudo-Riemannian manifolds. Circles are covariantly defined using the Frenet formula, but the definition of a sphere requires more work [Nom73]. This material is optional, although it gives an application of the general theory presented in the previous section, a geometric interpretation of spherical submanifolds, and gives some background for the results on the intrinsic properties of warped products to come. We also present

²This is because for any $p \in \overline{M}$, $(\overline{\nabla}_X Y)|_p$ depends only on the values of Y along any curve tangent to X_p . See Lemma 4.8 in [Lee97] and the following exercise, or Proposition 3.18 (3) in [O'N83]. ³More precisely it satisfies the properties in definition 3.9 in [O'N83] and is metric compatible.

this theory here because it's not covered in standard references, in contrast with the corresponding theory for geodesic submanifolds (see [O'N83]).

A proper circle⁴ in a pseudo-Riemannian manifold is defined using the Frenet formula as a unit speed curve whose first curvature is constant and non-zero and remaining curvatures vanish. To be precise, let $\gamma(t)$ be a unit speed curve in M, i.e. $\dot{\gamma}^2 = \pm 1$. Let $X := \dot{\gamma}$. Let $\kappa(t) := \|\nabla_X X\|$ be the (first) curvature of γ . Assuming $\kappa \neq 0$, we define Y to be the unit vector field over γ derived from $\nabla_X X$, that is Y satisfies the following equation

$$\nabla_X X = \kappa Y$$

A proper circle is defined to be a curve which satisfies $\nabla_X Y = cX$ for some $c \in \mathbb{R} \setminus \{0\}$. We observe that

$$\langle \nabla_X Y, X \rangle = - \langle Y, \nabla_X X \rangle$$

= $-\kappa \langle Y, Y \rangle$

The above equation implies that $c = -\varepsilon_0 \varepsilon_1 \kappa$ where $\varepsilon_0 := \operatorname{sgn} \langle X, X \rangle$ and $\varepsilon_1 := \operatorname{sgn} \langle Y, Y \rangle$. Thus a proper circle is defined by the equations

$$\nabla_X X = \kappa Y$$
$$\nabla_X Y = -\varepsilon_0 \varepsilon_1 \kappa X$$

where $\kappa \neq 0$ is a constant. A proper circle satisfies the following third order ODE [ANY90]:

$$\nabla_X \nabla_X X = - \langle \nabla_X X, \nabla_X X \rangle \langle X, X \rangle X \tag{3.2.1}$$

Conversely we will see shortly that any unit speed curve satisfying the above equation with $\langle \nabla_X X, \nabla_X X \rangle \neq 0$ is a proper circle. We define a *circle* in a pseudo-Riemannian manifold to be a unit speed curve satisfying the above equation, hereafter called the *circle equation*. The following lemma shows that any pseudo-Riemannian manifold admits circles:

Lemma 3.2.1 (Existence and Uniqueness of Circles [NY74])

Consider the following initial conditions: $p \in M$, a unit vector $X_p \in T_pM$ and $Y_p \in X_p^{\perp}$. There exists a unique locally defined unit speed curve $\gamma(t)$ in M satisfying Eq. (3.2.1) and

⁴Sometimes these are called geodesic circles [Ami03]. This name emphasizes the fact that we due not require the image of these curves to be a compact set, i.e. homeomorphic to \mathbb{S}^1 .

the initial conditions:

$$\gamma(0) = p$$
$$\dot{\gamma}(0) = X_p$$
$$\nabla_X X)|_p = Y_p$$

where $X := \dot{\gamma}$ and $Y := \nabla_X X$. Furthermore, $\langle Y, Y \rangle$ is constant along any circle.

(

PROOF It follows by the existence and uniqueness theorem for ODEs that there exists a unique locally defined curve $\gamma(t)$ satisfying Eq. (3.2.1) with the above initial conditions. Then observe the following:

$$\nabla_X \langle X, X \rangle = 2 \langle X, \nabla_X X \rangle = \langle X, Y \rangle$$
$$\nabla_X \langle X, Y \rangle = \langle Y, Y \rangle + \langle X, \nabla_X Y \rangle$$
$$\stackrel{(3.2.1)}{=} \langle Y, Y \rangle - \langle X, X \rangle^2 \langle Y, Y \rangle$$
$$= \langle Y, Y \rangle (\langle X, X \rangle^2 - 1)$$

The above two equations define a system of ODEs for $\langle X, X \rangle$ and $\langle X, Y \rangle$, with initial values $\langle X, X \rangle |_p = \varepsilon = \pm 1$ and $\langle X, Y \rangle |_p = 0$. Thus by the uniqueness of the solutions, it follows that $\langle X, X \rangle = \varepsilon$ and $\langle X, Y \rangle = 0$ wherever γ is defined. Hence γ is a unit speed curve.

Finally observe that

$$\nabla_X \langle Y, Y \rangle = 2 \langle \nabla_X Y, Y \rangle$$

$$\stackrel{(3.2.1)}{=} -2 \langle \nabla_X X, \nabla_X X \rangle \langle X, X \rangle \langle X, Y \rangle$$

$$= 0$$

Hence $\langle Y, Y \rangle$ is constant.

Note that k := ||Y|| in the above lemma is usually called the *curvature* of the circle. In Riemannian manifolds, circles are completely classified by their curvature, although this is not true for pseudo-Riemannian manifolds. Using the above lemma we can classify circles in a pseudo-Riemannian manifold as follows. Let $\gamma(t)$ be a circle in M and suppose γ satisfies the initial conditions of the above lemma. Then γ can be classified as follows depending on Y_p :

Geodesic: If $Y_p = 0$.

Proper Circle: If $\langle Y, Y \rangle |_p \neq 0$.

Null Circle: If $\langle Y, Y \rangle |_p = 0$ but $Y_p \neq 0$, i.e. Y_p is lightlike, hence Eq. (3.2.1) reduces to $\nabla_X \nabla_X X = 0$.

Note that this classification is well defined globally since $\langle Y, Y \rangle$ is a constant of a circle and the uniqueness theorem for ODEs forces any circle with $Y_p = 0$ to be a geodesic.

Example 3.2.2 (Geodesics in Spherical Submanifolds [Kas10])

Let M be a spherical submanifold of M. Suppose $\gamma(t)$ is a unit speed geodesic on M. We will show that γ is a circle in \overline{M} . By the Gauss equation, we have the following:

$$\bar{\nabla}_X X = \langle X, X \rangle H$$

Then by the Weingarten equation and using the fact that $\overline{\nabla}^{\perp} H = 0$ where $\overline{\nabla}^{\perp}$ is the induced normal connection over M, we have the following:

$$\begin{split} \bar{\nabla}_X \bar{\nabla}_X X &= \langle X, X \rangle \, \bar{\nabla}_X H \\ &= - \langle X, X \rangle \, A_H(X) \\ &= - \langle X, X \rangle \, \langle H, H \rangle \, X \\ &= - \langle X, X \rangle \, \langle \bar{\nabla}_X X, \bar{\nabla}_X X \rangle \, X \end{split}$$

since for any $Z \in \mathfrak{X}(M)^{\perp}$, $\langle A_H(X), Z \rangle \stackrel{(3.1.1)}{=} \langle h(X, Z), H \rangle = \langle X, Z \rangle \langle H, H \rangle.$

We note here that the above example in combination with Lemma 3.2.1 shows that the mean curvature vector field of a spherical submanifold is locally determined by its value at a single point. Also note that the proper circles in pseudo-Euclidean space are given in Example D.4.4.

We will now present some additional results that show how circles can be used to characterize spherical submanifolds. These results were first obtained for the Riemannian case by Nomizu and Yano in [NY74]. They were generalized to the Lorentzian case by Ikawa in [Ika85] and to the pseudo-Riemannian case by Abe, Nakanishi, and Yamaguchi in [ANY90].

For the following theorems we denote a pseudo-Riemannian manifold M with signature α by M_{α} . The following theorem characterizes spherical submanifolds in terms of circles, it is analogous to the corresponding theorem for geodesics and geodesic submanifolds (see [O'N83, section 4.4]).

Theorem 3.2.3 (Circles and Spheres [ANY90])

Let M_{α} be an *n* dimensional pseudo-Riemannian submanifold of \overline{M}_{β} . For any $\varepsilon_0 \in \{-1, 1\}$ and $\varepsilon_1 \in \{-1, 0, 1\}$ satisfying $2 - 2\alpha \leq \varepsilon_0 + \varepsilon_1 \leq 2n - 2\alpha - 2$ and $k \in \mathbb{R}^+$, the following are equivalent:

- (a) Every circle in M_{α} with $\langle X, X \rangle = \varepsilon_0$ and $\langle \nabla_X X, \nabla_X X \rangle = \varepsilon_1 k^2$ is a circle in \overline{M}_{β} .
- (b) M_{α} is a spherical submanifold of \overline{M}_{β} .

PROOF See [ANY90].

More intuitively, the above theorem states that a spherical submanifold M is precisely a submanifold in which all circles in M are circles in the ambient space. Also note that the above theorem shows that a circle is precisely a spherical submanifold of dimension one. The following theorem is a variant of the above theorem which is known to hold (in full generality) only in the strictly pseudo-Riemannian case.

Theorem 3.2.4 (Circles and Spheres II [ANY90])

Let M_{α} be an *n* dimensional $(1 \leq \alpha \leq n-1)$ pseudo-Riemannian submanifold of \overline{M}_{α} having the same signature α . For any $\varepsilon_0 \in \{-1, 1\}$, the following are equivalent:

- (a) Every geodesic in M_{α} with $\langle X, X \rangle = \varepsilon_0$ is a circle in \overline{M}_{α} .
- (b) M_{α} is a spherical submanifold of \overline{M}_{α} .

These results can be further generalized by considering more general types of curves such as helices (which we will not define here). See [Nak88] where a theorem analogous to Theorem 3.2.3 is proven characterizing helices in terms of geodesic submanifolds. Also in [JF94] results relating conformal circles to umbilical submanifolds are presented.

The following lemma describes how much information is required to specify a sphere. It is a partial generalization of the corresponding lemma for the Riemannian case proven in [Kas10].

Lemma 3.2.5 (Uniqueness of Spheres)

Suppose that M and N are connected and geodesically complete spherical submanifolds of \overline{M} both satisfying the following condition: For some $p \in M \cap N$, M and N are tangent and have the same mean curvature vectors. Then $M \equiv N$.

PROOF Our proof is a generalization of the proof of lemma 4.14 in [O'N83, P. 105].

Let $q \in M$ be arbitrary and suppose that $\gamma(t)$ is a geodesic segment in M running from p to q. Then observe that γ is a geodesic circle in \overline{M} with velocity X_p and acceleration $\langle X, X \rangle |_p H_p^M$ at p where H^M is the mean curvature vector field of M. By the uniqueness

of circles (see Lemma 3.2.1) and the hypothesis it follows that γ is also geodesic in N which is defined everywhere since N is geodesically complete. Note that this implies that mean curvature vector fields of M and N coincide over γ , so we denote this vector field by H.

Now suppose $Z_p \in T_p M \cap X_p^{\perp}$ and let Z be the parallel transport of Z_p over γ with respect to M. Since parallel transport is an isometry, $\langle Z, X \rangle = 0$. Thus by the Gauss equation,

$$\bar{\nabla}_X Z = \nabla^M_X Z + \langle Z, X \rangle H$$
$$= 0$$

where $\overline{\nabla}$ is the Levi-Civita connection on \overline{M} and ∇^M is the induced Levi-Civita connection on M. Thus Z is also the parallel transport of Z_p over γ with respect to \overline{M} .

Thus the parallel transport of $T_pM \cap X_p^{\perp}$ to q on \overline{M} is equal to $T_qM \cap X_q^{\perp}$. Similarly the parallel transport of $T_pN \cap X_p^{\perp}$ to q on \overline{M} is equal to $T_qN \cap X_q$. Since the parallel transport on \overline{M} is uniquely determined, we deduce that $T_qM \cap X_q^{\perp} = T_qN \cap X_q^{\perp}$. Since $X_q \in T_qM, T_qN$, we conclude that $T_qM = T_qN$. Thus since M is connected, one can apply this argument to an arbitrary broken geodesic (see [O'N83]) to conclude that $M \subseteq N$.

Finally by applying the argument for M interchanged with N, we see that $M \equiv N_{\bullet}$

Let M be a space of constant curvature. We will show in this thesis that for every $p \in M$, non-degenerate subspace $V \subset T_p M$, and normal vector $H \in (T_p M)^{\perp}$ there exists a connected and geodesically complete spherical submanifold passing through p with tangent space V and mean curvature vector H at p. In the following theorem, we will show that this property characterizes Riemannian spaces of constant curvature. For the following theorem, we say a Riemannian manifold M satisfies the *axiom of r-spheres* if: for every $p \in M$ and any r dimensional subspace $V \subset T_p M$ there exists a spherical submanifold passing through p and tangent to V.

Theorem 3.2.6 (Spheres in spaces of constant curvature [LN71])

Let M be a Riemannian manifold with dimension $n \ge 3$ and fix $2 \le r < n$. Then M is a space of constant curvature iff it satisfies the axiom of r-spheres (see above).

PROOF See [LN71].

3.3 Product Manifolds

In this section we will briefly introduce some notations used on product manifolds. Suppose $M = \prod_{i=1}^{k} M_i$ is a product of pseudo-Riemannian manifolds (M_i, g_i) . We denote $M_{i\perp} := M_1 \times \cdots \times M_{i-1} \times M_{i+1} \times \cdots \times M_k$ and the canonical projections $\pi_i : M \to M_i$ by $p \to p_i$ for each *i*. We also let $\pi_{i\perp} : M \to M_{i\perp}$ be the canonical projection associated with the decomposition $M = M_i \times M_{i\perp}$. We denote by L_i the canonical foliation of Minduced by M_i . For $\bar{p} \in M$, the leaf of $L_i(\bar{p})$ through \bar{p} and the canonical embedding of M_i in M denoted τ_i are given by

$$\tau_i(p) := (\bar{p}_1, \dots, \bar{p}_{i-1}, p, \bar{p}_{i+1}, \dots, \bar{p}_k), \quad p \in M_i$$
$$L_i(\bar{p}) := \tau_i(M_i) = \{ p \in M : p = \tau_i(p_i), \ p_i \in M_i \}$$

We let E_i denote the integrable distribution induced by L_i .

We can naturally "lift" any tensor defined on the manifolds M_i to M. For example if $\tilde{\varphi} \in \mathcal{F}(M_i)$ then the lift is $\varphi := \tilde{\varphi} \circ \pi_i \in \mathcal{F}(M)$, we denote the set of all such functions on M of this form by $\hat{\mathcal{F}}(M_i)$. For $\tilde{v} \in \mathfrak{X}(M_i)$, the lift is the unique vector field $v \in \mathfrak{X}(M)$ such that $(\pi_i)_*v = \tilde{v}$ and $(\pi_{i\perp})_*v = 0$. Analogously we denote the set of all such vector fields on M of this form by $\hat{\mathfrak{X}}(M_i)$; note that $\hat{\mathfrak{X}}(M_i)$ is in general a proper subspace of $\Gamma(E_i)$. $\hat{S}^p(M_i)$ is defined similarly.

Example 3.3.1

Suppose $M = \prod_{i=1}^{k} M_i$ is a product manifold. In adapted coordinates this lifting operation is very simple. Indeed, let (y_j^i) be coordinates for M_j and consider the product coordinates $(x) = (y_1, \ldots, y_k)$ for M. If $T \in S^2(M_j)$, then the lift, \tilde{T} , satisfies the following equation:

$$\tilde{T}(\mathrm{d}y_i^k, \mathrm{d}y_i^l) = \begin{cases} T(\mathrm{d}y_i^k, \mathrm{d}y_i^l) & \text{if } i = j \\ 0 & \text{else} \end{cases}$$

Hence note that if E_j denotes the distribution induced by M_j , then \tilde{T} is tangent to E_j , i.e. \tilde{T} can locally be written as a sum of 2-fold symmetrized products of elements in $\Gamma(E_j)$. Furthermore, the non-zero components in product coordinates are functions on M_j .

If $v \in \hat{\mathfrak{X}}(M_i)$ and $u \in \hat{\mathfrak{X}}(M_j)$, then $(\pi_i)_*[v, u] = [\tilde{v}, \tilde{u}]$ if i = j and [v, u] = 0 if $i \neq j$. Also note that usually we will use the same symbol for a tensor and its lift. For $\varphi \in \mathcal{F}(M)$, we say that φ is independent of M_i (or E_i) if $\varphi \in \hat{\mathcal{F}}(M_{i\perp})$; if M is connected this is equivalent to $\varphi_*E_i = 0$. We say that φ depends only on M_i (or E_i) if $\varphi \in \hat{\mathcal{F}}(M_i)$.

3.4 Nets and their Integrability

The following notion of (orthogonal) nets will be useful:

Definition 3.4.1 (Nets [MRS99])

A family $\mathcal{E} = (E_i)_{i=1}^k$ of integrable distributions E_i on a manifold M is called a *net* on M if the tangent bundle TM can be decomposed as:

$$TM = \bigoplus_{i=1}^{k} E_i$$

If M is a pseudo-Riemannian manifold, and the direct sum in the above equation is replaced with the orthogonal direct sum, then \mathcal{E} is called an *orthogonal net*. \Box

Remark 3.4.2

If M is a pseudo-Riemannian manifold, then unless specified otherwise, all nets are assumed to be orthogonal.

A net \mathcal{E} is said to be *(locally) integrable* (or *locally decomposable* in [MRS99]) if for every $p \in M$ there exists a neighborhood $U \subseteq M$ of p and a C^{∞} -diffeomorphism f from a product manifold $\prod_{i=1}^{k} M_i$ onto U such that for every $q \in \prod_{i=1}^{k} M_i$ and every i = 1, ..., kthe slice $(q_1, ..., q_{i-1}) \times M_i \times (q_{i+1}, ..., q_k)$ gets mapped into an integral manifold of E_i . In this case, the product manifold $\prod_{i=1}^{k} M_i$ is said to be (locally) *adapted* to \mathcal{E} . An (orthogonal) net \mathcal{E} is called an *(orthogonal) web* if it is integrable and dim $E_i = 1$ for each i. Given a collection of distributions $\mathcal{E} = (E_i)_{i=1}^k$ on a pseudo-Riemannian manifold, we say the collection is *orthogonally integrable* if \mathcal{E} forms an integrable orthogonal net. For the following theorem, if \mathcal{E} is not assumed to be orthogonal, then $E_i^{\perp} := \bigoplus_{j \neq i} E_j$. In [RS99, Theorem 1] the following has been shown, which justifies the term "orthogonally integrable"

Theorem 3.4.3 (Characterizations of integrable nets [RS99])

For the decomposition $TM = \bigoplus_{i=1}^{k} E_i$ by the family of distributions $\mathcal{E} = (E_i)_{i=1}^{k}$, the following are equivalent

- 1. \mathcal{E} is an integrable net.
- 2. The orthogonal distributions E_i^{\perp} are integrable for i = 1, ..., k.
- 3. The distributions E_i and their direct sums $E_i \oplus E_j$ are integrable for i, j = 1, ..., k.

The above theorem also proves the following well known fact:

Corollary 3.4.4

Any net \mathcal{E} with two factors, i.e. $\mathcal{E} = (E_1, E_2)$, is integrable.

When the net has more than two factors, it's easy to find non-integrable cases:

Example 3.4.5 (Non-integrable nets)

Suppose M^n is a pseudo-Riemannian manifold with n = 3. Let u be any non-null vector field such that u^{\perp} is not an integrable distribution. Extend this to a local orthonormal basis, $\{u, v, w\}$ for TM, then clearly these vector fields form a net which is not integrable by the above theorem.

As a concrete example, one can take $M = \mathbb{E}^3$ and u to be the Killing vector field whose integral curves are helices.

The following example gives the simplest way to obtain integrable nets.

Example 3.4.6 (Product Nets)

Suppose $M = \prod_{i=1}^{k} M_i$ is a product of manifolds M_i . If E_i denotes the canonical foliation induced by M_i then $\mathcal{E} = (E_i)_{i=1}^k$ is called the *product net* of $\prod_{i=1}^k M_i$. Note that by definition, \mathcal{E} is an integrable net. If each M_i is a pseudo-Riemannian manifold equipped with covariant metric g_i , then equipping M with the metric $g = \sum_{i=0}^k \pi_i^* g_i$ makes \mathcal{E} into an orthogonal net.

3.5 Warped and Twisted Products

Warped products are ubiquitous in applications of pseudo-Riemannian geometry. Most of the separable coordinate systems in spaces of constant curvature are built up using them [Kal86], and some exact solutions in general relativity are composed of them [DU05; Zeg11]. They can intuitively be thought of as a partial generalization of the spherical coordinate system to arbitrary pseudo-Riemannian manifolds. Indeed, it will eventually become clear that all the spherical coordinate systems (on any space of constant curvature) can be constructed iteratively using warped products, and that they share several properties with these coordinate systems. Similarly it will be clear that the well known Schwarzschild metric in relativity can be constructed by using warped products. In this section we will give a brief introduction to these products by studying them as special cases of twisted products. The content of this section is primarily from [MRS99] where the notion of a twisted product is studied. For more on warped products and applications see [O'N83; MRS99; Zeg11].

The following general definition of a twisted product is useful in the study of conformal Killing tensors.

Definition 3.5.1 (Warped and Twisted Products)

Let $M = \prod_{i=0}^{k} M_i$ be a product of pseudo-Riemannian manifolds (M_i, g_i) where dim $M_i > 0$ for i > 0. Suppose for $i = 0, ..., k, \pi_i : M \to M_i$ is the projection map and $\rho_i : M \to \mathbb{R}^+$ is a function. The following metric g on M is called a *twisted product metric*

$$g(X,Y) = \sum_{i=0}^{k} \rho_i^2 g_i(\pi_{i*}X, \pi_{i*}Y) \quad \text{for } X, Y \in \mathfrak{X}(M)$$

In this case (M, g) is called a *twisted product* and is denoted by $\prod_{i=0}^{k} M_i$ where $\rho = (\rho_0, ..., \rho_k)$. Furthermore the ρ_i are called twist functions of the twisted product. If each ρ_i depends only on M_0 and $\rho_0 \equiv 1$ then g is called a *warped product metric* and (M, g) is called a *warped product*. The warped product is denoted by $M_0 \times_{\rho_1} M_1 \times \cdots \times_{\rho_k} M_k$.

 M_0 is called the geodesic factor of the warped product and the M_i for i > 0 are called spherical factors.

Example 3.5.2

By taking M_0 to be a point and k = 1 in the definition of a twisted product, we get a *conformal product*.

Example 3.5.3

By taking M_0 to be a point and k > 1 in the definition of a warped product, we get a *pseudo-Riemannian product*. Throughout this thesis we will treat pseudo-Riemannian products as special cases of warped products this way.

Example 3.5.4

If dim $M_i = 1$ for each i, then the twisted product metric is locally the metric of an orthogonal coordinate system.

Example 3.5.5 (Prototypical warped product)

The prototypical example of a warped product is the following warped product defined in (an open subset of) \mathbb{E}^n , which is the product manifold $\mathbb{R}^+ \times S^{n-1}$ equipped with the metric $g = d\rho^2 + \rho^2 \tilde{g}$ where \tilde{g} is the metric of the (n-1)-sphere S^{n-1} .

Note that a twist function ρ_i of a twisted product is only uniquely defined modulo products of functions $f \in \hat{\mathcal{F}}(M_i)$. To elaborate, from the above definition one sees that we can multiply ρ_i^2 by $f \in \hat{\mathcal{F}}(M_i)$ if we divide g_i by f. The geometry of the twisted product is not altered by such transformations as we will see. We say that the twist functions are *normalized* (with respect to a point $\bar{p} \in M$), if for each $i, \rho_i(p) = 1$ for all $p \in L_i(\bar{p})$.

First we give the formulas for the Levi-Civita connection and Riemann tensor of a twisted product; it is from Proposition 1 in [MRS99]. We will make use of the following notation: Given a collection of distributions $(E_i)_{i=1}^k$ satisfying $TM = \bigoplus_{i=1}^k E_i$, then for any vector $X \in \mathfrak{X}(M)$, we have the orthogonal splitting $X = \sum_i X^i$ where each $X^i \in \Gamma(E_i)$.

Proposition 3.5.6

Let ${}^{\rho}\prod_{i=0}^{k}M_{i}$ be a twisted product with product net $\mathcal{E} = (E_{i})_{i=0}^{k}$. Let $\tilde{\nabla}$ be the Levi-Civita connection associated with the ordinary pseudo-Riemannian product metric of $\prod_{i=0}^{k}M_{i}$ with Riemann tensor \tilde{R} and ∇ be the Levi-Civita connection of the twisted product metric. Let $U_{i} := -\nabla \log \rho_{i}$ and $X, Y \in \mathfrak{X}(M)$, then ∇ is given as follows

$$\nabla_X Y = \tilde{\nabla}_X Y + \sum_{i=0}^k (\langle X^i, Y^i \rangle U_i - \langle X, U_i \rangle Y^i - \langle Y, U_i \rangle X^i)$$
(3.5.1)

Note that $\tilde{\nabla}$ satisfies $(\tilde{\nabla}_X Y)^i = \tilde{\nabla}_X Y^i$. The Riemann tensor R of the twisted product is given by:

$$R(X,Y) = \tilde{R}(X,Y) + \sum_{i=0}^{k} ((\nabla_X U_i - \langle X, U_i \rangle U_i) \wedge Y^i + X^i \wedge (\nabla_Y U_i - \langle Y, U_i \rangle U_i))$$
(3.5.2)

$$+\sum_{i,j=0}^{k} \left\langle U_i, U_j \right\rangle X^i \wedge Y^j$$

for any $X, Y \in \mathfrak{X}(M)$. For $X, Y \in \mathfrak{X}(M)$, the linear operator $X \wedge Y$ is the one metrically equivalent to the bivector.

The following corollary gives the corresponding formulas for a warped product.

Corollary 3.5.7

Let $M_0 \times_{\rho_1} M_1 \times \cdots \times_{\rho_k} M_k$ be a warped product. Let $\tilde{\nabla}$ be the Levi-Civita connection associated with the ordinary pseudo-Riemannian product metric of $\prod_{i=0}^k M_i$ with Riemann tensor \tilde{R} and ∇ be the Levi-Civita connection of the warped product metric. Let $H_i :=$ $-\nabla \log \rho_i$ and $X, Y \in \mathfrak{X}(M)$, then ∇ is given as follows

$$\nabla_X Y = \tilde{\nabla}_X Y + \sum_{i=1}^k (\langle X^i, Y^i \rangle H_i - \langle X, H_i \rangle Y^i - \langle Y, H_i \rangle X^i)$$

Note that $\tilde{\nabla}$ satisfies $(\tilde{\nabla}_X Y)^i = \tilde{\nabla}_X Y^i$. The Riemann tensor R of the warped product is given by:

$$R(X,Y) = \tilde{R}(X,Y) + \sum_{i=1}^{k} ((\nabla_{X^{0}}H_{i} - \langle X, H_{i} \rangle H_{i}) \wedge Y^{i} + X^{i} \wedge (\nabla_{Y^{0}}H_{i} - \langle Y, H_{i} \rangle H_{i}))$$
$$- \sum_{i,j=1}^{k} \langle H_{i}, H_{j} \rangle X^{i} \wedge Y^{j}$$

for any $X, Y \in \mathfrak{X}(M)$. For $X, Y \in \mathfrak{X}(M)$, the linear operator $X \wedge Y$ is the one metrically equivalent to the bivector. Furthermore, the Riemann tensor \tilde{R} satisfies $(\tilde{R}(X,Y)Z)^i = \tilde{R}(X^i, Y^i)Z^i$.

PROOF The formula for Riemann tensor follows from Eq. (3.5.2) by expanding $\nabla_X H_j$ as follows:

$$\nabla_X H_j = \nabla_{X^0} H_j - \sum_{i=1}^k \langle H_i, H_j \rangle X^i$$

The remaining facts follow from Proposition 3.5.6 and Corollary 2 in [MRS99].

The above formula for the curvature tensor can be used to obtain general formulas for the sectional curvature of warped products. First we need some definitions. If $f \in \mathcal{F}(M)$, we denote the Hessian of f [O'N83, P. 86], by $S_{ij}^f = \nabla_i \nabla_j f$. If $X, Y \in T_p M$ span a non-degenerate 2-plane then the sectional curvature of the 2-plane, K(X, Y), is given in terms of the curvature tensor R as [O'N83, lemma 3.39]:

$$K(X,Y) = \frac{\langle R(X,Y)Y,X\rangle}{\|X \wedge Y\|^2}, \qquad \|X \wedge Y\|^2 = \langle X,X\rangle \langle Y,Y\rangle - \langle X,Y\rangle^2$$

Now we have the following:

Corollary 3.5.8 (Sectional curvature of warped products)

Suppose $M = M_0 \times_{\rho_1} M_1 \times \cdots \times_{\rho_k} M_k$ is a warped product with product net $\mathcal{E} = (E_i)_{i=0}^k$. Let $X, Y \in \Gamma(E_0), V \in \Gamma(E_i)$ and $U \in \Gamma(E_k)$ for i, k > 0. If $H_i = -\nabla \log \rho_i$ denotes the mean curvature normal of E_i , then we have the following:

$$K_{XY} = K_{XY}^{M_0}$$

$$K_{XV} = -\frac{S^{\rho_i}(X, X)}{\rho_i X^2}$$
(3.5.3)

$$K_{UV} = -\langle H_i, H_k \rangle \quad (i \neq k)$$
(3.5.4)

$$K_{UV} = \frac{K_{UV}^{m_i} - (\nabla \rho_i)^2}{\rho_i^2} \quad (i = k)$$
(3.5.5)

PROOF This follows from the formula for the curvature tensor in Corollary 3.5.7.

The following properties of the twisted product can be found in Proposition 2 in [MRS99].

Proposition 3.5.9 (Properties of the Twisted Product [MRS99])

Let ${}^{\rho}\prod_{i=0}^{k}M_{i}$ be a twisted product with product net $\mathcal{E} = (E_{i})_{i=0}^{k}$ and $U_{i} := -\nabla \log \rho_{i}$.

- 1. \mathcal{E} is an orthogonally integrable net.
- 2. For each *i* the distribution E_i is umbilical with mean curvature normal $H_i = U_i^{\perp i}$.
- 3. E_i is geodesic iff ρ_i is independent of M_j for $j \neq i$. E_i^{\perp} is geodesic iff ρ_j is independent of M_i for $j \neq i$.
- 4. If ρ is independent of M_i then E_i is Killing. The converse is also true if the twisted product is normalized.

The following theorem characterizes twisted and warped products in terms of the geometry of their canonical foliations.

Theorem 3.5.10 (Geometric Characterization of Twisted and Warped Products [MRS99]) Let $M = \prod_{i=0}^{k} M_i$ be a connected product manifold equipped with metric g and orthogonal product net $\mathcal{E} = (E_i)_{i=0}^k$. Then g is the metric of

- 1. a twisted product ${}^{\rho}\prod_{i=0}^{k} M_i$ iff E_i are umbilical distributions
- 2. a warped product $M_0 \times_{\rho_1} M_1 \times \cdots \times_{\rho_k} M_k$ iff E_i are Killing distributions for i = 1, ..., k
- 3. a pseudo-Riemannian product iff E_i are geodesic distributions

PROOF The characterization of the twisted product follows from Proposition 4 in [MRS99]. We note here that the relationship between the second fundamental form and Lie derivatives of the metric given in Proposition 3.1.2 is crucial to the proof of this fact. The other characterizations follow from the first and Proposition 3.5.9 above.

Remark 3.5.11

It follows by definition of the twisted product, that they are invariant under conformal transformations. The conformal generalizations of warped and pseudo-Riemannian products and their characterizations are given in [Toj04].

The following notions of twisted and warped product nets will be especially useful for studying conformal Killing tensors. It was originally Definition 3 in [MRS99].

Definition 3.5.12 (Twisted and warped product nets)

Let M be a pseudo-Riemannian manifold and suppose $\mathcal{E} = (E_i)_{i=0}^k$ is an orthogonal net.

- 1. \mathcal{E} is called a *twisted product net (TP-net)* if it is integrable and each distribution E_i is umbilical.
- 2. \mathcal{E} is called a *warped product net (WP-net)* if E_i is Killing for i = 1, ..., k.

Remark 3.5.13

In all applications, dim $E_i > 0$ for i > 0. Although we will allow dim $E_0 = 0$ for a WP-net since this gives us a *pseudo-Riemannian product net (RP-net)*.

It can be shown that if \mathcal{E} is a WP-net, then it is a TP-net with $E_0 = \bigcap_{i=1}^{k} E_i^{\perp}$ a geodesic distribution [MRS99, Proposition 3]. Also in the case \mathcal{E} is a WP-net we refer to E_0 as the geodesic distribution of the WP-net and the E_i for i > 0 as the Killing distributions of the WP-net. The following theorem, which is Corollary 1 in [MRS99], gives the motivation for the above definition. It shows that every TP-net (resp. WP-net) admits a locally adapted twisted product (resp. warped product).

Theorem 3.5.14 (Twisted and warped product nets [MRS99])

Let (M, g) be a pseudo-Riemannian manifold and suppose $\mathcal{E} = (E_i)_{i=0}^k$ is a TP-net (resp. WP-net). Then for every $p \in M$ there exists an open set $U \subseteq M$ containing p and a map $f: \prod_{i=0}^k M_i \to U$ which is an isometry with respect to a twisted (resp. warped) product metric on $\prod_{i=0}^k M_i$.

PROOF The existence of the map $f : \prod_{i=0}^{k} M_i \to U$ which is a diffeomorphism is guaranteed by the integrability of the net \mathcal{E} , see Theorem 3.4.3. Once $\prod_{i=0}^{k} M_i$ is equipped with f^*g , the result then follows by Theorem 3.5.10.

Remark 3.5.15

One can also check that a similar theorem holds for a pseudo-Riemannian product net and metric. $\hfill \Box$

Now we can give some justification to the name "Killing" for a non-degenerate distribution which is spherical and has a geodesic orthogonal complement. By the above corollary, we see that a one dimensional Killing distribution is always spanned by a Killing vector field. Conversely any normal⁵ non-null Killing vector field spans a Killing distribution. The following can be said about multidimensional Killing distributions via the warped products they induce [Zeg11]:

Proposition 3.5.16 (Lifting isometries from Killing distributions)

Let $M = B \times_{\rho} F$ be a warped product and suppose $\tilde{f} : F \to F$ is an isometry of F. Then the lift f defined by

$$f(x,y) := (x, \tilde{f}(y)), \quad (x,y) \in B \times F$$

is an isometry of M.

⁵We say a non-null vector field is normal if the orthogonal distribution is Frobenius integrable.

Chapter 4

Killing tensors

In this chapter we thoroughly study the geometric objects encoding separation, (conformal) Killing tensors. In the first section we will first review a formalism we will use in this chapter and the next. In the following section, we introduce Killing tensors and give some equivalent definitions of them. In Section 4.3 we consider a conformal generalization of Killing tensors which will also be of use. In Section 4.4 we study orthogonal conformal Killing tensors systematically. We give a characterization of them based on the geometry of their eigenspaces and present some consequences of it. We then present other miscellaneous results which will be used later. Finally, in Section 4.5 we describe the Killing tensors which have a canonical algebraic decomposition in warped products.

4.1 Hamiltonian mechanics on the Cotangent bundle

We will be working on the cotangent bundle T^*M , which is the natural geometric setting for Hamiltonian mechanics, and Hamilton-Jacobi theory [Arn89]. We assume the reader is familiar with the basic notions of Hamiltonian mechanics on the cotangent bundle T^*M , see [Lee12] for the basics and [Arn89] for more details. We review the basics to fix our notations, following [Woo75] and [Ben89]. We denote the natural projection map by $\pi : T^*M \to M$ which acts on a point $(q, p) \in T^*M$ as $\pi(q, p) = q$. Any local coordinate system (q^i) on M induces coordinates (q^i, p_j) on T^*M , hereafter called *canonical coordinates*. The coordinates (p_j) are called *momenta*.

A Hamiltonian is simply a function $H \in \mathcal{F}(T^*M)$. If M is a pseudo-Riemannian manifold with metric $\langle \cdot, \cdot \rangle$, the natural Hamiltonian H with potential $V \in \mathcal{F}(M)$ is defined by:

$$H(q,p) := \frac{1}{2} \langle p, p \rangle + V(q) \quad (q,p) \in T^*M$$

The geodesic Hamiltonian is obtained by setting $V \equiv 0$ in the above equation.

The Liouville 1-form θ is defined by:

$$\theta_{(q,p)}(X) = p(\pi_*X) \quad (q,p) \in T^*M, \ X \in T_{(q,p)}T^*M$$

In canonical coordinates, $\theta = p_i dq^i$. The canonical symplectic form on T^*M is then $\omega := d\theta$. A crucial property of ω is that it's non-degenerate, i.e. at each point (q, p) the quadratic form ω on $T_{(q,p)}T^*M$ is non-degenerate.

This means to each function $F \in \mathcal{F}(T^*M)$, ω induces a vector field X_F , called the *Hamiltonian vector field* of F defined by the following equation:

$$\omega(X_F, Y) = \mathrm{d}F(Y) = YF, \quad Y \in \mathfrak{X}(T^*M)$$
(4.1.1)

If we take F = H where H is the Hamiltonian, then an integral curve of X_H satisfies the classical Hamilton's equations [LL76] in canonical coordinates (q, p):

$$\dot{q}^i = \frac{\partial H}{\partial p_i} \qquad \qquad \dot{p}_i = -\frac{\partial H}{\partial q^i}$$

For any $F, G \in \mathcal{F}(T^*M)$, the *Poisson bracket* is defined by:

$$\{F,G\} := \omega(X_F, X_G)$$

In canonical coordinates

$$\{F,G\} = \sum_{i=1}^{n} \left(\frac{\partial F}{\partial q^{i}} \frac{\partial G}{\partial p_{i}} - \frac{\partial G}{\partial q_{i}} \frac{\partial F}{\partial p^{i}} \right)$$
(4.1.2)

If the Poisson bracket vanishes identically, then we say the functions F and G Poisson commute. A set of functions which Poisson commute are said to be in *involution*. We say a function $F \in \mathcal{F}(T^*M)$ is a first integral if it satisfies

$$\{F,H\} = 0$$

where H is the Hamiltonian. Note that it follows from Eq. (4.1.1), that first integrals are constant along the integral curves of X_H , i.e. a first integral F satisfies $X_H F = 0$. In particular we note that the Hamiltonian is a first integral, known as the *energy* for natural Hamiltonian systems.

4.2 Definition in terms of Poisson and Schouten brackets and Covariant Derivative

In this section we give three equivalent definitions of a Killing tensor on a pseudo-Riemannian manifold. We first start with the definition on the cotangent bundle.

Poisson bracket Fix canonical coordinates (q, p) for T^*M . Each $K \in S^r(M)$ induces a homogeneous polynomial of the momenta defined by

$$E_K := K^{j_1 \dots j_r} p_{j_1} \dots p_{j_r} \in \mathcal{F}(T^*M)$$
(4.2.1)

This process can be inverted to obtain K from E_K . Indeed, if we denote the dependence of $E_K(p)$ on p explicitly, then observe that for any covectors $p_1, \ldots, p_r \in T_q^*M$, we recover K using the generalized polarization identity:

$$K_{i_1\dots i_r} p_1^{i_1} \dots p_r^{i_r} = \frac{1}{r!} \frac{\partial}{\partial t_1} \cdots \frac{\partial}{\partial t_r} E_K(t_1 p_1 + \dots + t_r p_r)$$
(4.2.2)

We also note that if $K \in S^r(M)$ and $L \in S^t(M)$, then

$$E_{K\odot L} = E_K E_L \tag{4.2.3}$$

A tensor $K \in S^r(M)$ is called a *Killing tensor* on M if E_K is a first integral for the geodesic Hamiltonian on T^*M . Hence from the previous section we see that the function E_K is constant on the trajectories of the geodesic flow.

Schouten bracket We now introduce the Schouten bracket and use it to obtain a condition on M which characterizes when $K \in S^r(M)$ is a Killing tensor. In analogy with Proposition 3.1 in [Mar97], the Schouten bracket (for symmetric contravariant tensors) is defined as follows:

Theorem 4.2.1 (Schouten bracket [Sch53])

There exists a unique \mathbb{R} -bilinear operator, mapping $S(M) \times S(M) \to S(M)$, called the Schouten bracket, denoted by $(P,Q) \mapsto [P,Q]$, and determined by the following properties:

- (a) For $f, g \in S^0(M)$, [f, g] = 0.
- (b) For a vector $X \in S^1(M)$, and $Q \in S(M)$ we have $[X,Q] = \mathcal{L}_X Q$.
- (c) For $P, Q \in S(M)$

$$[P,Q] = -[Q,P]$$

(d) For $P, Q, R \in S(M)$

$$[P, Q \odot R] = [P, Q] \odot R + Q \odot [P, R]$$

The proof of the above fact follows by expanding the tensors in a local basis (see [Mar97] for the details of a rigorous proof). Indeed, in local coordinates (x^i) , for $P \in S^p(M)$ and $Q \in S^q(M)$ one can derive the following:

$$[P,Q]^{i_1\dots i_p j_1\dots j_{q-1}} = pP^{k(i_1\dots i_{p-1})} \partial_k Q^{i_p j_1\dots j_{q-1})} - qQ^{k(j_1\dots j_{q-1})} \partial_k P^{i_1\dots i_p)}$$
(4.2.4)

The Schouten bracket has the following fundamental property.

Theorem 4.2.2 (Schouten and Poisson brackets)

If $K \in S^p(M)$ and $G \in S^q(M)$, the Schouten bracket satisfies the following identity:

$$E_{[K,G]} = -\{E_K, E_G\} \tag{4.2.5}$$

PROOF This result follows by a straightforward calculation using Eq. (4.1.2) and Eq. (4.2.4).

Due to the connection above, the Schouten bracket is often defined in terms of the Poisson bracket (for example, in [Ben89]). With this connection, the properties of the Schouten bracket can be derived from similar properties of the Poisson bracket. We will give an example of this in the proof of the following result.

Proposition 4.2.3 (Properties of the Schouten bracket)

For $P, Q, R \in S(M)$, the Schouten bracket satisfies the following:

- (a) If $P \in S^p(M)$ and $Q \in S^q(M)$ then $[P,Q] \in S^{p+q-1}(M)$
- (b) The Jacobi identity is satisfied:

$$[P, [Q, R]] + [R, [P, Q]] + [Q, [R, P]] = 0$$

(c) If $P \in S^p(M)$ and $f \in \mathcal{F}(M)$ then

$$[P,f]^{i_1\dots i_{p-1}} = pP^{i_1\dots i_{p-1}j}\partial_j f$$
(4.2.6)

PROOF The first property follows from the coordinate formula. The second follows by a direct calculation (see [Nij55] for more details), or using the Jacobi identity for the Poisson bracket (see [Lee12]) and Theorem 4.2.2. The third property, which is a useful fact, follows immediately from the coordinate formula.

In analogy with the Poisson bracket, we say two tensors $K \in S^p(M)$ and $G \in S^q(M)$ Schouten commute if they satisfy [K, G] = 0. By Theorem 4.2.2 and the generalized polarization identity (Eq. (4.2.2)), we see that a tensor $K \in S^p(M)$ is a Killing tensor (KT) iff it satisfies:

$$[K,G] = 0$$

where G is the (inverse) contravariant metric. If p = 1, then K is called a *Killing vector* (KV) and the above equation reduces to $\mathcal{L}_K G = 0$ (see Theorem 4.2.1). The properties of the Schouten bracket imply that Killing tensors form a Lie algebra (with respect to the Schouten bracket) which is closed under the symmetric product.

We conclude with some historical remarks on the Schouten bracket. The Schouten bracket was discovered originally by Schouten in [Sch53]. Its properties have been studied by his student Nijenhuis in [Nij55]. In Schouten's original work he introduced a more general bracket defined on the space of contravariant tensors, also called the *Schouten bracket*. This bracket naturally breaks down into two brackets, one for symmetric tensors discussed above, and another for anti-symmetric tensors. Indeed, if P, Q are contravariant tensors with symmetric and anti-symmetric parts P_s, Q_s and P_a, Q_a respectively, then the Schouten bracket can be written [Nij55]:

$$[P,Q] = [P_s,Q_s] + [P_a,Q_a]$$

where $[P_s, Q_s]$ is the Schouten bracket for symmetric tensors and $[P_a, Q_a]$ is the one for anti-symmetric tensors. The one for anti-symmetric tensors satisfies a theorem similar to Theorem 4.2.1 (see Proposition 3.1 in [Mar97]), the main difference being that the symmetric product is replaced with the wedge product. The Schouten bracket for anti-symmetric tensors appears more often in the literature because of its role in the coordinate-independent definition of a Poisson manifold.

Levi-Civita Connection We will give our last definition of a Killing tensor in terms of the Levi-Civita connection, ∇ , of the metric g. First we need the following fact [Woo75]. **Proposition 4.2.4 (Schouten bracket and Connections)**

If $P \in S^p(M)$ and $Q \in S^q(M)$, and ∇ is a torsion-free connection, then the Schouten bracket has the following form:

$$[P,Q]^{i_1\dots i_p j_1\dots j_{q-1}} = pP^{k(i_1\dots i_{p-1}}\nabla_k Q^{i_p j_1\dots j_{q-1}}) - qQ^{k(j_1\dots j_{q-1}}\nabla_k P^{i_1\dots i_p)}$$

PROOF The proof follows by a straightforward calculation. Fix a local coordinate system (x^i) and let Γ^i_{kl} be the Christoffel symbols of the connection ∇ . Then observe that for $K \in S^q(M)$ we can write:

$$\nabla_i K^{j_1 \dots j_q} = \partial_i K^{j_1 \dots j_q} + \sum_r \Gamma^{j_r}{}_{il} K^{j_1 \dots l_m j_q}$$
$$= \partial_i K^{j_1 \dots j_q} + q \Gamma^{j_q}{}_{il} K^{j_1 \dots j_{q-1}l}$$

Thus

$$\begin{split} [P,Q]^{i_1\dots i_p j_1\dots j_{q-1}} &= pP^{k(i_1\dots i_{p-1})}\partial_k Q^{i_p j_1\dots j_{q-1})} - qQ^{k(j_1\dots j_{q-1})}\partial_k P^{i_1\dots i_p)} \\ &= pP^{k(i_1\dots i_{p-1})}\nabla_k Q^{i_p j_1\dots j_{q-1})} - qQ^{k(j_1\dots j_{q-1})}\nabla_k P^{i_1\dots i_p)} \\ &- qpP^{k(i_1\dots i_{p-1})}\Gamma^{i_p}{}_{kl}Q^{j_1\dots j_{q-1})l} + qpQ^{k(j_1\dots j_{q-1})}\Gamma^{i_p}{}_{kl}P^{i_1\dots i_{p-1})l} \\ &= pP^{k(i_1\dots i_{p-1})}\nabla_k Q^{i_p j_1\dots j_{q-1})} - qQ^{k(j_1\dots j_{q-1})}\nabla_k P^{i_1\dots i_p)} \\ \blacksquare \end{split}$$

We have the following corollary.

Corollary 4.2.5

If $K \in S^q(M)$ and G is the (inverse) contravariant metric, then:

$$[K,G]^{i_1\dots i_{q-1}j_k} = -2\nabla^{(j}K^{ki_1\dots i_{q-1})}$$
(4.2.7)

Equation (4.2.7) implies that $K \in S^p(M)$ is a Killing tensor iff the following is satisfied:

$$\nabla_{(i}K_{i_1\dots i_p)} = 0$$

This is the standard definition of a Killing tensor. Using the above equation, we can give another characterization of Killing tensors as follows.

Theorem 4.2.6 (Killing tensors and constants of motion)

A symmetric tensor $K \in S^p(M)$ is a Killing tensor iff for any unit speed geodesic $\gamma(t)$, the quantity

$$K_{i_1\dots i_p}\dot{\gamma}^{i_1}\dots\dot{\gamma}^{i_p}$$

is a constant along the geodesic.

PROOF This follows from a straightforward calculation. The converse follows with the help of the generalized polarization identity (Eq. (4.2.2)).

4.3 Conformal Killing tensors and special classes

The conformally covariant generalization of a Killing tensor is known as *conformal Killing* tensor (CKT). A tensor $K \in S^p(M)$ is said to be a conformal Killing tensor of valence p if there exists $C \in S^{p-1}(M)$ (called the *conformal factor*) such that

$$[K,G] = -2C \odot G$$

As an immediate consequence of the definition, one can check that a Killing tensor is a conformal Killing tensor on any conformally related manifold. In analogy with Killing tensors, we have the following equivalent characterizations of a conformal Killing tensor:

Theorem 4.3.1 (Characterizations of conformal Killing tensors)

Suppose $K \in S^p(M)$, and let $E_K \in \mathcal{F}(T^*M)$ denote the corresponding homogeneous polynomial on T^*M and H be the geodesic Hamiltonian. The following are equivalent:

- (a) K is a conformal Killing tensor with conformal factor $C \in S^{p-1}(M)$.
- (b) On T^*M we have $\{E_K, H\} = 2E_CH$.
- (c) With respect to the Levi-Civita connection, ∇ , we have:

$$\nabla_{(i}K_{ji_1\dots i_{p-1})} = C_{(i_1\dots i_{p-1}}g_{ij)} \tag{4.3.1}$$

PROOF The equivalence of the first and second characterizations follows from Eqs. (4.2.3) and (4.2.5). The equivalence of the first and third characterizations follows from Eq. (4.2.7).

In the literature, CKTs are often assumed to be traceless. This is due to the fact that for any $f \in \mathcal{F}(M)$, fG is a CKT. Though we do not make this assumption.

An important class of CKTs are those of valence two for which the conformal factor $C = \nabla f$ for some $f \in \mathcal{F}(M)$. If L is such a CKT, then it is said to be of gradient-type, and one can show that the following tensor is a KT:

$$K = fG - L$$

If in addition, f = tr(L), then it said to be of *trace-type*.

4.4 Orthogonal conformal Killing tensors

In this section we study the most important class of conformal Killing tensors for our purposes, the *orthogonal conformal Killing tensors*. An orthogonal conformal Killing tensor, is a conformal Killing tensor which is also an orthogonal tensor. In the remainder of this chapter, all tensors are assumed to be of valence two.

For now, by a conformal Killing tensor we will mean an orthogonal conformal Killing tensor. In this section we first present a formulation of the conformal Killing equation in terms of the eigenspaces of a conformal Killing tensor given in [CFS06]. This formulation will be the most useful in our study. We then use the theory of twisted and warped products given in [MRS99] (which is reviewed in Chapter 3) to show that an orthogonal conformal Killing tensor naturally induces a twisted product, and then derive the well known conformal Killing equation in the eigenframe. We then give necessary and sufficient conditions on an eigenfunction of the tensor for the associated eigenspace to be geodesic or Killing. We then end the section with some miscellaneous results on CKTs which will be applied in the next chapter.

We denote $\{x, y\} := \frac{1}{2}(\nabla_x y + \nabla_y x)$ for $x, y \in \mathfrak{X}(M)$ which is sometimes called the Jordan bracket [Rov98]. Note that the following statements are made for a CKT with conformal factor t, hence the corresponding statements for KTs can be obtained by setting t = 0.

Lemma 4.4.1 ([CFS06])

Let E_{λ} be a non-degenerate eigenspace of a CKT, T, associated with eigenfunction λ . Then the following equation holds for all $x, y \in \Gamma(E_{\lambda})$

$$(T - \lambda I) \{x, y\} = \frac{1}{2} \langle x, y \rangle (\nabla \lambda - t)$$

where t is the conformal factor of T. Moreover $\nabla \lambda - t \in \Gamma(E_{\lambda}^{\perp})$. The following equation holds for eigenvectors x, y, z with different eigenfunctions

$$T(\{x,y\},z) + T(\{z,x\},y) + T(\{y,z\},x) = 0$$

PROOF We give some details of the proof, following [CFS06], by using Eq. (4.3.1) as the defining equation of a CKT.

Suppose $x, y \in \Gamma(E_{\lambda})$ and $z \in \mathfrak{X}(M)$. A direct calculation shows the following:

$$3! \nabla_{(i} T_{jk)} x^{i} y^{j} z^{k} = 2(x(\lambda)g(y,z) + y(\lambda)g(x,z) + z(\lambda)g(x,y) - 2(T_{jk} - \lambda g_{jk})\{x,y\}^{j} z^{k})$$

$$3! g \odot t(x,y,z) = 2(g(x,y)t(z) + g(z,x)t(y) + g(y,z)t(x))$$

If we take x = y = z, then equating the above equations implies that:

$$x^2 g(\nabla \lambda - t, x) = 0$$

By non-degeneracy of E_{λ} , we see that $\nabla \lambda - t \in \Gamma(E_{\lambda}^{\perp})$. The first equation in the lemma then follows by equating the first set of equations. The second equation in the lemma holds since if x, y, z have different eigenfunctions, then

$$3! \nabla_{(i} T_{jk)} x^{i} y^{j} z^{k} = -2(T(\{x, y\}, z) + T(\{z, x\}, y) + T(\{y, z\}, x))$$

$$3! g \odot t(x, y, z) = 0$$

The following proposition is an immediate consequence of the above lemma.

Proposition 4.4.2 ([CFS06])

Let T be an orthogonal tensor and let E_i be the eigenspaces corresponding to the eigenfunctions λ_i . Then T is a conformal Killing tensor with conformal factor t iff

- 1. $(T \lambda_i I)\{x, y\} = \frac{1}{2} \langle x, y \rangle (\nabla \lambda_i t) \text{ for all } x, y \in \Gamma(E_i)$
- 2. $T(\{x, y\}, z) + T(\{z, x\}, y) + T(\{y, z\}, x) = 0$ for eigenvectors x, y, z with different eigenfunctions

The following theorem gives an equivalent characterization of condition 1 of the above proposition which will allow us to more directly study the geometrical properties of CKTs. It was originally Theorem 2 in [CFS06]. Before we state it, we remind the reader of some notation. Given a collection of distributions $(E_i)_{i=1}^k$ satisfying $TM = \bigoplus_{i=1}^k E_i$, then for any vector $x \in \mathfrak{X}(M)$, we have the orthogonal splitting $x = \sum_i x^i$ where each $x^i \in \Gamma(E_i)$.

Theorem 4.4.3 (Geometric Characterization of Orthogonal CKTs [CFS06]) Let T be an orthogonal tensor and let E_i be the eigenspaces corresponding to the eigenfunctions λ_i . Then T is a conformal Killing tensor with conformal factor t iff

- 1. The eigenspaces E_i are almost umbilical.
- 2. The mean curvature normals of the eigenspaces satisfy the following equation:

$$H_i = -\frac{1}{2} \sum_{j \neq i} (\nabla \log |\lambda_i - \lambda_j|)^j$$
(4.4.1)

3. The conformal factor satisfies the following equation:

$$t = \sum (\nabla \lambda_i)^i$$

4. $T(\{x, y\}, z) + T(\{z, x\}, y) + T(\{y, z\}, x) = 0$ for eigenvectors x, y, z with different eigenfunctions

Remark 4.4.4

For a Killing tensor the second condition can be simplified to:

$$H_i = -\frac{1}{2} \sum_{j \neq i} \frac{1}{(\lambda_i - \lambda_j)} (\nabla \lambda_i)^j \qquad \Box$$

PROOF By projecting condition 1 in Proposition 4.4.2 onto E_i , we see that $(\nabla \lambda_i - t)^i = 0$, thus

$$t = \sum (\nabla \lambda_i)^i \tag{4.4.2}$$

By projecting condition 1 in Proposition 4.4.2 onto every eigenspace E_j with $j \neq i$, one obtains for $x, y \in \Gamma(E_i)$

$$(\lambda_j - \lambda_i) \{x, y\}^j = \frac{1}{2} g(x, y) (\nabla \lambda_i - t)^j$$

Then summing over $j \neq i$, one obtains

$$\{x, y\}^{\perp i} = g(x, y) \sum_{j \neq i} \frac{1}{2(\lambda_j - \lambda_i)} (\nabla \lambda_i - t)^j$$

Since $t = \sum (\nabla \lambda_i)^i$ the above equation can be written

$$\{x, y\}^{\perp i} = g(x, y) \sum_{j \neq i} \frac{1}{2(\lambda_j - \lambda_i)} (\nabla \lambda_i - \nabla \lambda_j)^j$$
$$= \frac{-g(x, y)}{2} \sum_{j \neq i} (\nabla \log |\lambda_i - \lambda_j|)^j$$

Thus if h_i denotes the second fundamental form of E_i , the above equation is equivalent to the following:

$$h_i(x,y) = \{x,y\}^{\perp i} = g(x,y)H_i$$
(4.4.3)

where H_i is given by Eq. (4.4.1). This last equation is equivalent to saying that E_i is almost umbilical with mean curvature normal H_i . Equations (4.4.2) and (4.4.3) are equivalent to condition 1 in Proposition 4.4.2 as we just projected that condition onto all the eigenspaces to derive the above equations. Hence by Proposition 4.4.2, the theorem is proven.

We will now proceed to show that when the eigenspaces are orthogonally integrable,
Condition 4 of the above theorem is automatically satisfied. The following lemma can be deduced from a knowledge of rotation coefficients, although we state it for completeness.

Lemma 4.4.5

Suppose $(E_i)_{i=0}^k$ is an integrable net. Then for $x \in \Gamma(E_i)$ and $y \in \Gamma(E_j)$ with $j \neq i$, $\nabla_x y \in \Gamma(E_i \oplus E_j)$.

PROOF Suppose $z \in \Gamma(E_k)$ where k is different from i, j. Observe that

$$g(\nabla_x y, z) - g(\nabla_y x, z) = g([x, y], z) = 0$$

Also

$$g(\nabla_y x, z) + g(x, \nabla_y z) = \nabla_y g(x, z) = 0$$

The above two equations hold for all permutations of x, y, z. Thus

$$g(\nabla_x y, z) = g(\nabla_y x, z) = -g(x, \nabla_y z) = -g(x, \nabla_z y)$$
$$= g(\nabla_z x, y) = g(\nabla_x z, y) = -g(z, \nabla_x y)$$

Thus $g(\nabla_x y, z) = 0.$

The following corollary gives a version of the above theorem for orthogonal tensors with orthogonally integrable eigenspaces.

Corollary 4.4.6

Suppose T is an orthogonal tensor with orthogonally integrable eigenspaces and let E_i be the eigenspaces corresponding to the eigenfunctions λ_i . Then T is a conformal Killing tensor with conformal factor t iff

- 1. The eigenspaces E_i are umbilical.
- 2. The mean curvature normals of the eigenspaces satisfy the following equation:

$$H_i = -\frac{1}{2} \sum_{j \neq i} (\nabla \log |\lambda_i - \lambda_j|)^j$$
(4.4.4)

3. The conformal factor satisfies the following equation:

$$t = \sum (\nabla \lambda_i)^i \tag{4.4.5}$$

Remark 4.4.7

The Haantjes theorem (Theorem B.0.19) gives a simple necessary and sufficient condition to check if an orthogonal tensor has orthogonally integrable eigenspaces. \Box

PROOF Since $A^{E_i} = 0$ for each i, the eigenspaces are almost umbilical iff they are umbilical. Condition 4 of Theorem 4.4.3 is automatically satisfied due to Lemma 4.4.5, hence the result holds by Theorem 4.4.3.

Now we use a result from [MRS99] which characterizes twisted products to show that orthogonally integrable CKTs naturally give rise to a twisted product structure.

Corollary 4.4.8 (Conformal Killing tensors induce twisted product nets)

Suppose T is an orthogonal tensor with orthogonally integrable eigenspaces $(E_i)_{i=1}^k$ and associated eigenfunctions $(\lambda_i)_{i=1}^k$. Let $M = \prod_{i=1}^k M_i$ be a connected product manifold locally adapted to the eigenspaces of T. Then T is a CKT iff (M,g) is a twisted product with twist functions ρ_i satisfying the following equation:

$$(\nabla \log \rho_i)^{\perp i} = \frac{1}{2} \sum_{j \neq i} (\nabla \log |\lambda_i - \lambda_j|)^j \tag{4.4.6}$$

PROOF This result follows from the above corollary together with Theorem 3.5.10(1) and Proposition 3.5.9(2).

Remark 4.4.9

As a direct consequence of the above two corollaries, we have the following. In local coordinates (x^i) , a tensor T diagonalized in these coordinates with eigenfunctions (λ^i) (counted with multiplicity) is a conformal Killing tensor with conformal factor t iff the following equations are satisfied:

$$\partial_i \lambda_j = (\lambda_i - \lambda_j) \partial_i \log \left| g^{jj} \right| + t_i \qquad \qquad \partial_i \lambda_i = t_i \qquad (4.4.7)$$

Later on, we will use the above corollary to show how to encode the orthogonal separation of the Hamilton-Jacobi equation in terms of Killing tensors.

The above corollary motivates us to define a *Killing net* (K-net) (resp. *Conformal Killing net* (CK-net)) as the TP-net formed by the eigenspaces of a Killing tensor (resp. conformal Killing tensor) when the eigenspaces are orthogonally integrable. The following lemma shows that CK-nets are a special class of TP-nets. In particular, it will give us a simple way to check when an eigenspace of a CKT is Killing.

Lemma 4.4.10

Suppose $(E_i)_{i=1}^k$ is an orthogonally integrable CK-net and let λ_i be the associated eigenfunctions. If E_i^{\perp} is geodesic, then E_i is spherical.

PROOF Suppose $x \in \hat{\mathfrak{X}}(M_i)$ and $y \in \hat{\mathfrak{X}}(M_j)$ where $j \neq i$. Recall that this implies [x, y] = 0. Then by Eq. (4.4.4)

$$x \langle H_i, y \rangle = -\frac{1}{2} x \langle \nabla \log |\lambda_i - \lambda_j|, y \rangle$$
$$= -\frac{1}{2} x y \log |\lambda_i - \lambda_j|$$
$$= -\frac{1}{2} y x \log |\lambda_i - \lambda_j|$$
$$= -\frac{1}{2} y \langle \nabla \log |\lambda_i - \lambda_j|, x \rangle$$
$$= y \langle H_i, x \rangle$$

Now, since E_i^{\perp} is geodesic, one can show that $H_j^i = 0$ for $j \neq i$. This can be seen for example, by working in a local twisted product given by Corollary 4.4.8 and then using Proposition 3.5.9 (3). Hence by the above calculation, $x \langle H_i, y \rangle = y \langle H_j, x \rangle = y \langle H_j^i, x \rangle = 0$. Thus

$$\langle \nabla_x H_i, y \rangle = x \langle H_i, y \rangle - \langle H_i, \nabla_x y \rangle$$

$$= - \langle H_i, \nabla_y x \rangle$$

$$= \langle \nabla_y H_i, x \rangle - y \langle H_i, x \rangle$$

$$= \langle \nabla_y H_i, x \rangle$$

$$= 0$$

where the last line follows since E_i^{\perp} is geodesic. Hence $\langle \nabla_x H_i, y \rangle = 0$ for all $x \in \Gamma(E_i)$ and $y \in \Gamma(E_i^{\perp})$, thus E_i is spherical.

The following corollary allows us to determine the geometry of the eigenspaces of a CKT with orthogonally integrable eigenspaces using its eigenfunctions.

Corollary 4.4.11

Suppose T is a CKT with conformal factor t and orthogonally integrable eigenspaces $(E_i)_{i=1}^k$.

1. E_i is Killing iff

$$(\nabla \lambda_j)^i = t^i \quad for \ all \ j \neq i$$

2. E_i is geodesic iff

$$(\nabla \lambda_i)^j = t^j \quad for \ all \ j \neq i$$

In particular for a KT, E_i is Killing iff all the eigenfunctions are independent of E_i and E_i is geodesic iff λ_i is a constant.

PROOF This follows from the above lemma together with Corollary 4.4.6 and the definitions of Killing and geodesic distributions.

From the above corollary, it follows immediately that if M admits a KT with orthogonally integrable eigenspaces $\mathcal{E} = (E_i)_{i=0}^k$ and respective eigenfunctions $(\lambda_i)_{i=0}^k$ such that λ_0 is constant and λ_i depends only on E_0 for each i > 0, then \mathcal{E} is a WP-net. One can easily use Corollary 4.4.6 and Corollary 4.4.8 to show conversely that any WP-net admits a KT. Although in the next section, we will give a different proof of this fact (see Corollary 4.5.3).

We can also deduce the following important fact from Corollary 4.4.8 [Ben93].

Proposition 4.4.12

Let K, J be Killing tensors. Suppose there exists an orthogonal web $(E_i)_{i=1}^n$ such that K and J are simultaneously diagonalized in any coordinates adapted to this web. Then

$$[K, J] = 0 \qquad \qquad \square$$

PROOF The proof follows by a straightforward calculation. Let (x^i) be local coordinates adapted to $(E_i)_{i=1}^n$ and (λ_i) (resp. (μ_i)) be the eigenfunctions of K (resp. J) counted with multiplicity. In these coordinates the remark following Corollary 4.4.8 gives the equations satisfied by the eigenfunctions of these KTs. Using Eq. (4.2.4), we calculate the possibly non-zero terms of the Schouten bracket as follows¹:

$$\begin{split} K^{jj}\partial_{j}J^{kk} - J^{jj}\partial_{j}K^{kk} &= g^{jj}(\lambda_{j}\partial_{j}(\mu_{k}g^{kk}) - \mu_{j}\partial_{j}(\lambda_{k}g^{kk})) \\ &= g^{jj}(\lambda_{j}[(\partial_{j}\mu_{k})g^{kk} + \mu_{k}\partial_{j}g^{kk}] - \mu_{j}[(\partial_{j}\lambda_{k})g^{kk} + \lambda_{k}\partial_{j}g^{kk}]) \\ &= g^{jj}g^{kk}(\lambda_{j}[\partial_{j}\mu_{k} + \mu_{k}\partial_{j}\log|g^{kk}|] - \lambda_{j}\mu_{j}\partial_{j}\log|g^{kk}| \\ &+ \lambda_{j}\mu_{j}\partial_{j}\log|g^{kk}| - \mu_{j}[\partial_{j}\lambda_{k} + \lambda_{k}\partial_{j}\log|g^{kk}|]) \\ &= g^{jj}g^{kk}(\lambda_{j}[\partial_{j}\mu_{k} + (\mu_{k} - \mu_{j})\partial_{j}\log|g^{kk}|] \\ &- \mu_{j}[\partial_{j}\lambda_{k} + (\lambda_{k} - \lambda_{j})\partial_{j}\log|g^{kk}|]) \\ &\stackrel{(4.4.7)}{=} 0 \end{split}$$

¹Note that there is no sum on the index j.

Thus it follows by Eq. (4.2.4) that K and J Schouten commute.

The above proposition has a converse, given as follows.

Proposition 4.4.13 ([KMJ80])

Suppose K_1, \ldots, K_n are point-wise independent Killing tensors which pair-wise Schouten commute. Suppose $\mathcal{E} = (E_i)_{i=1}^n$ is an orthogonal net which simultaneously diagonalizes these tensors. Then \mathcal{E} is integrable, i.e. it is an orthogonal web.

Remark 4.4.14

This fact was originally discovered in [KMJ80]. A proof of this result can be found in [BCR02], where it was shown that the assumption that the tensors are Killing tensors is redundant. An obvious possible generalization is to replace n in the above proposition with some positive integer $k \leq n$.

If D is a distribution then we denote by $S^p(D)$ the set of symmetric contravariant tensors of valence p over D, i.e. each element $T \in S^p(D)$ can locally be written as a sum of p-fold symmetrized products of elements in $\Gamma(D)$. The following proposition on restriction of CKTs to submanifolds will be of use later on.

Proposition 4.4.15 (Restriction of CKTs to Invariant Submanifolds)

Let T be a CKT with conformal factor t and suppose D is an integrable non-degenerate T-invariant distribution. If \tilde{M} is an integral manifold of D regarded as a pseudo-Riemannian manifold with the induced metric, then T restricts to a CKT on \tilde{M} with the induced conformal factor.

PROOF By hypothesis $TM = D \oplus D^{\perp}$, hence we can write

$$T = T_D + T_{D^{\perp}}$$
$$t = t_D + t_{D^{\perp}}$$
$$G = G_D + G_{D^{\perp}}$$

Let $\iota: \tilde{M} \to M$ be the inclusion map, then note that $T_D = \iota_* \tilde{T}$ for some $\tilde{T} \in S^2(\tilde{M})$. Similar equations hold for t_D and G_D . Thus we observe that the following equation holds over \tilde{M} , $[T_D, G_D] = [\iota_* \tilde{T}, \iota_* \tilde{G}] = \iota_* [\tilde{T}, \tilde{G}]$ by naturality of the Schouten bracket. In particular, we see that $[T_D, G_D] \in S^2(D)$. Now

$$[T,G] = [T_D,G_D] + [T_D,G_{D^{\perp}}] + [T_{D^{\perp}},G_D] + [T_{D^{\perp}},G_{D^{\perp}}]$$

also

$$t \odot G = t_D \odot G_D + t_D \odot G_{D^\perp} + t_{D^\perp} \odot G_D + t_{D^\perp} \odot G_{D^\perp}$$

By projecting onto $S^2(D)$ we find that $[T_D, G_D] = -2t_D \odot G_D$, thus $[\tilde{T}, \tilde{G}] = -2\tilde{t} \odot \tilde{G}$ by injectivity of ι_* .

4.5 Killing tensors in Warped Products

In the previous section we have seen that a multidimensional eigenspace of an orthogonally integrable Killing tensor is necessarily umbilical. The ideal case where this eigenspace is Killing is amenable to analysis. We will also see later on that this case is important to the study of certain Killing tensors in spaces of constant curvature.

In this section we give conditions under which a tensor $K \in S^2(M)$ that admits a K-invariant Killing distribution is a KT. Our first application of this result is to find necessary and sufficient conditions for extending Killing tensors defined on the geodesic and spherical factors of a warped product. These results are very useful for constructing Killing tensors.

The following lemma won't be directly used but it's useful to keep it in mind for proofs to come.

Lemma 4.5.1 (Schouten bracket on Product Manifolds)

Let $M = B \times F$ be a product manifold and suppose $K \in \hat{S}^p(B)$, $G \in \hat{S}^q(F)$. Then the following holds:

$$[K,G] = 0 \qquad \qquad \square$$

PROOF This follows from the naturality of the Schouten bracket, i.e. the proof is similar to that when K and G are vector fields.

In the follow proposition we will characterize KTs in warped products.

Proposition 4.5.2 (Killing tensors in Warped Products)

Suppose $K \in S^2(M)$ and D is a K-invariant Killing distribution. Let $B \times_{\rho} F$ be a local warped product adapted to the WP-net (D^{\perp}, D) with contravariant metric $G = G_0 + \kappa G_1$ where $\kappa := \rho^{-2}$.

Then K is a KT iff there exist KTs $K' \in S^2(B)$, $\tilde{K} \in S^2(F)$ and $t \in \mathcal{F}(B)$ such that the following equations hold:

$$K = K' + tG_1 + \tilde{K}$$

$$dt = K' d\kappa$$
(4.5.1)

 \diamond

Furthermore \tilde{K} is also a KT on $B \times_{\rho} F$.

PROOF By hypothesis, we can write $K = K_0 + K_1$ where $K_0 \in S^2(D^{\perp})$ and $K_1 \in S^2(D)$. Thus,

$$[K,G] = [K_0 + K_1, G_0 + \kappa G_1]$$

= $[K_0, G_0] + [K_0, \kappa G_1] + [K_1, G_0] + [K_1, \kappa G_1]$
$$\stackrel{(4.2.6)}{=} [K_0, G_0] + \kappa [K_0, G_1] + 2K_0(\mathrm{d}\kappa) \odot G_1 + [K_1, G_0] + \kappa [K_1, G_1]$$

Note that $[K_0, G_0] \in S^3(D^{\perp})$ (see Eq. (4.2.4)), then by linear independence, [K, G] = 0 iff

$$[K_0, G_0] = 0 \tag{4.5.2}$$

$$[K_1, G_1] = 0 \tag{4.5.3}$$

$$\kappa[K_0, G_1] + 2K_0(\mathrm{d}\kappa) \odot G_1 + [K_1, G_0] = 0 \tag{4.5.3}$$

Suppose $(x^i) = (x^a, x^{\alpha})$ are local coordinates adapted to the warped product $B \times_{\rho} F$. We denote coordinates for B using Latin letters such as a, b, coordinates for F using Greek letters such as α, β and the letters i, j, k are reserved for generic indices. Let $X_i := \partial_i$, then by Eq. (4.2.4) we have:

$$[K_1, G_0] = 2(K_1 dG_0^{jk} - G_0 dK_1^{jk}) \odot X_j \odot X_k$$
$$= -2G_0 dK_1^{\alpha\beta} \odot X_\alpha \odot X_\beta$$

and

$$[K_0, G_1] = 2(K_0 \mathrm{d} G_1^{jk} - G_1 \mathrm{d} K_0^{jk}) \odot X_j \odot X_k$$
$$= -2G_1 \mathrm{d} K_0^{ab} \odot X_a \odot X_b$$

Thus by linear independence, Eq. (4.5.3) is satisfied iff

$$[K_0, G_1] = 0$$

2K_0(d\kappa) \odot G₁ + [K₁, G₀] = 0

The first of the above equations are satisfied iff $G_1 dK_0^{ab} = 0$, i.e. $K_0 \in \hat{S}^2(B)$. The second becomes

$$2K_0(\mathrm{d}\kappa) \odot G_1 + [K_1, G_0] = 2(K_0(\mathrm{d}\kappa)G_1^{\alpha\beta} - G_0\mathrm{d}K_1^{\alpha\beta}) \odot X_\alpha \odot X_\beta$$

which is identically zero iff

$$d^{0}K_{1}^{\alpha\beta} = K_{0}(d\kappa)G_{1}^{\alpha\beta}$$

$$\Rightarrow d^{0}(K_{0}(d\kappa)) = 0 \text{ by non-degeneracy of } G_{1}$$

where d⁰ is d followed by the (point-wise) orthogonal projection onto $(D^{\perp})^*$. So, $K_0(d\kappa) = d^0 t$ for some $t \in \mathcal{F}(B)$, thus

$$\mathrm{d}^0 K_1^{\alpha\beta} = \mathrm{d}^0(tG_1^{\alpha\beta})$$

Hence $\tilde{K}^{\alpha\beta} := K_1^{\alpha\beta} - tG_1^{\alpha\beta} \in \mathcal{F}(F)$, i.e. $\tilde{K} \in \hat{S}^2(F)$. Equation (4.5.3) is satisfied iff \tilde{K} is a KT on F and Eq. (4.5.2) is satisfied iff K_0 is a KT on B. Finally if we let $K' := K_0$, the result follows. The last statement that \tilde{K} is a KT on $B \times_{\rho} F$ can be readily verified from the above equations.

Two important special cases of the above proposition are the following:

- 1. By taking K', t = 0, we see that $\tilde{K} \in \hat{S}^2(F)$ is a KT on F iff it is a KT on $B \times_{\rho} F$.
- 2. By taking $\tilde{K} = 0$ we find that a necessary and sufficient condition for $K' \in S^2(B)$ to be lifted into a KT on $B \times_{\rho} F$ is that

$$d(K'd\kappa) = 0$$

We can also prove the following corollary cf. [Jel00], which shows that a WP-net is a K-net.

Corollary 4.5.3 (WP-nets always admit KTs)

A pseudo-Riemannian manifold M admits a WP-net $\mathcal{E} = (E_i)_{i=0}^k$ iff there exists a KT, K on M whose eigen-net is \mathcal{E} and the corresponding eigenfunctions λ_i satisfy:

- 1. λ_0 is a constant
- 2. λ_i depends only on E_0 for each i > 0

Furthermore if such a KT exists, then the warping functions can locally be chosen to satisfy the following equation $\rho_i^2 = |\lambda_i - \lambda_0|$ for i > 0.

PROOF If M admits a KT with orthogonally integrable eigenspaces and eigenfunctions satisfying the above conditions, then it follows from Corollary 4.4.11 that its eigenspaces form a WP-net.

Conversely suppose \mathcal{E} is a WP-net, and suppose $G = G_0 + \sum_{i=1}^k \kappa_i G_i$ is an adapted warped product metric. The above proposition shows that each G_i for i > 0 is a KT on M. Hence for each i if we choose $c_i \in \mathbb{R}$, then $K := c_0 G + \sum_{i=1}^k c_i G_i$ is a KT on M. Thus, locally we can always choose the c_i such that K is a KT with eigenspaces equal to \mathcal{E} and clearly the eigenfunctions satisfy the above conditions.

Now if such a KT exists, by Eq. (4.4.6) in Corollary 4.4.8, we have for i > 0

$$(\nabla \log \rho_i^2)^{\perp i} = \sum_{j \neq i} (\nabla \log |\lambda_i - \lambda_j|)^j$$
$$= \nabla \log |\lambda_i - \lambda_0|$$

Thus it follows that locally we can choose the warping functions as stated.

The following corollary follows immediately by inductively applying Proposition 4.5.2. Corollary 4.5.4

Suppose $\mathcal{E} = (D_i)_{i=0}^k$ is a WP-net and K is KT with D_i a K-invariant distribution for i = 1, ..., k. Let $M = M_0 \times_{\rho} \prod_{i=1}^k M_i$ be a local warped product adapted to \mathcal{E} . Then in contravariant form, K can be decomposed as follows:

$$K = K_0 + \sum_{i=1}^k K_i$$

where each $K_i \in \hat{S}^2(M_i)$ is a KT for i = 1, ..., k. Furthermore K_0 is a KT and each D_i is an eigenspace of K_0 for i = 1, ..., k (see Corollary 4.4.11 for more on K_0).

Chapter 5

Hamilton-Jacobi separation via Characteristic Killing tensors

In this chapter we present the geometric theory of the Hamilton-Jacobi equation, its separation, and the intrinsic characterization of separation. In the first section, we introduce Hamilton-Jacobi theory. This section is mainly included for completeness and not necessary to read later chapters. In Section 5.2 we introduce the separation of variables method for the Hamilton-Jacobi equation, and present the Levi-Civita equations which characterize separable coordinates. In the last two sections, we present the intrinsic characterization of separation for geodesic and natural Hamiltonians respectively. In these sections we also add more details for the case when the separable coordinates are decomposable in a warped product. In this chapter we will be using the formalism introduced in Section 4.1. In the first two sections of this chapter, we consider an arbitrary Hamiltonian H. Finally, we note that this chapter can be read fairly lightly if one is not particularly interested in the general theory.

5.1 Hamilton-Jacobi Theory

We present the fundamental results of Hamilton-Jacobi theory. We will not go into much detail, just presenting the results of interest to us. Motivation for this theory can be found in classical references such as [LL76]. The exposition here mainly follows [Woo75], with help from [Ben89] and [Arn89].

Given a 1-form $\phi \in A_1(M)$, we denote by $\Phi : M \to T^*M$, the associated cross-section of T^*M . We observe that by definition, $\pi \circ \Phi = \text{Id.}$ Furthermore ϕ is said to be *closed* if $d\phi = 0$. A closed 1-form $\phi \in A_1(M)$ is called a *solution* of the Hamilton-Jacobi (HJ) equation if [Woo75] $H \circ \Phi = E$

for some $E \in \mathbb{R}$ where H is the Hamiltonian. In canonical coordinates (q, p), locally we can assume that there exists $W \in \mathcal{F}(M)$ such that $\phi = dW$, then the HJ equation takes its usual form:

$$H(q^i, \frac{\partial W}{\partial q^i}) = E$$

A local complete solution of the Hamilton-Jacobi equation is a diffeomorphism Ψ : $U \times V \subseteq M \times \mathbb{R}^n \to T^*M$ (onto some open subset of T^*M), such that for each $v \in V$, the restriction $\Psi_v : U \to T^*M$ defines a closed 1-form $\psi_v \in A_1(M)$ which is a solution of the Hamilton-Jacobi equation [Woo75].

In local canonical coordinates $(q, p), \Psi$ takes the form

$$\Psi(q^i,c^j) = (q^i,\frac{\partial W}{\partial q^i})$$

The condition that Ψ is locally invertible is equivalent to the condition

$$\det(\frac{\partial^2 W}{\partial q^i \partial c_j}) \neq 0$$

which recovers the standard definition of a complete solution [Ben89]. We can now state the central theorem of Hamilton-Jacobi theory:

Theorem 5.1.1 (Jacobi Theorem)

Let Ψ be a local complete solution of the Hamilton-Jacobi equation as above. Then the Hamiltonian admits n functionally independent first integrals F_1, \ldots, F_n which Poisson commute. In fact, if $\pi_2 : U \times V \to V$ is the projection and if $k_1, \ldots, k_n \in \mathcal{F}(V)$ are functionally independent, then the F_i are given explicitly by:

$$F_i = k_i \circ \pi_2 \circ \Psi^{-1}$$

For the proof we need the following lemmas from [Woo75].

Lemma 5.1.2

If X and Y are tangent vectors to T^*M at some point $p \in T^*M$, then the following hold:

$$\omega(\Phi_*\pi_*X, \Phi_*\pi_*Y) = d\phi(\pi_*X, \pi_*Y)$$
$$\omega(\Phi_*\pi_*X, Y) + \omega(X, \Phi_*\pi_*Y) = \omega(X, Y) + d\phi(\pi_*X, \pi_*Y)$$

PROOF To prove the first equation, we note first note that $\Phi^*\theta = \phi$ (θ is the Liouville form). Indeed, for $p \in M$ and $X \in T_pM$ then

$$\Phi^*\theta(X) = \theta(\Phi_*X) = \phi(\pi_*\Phi_*X) = \phi(X)$$

where the last equality follows from the fact that $\pi_* \Phi_* = (\pi \circ \Phi)_* = \text{Id.}$ Thus

$$\omega(\Phi_*\pi_*X, \Phi_*\pi_*Y) = (\Phi^*\omega)(\pi_*X, \pi_*Y)$$
$$= d(\Phi^*\theta)(\pi_*X, \pi_*Y)$$
$$= d\phi(\pi_*X, \pi_*Y)$$

Now, for the second equation we first make the following observation, if $p \in T^*M$ then for any $X, Y \in T_p T^*M$ satisfying $\pi_* X = \pi_* Y = 0$, we have by definition of ω that:

$$\omega(X,Y) = 0$$

We also note that $X' := X - \Phi_* \pi_* X$ satisfies $\pi_* X' = 0$. Hence the above equation applied to the vectors X' and Y' implies:

$$\omega(X,Y) = \omega(\Phi_*\pi_*X,Y) + \omega(X,\Phi_*\pi_*Y) - \omega(\Phi_*\pi_*X,\Phi_*\pi_*Y)$$

The second equation then follows from the above and first equations. ∎ Lemma 5.1.3

If $\phi \in A_1(M)$ is closed and if $F, G \in \mathcal{F}(T^*M)$ satisfy:

$$F \circ \Phi = \text{const},$$
 $G \circ \Phi = \text{const}$

Then $\{F, G\}$ vanishes on $\Phi(M)$.

PROOF This is a consequence of the second equation in the above lemma. Indeed, since ϕ is closed, we have from the above lemma that

$$\{F, G\} = \omega(X_F, X_G) = \omega(\Phi_* \pi_* X_F, X_G) + \omega(X_F, \Phi_* \pi_* X_G)$$
(5.1.1)

Then by hypothesis, for points in $\Phi(M)$, we observe that

$$\omega(\Phi_*\pi_*X_F, X_G) = (\Phi_*\pi_*X_F)g = \pi_*X_F(G \circ \Phi) = 0$$

Similarly, $\omega(X_F, \Phi_*\pi_*X_G) = 0$, hence the result follows from Eq. (5.1.1).

We are now ready to prove the Jacobi theorem:

PROOF (THEOREM 5.1.1) Fix $f, g \in \mathcal{F}(V)$ then let $F := f \circ \pi_2 \circ \Psi^{-1}$ and $G := g \circ \pi_2 \circ \Psi^{-1}$. As usual let H be the Hamiltonian.

Fix $v \in V$. By construction it follows that both F and G are constant on $\Psi_v(M)$. Since Ψ_v is induced by a closed 1-form on M, the above lemma implies that $\{F, G\} = 0$ on $\Psi_v(M)$. By assumption, $H \circ \Psi_v$ is constant, hence the same argument shows that $\{F, H\} = 0$ on $\Psi_v(M)$. Since Ψ is a bijection onto its image, $\operatorname{Im}(\Psi)$, $\{F, G\} = \{F, H\} = 0$ on $\operatorname{Im}(\Psi)$.

The conclusions of the theorem immediately follow from these observations.

A natural question arises: Under what conditions does the Jacobi theorem have a converse? More precisely, if $F_1, \ldots, F_n \in \mathcal{F}(T^*M)$ are functionally independent commuting first integrals, then when do these integrals arise from a complete solution of the Hamilton-Jacobi equation?

It turns out that a necessary and sufficient condition is that $\{\pi_*X_{F_1}, \ldots, \pi_*X_{F_n}\}$ are point-wise independent [Woo75]. Motivated by this, we say that functions $F_1, \ldots, F_n \in \mathcal{F}(T^*M)$ are vertically independent if $\{\pi_*X_{F_1}, \ldots, \pi_*X_{F_n}\}$ are point-wise independent. Note that this condition implies the functions are functionally independent. Furthermore, in canonical coordinates, this condition is equivalent to:

$$\det(\frac{\partial F_i}{\partial p_j}) \neq 0 \tag{5.1.2}$$

To show that this is in fact necessary, suppose Ψ is a complete solution of the HJ equation. If $\varphi := \Psi^{-1} : T^*M \to M \times \mathbb{R}^n$, in canonical coordinates, we see that:

$$\varphi(q,p) = (q, F(q,p))$$

Then by the inverse function theorem, φ is locally invertible iff Eq. (5.1.2) holds.

Conversely, suppose $F_1, \ldots, F_n \in \mathcal{F}(T^*M)$ are functionally independent commuting first integrals. These functions define a foliation, L, whose leaves are given by:

$$L_v = \{ p \in T^*M : F_i(p) = v_i, i = 1, \dots, n \}$$

for a constant vector $v \in \mathbb{R}^n$. The following theorem shows that these integrals arise from a complete solution to the HJ equation:

Theorem 5.1.4 (Complete solutions via First integrals)

If $F_1, \ldots, F_n \in \mathcal{F}(T^*M)$ are vertically independent commuting first integrals, then these integrals arise from a local complete solution to the HJ equation.

PROOF Let *L* be the foliation of T^*M obtained from these functions as defined above. Then since $X_{F_i}F_j = \{F_i, F_j\} = 0$ and because of the functional independence condition, $\{X_{F_1}, \ldots, X_{F_n}\}$ form a (point-wise) basis for *TL*. Take local canonical coordinates (q, p)for T^*M , then consider the map $\varphi : T^*M \to M \times \mathbb{R}^n$ given by

$$\varphi(q,p) = (q, F_1(q,p), \dots, F_n(q,p))$$

By the inverse function theorem, this map is invertible iff

$$\det(\frac{\partial F_i}{\partial p_j}) \neq 0$$

which is precisely the condition that $\{\pi_*X_{F_1}, \ldots, \pi_*X_{F_n}\}$ are point-wise independent. Let $\Psi: M \times \mathbb{R}^n \to T^*M$ be the inverse, then it must have the form

$$\Psi(q, F) = (q, w(q, F))$$

Now observe that for any $v \in \mathbb{R}^n$ that $L_v = \Psi_v(M)$. Hence the symplectic form ω vanishes on $\Psi_v(M)$ since $\omega(X_{F_i}, X_{F_j}) = \{F_i, F_j\} = 0$. It follows by the first equation in Lemma 5.1.2 that the form $w_i dq^i$ is closed. We must have that $H \circ \Psi$ is a constant, since $X_{F_i}H = \{F_i, H\} = 0$. Thus Ψ is a (local) complete solution of the Hamilton-Jacobi equation which induces the integrals F_i .

Liouville first showed that one can relax the vertical independence condition to functional independence in the hypothesis of the above theorem and obtained a method to integrate Hamilton's equations by quadratures [Arn89]. This method for integrating Hamilton's equations is known as *Liouville integrability*. Also, this formulation of a complete solution of the Hamilton-Jacobi equation is more intuitive and will help motivate the intrinsic characterization of separation which will be given later on.

We end this section with a remark on a characterization of a complete solution of the HJ equation in contemporary terms [Ben89]. First, a submanifold of T^*M is called *Lagrangian* if the symplectic form vanishes over it and it has maximal dimension n [Lee12]. The proof of the above theorem shows that the leaves of the foliation induced by the functions $F_1, \ldots, F_n \in \mathcal{F}(T^*M)$ arising from a complete solution to the HJ equation are Lagrangian, level sets of the Hamiltonian, and transverse to the fibers of T^*M . It's an easy exercise to show that this locally characterizes a complete solution to the HJ equation. From this characterization, we can define when two complete solutions of the Hamilton-Jacobi equation are equivalent [Ben91]:

Definition 5.1.5

Two complete solutions of the Hamilton-Jacobi are called *equivalent* if on the subset where they are both defined, their induced Lagrangian foliations coincide. \Box

5.2 Separation of the Hamilton-Jacobi equation

The standard method for solving the Hamilton-Jacobi equation is the method of separation of variables. In this section we will briefly describe precisely what this means and then obtain the Levi-Civita equations, which characterize separable coordinates. In this section we will work in canonical coordinates (q^i, p_j) on T^*M and all considerations are local. Furthermore, we use the following notations:

$$\partial_i = \frac{\partial}{\partial q^i} \qquad \qquad \partial^j = \frac{\partial}{\partial p_j}$$

In canonical coordinates, a local complete solution to the Hamilton-Jacobi equation is a (generating) function W(q, c) where $c = (c_1, \ldots, c_n)$ are constants of integration, satisfying:

$$H(q^i, \frac{\partial W}{\partial q^i}) = E$$

for some $E \in \mathbb{R}$ and the completeness condition:

$$\det(\frac{\partial^2 W}{\partial q^i \partial c_j}) \neq 0$$

Such a solution is called a *separable solution* if it additionally has the form:

$$W = W_1(q^1, c_j) + W_2(q^2, c_j) + \dots + W_n(q^n, c_j)$$
(5.2.1)

The idea behind this ansatz is that, if one can break up the Hamiltonian as follows:

$$H_1(q^1, \frac{\partial W_1}{\partial q^1}) + H_2(q^2, \frac{\partial W_2}{\partial q^2}) + \dots + H_n(q^n, \frac{\partial W_n}{\partial q^n}) = E$$

then one obtains the following system of decoupled ODEs:

$$H_1(q^1, \frac{\partial W_1}{\partial q^1}) = E_1$$

$$\vdots$$
$$H_n(q^n, \frac{\partial W_n}{\partial q^n}) = E_n$$

which can be integrated by quadratures to obtain W, provided $\partial^i H \neq 0$. See [Arn89; LL76] for some classical examples on explicitly separating the HJ equation. This is only the very start of our work, and so examples at this stage are largely irrelevant.

Now the natural question is: when does the Hamilton-Jacobi equation admit a separable solution? We first need a definition:

Definition 5.2.1 (Separable Coordinates)

A coordinate system (q^i) for M is called *separable* if the Hamilton-Jacobi equation admits a separable solution in the induced canonical coordinates (q^i, p_j) on T^*M and $\partial^i H \neq 0$. These coordinates are called *orthogonally separable* if the metric is orthogonal, i.e. it satisfies $g_{ij} = 0$ for $i \neq j$.

An important observation to be made is the following. If (q^i) are separable coordinates, then any coordinate system (\bar{q}^i) having a transformation formula of the form $(\bar{q}^1, \ldots, \bar{q}^n) =$ $(f_1(q^1), \ldots, f_n(q^n))$ is also separable. Hence the separable property is dependent only on the web formed by the coordinates (q^i) . Motivated by this observation, we define a *separable web* to be the orthogonal web formed by orthogonally separable coordinates.

The next step is to obtain the Levi-Civita equations. These equations originally obtained by Levi-Civita in [LC04] give necessary and sufficient conditions to determine if a given coordinate system on M is separable.

Theorem 5.2.2 (Levi-Civita equations [LC04])

Suppose H is a Hamiltonian on T^*M . Let (q^i) be local coordinates for M and (q^i, p_j) be the induced canonical coordinates on T^*M . Then the coordinates (q^i) are separable iff the following equations¹ are satisfied:

$$\partial^{i}H\partial^{j}H\partial_{ij}H + \partial_{i}H\partial_{j}H\partial^{ij}H - \partial^{i}H\partial_{j}H\partial^{j}H - \partial^{j}H\partial_{i}H\partial^{j}H = 0 \quad (i \neq j)$$

which are called the Levi-Civita equations.

PROOF Our proof is a modification of that in [DR07], where it is used in a somewhat different context. See also [Ben91] or [Kal86]. Let W(q, c) be a separable solution of the HJ equation. Then the following equations are satisfied:

$$H(q,p) = E \qquad \qquad p_i = \frac{\partial W}{\partial q^i}$$

Upon differentiating the first of these equations, one obtains:

$$\frac{\partial H}{\partial q^i} + \frac{\partial p_i}{\partial q^i} \frac{\partial H}{\partial p_i} = 0$$

Let $w_i := \frac{\partial W}{\partial q^i}$, then w_i satisfies the following system of PDEs

¹There is no summation over the indices.

$$\frac{\partial w_i}{\partial q^i} = -\frac{\partial_i H}{\partial^i H} = R_i(q, w) \qquad \qquad \frac{\partial w_i}{\partial q^j} = 0 \quad (i \neq j) \tag{5.2.2}$$

This system has a complete solution $w_i(q, c)$ iff the integrability conditions

$$0 = \frac{dR_i}{dq^j} = \frac{\partial R_i}{\partial q^j} + \frac{\partial w_j}{\partial q^j} \frac{\partial R_i}{\partial p_j} = \frac{\partial R_i}{\partial q^j} + R_j \frac{\partial R_i}{\partial p_j} \quad (i \neq j)$$

are satisfied. Upon expanding the right hand side of the above equation, one obtains the Levi-Civita equations. Conversely, assume $w_i(q,c)$ is a complete solution of the system Eq. (5.2.2). Then clearly there exists a function W(q,c) of the form in Eq. (5.2.1) such that $w_i = \frac{\partial W}{\partial q^i}$. The first of Eq. (5.2.2) implies that W is a solution of the HJ equation. Finally, w_i is a complete solution of Eq. (5.2.2) iff

$$\det(\frac{\partial^2 W}{\partial q^i \partial c_j}) \neq 0$$

i.e. W is a complete solution of the HJ equation.

The Levi-Civita equations evaluated with a natural Hamiltonian in orthogonal coordinates (q^i) are equivalent to the following PDEs:

$$\partial_i \partial_j g^{kk} - \partial_i \log \left| g^{jj} \right| \partial_j g^{kk} - \partial_j \log \left| g^{ii} \right| \partial_i g^{kk} = 0 \quad (i \neq j) \tag{5.2.3a}$$

$$\partial_i \partial_j V - \partial_i \log \left| g^{jj} \right| \partial_j V - \partial_j \log \left| g^{ii} \right| \partial_i V = 0 \quad (i \neq j)$$
(5.2.3b)

An important observation to be made here is that separation of the geodesic Hamiltonian is necessary for the separation of a natural Hamiltonian. Thus our main focus will be on the separation of geodesic Hamiltonians. The theory for natural Hamiltonians will be added in afterwards. In the next section we will build on the Levi-Civita equations and obtain an intrinsic characterization of separation for geodesic Hamiltonians.

We now proceed to find an analogue of Theorem 5.1.4 for separable solutions, i.e. characterize these solutions in terms of the first integrals they induce. Benenti has shown in [Ben89, Theorem 2.1] that the correct additional condition is that the integrals be in separable involution. Two first integrals, $F, G \in \mathcal{F}(T^*M)$, are said to be in *separable involution* if there exists coordinates (q^i) on M such that

$$\{F,G\}_i := \partial^i F \partial_i G - \partial_i F \partial^i G = 0 \quad i = 1, \dots, n$$

in the induced canonical coordinates on T^*M . We first show that the integrals generated from a separable solution are in separable involution:

Proposition 5.2.3 ([Ben89])

Suppose $F_1, \ldots, F_n \in \mathcal{F}(T^*M)$ are the first integrals generated by a separable solution to the Hamilton-Jacobi equation. Then with respect to the associated separable coordinates (q^i) , we have that

$$\{F_i, F_j\}_k = 0$$

PROOF See the proof of Theorem 2.1 in [Ben89].

The following theorem is the analogue of Theorem 5.1.4 for separable solutions.

Theorem 5.2.4 (Separable solutions via First integrals [Ben89])

If $F_1, \ldots, F_n \in \mathcal{F}(T^*M)$ are vertically independent first integrals in separable involution with respect to coordinates (q^i) for M, then these coordinates are separable and they generate these integrals via a separable solution to the HJ equation.

PROOF See the proof of Theorem 2.1 in [Ben89].

5.3 Intrinsic characterization of Separation for geodesic Hamiltonians

In this section we consider a geodesic Hamiltonian H. We will present an intrinsic characterization of separation for this Hamiltonian. Eisenhart was the first to obtain this characterization in [Eis34]. Although we will follow a more recent proof of this fact by Benenti in [Ben97] which uses the Levi-Civita equations.

This characterization of separation is motivated by the characterization of a complete solution of the Hamilton-Jacobi equation in terms of commuting first integrals F_1, \ldots, F_n , see Theorem 5.1.4. We assume that the Hamilton-Jacobi equation admits a separable solution. Stackel proved remarkably in [Sta93], that each of these integrals are necessarily quadratic in momenta. Hence, in the notation of Section 4.2, there exist Killing tensors K_1, \ldots, K_n on M such that each $F_i = \frac{1}{2}E_{K_i}$. It was additionally shown in [Sta93] that these KTs are simultaneously diagonalized in the separable coordinates. Note that all the properties satisfied by the integrals F_i in Theorem 5.1.4 translate to properties satisfied by the KTs on M. Indeed, the tensors $K_1, \ldots, K_n \in S^2(M)$:

- 1. Are Killing tensors which pair-wise Schouten commute.
- 2. Are point-wise independent.
- 3. Are simultaneously diagonalized in a coordinate system (q^i) .

_

Note that the last point is due to the separability condition, as mentioned before. It follows by Theorem 5.1.4 that such a set of tensors induce a complete solution to the Hamilton-Jacobi equation on T^*M . We will prove later in this section that the above conditions are sufficient to ensure that the solution is separable. But for now, the key observation to be made is that among the KTs in the vector space spanned by K_1, \ldots, K_n , at least one of them, say K, locally has simple eigenvalues. This follows immediately by using the point-wise independence of these tensors in the coordinate system (q^i) which diagonalizes them. Furthermore the eigenspaces of K form an orthogonal web which is identical to the separable web formed by the separable coordinates. This motivates the following definition:

Definition 5.3.1

A characteristic Killing tensor (ChKT) is Killing tensor with point-wise real simple eigenvalues and orthogonally integrable eigenspaces. \Box

The KT, K, is a ChKT. The following theorem shows that the existence of a ChKT is necessary and sufficient for separation and thereby gives an intrinsic characterization of separation.

Theorem 5.3.2 (Orthogonal Separation of Geodesic Hamiltonians [Eis34])

The geodesic Hamiltonian is separable in an orthogonal web \mathcal{E} iff there exists a ChKT whose eigenspaces form \mathcal{E} .

PROOF Our proof follows that in [Ben97, proposition 3]. Suppose K is a ChKT and (q^i) are coordinates adapted to the eigenspaces of K. If $\lambda_1, \ldots, \lambda_n$ are the eigenfunctions of K, it follows by Eq. (4.4.7) that they satisfy the following equations in these coordinates:

$$\partial_i \lambda_j = (\lambda_i - \lambda_j) \partial_i \log \left| g^{jj} \right| \qquad \qquad \partial_i \lambda_i = 0 \qquad (5.3.1)$$

The integrability conditions of the above system of PDEs are:

$$(\lambda_i - \lambda_j)(\partial_i \partial_j g^{kk} - \partial_i \log \left| g^{jj} \right| \partial_j g^{kk} - \partial_j \log \left| g^{ii} \right| \partial_i g^{kk}) = 0 \quad (i \neq j)$$

Since K has simple eigenfunctions, one observes that the above equations are identical to the Levi-Civita equations for a geodesic Hamiltonian Eq. (5.2.3a). Hence it follows by Theorem 5.2.2 that the coordinates (q^i) are separable iff there exists a ChKT diagonalized in the coordinates.

This gives our first and most fundamental characterization of separation. Although, this characterization is still computationally difficult to work with. One simplification is offered by Haantjes theorem (Theorem B.0.19), which gives a simpler necessary and

sufficient condition on an orthogonal tensor to determine if it has orthogonally integrable eigenspaces. Also on a Riemannian manifold (or in dimensions less than four), one can use the discriminant to check if a linear operator has simple eigenvalues, see for example [Ben04, Theorem 3.6]. In spaces of constant curvature, we will not use this theorem directly to obtain parameterizations of the separable webs. This will depend on deeper insights which we will discuss in later chapters.

We have the following corollary of the proof:

Corollary 5.3.3

Suppose $\mathcal{E} = (E_i)_{i=1}^n$ is a separable web. Then there exists an n dimensional space, \mathcal{K} , of Killing tensors which pair-wise Schouten commute, are point-wise independent and simultaneously diagonalized in \mathcal{E} . A necessary and sufficient condition for an arbitrary KT, K, to be an element of \mathcal{K} is that it is diagonalized in the separable web \mathcal{E} .

Furthermore, the induced quadratic first integrals on T^*M are precisely those guaranteed by the Jacobi theorem.

PROOF In adapted coordinates (q^i) , because the integrability conditions of Eq. (5.3.1) are satisfied, it follows that there exist point-wise independent KTs K_1, \ldots, K_n which are simultaneously diagonalized in (q^i) . It follows by Proposition 4.4.12 that they pair-wise Schouten commute. Furthermore, because of the linearity of Eq. (5.3.1) it follows that the KTs K_1, \ldots, K_n span an n dimensional space \mathcal{K} .

If a KT, K, is diagonalized in the coordinates (q^i) , then by the uniqueness of the solutions to the PDE system (Eq. (5.3.1)) it follows that $K \in \mathcal{K}$.

To prove the last remark, we must show that the first integrals induced by elements of \mathcal{K} arise from a separable solution to the HJ equation. By Theorem 5.2.4, we only need to show that these first integrals are in separable involution. Let $K, J \in \mathcal{K}$ with eigenfunctions $(\lambda_i)_{i=1}^n$ and $(\mu_i)_{i=1}^n$ respectively, and let $F, G \in \mathcal{F}(T^*M)$ be the induced first integrals (see Eq. (4.2.1)). In the induced canonical coordinates (q^i, p_j) on T^*M , we calculate

$$\{F,G\}_{j} = \partial^{j}F\partial_{j}G - \partial_{j}F\partial^{j}G$$

$$= \frac{1}{2}(\lambda_{j}g^{jj}p_{j}\sum_{i=1}^{n}\partial_{j}(\mu_{i}g^{ii})p_{i}p_{i} - \mu_{j}g^{jj}p_{j}\sum_{i=1}^{n}\partial_{j}(\lambda_{i}g^{ii})p_{i}p_{i})$$

$$= \frac{1}{2}\sum_{i=1}^{n}(\lambda_{j}g^{jj}\partial_{j}(\mu_{i}g^{ii}) - \mu_{j}g^{jj}\partial_{j}(\lambda_{i}g^{ii}))p_{j}p_{i}p_{i}$$

$$= \frac{1}{2}\sum_{i=1}^{n}(K^{jj}\partial_{j}(J^{ii}) - J^{jj}\partial_{j}(K^{ii}))p_{j}p_{i}p_{i}$$

$$= \frac{1}{4}\sum_{i=1}^{n}[K, J]^{jii}p_{j}p_{i}p_{i}$$

$$= 0$$

The last equality follows from Proposition 4.4.12.

The vector space of KTs, \mathcal{K} , in the above corollary is called the *KS-space* associated with the separable web \mathcal{E} .

We now give an application of this characterization. It is particularly useful to prove separability of certain warped product metrics. We will consider a generalization of a well known metric from Relativity:

Example 5.3.4 (Separability of The Schwarzschild metric)

This example is from [Ben91, section 5], where more examples from Relativity can be found. Consider the *Reissner-Nordström metric*

$$\mathrm{d}s^2 = \frac{r^2}{\Delta}\mathrm{d}r^2 - \frac{\Delta}{r^2}\mathrm{d}t^2 + r^2(\mathrm{d}\theta^2 + \sin^2\theta\mathrm{d}\phi^2)$$

where $\Delta = r^2 + e^2 - 2mr$, which models the gravitational field outside a spherically symmetric body of charge *e* and mass *m*. The *Schwarzschild metric* is obtained by setting e = 0. We first note that this metric is a warped product

$$\mathbb{E}^1 \times_{\frac{\Delta}{r^2}} \mathbb{E}^1_1 \times_{r^2} \mathbb{S}^2$$

where the 2-sphere \mathbb{S}^2 is equipped with spherical coordinates (θ, ϕ) and metric $g_2 := d\theta^2 + \sin^2\theta d\phi^2$. It is well known that the spherical coordinate system is separable, and hence by Theorem 5.3.2 it admits a ChKT K_2 diagonalized in these coordinates. By Proposition 4.5.2, this ChKT can be lifted to KT on M. Similarly, the contravariant metric G_1 (resp. G_2), of \mathbb{E}_1^1 (resp. \mathbb{S}^2) can be lifted to a KT on M. Hence, locally one can obtain a ChKT (diagonalized in these coordinates) by taking an appropriate linear combination of the KTs in $\mathcal{K} = \text{span}\{G, G_1, G_2, K_2\}$. Thus this metric is separable by Theorem 5.3.2. We also note that \mathcal{K} is the KS-space associated with these separable coordinates.

Motivated by Corollary 5.3.3, using Proposition 4.4.13 we can obtain another intrinsic characterization of separation due originally to [KMJ80]:

Theorem 5.3.5 (Orthogonal Separation of Geodesic Hamiltonians II [KMJ80]) Suppose K_1, \ldots, K_n are point-wise independent Killing tensors which pair-wise Schouten commute. Suppose there exists an orthogonal net $\mathcal{E} = (E_i)_{i=1}^n$ which simultaneously diagonalizes these tensors. Then \mathcal{E} is a separable web.

PROOF From Proposition 4.4.13, we see that \mathcal{E} is an orthogonal web. Using the point-wise independence condition one can construct a ChKT in a neighborhood of each point by taking a constant linear combination of the KTs K_1, \ldots, K_n . The eigenspaces of this ChKT locally form the net \mathcal{E} , hence it follows from Theorem 5.3.2 that \mathcal{E} is separable.

In a Riemannian manifold, the above theorem can be strengthened [KMJ80]:

Corollary 5.3.6

Suppose M is a Riemannian manifold, and K_1, \ldots, K_n are point-wise independent Killing tensors which pair-wise Schouten commute and commute as linear operators. Then locally there exists a separable web $\mathcal{E} = (E_i)_{i=1}^n$ which simultaneously diagonalizes these tensors. \Box

PROOF Since the tensors K_1, \ldots, K_n pair-wise commute as linear operators, they can be simultaneously diagonalized at each point. Using the point-wise independence condition, we can assume that there locally exists an orthogonal net $\mathcal{E} = (E_i)_{i=1}^n$ which simultaneously diagonalizes these tensors. Then the result follows from the above theorem.

The following example shows that the assumption that M is a Riemannian manifold in the above corollary is necessary:

Example 5.3.7 (Complex Separation [DR07])

Let $M = \mathbb{E}_1^2$ with coordinates (t, x). Consider the following contravariant Killing tensors:

$$K_1 := G = \begin{pmatrix} -1 & 0 \\ 0 & 1 \end{pmatrix}$$
 $K_2 := \begin{pmatrix} 0 & 1 \\ 1 & 0 \end{pmatrix}$

One can check that these KTs satisfy the hypothesis of Corollary 5.3.6, but they don't arise from a separable solution because the linear operator associated with K_2 is not diagonalizable.

Note that the Killing tensors in the above example induce first integrals on T^*M which satisfy the hypothesis of Theorem 5.1.4. Hence they arise from a complete solution to the HJ equation. It was shown in [DR07] that one can define a notion of complex separation associated with this complete solution.

We end this section with a remark on the Stäckel form, which was introduced in [Sta93]. He gives a complete (non-intrinsic) characterization of orthogonal separation, by specifying the exact form of the KS-space in separable coordinates in terms of the Stäckel matrix. Thus, he has implicitly obtained the general solution of the system of PDEs given by Eq. (5.3.1). See for example [Ben91, Theorem 3.1] or [Kal86, Stäckel's Theorem] or [Par65] for details. Eisenhart's original solution in [Eis34] was based on Stäckel's work. Although, due to the non-intrinsic nature of Stäckel's results, they are not of much use for our purposes.

5.3.1 Killing-Stackel spaces in Warped Products

In this section we will study the KS-space of a separable web when it's decomposable in a warped product. We wish to further understand the structure of the KS-space associated to a ChKT K which admits a K-invariant Killing distribution. We first need a definition:

Definition 5.3.8

We define the dKdV equation with Killing tensor K and potential² $V \in \mathcal{F}(M)$ as:

$$d(KdV) = 0$$

We will first do some calculations in a more general setting to study the dKdV equation. Suppose K is a KT with orthogonally integrable eigenspaces $(E_i)_{i=1}^k$ with associated eigenfunctions $\lambda_1, ..., \lambda_k$. We work in the local twisted product $\bigcap_{i=1}^{\rho} \prod_{i=1}^k M_i$ adapted to the eigenspaces of K given by Corollary 4.4.8. Fix $x \in \hat{\mathfrak{X}}(M_i)$ and $y \in \hat{\mathfrak{X}}(M_j)$ such that [x, y] = 0, then letting $\sigma_i := \log \rho_i^2$, it follows from Eq. (4.4.6) that the eigenfunctions satisfy

$$x\lambda_j = (\lambda_j - \lambda_i)x\sigma_j$$

Fixing $V \in \mathcal{F}(M)$ and using the above equation we have

 $^{^2\}mathrm{The}$ reason for this terminology will become apparent in the next section.

$$d(KdV)(x,y) = x(K(y,\nabla V)) - y(K(x,\nabla V))$$

= $x(\lambda_j yV) - y(\lambda_i xV)$
= $x\lambda_j yV - y\lambda_i xV + \lambda_j xyV - \lambda_i yxV$
= $(\lambda_j - \lambda_i)x\sigma_j yV - (\lambda_i - \lambda_j)y\sigma_i xV + (\lambda_j - \lambda_i)xyV$
= $(\lambda_j - \lambda_i)(xyV + x\sigma_j yV + y\sigma_i xV)$

Hence we have proven the following:

Lemma 5.3.9 (The dKdV Equation in the eigenframe)

Given K and V as above, $d(KdV) \equiv 0$ iff for each $x \in \hat{\mathfrak{X}}(M_i)$ and $y \in \hat{\mathfrak{X}}(M_j)$ with $i \neq j$ the following holds:

$$xyV + x\log\rho_{i}^{2}yV + y\log\rho_{i}^{2}xV = 0$$
(5.3.2)

From which we can deduce the following:

1. If E_i^{\perp} is geodesic, hence E_i is Killing (see Lemma 4.4.10), we have for all $y \in \hat{\mathfrak{X}}(M_{i\perp})$:

$$y(\rho_i^2 x V) = 0 (5.3.3)$$

2. In particular, if E_i and E_j are Killing and $i \neq j$, we have for $x \in \hat{\mathfrak{X}}(M_i)$ and $y \in \hat{\mathfrak{X}}(M_j)$:

$$xyV = 0$$

PROOF The first equation immediately follows from the above calculations. Now for the consequences, if E_i^{\perp} is geodesic, then $x(\log \rho_j^2) = 0$ for $j \neq i$ by Proposition 3.5.9 (3), hence

$$\begin{split} xyV + x\log\rho_j^2 yV + y\log\rho_i^2 xV &= xyV + y\log\rho_i^2 xV \\ &= xyV + \frac{y\rho_i^2}{\rho_i^2} xV \\ &= \frac{1}{\rho_i^2}(\rho_i^2 xyV + y\rho_i^2 xV) \\ &= \frac{1}{\rho_i^2}y(\rho_i^2 xV) \end{split}$$

Hence $y(\rho_i^2 x V) = 0$. The second statement also follows immediately.

We now obtain a necessary and sufficient condition for extending a Killing-Stäckel space from the geodesic factor of a warped product.

Proposition 5.3.10 (Extending a Killing-Stäckel space into a warped product) Suppose $M = B \times_{\rho} F$ is a warped product and \mathcal{K} is Killing-Stäckel space in B. If there exists a ChKT $K \in \mathcal{K}$ that can be extended into a KT on M (via the method of Proposition 4.5.2) then all KTs in \mathcal{K} can be extended into KTs on M.

PROOF Suppose $K \in \mathcal{K}$ is a ChKT that can be extended into a KT on M. Then from Proposition 4.5.2, K satisfies the dKdV equation with $\kappa = \rho^{-2}$. Then by Eq. (5.3.2) in Lemma 5.3.9 it follows that every $K \in \mathcal{K}$ satisfies the dKdV equation with ρ^{-2} . Hence by Proposition 4.5.2 every $K \in \mathcal{K}$ can be extended into a KT on M.

The above proposition motivates the following notion of a reducible separable web, which is characterized intrinsically by the invariant distributions of an associated ChKT. **Definition 5.3.11 (Reducible separable web)**

Suppose \mathcal{E} is a separable web locally characterized by a ChKT, K. \mathcal{E} is said to be *reducible* if it admits a K-invariant Killing distribution.

First note that since all KTs in the KS-space of a separable web are simultaneously diagonalized, the above definition doesn't depend on the choice of the ChKT, hence is well-defined. The following proposition states clearly why we introduce the notion of a reducible separable web.

Proposition 5.3.12 (The Killing-Stäckel space of a reducible separable web) Suppose K is a ChKT with associated KS-space \mathcal{K} inducing a reducible separable web, i.e. there exists a K-invariant Killing distribution D. Let $M = B \times_{\rho} F$ be a local warped product adapted to the WP-net (D^{\perp}, D) with adapted contravariant metric $G = G_B + \rho^{-2}G_F$. Then there are KS-spaces \mathcal{K}_B and \mathcal{K}_F on B and F respectively such that $L \in \mathcal{K}$ iff there exists $L_B \in \mathcal{K}_B, L_F \in \mathcal{K}_F$ and $l \in \hat{\mathcal{F}}(B)$ such that the following equations hold

$$L = L_B + lG_F + L_F$$
$$dl = L_B d\rho^{-2} \qquad \diamondsuit$$

PROOF By Proposition 4.4.15 it follows that \mathcal{K} induces a KS-space \mathcal{K}_B in B and a KS-space \mathcal{K}_F in F. If $L \in \mathcal{K}$, then it follows from Proposition 4.5.2 that L is determined up to constants by KTs in \mathcal{K}_B and \mathcal{K}_F satisfying the above equations. Conversely from Proposition 4.5.2 it follows that every KT in \mathcal{K}_F can be extended to a KT in \mathcal{K} . Furthermore it follows from Proposition 4.5.2 that K can be decomposed into a KT on M to satisfy the hypothesis of the above corollary. Hence from the above corollary it follows that each $L_B \in \mathcal{K}_B$ can be extended into a KT in \mathcal{K} given by the above equation by taking $L_F = 0$.

One usually determines if an orthogonal separable web is reducible by inspecting the metric in adapted coordinates by using Proposition 3.5.9 (4) and keeping in mind that all KTs in the KS-space are diagonalized in adapted coordinates. We give some examples to illustrate this.

Example 5.3.13

The dimension of the Killing distribution is one in the above definition iff there is a Killing vector spanning one of the distributions of the web. This is sometimes called a web symmetry [HMS09]. \Box

Example 5.3.14

There is an abundant supply of reducible separable webs in spaces of constant curvature [Kal86]. These are a special case of KEM webs which will be introduced in Section 6.5. Concrete examples can be found in Section 9.6.2.

5.4 Intrinsic characterization of Separation for natural Hamiltonians

In this section we consider a natural Hamiltonian H. We will present an intrinsic characterization of separation for this Hamiltonian. Following [Ben97], this will reduce to the intrinsic characterization of separation for geodesic Hamiltonians.

In order to reduce this to the geodesic case, consider the following construction. Let $V \in \mathcal{F}(M)$ be the potential function of the natural Hamiltonian and assume locally that $V \neq 0$. Consider the local warped product $\overline{M} := M \times_{\rho} \mathbb{E}^{1}_{\nu}$, with adapted contravariant metric $\overline{G} := G + \rho^{-1}G_{1}$ where ρ, ν are defined as follows³:

$$\frac{1}{\rho} := 2V \qquad \qquad \nu := \operatorname{sgn} V$$

We let $(\bar{q}^j) = (q^0, q^i)$ be product coordinates on \bar{M} , where (q^i) are coordinates for M. This warped product metric is called an *Eisenhart metric*, since Eisenhart showed that geodesics $\bar{q}^j(t)$ in this warped product with $\dot{\bar{q}}^0 = 1$ project onto solutions of Hamilton's Equations for the natural Hamiltonian associated with V [Eis28].

It was a remarkable observation by Benenti in [Ben97], that showed that the Hamilton-Jacobi equation associated with potential V is separable in coordinates (q^i) on M iff the geodesic Hamilton-Jacobi equation is separable in the induced product coordinates (q^0, q^i) on $M \times_{\rho} \mathbb{E}^1_{\nu}$. This follows for example, by an inspection of the Levi-Civita equations (Eq. (5.2.3)) associated with the respective Hamiltonians. This observation allows us to prove the following theorem:

³Note the difference in the use of the warping function here, this will simply following calculations.

Theorem 5.4.1 (Benenti's Theorem [Ben97])

A natural Hamiltonian with potential V is separable in a web \mathcal{E} iff there exists a ChKT K whose eigenspaces form \mathcal{E} which satisfies the dKdV equation:

$$d(KdV) = 0$$

Furthermore if V separates in the separable web \mathcal{E} , then all K in the KS-space associated with \mathcal{E} satisfy the dKdV equation with V.

PROOF By the preceding observations, a necessary and sufficient condition for the separability of the potential V is that the geodesic Hamiltonian on $\overline{M} = M \times_{\rho} \mathbb{E}_{\nu}^{1}$ be separable in product coordinates. By Theorem 5.3.2, this is equivalent to the existence of a ChKT on \overline{M} which has the Killing distribution, $T\mathbb{E}_{\nu}^{1}$, as an invariant distribution. It follows by Proposition 4.5.2 that any such ChKT, \tilde{K} , can be put into the form:

$$\tilde{K} = K + 2tG_1$$

where $K \in \hat{S}^2(M)$ and $t \in \mathcal{F}(M)$ satisfies:

$$\mathrm{d}t = K\mathrm{d}V$$

Thus it follows by Proposition 4.5.2, that a necessary and sufficient condition is the existence of a ChKT $K \in S^2(M)$ satisfying the dKdV equation with V.

The last remark follows by Eq. (5.3.2) in Lemma 5.3.9 as in the proof of Proposition 5.3.10.

We now examine the form of the first integrals guaranteed by the Jacobi theorem (Theorem 5.1.1). On the Eisenhart manifold \overline{M} , we are guaranteed n + 1 commuting, point-wise independent Killing tensors by Corollary 5.3.3. Let (q^0, q^i) be the associated separable coordinates. By Proposition 4.5.2 each of these KTs can be put into the form:

$$\tilde{K} = K + 2UG_1$$

where $K \in \hat{S}^2(M)$ and $U \in \mathcal{F}(M)$. Choose a basis $\tilde{K}_0, \ldots, \tilde{K}_n$ for this KS-space such that $\tilde{K}_0 := G_1$ and $\tilde{K}_1 := \bar{G}$. On $T^*\bar{M}$, in the induced canonical coordinates (q^i, p_j) , consider the following first integrals:

$$E_1 := H$$
 $E_k := \frac{1}{2} K_k^{ij} p_i p_j + U_k p_0^2 \quad k = 2, \dots, n$

In the following corollary, we will show that the induced functions on T^*M , obtained by taking $p_0 = 1$, are commuting first integrals:

Corollary 5.4.2

Suppose V is a potential separable in the web associated with a ChKT K. Let (q^i) be the associated separable coordinates and (q^i, p_j) be the induced canonical coordinates on T^*M . Then there exist functionally independent commuting first integrals F_1, \ldots, F_n (where $F_1 := H$) each having the form:

$$F = \frac{1}{2}K^{ij}p_ip_j + U(q^i)$$

where each quadratic polynomial in the momenta is induced by a KT in the KS-space associated with the separable coordinates. Furthermore, these integrals are precisely those guaranteed by the Jacobi theorem. \Box

PROOF Let $\tilde{K}_1, \ldots, \tilde{K}_n$ be the KTs from the preceding discussion. Then it follows from Proposition 4.4.15 that the projected tensor, K_i , is a KT on M. Then note that the KTs K_1, \ldots, K_n form a basis for the KS-space associated with the ChKT K. Furthermore, since \tilde{K}_i is a KT on \bar{M} , it follows from Proposition 4.5.2 that each K_i satisfies:

$$\mathrm{d}U_i = K_i \mathrm{d}V \tag{5.4.1}$$

Now, as mentioned earlier, we define each $F_k \in \mathcal{F}(T^*M)$ by:

$$F_k := E_{K_k} + U_k = \frac{1}{2} K_k^{ij} p_i p_j + U_k \quad k = 1, \dots, n$$

with $F_1 = H$. Then,

$$\{F_i, F_j\} = \{E_{K_i}, E_{K_j}\} + \{E_{K_i}, U_j\} + \{U_i, E_{K_j}\}$$

$$\stackrel{(4.2.5)}{=} -\frac{1}{2} (\frac{1}{2} E_{[K_i, K_j]} + E_{[K_i, U_j]} + E_{[U_i, K_j]})$$

Now note that $[K_i, K_j] = 0$. Also by Proposition 4.2.3,

$$[K_i, U_j] = 2K_i \mathrm{d}U_j$$

Thus one can immediately verify that $[K_i, U_j] = [K_j, U_i]$, due to Eq. (5.4.1) and because the KTs, K_i , commute as linear operators. Thus we conclude that $\{F_i, F_j\} = 0$. The functional independence of the integrals follows from that fact that the KTs K_1, \ldots, K_n are point-wise independent.

The proof of the last remark (showing that these integrals arise from a separable solution) is a simple generalization of that in Corollary 5.3.3.

For completeness sake, we also mention that given a ChKT K, the most general potential satisfying the dKdV equation with K is known in separable coordinates. Indeed,

if (q^i) are coordinates which diagonalize K, then the dKdV equation in these coordinates is Eq. (5.2.3b). This follows from the proof of Theorem 5.4.1 or from Eq. (5.3.2). The general solution, V, of this PDE is easily obtained from Stäckel theory (see references at the end of Section 5.3 for a proof), and is given as follows:

$$V = V_i g^{ii} \tag{5.4.2}$$

where each V_i depends only on q^i .

5.4.1 Separation of natural Hamiltonians in Warped Products

In this section we are concerned with the separation of the Hamilton-Jacobi equation in reducible separable webs. So we fix a natural Hamiltonian H with potential V. Kis assumed to be a KT with orthogonally integrable eigenspaces $(E_i)_{i=1}^k$ with associated eigenfunctions $\lambda_1, ..., \lambda_k$. We work in the local twisted product $\prod_{i=1}^{\rho} M_i$ adapted to the eigenspaces of K given by Corollary 4.4.8.

Now suppose E_i is Killing and that K_i is a KT on M_i . Then by Proposition 4.5.2, the lift K_i , is a KT on M. The following proposition will allow us to reduce the calculation of the dKdV equation with K_i on M to the restriction of the equation on M_i . To make this precise, we fix $\bar{p} \in M$ and let $L_i(\bar{p})$ be the leaf of the canonical foliation of M_i through \bar{p} . Furthermore let $\tau_i : M_i \to L_i(\bar{p})$ be the embedding of M_i in M.

Proposition 5.4.3 (Reduction of The dKdV equation on warped products) Suppose K and K_i are as above, E_i is Killing and additionally assume that M is connected. For a potential $V \in \mathcal{F}(M)$, let $V_i := \tau_i^* V \in \mathcal{F}(M_i)$. Suppose d(KdV) = 0 holds on M, then the following is true:

$$d(K_i dV) = 0 \quad \Leftrightarrow \quad d(\tilde{K}_i dV_i) = 0 \qquad \diamondsuit$$

PROOF The first implication follows trivially by naturality of the exterior derivative, so now we prove the converse. First we note that as endomorphisms of T^*M , $K_i = \rho_i^2 \tilde{K}_i$ where \tilde{K}_i is the lift of an endomorphism of T^*M_i . We also note that for $y \in \hat{\mathfrak{X}}(M_{i\perp})$

$$\mathcal{L}_y(\rho_i^2(\mathrm{d}V)_i) = 0$$

where $(dV)_i$ is the orthogonal projection of dV onto T^*M_i . To prove this, we first note that since d(KdV) = 0, $y(\rho_i^2 x V) = 0$ for all $x \in \hat{\mathfrak{X}}(M_i)$ by Eq. (5.3.3) in Lemma 5.3.9. This implies that $d(\rho_i^2(dV)_i) = 0$. Hence the above equation follows by Cartan's Formula which relates the exterior derivative of forms to their Lie derivatives. Now by hypothesis, for $x \in \hat{\mathfrak{X}}(M_i)$ and $y \in \hat{\mathfrak{X}}(M_i)$ with [x, y] = 0 we have that $\tau_i^*(\mathrm{d}(K_i \mathrm{d}V)(x, y)) = 0$. Then for $z \in \hat{\mathfrak{X}}(M_{i\perp})$,

$$zd(K_idV)(x,y) = z[x(K_i(y,dV)) - y(K_i(x,dV))]$$

= $z[x(\tilde{K}_i(y,\rho_i^2(dV)_i))) - y(\tilde{K}_i(x,\rho_i^2(dV)_i)))]$
= $x(\tilde{K}_i(y,\mathcal{L}_z(\rho_i^2(dV)_i))) - y(\tilde{K}_i(x,\mathcal{L}_z(\rho_i^2(dV)_i))))$
= 0

where the last equation follows since $\mathcal{L}_z(\rho_i^2(\mathrm{d}V)_i) = 0$. Thus since M is connected we conclude that $\mathrm{d}(K_i\mathrm{d}V)(x,y) = 0$ on M.

For $x \in \mathfrak{X}(M_i)$ and $y \in \mathfrak{X}(M_{i\perp})$

$$d(K_i dV)(x, y) = x(\tilde{K}_i(y, \rho_i^2(dV)_i)) - y(\tilde{K}_i(x, \rho_i^2(dV)_i))$$
$$= -\tilde{K}_i(x, y(\rho_i^2(dV)_i))$$
$$= 0$$

Also it easily follows that for $x \in \hat{\mathfrak{X}}(M_{i\perp})$ and $y \in \hat{\mathfrak{X}}(M_{i\perp})$, that $d(K_i dV)(x, y) = 0$. Thus the result is proven.

We now consider the problem of separation in warped products. To be precise, suppose $N = N_0 \times_{\rho} \prod_{i=1}^{l} N_i$ is a warped product and $\mathcal{E} = (D_i)_{i=0}^{l}$ is the associated WP-net. Suppose K is a ChKT such that each Killing distribution defining \mathcal{E} is K-invariant. According to Benenti's Theorem (Theorem 5.4.1), for a potential $V \in \mathcal{F}(M)$ to be separable in the web associated with K, we need to check that the dKdV equation is satisfied. Although in this case we have some more information. Due to Corollary 4.5.4, K can be decomposed as follows in contravariant form:

$$K = K_0 + \sum_{i=1}^{l} K_i$$

where each $K_i \in \hat{S}^2(N_i)$ is a KT for i = 1, ..., l, each D_i is an eigenspace of K_0 for i = 1, ..., land K_0 restricted to D_0 is characteristic. By Benenti's Theorem, if V satisfies the dKdV equation with K, then it must satisfy the dKdV equation with each K_i . In particular it must satisfy the dKdV equation with K_0 . Since K_0 invariantly encodes the warped product through it's eigenspaces and a partial separable web on D_0 , one could ask if the converse holds. If V satisfies the dKdV equation with a given KT K_0 with eigenspaces as just stated, is it possible to build up a separable web for V by reducing the problem to one on the spherical factors of N? The following theorem shows that we can.

Theorem 5.4.4 (Separation in Warped Products)

Suppose $(D_i)_{i=0}^l$ is a WP-net and K_0 is a KT with eigenspaces D_i for i = 1, ..., l and characteristic on D_0 . Fix $\bar{p} \in M$ and let $N = \prod_{i=0}^l N_i$ be a connected product manifold passing through \bar{p} adapted to the WP-net $(D_i)_{i=0}^l$. Then the following holds:

Suppose $V \in \mathcal{F}(M)$ satisfies $d(K_0 dV) = 0$. Let $V_i := \tau_i^* V \in \mathcal{F}(N_i)$ and suppose for each i = 1, ..., k there exists a ChKT \tilde{K}_i on N_i such that $d(\tilde{K}_i dV_i) = 0$.

Then V is separable in the web formed by the simple eigenspaces of K_0 together with the lifts of the simple eigenspaces of $\tilde{K}_1, ..., \tilde{K}_l$.

PROOF For i = 1, ..., l, let K_i be the lift of \tilde{K}_i to N. Consider the tensor

$$K := K_0 + \sum_{i=1}^l K_i$$

By Proposition 4.5.2, K is a Killing tensor on N. Let \tilde{G}_i be the contravariant metric on N_i , then by replacing \tilde{K}_i with $a_i \tilde{K}_i + b_i \tilde{G}_i$ for some $a_i \in \mathbb{R} \setminus \{0\}$ and $b_i \in \mathbb{R}$, we can assume K locally has simple eigenfunctions. Let q_0 be coordinates which diagonalize the ChKT induced by K_0 on N_0 . Let q_j be coordinates which diagonalize \tilde{K}_j on N_j for each j > 0. Then one can check that the product coordinates (q_0, q_1, \ldots, q_l) are orthogonal and diagonalize K, hence K is a ChKT. By Proposition 5.4.3, $d(K_i dV) = 0$ on N for each i > 0, hence K satisfies the dKdV equation with V. Thus it follows by Theorem 5.4.1 that V separates in the product coordinates (q_0, q_1, \ldots, q_l) , which proves the claim.

The above theorem and the preceding discussion shows that reducible separable webs enable one to reduce the problem of separation to certain spherical submanifolds after one finds a KT with the same eigenspaces as K_0 in the above theorem.

The motivating application of the above theorem is to devise a recursive algorithm (The BEKM separation algorithm) to separate natural Hamiltonians defined on spaces of constant curvature. Before we can do this, we have to first introduce concircular tensors; this is done in the next chapter.

5.5 Notes

Much of the theory on the Hamilton-Jacobi equation and its separation that we have presented is based on contemporary formulations [Woo75; Ben89]. Most of the theory on the intrinsic characterization of separation is due to Benenti [Ben97], following Eisenhart's lead [Eis34]. Much of the recent interest in the separation of the Hamilton-Jacobi equation was due to the discovery of the separability of the Hamilton-Jacobi equation for the geodesics in the Kerr solution from general relativity [Car68].

Before the 1960s, the fundamental result was due to Stäckel in [Sta93]. He obtained the general form of the orthogonal separable metric in separable coordinates. Eisenhart's intrinsic characterization in [Eis34] is based on Stäckel's work and the proof is much more complicated than the one presented here.

We have omitted the theory for general (possibly non-orthogonal) separation which is covered in [Ben97]. Furthermore, as hinted at by Example 5.3.7, a notion of complex separation is possible on strictly pseudo-Riemannian manifolds. See [DR07] for details.

The material on warped products is new and is from the article [RM14b].

Chapter 6

Concircular tensors and KEM webs

In this chapter we study concircular tensors and the orthogonal (separable) webs which can be constructed using them: Kalnins-Eisenhart-Miller (KEM) webs. As stated in the introduction (see Section 2.2), we study these tensors because of their computational value in working with KEM webs.

 $L \in S^{p}(M)$ is called a *concircular tensor also called a C-tensor (CT)* of valence p if there exists $C \in S^{p-1}(M)$ (called the *conformal factor*) such that

$$\nabla_x L = C \odot x \tag{6.0.1}$$

for all $x \in \mathfrak{X}(M)$. The reason for the name "concircular" will be given in Section 6.4. Sometimes we denote the space of concircular tensors of valence p by $C^{p}(M)$ and the subspace of covariantly constant tensors by $C_{0}^{p}(M)$. Concircular tensors of arbitrary valence were originally defined in [Cra08], where they were called special conformal Killing tensors. This is because concircular tensors are conformal Killing tensors as we shall show shortly.

In the first four sections, we study CTs in general. In the last three sections, we present the application of CTs to problem of orthogonal separation of the Hamilton-Jacobi equation.

6.1 General Valence

The theory of general valence CTs has been studied in [Cra08]. We give a brief outline here.

We first observe that the defining equation implies that CTs form a vector space which is closed under the symmetric product. Indeed, if L_1 and L_2 are CTs with conformal factors C_1 and C_2 respectively, then a short calculation shows that $L_1 \odot L_2$ is a CT with conformal factor $C_1 \odot L_2 + C_2 \odot L_1$.

Proposition 6.1.1 (Properties of Concircular tensors [Cra08])

Suppose L is a CT of arbitrary valence with conformal factor C. Then L is a CKT with conformal factor C and C is given as follows:

$$C = \frac{r}{n+r-1} \nabla \cdot L \qquad \qquad \square$$

PROOF In coordinates the defining equation of L reads:

$$\nabla_j L_{i_1,\dots,i_r} = C_{(i_1,\dots,i_{r-1}} g_{i_r)j} \tag{6.1.1}$$

Thus

$$\nabla_{(j} L_{i_1,\dots,i_r)} = C_{(i_1,\dots,i_{r-1}} g_{i_r j)}$$

which proves that L is a CKT. Also to obtain the equation for C, we get from Eq. (6.1.1) that

$$\nabla \cdot L = \nabla_j L^{i_1, \dots, i_{r-1}j}$$
$$= C^{(i_1, \dots, i_{r-1}} \delta^{j)}{}_j$$
$$= \frac{(n+r-1)}{r} C^{i_1, \dots, i_{r-1}}$$

In [Cra08], Crampin has derived structural equations for CTs of arbitrary valence and as a consequence, he has proven the following theorem:

Theorem 6.1.2 (The Vector Space of Concircular tensors [Cra08])

Suppose n > 2. Then the C-tensors of valence r form a finite dimensional real vector space with maximal dimension equal to the dimension of the space of constant symmetric r-tensors in \mathbb{R}^{n+1} . Furthermore the maximal dimension is achieved if and only if the space has constant curvature.

Remark 6.1.3

When $r \leq 2$ the above result holds for n = 2 as well [TCS05; Cra07]. In particular, if r = 1 (resp. r = 2) the maximal dimension is n + 1 (resp. $\frac{1}{2}(n+1)(n+2)$).

The above theorem implies the following:

Corollary 6.1.4 (Concircular tensors in spaces of constant curvature)

Suppose M^n is a space of constant curvature. Let $\beta = \{v_1, \ldots, v_{n+1}\}$ be a basis for the space of concircular vectors, then a given C-tensor of valence r can be written uniquely as a linear combination of r-fold symmetric products of the vectors in β .

6.2 Torsionless Conformal Killing tensors

Before moving on to study concircular 2-tensors (the main object of interest), we will first study torsionless (orthogonal) conformal Killing tensors. The main reason for studying torsionless orthogonal CKTs is because they will help us to study concircular tensors, as we will see in the next section.

Historically, Benenti originally showed that a torsionless CKT, L, with n functionally independent eigenfunctions can be used to generate a basis for a KS-space in a coordinate independent way [Ben92a, Proposition 2.1]. In [Ben05] he showed that his method worked when the eigenfunctions were just assumed to be simple. For this reason, we refer to any torsionless CKT with simple (real) eigenfunctions as a *Benenti tensor* (also called an L-tensor by Benenti). In [Ben05], Benenti extended his study to include general torsionless orthogonal CKTs. Most of the results in this section are based on Benenti's, but we arrive at them using the characterization of orthogonal CKTs in terms of their eigenspaces (given by Corollary 4.4.6).

In this section, L is assumed to be a torsionless orthogonal CKT unless otherwise stated. We now recall the implications of the torsionless property, which are studied in detail in Appendix B. Suppose $(E_i)_{i=1}^k$ are the eigenspaces of L and $(\lambda_i)_{i=1}^k$ are the associated eigenfunctions. Then by Theorem B.0.20, the eigenspaces are orthogonally integrable and each eigenfunction satisfies

$$(\nabla\lambda_i)^j = 0 \quad j \neq i \tag{6.2.1}$$

We will see that the above equations satisfied by the eigenfunctions make these CKTs highly amenable to analysis. Due to Eq. (4.4.5), we have to assume L is a CKT, not just a KT, in order to deal with non-trivial cases.

By Corollary 4.4.8 there is a twisted product $\prod_{i=1}^{\rho} M_i$ which is adapted to the eigenspaces of L. We can explicitly solve for the twist function ρ_i in this case. Indeed, from Eq. (4.4.4), we have

$$H_{i} = -\frac{1}{2} \sum_{j \neq i} (\nabla \log |\lambda_{i} - \lambda_{j}|)^{j}$$

$$\stackrel{(6.2.1)}{=} -\frac{1}{2} \sum_{j \neq i} (\sum_{k \neq i} \nabla \log |\lambda_{i} - \lambda_{k}|)^{j}$$

$$= -\frac{1}{2} \sum_{j \neq i} (\nabla \log \prod_{k \neq i} |\lambda_{i} - \lambda_{k}|)^{j}$$

$$= -\frac{1}{2} (\nabla \log \prod_{k \neq i} |\lambda_{i} - \lambda_{k}|)^{\perp i}$$

Hence by Eq. (4.4.6), we have

$$(\nabla \log \rho_i^2)^{\perp i} = (\nabla \log \prod_{k \neq i} |\lambda_i - \lambda_k|)^{\perp i}$$

Thus $\log \rho_i^2 - \log \prod_{k \neq i} |\lambda_i - \lambda_k| = f_i$ where f_i is a function of M_i only. Hence we have the following (cf. [Ben05, Theorem 18.1]):

Proposition 6.2.1 (Characterization of torsionless orthogonal CKTs) Suppose L is a torsionless orthogonal tensor with eigenspaces $(E_i)_{i=1}^k$ and associated eigenfunctions $(\lambda_i)_{i=1}^k$. Then L is a CKT iff there is a twisted product adapted to its

eigenspaces such that each twist function ρ_i can be chosen to be:

$$\rho_i^2 = \prod_{k \neq i} |\lambda_i - \lambda_k| \qquad \qquad \square$$

PROOF The characterization follows from the preceding calculation and Corollary 4.4.8.

We now deduce some important geometric properties of the eigenspaces of torsionless orthogonal CKTs.

Proposition 6.2.2 (Properties of torsionless orthogonal CKTs [Ben05]) Suppose that L is a torsionless orthogonal CKT, then the following statements are true:

1. L is of gradient-type and the conformal factor α is given as follows:

$$\alpha = \sum_{i} \nabla \lambda_i \tag{6.2.2}$$

- 2. An eigenfunction λ_i is constant iff its associated eigenspace E_i is Killing.
- 3. L is of trace-type iff each multidimensional eigenspace is Killing.
- 4. If L has simple eigenfunctions, then it is of trace-type.
- 5. If $S := \bigoplus_{\dim E_i=1} E_i$ is the space of simple eigenspaces of L, then L restricted to any integral manifold of S is a Benenti tensor.

PROOF The first statement follows from condition 3 of Corollary 4.4.6 and Eq. (6.2.1). The second statement follows from Corollary 4.4.11 and Eq. (6.2.1) or from Proposition 3.5.9 (4) using the formula for the twist function from Proposition 6.2.1.

For the third statement assume L is trace-type. Due to the second statement, we need only show that λ_i is constant when dim $E_i > 1$. The trace-type condition implies
that $\alpha = \nabla \operatorname{tr}(L) = \sum_{i} m_i \nabla \lambda_i$ where $m_i = \dim E_i$. This together with condition 3 in Corollary 4.4.6 gives the following:

$$m_i \nabla \lambda_i = \alpha^i = \nabla \lambda_i$$

Hence when dim $E_i > 1$, λ_i must be constant. The converse follows from Eq. (6.2.2). The fourth statement is an immediate corollary of the third statement.

To prove the fifth statement note that due to Proposition 4.4.15, L restricts to a torsionless CKT with simple eigenfunctions on any integral manifold of S. Hence the fifth statement follows from the fourth statement.

Remark 6.2.3

It follows from statement 1 of the above proposition that if L is a torsionless orthogonal CKT, then it has a conformal factor ∇f for some $f \in \mathcal{F}(M)$. Then one can show that the following tensor is a KT:

$$K = fG - L$$

Note that such a KT shares the same eigenspaces as L and the above is a KT for all gradient-type CKTs; it will be useful to us later.

Using the above remark, we can prove the following:

Corollary 6.2.4 (Benenti tensors induce Separable Webs)

If L is a Benenti tensor then the web formed by its eigenspaces is separable. \Box

PROOF If ∇f is the conformal factor of L, then the above remark implies that K = fG - L is a KT with the same eigenspaces as L, hence a ChKT. The result then follows by Theorem 5.3.2.

We call the separable web induced by a Benenti tensor L (as in the above corollary) the separable web generated by L. Similarly we call the associated KS-space the KSspace generated by L. Theorem 7.1 in [Ben05] shows that a basis for the KS-space can be generated using only L and the metric; a slight generalization of this result will be presented in Section 6.6.

One can show that given a separable web $\mathcal{E} = (E_i)_{i=1}^n$, there are many ChKTs whose eigenspaces form this web. In fact, Corollary 6.6.8 will show that the KS-space generated by an ICT contains an (n-1)-dimensional subspace of ChKTs. The following proposition shows that this is not the case when we restrict ourselves to torsionless orthogonal CKTs. It shows that for all non-trivial cases the eigenfunctions of a torsionless orthogonal CKT are essentially uniquely determined by its eigenspaces.

Proposition 6.2.5 (Equivalent Torsionless Orthogonal CKTs)

Suppose L and L are torsionless orthogonal CKTs and assume M is connected. Suppose $(E_i)_{i=1}^k$ are the eigenspaces of L.

If L is not covariantly constant (on any open set), then: The eigenspaces of \tilde{L} are the same as those of L iff there exists $a, b \in \mathbb{R}$ with $a \neq 0$ such that

$$L = a\tilde{L} + bG$$

If L is covariantly constant, then: The eigenspaces of \tilde{L} are the same as those of L iff there exists $c_i \in \mathbb{R}$ for i = 1, ..., k such that¹

$$L = \tilde{L} + \sum_{i} c_i G_i$$

where G_i denotes the restriction of the metric G to E_i , which is a KT in this case.

PROOF Suppose L and \tilde{L} share the same eigenspaces $\mathcal{E} = (E_i)_{i=1}^k$ with respective eigenfunctions $(\lambda_i)_{i=1}^k$ and $(\tilde{\lambda}_i)_{i=1}^k$. Then since H_i is uniquely determined by E_i for each i, from Eq. (4.4.4) we have the following for $j \neq i$

$$(\nabla \log |\lambda_i - \lambda_j|)^j = (\nabla \log \left| \tilde{\lambda}_i - \tilde{\lambda}_j \right|)^j$$
$$\Leftrightarrow (\nabla \log \left| \frac{\lambda_i - \lambda_j}{\tilde{\lambda}_i - \tilde{\lambda}_j} \right|)^j = 0$$

Similarly by permuting $i \leftrightarrow j$, we have

$$(\nabla \log \left| \frac{\lambda_i - \lambda_j}{\tilde{\lambda}_i - \tilde{\lambda}_j} \right|)^i = 0$$

Hence

$$\frac{\lambda_i - \lambda_j}{\tilde{\lambda}_i - \tilde{\lambda}_j} \stackrel{(6.2.1)}{=} a_{ij} \in \mathbb{R}$$
$$\Rightarrow \lambda_i - a_{ij}\tilde{\lambda}_i = \lambda_j - a_{ij}\tilde{\lambda}_j \stackrel{(6.2.1)}{=} b_{ij} \in \mathbb{R}$$
(6.2.3)

If i, j, k are distinct, by differentiating the above equation we get:

¹There are additional technical restrictions on the constants c_i which ensure that \tilde{L} and L have the same eigenspaces.

$$a_{ij}\nabla\lambda_i = \nabla\lambda_i = a_{ik}\nabla\lambda_i$$
$$\Rightarrow (a_{ij} - a_{ik})\nabla\tilde{\lambda}_i = 0$$

From Eq. (4.4.4) and Proposition 6.2.2 (1) and (2), we see that L (or \tilde{L}) is covariantly constant iff all its eigenfunctions are constants iff \mathcal{E} is a pseudo-Riemannian product net. Hence if L is not covariantly constant, then there exists an i such that $\nabla \tilde{\lambda}_i \neq 0$. Then the above equation implies for each $j, k \neq i$ that $a_{ij} = a_{ik}$. So let $a := a_{ij}$ for some $j \neq i$. Then from Eq. (6.2.3), we see that

$$b_{ij} = \lambda_i - a\tilde{\lambda}_i = \lambda_k - a\tilde{\lambda}_k = b_{ik}$$

Thus we can let $b := b_{ij}$ for some $j \neq i$. Then Eq. (6.2.3) shows that for all i

$$\lambda_i = a\tilde{\lambda}_i + b$$

This proves the first part of the proposition. Finally if L is covariantly constant, then \mathcal{E} is a pseudo-Riemannian product net, thus the eigenfunctions are forced to be constants and the second part follows.

Remark 6.2.6

We should mention here that Theorem 10.1 in [Ben05] is incorrect as stated. The mistake can be seen by comparing the statement of Theorem 10.1 with that of the above theorem, while keeping in mind that L-tensors are torsionless CKTs with simple eigenfunctions. \Box

6.3 Concircular 2-tensors

Hereafter by concircular tensor, we mean a concircular 2-tensor. In this section we will develop the basic theory of concircular tensors. This class of concircular tensors are the most important for separating the Hamilton-Jacobi equation and so these tensors are the most studied. We will assume the reader is familiar with Appendix B.

Much of the theory is due to Benenti [Ben05] and Crampin [Cra03]. They were first formally introduced within the context of separation of variables by Crampin in [Cra03], where he referred to them as special conformal Killing tensors, cf. [TCS05; Cra07]. They have also been studied in [Ben05] where they are called J-tensors. The theory regarding the cases when these tensors have multidimensional eigenspaces is new and was originally presented in [RM14b].

First note that the defining equations for CTs can be written in index notation as follows:

$$\nabla_k L_{ij} = \alpha_{(i}g_{j)k}$$

When n > 1, we say a concircular tensor is *non-trivial* if it's not a multiple of the metric.

The following proposition presents two key properties of concircular tensors, from which much else can be deduced.

Proposition 6.3.1 (Properties of Concircular 2-tensors [Cra03]) Suppose L is a CT with conformal factor α . Then the following hold:

1. L is a trace-type CKT, i.e. the conformal factor α is given as follows:

$$\alpha = \nabla \operatorname{Tr} L$$

2. The Nijenhuis tensor of L vanishes.

PROOF The first property follows by taking the trace of the defining equation

$$\nabla_k L_{ij} = \frac{1}{2} (\alpha_i g_{jk} + \alpha_j g_{ik})$$

over the indices i, j.

To show that L is torsionless, by Proposition B.0.14 (2) we only need to prove that $(\nabla_{Lu}L)v - L(\nabla_uL)v$ is symmetric with respect to u, v. Then

$$(\nabla_{Lu}L)v - L(\nabla_uL)v = \frac{1}{2}[(Lu \otimes \mathrm{d}\operatorname{Tr} L + \nabla \operatorname{Tr} L \otimes (Lu)^{\flat})v - L(u \otimes \mathrm{d}\operatorname{Tr} L + \nabla \operatorname{Tr} L \otimes u^{\flat})v]$$

$$= \frac{1}{2}[(v\operatorname{Tr} L)Lu + \langle Lu, v \rangle \nabla \operatorname{Tr} L - (v\operatorname{Tr} L)Lu - \langle u, v \rangle L(\nabla \operatorname{Tr} L)]$$

$$= \frac{1}{2}[\langle Lu, v \rangle \nabla \operatorname{Tr} L - \langle u, v \rangle L(\nabla \operatorname{Tr} L)]$$

which is symmetric with respect to u, v since L is self-adjoint, i.e. $\langle Lu, v \rangle = \langle u, Lv \rangle$.

An orthogonal concircular tensor (OCT) is a concircular tensor which is also an orthogonal tensor. The above proposition will allow us to study OCTs as special cases of orthogonal CKTs. Hence we can apply the results of the previous section. Another key observation is the following: by Proposition 6.2.2 (3) any multidimensional eigenspace of an OCT is Killing. This will allow us to develop a constructive theory (in Section 6.5) to separate the HJ equation using these tensors. We summarize here the results following from Proposition 6.2.2:

Corollary 6.3.2 (OCTs induce Warped Products)

Suppose L is an OCT. Let $S := \bigoplus_{\dim E_i=1} E_i$ be the space of simple eigenspaces of L. Then S is the geodesic distribution of a warped product net with the (multidimensional) eigenspaces complementary to S as the Killing distributions. Furthermore L restricted to any integral manifold of S is a Benenti tensor. \diamond

PROOF Proposition 6.2.2 shows that each eigenspace satisfying dim $E_i > 1$ is Killing, also $\bigcap_{\dim E_i > 1} E_i^{\perp} = \bigoplus_{\dim E_i = 1} E_i = S$ is geodesic since each E_i^{\perp} is. Thus S together with the complementary eigenspaces forms a warped product net. Then Proposition 6.2.2 (5) completes the proof.

Since Benenti tensors have been well studied in the literature (see for example [Ben05]), the above proposition implies that much of this theory can still be applied to OCTs (provided $S \neq 0$).

The following class of CTs are the basic building blocks of all OCTs.

Definition 6.3.3 (Irreducible concircular tensors)

An OC-tensor with functionally independent eigenfunctions is referred to as an *irreducible* concircular tensor (ICT) or more succinctly an IC-tensor. To be precise, an IC-tensor has real eigenfunctions $u^1, ..., u^k$ (counted without multiplicity) satisfying:

$$\mathrm{d}u^1\wedge\cdots\wedge\mathrm{d}u^k\neq 0$$

Furthermore an OC-tensor which is not irreducible is called reducible.

Remark 6.3.4

IC-tensors were the class of C-tensors mainly studied in [Cra03].

Since by Proposition 6.2.2, the eigenfunction associated with a multidimensional eigenspace of an OCT is constant, it follows that an ICT must have simple eigenfunctions, hence ICTs are Benenti tensors. The special property that ICTs have is that their eigenfunctions can be used as (local) coordinates for the separable web they induce [Cra03]. We will refer to these coordinates as the *canonical coordinates* induced by these tensors. See Example 2.3.1 which shows how elliptic coordinates in \mathbb{E}^2 can be obtained from an ICT.

Locally, we can assume a reducible OC-tensor has eigenfunctions $u^1, ..., u^k$ which are functionally independent and the rest of which are constants. Hence as in the proof of Corollary 6.3.2, since each eigenspace corresponding to the constant eigenfunctions are Killing (see Proposition 6.2.2), there exists a warped product in which the functions $u^1, ..., u^k$ can be taken as coordinates on the geodesic factor. We add one final remark. If L is a CT then the Killing tensor defined in Remark 6.2.3 is called the *Killing Bertrand-Darboux tensor (KBDT)* generated by L and has the following form:

$$K = \operatorname{tr}(L)G - L \tag{6.3.1}$$

This tensor will be useful in some constructions done later on.

6.3.1 Characterizations of OCTs

In this section, our goal is to give some characterizations of OCTs which will be useful in their application. The following proposition allows us to do this. It is a converse to Proposition 6.3.1.

Proposition 6.3.5 ([Ben05])

If L is an orthogonal trace-type CKT with vanishing torsion then L is a CT. \Box

PROOF We prove this by generalizing a proof by Crampin of a special case when the eigenfunctions are simple [Cra03, Proposition 1] (see [Ben05, Theorem A.5.3] for an alternate proof). First we need to do some calculations in index notation.

By hypothesis, L satisfies $\nabla_{(k}L_{ij)} = \alpha_{(i}g_{jk)}$. Define T by

$$T_{ijk} = \nabla_k L_{ij} - \alpha_{(i}g_{j)k} = \nabla_k L_{ij} - \frac{1}{2}(\alpha_i g_{jk} + \alpha_j g_{ik})$$

then observe that $T_{(ijk)} = 0$ and $T_{ijk} = T_{jik}$.

Now the Nijenhuis torsion N_L is given by (see Proposition B.0.14 and the following remark)

$$(N_L)_{kij} = L^l_{\ i} \nabla_l L_{kj} - L^l_{\ j} \nabla_l L_{ki} - L^l_k (\nabla_i L_{lj} - \nabla_j L_{li})$$

$$(6.3.2)$$

First we express the first two terms in terms of T as follows:

$$L_{i}^{l} \nabla_{l} L_{kj} - L_{j}^{l} \nabla_{l} L_{ki} = L_{i}^{l} (T_{kjl} + \frac{1}{2} (\alpha_{k} g_{jl} + \alpha_{j} g_{kl})) - L_{j}^{l} (T_{kil} + \frac{1}{2} (\alpha_{k} g_{il} + \alpha_{i} g_{kl}))$$

$$= L_{i}^{l} T_{kjl} - L_{j}^{l} T_{kil} + \frac{1}{2} (L_{i}^{l} \alpha_{j} g_{kl} - L_{j}^{l} \alpha_{i} g_{kl})$$

$$= L_{i}^{l} T_{kjl} - L_{j}^{l} T_{kil} + \frac{1}{2} (L_{ki} \alpha_{j} - L_{kj} \alpha_{i})$$

Now we express the last two terms in terms of T as follows:

$$\nabla_{i}L_{lj} - \nabla_{j}L_{li} = T_{lji} + \frac{1}{2}(\alpha_{l}g_{ji} + \alpha_{j}g_{li}) - (T_{lij} + \frac{1}{2}(\alpha_{l}g_{ij} + \alpha_{i}g_{lj}))$$

$$= T_{lji} - T_{lij} + \frac{1}{2}(\alpha_{j}g_{li} - \alpha_{i}g_{lj})$$

$$\Rightarrow L_{k}^{\ l}(\nabla_{i}L_{lj} - \nabla_{j}L_{li}) = L_{k}^{\ l}T_{lji} - L_{k}^{\ l}T_{lij} + L_{k}^{\ l}\frac{1}{2}(\alpha_{j}g_{li} - \alpha_{i}g_{lj})$$

$$= L_{k}^{\ l}T_{lji} - L_{k}^{\ l}T_{lij} + \frac{1}{2}(L_{ki}\alpha_{j} - L_{kj}\alpha_{i})$$

Hence Eq. (6.3.2) becomes

$$(N_L)_{kij} = L^l_{\ i} T_{kjl} - L^l_{\ j} T_{kil} - L^l_{\ k} (T_{lji} - T_{lij}) = L^l_{\ i} T_{jkl} - L^l_{\ j} T_{ikl} + L^l_k (T_{lij} - T_{lji})$$

Now $T_{(lij)} = 0$ implies that $T_{lij} = -T_{jli} - T_{ijl}$, thus the vanishing of N_L implies:

$$2L_{k}^{\ l}T_{lji} = L_{\ i}^{l}T_{jkl} - L_{\ j}^{l}T_{ikl} - L_{k}^{\ l}T_{ijl}$$

Since the right hand side is symmetric in j, k it follows that $L_k^{\ l}T_{lji} = L_j^{\ l}T_{kli}$. Now, in invariant notation we evaluate T with different combinations of eigenvectors to show that it vanishes. First observe that for $x, y, z \in \mathfrak{X}(M)$ this equation takes the following form

$$T(Lx, y, z) = T(x, Ly, z)$$

Hence the above equation readily implies that for eigenvectors x, y with different eigenfunctions and any z, T(x, y, z) = 0.

Now suppose E is a multi-dimensional eigenspace with eigenfunction λ . Then λ must be a constant due to the trace-type condition by Proposition 6.2.2 (3). Let $x, y \in \Gamma(E)$ and $z \in \mathfrak{X}(M)$. First note that $L(z, y) = \lambda g(z, y)$. Then

$$(\nabla_z L)(x, y) = \nabla_z L(x, y) - L(\nabla_z x, y) - L(x, \nabla_z y)$$
$$= \lambda(\nabla_z g(x, y) - g(\nabla_z x, y) - g(x, \nabla_z y))$$
$$= \lambda(\nabla_z g)(x, y)$$
$$= 0$$

Let $m = \dim E$, then note that $\alpha(x) = m x\lambda = 0$ since λ is constant and because of the torsionless condition. Hence T(x, y, z) = 0 for all $x, y \in \Gamma(E)$ and $z \in \mathfrak{X}(M)$.

Now suppose E is a one dimensional eigenspace, $x \in \Gamma(E)$ and z is an eigenvector with a different eigenfunction than x. Then the cyclic condition implies that T(x, x, z) = -2T(z, x, x) = 0 and that T(x, x, x) = 0.

Thus since L has a basis of eigenvectors by hypothesis, it follows that $T \equiv 0$.

As a consequence of the above result, we have our first characterization of OCTs. **Proposition 6.3.6 (Characterization of orthogonal CTs)**

Suppose L is a torsionless orthogonal tensor with eigenspaces $(E_i)_{i=1}^k$ and associated eigenfunctions $(\lambda_i)_{i=1}^k$. Then L is an OCT iff there is a twisted product adapted to its eigenspaces such that each twist function ρ_i can be chosen to be:

$$\rho_i^2 = \prod_{k \neq i} |\lambda_i - \lambda_k|$$

and each multidimensional eigenspace E_i is a Killing distribution, or equivalently the eigenfunction corresponding to E_i is constant.

PROOF First note that Proposition 6.3.1 together with the above proposition shows that an orthogonal CT is precisely an orthogonal trace-type CKT with vanishing torsion. Since imposing the trace-type condition is equivalent to requiring the multidimensional eigenspaces to be Killing, the result then follows from Proposition 6.2.1.

Also since any torsionless CKT with simple eigenfunctions is necessarily of trace-type (see Proposition 6.2.2), the above characterization implies that a Benenti tensor is precisely a CT with simple eigenfunctions.

The following proposition gives another characterization of OCTs which is designed to answer the following question: Given a warped product and an ICT \tilde{L} on the geodesic factor of the warped product, can we extend \tilde{L} to an OCT on the warped product and if so what is this extension?

Proposition 6.3.7 (Characterization of Reducible OCTs)

Suppose $L \in S^2(M)$ is an orthogonal tensor. Then L is a reducible OCT iff there exists a warped product decomposition $M = M_0 \times_{\rho_1} M_1 \times \cdots \times_{\rho_k} M_k$ with adapted contravariant metric $G = \sum_{i=0}^k G_i$ such that L has the following contravariant form:

$$L = \tilde{L} + \sum_{i=1}^{k} \lambda_i G_i$$

where each $\lambda_i \in \mathbb{R}$ and $\tilde{L} \in \hat{S}^2(M_0)$ is the canonical lift (see Section 3.3) of an ICT $\tilde{L} \in S^2(M_0)$ satisfying the following equation on M_0 for each i > 0

$$\tilde{L}(d\log \rho_i) = d(\lambda_i \log \rho_i + \frac{1}{2} \operatorname{tr}(\tilde{L}))$$
(6.3.3)

PROOF Suppose L is an OCT. Let D_1, \ldots, D_l be the eigenspaces of L associated with constant eigenfunctions and let $M = M_0 \times_{\rho_1} M_1 \times \cdots \times_{\rho_k} M_k$ be a warped product adapted to $(\bigcap_{i=1}^l D_i^{\perp}, D_1, \ldots, D_l)$ which exists by Proposition 6.3.6. We define \tilde{L} to be the restriction of L to M_0 ; it follows by Proposition 6.3.6 that \tilde{L} is an ICT in M_0 . It also follows by Proposition 6.3.6 that we can assume

$$\rho_i^2 = \prod_a |\lambda_i - \lambda_a| \tag{6.3.4}$$

where a ranges over all eigenfunctions of \tilde{L} . If dim $M_0 = 0$, i.e. L induces a pseudo-Riemannian product, the conclusion follows. Otherwise, since λ_i is constant and because \tilde{L} is torsionless, we see that on M_0

$$\begin{split} \tilde{L}(\mathrm{d}\log\rho_i) &= \frac{1}{2}\sum_a \lambda_a \mathrm{d}\log|\lambda_i - \lambda_a| \\ &= \frac{1}{2}\sum_a \lambda_a \frac{\mathrm{d}\lambda_a}{\lambda_a - \lambda_i} \\ &= \frac{\lambda_i}{2}\sum_a \frac{\mathrm{d}\lambda_a}{\lambda_a - \lambda_i} + \frac{1}{2}\sum_a \mathrm{d}\lambda_a \\ &= \mathrm{d}(\lambda_i\log\rho_i + \frac{1}{2}\operatorname{tr}(\tilde{L})) \end{split}$$

Conversely, it is easily checked that if \tilde{L} is an ICT and ρ_i satisfies the above equation, then $c\rho_i$ must satisfy Eq. (6.3.4) for some $c \in \mathbb{R}^+$. Hence it follows that L defined in the statement is torsionless and then by Proposition 6.3.6 that L is a reducible OCT.

The above proposition will be applied to classify reducible OCTs in spaces of constant curvature (see Section 9.5).

6.3.2 Relation to Geodesically Equivalent Metrics

In this section we will briefly describe how concircular tensors appear in the study of geodesically equivalent metrics. This is an important connection as geometers have studied CTs with this interest in mind [Sha00]. We follow [BM03] in our exposition, see also [Ben05].

First, we have the following definition from [BM03].

Definition 6.3.8

Two metrics g and \bar{g} are *geodesically equivalent* if they have the same geodesics (considered as unparameterized curves).

Note that some authors use the term *projectively equivalent* instead. The study of such metrics dates back to the late nineteenth century, see [BM03] for historical references. The following theorem significantly simplifies the search for geodesically equivalent metrics [Ben05]:

Theorem 6.3.9

Given a concircular tensor L for a metric g, one can construct another metric, \overline{g} , which is geodesically equivalent to g. Conversely, given two geodesically equivalent metrics g and \overline{g} , a concircular tensor for g can be constructed from them.

We will not get into the details of this construction, see [BM03; Ben05]. This connection will allow us to obtain results on CTs from the theory of geodesically equivalent metrics, as we will see in the next section.

On a related note, concircular tensors also appear in classical mechanics in the following way. Given a Riemannian manifold (M, g), and a force vector field $F \in \mathfrak{X}(M)$, consider the following dynamical equation:

$$\nabla_X X = F, \qquad X = \dot{\gamma}$$

where $\gamma(t)$ is a curve in M. A solution, γ , of the above equation is called a *geodesic* for the system (M, g, F). When $M = \mathbb{E}^3$, the above equation is simply Newton's equation with a position dependent force F.

A natural question in this context is: when are the geodesics of the above system equivalent to those of a natural Hamiltonian system on M? It turns out that a necessary condition is that g admits a non-singular concircular tensor L such that (see Theorem 13.2 in [Ben05]):

$$F = -A^{-1}\nabla V, \qquad A = \operatorname{cof}(L)$$

where cof(L) is the cofactor tensor of L (see Eq. (A.0.3)). Locally the above condition is equivalent to d(AF) = 0. The geodesically equivalent natural Hamiltonian system is the one obtained from the geodesically equivalent metric \bar{g} constructed from g and L (see Theorem 6.3.9) and the potential V in the above equation. Systems admitting such force vectors generalize conservative mechanical systems and are called *cofactor systems* in the literature; see [Ben05] and references therein for more on these systems.

Since concircular tensors appear in a few different areas of research, they go by several different names. They have been called special conformal Killing tensors in [Cra03], J-tensors in [Ben05], Sinyukov mappings in [Mat05], Benenti tensors in [BM03], and finally elliptic coordinates matrices in [Lun03].

6.3.3 Existence in arbitrary manifolds

In this section we will briefly review some results on the existence of concircular tensors in general Riemannian manifolds. The connection to the theory of geodesically equivalent metrics provides for some fruitful cross-pollination of results, the following being one of them [Cra07]:

Theorem 6.3.10 (Lacunae in the dimension of the space of CTs [Sha00])

Let M^n be a Riemannian manifold with n > 2 and set $m := \dim C^1(M)$ which is the dimension of the space of concircular vectors. If $m \neq n+1$, then we have the following estimate:

$$\frac{1}{2}m(m+1) + 1 \le \dim \mathbf{C}^2(M) \le \frac{1}{2}m(m+1) + \operatorname{int}(\frac{1}{3}(n+1-m))$$

where int(r) is the integer part of r.

In order for the above result to be of maximum value, it helps to have similar results for concircular vectors:

Theorem 6.3.11 (Lacunae in the dimension of the space of CVs [Cra07]) Let M^n be a Riemannian manifold with n > 2. Then the dimension of the space of concircular vectors, dim $C^1(M)$, satisfies: dim $C^1(M) \le n - 2$ or dim $C^1(M) = n + 1$. \Box

Table 6.1: dim $C^2(M)$ if $n=3$	Table 6.2: dim	$C^2(M)$ if $n = -$	4
$\frac{\operatorname{dim} C^1(M)}{\operatorname{dim} C^2(M)}$	$\dim \mathrm{C}^1(M)$	$\dim \mathrm{C}^2(M)$	
$\frac{\operatorname{dim} \mathbb{C}(M)}{0} = \frac{1}{1}$	0	1	
	1	2	
	2	4	
4 10	5	15	

Combining the above theorems, we have summarized their implications for low dimensions in Tables 6.1 and 6.2. One can deduce from the above table that any Riemannian 3-manifold admitting a Benenti tensor is a space of constant curvature [Cra07]. We also mention the following fact, which is Theorem 2 in [Cra03].

Theorem 6.3.12 ([Cra03])

Suppose M^n is a pseudo-Riemannian manifold with n > 2. If M admits two CTs, one of which is an ICT, and at each point they have no non-trivial common invariant subspaces, then M is a space of constant curvature.

For applications to general relativity, see [Gro11].

6.4 Concircular vectors*

Concircular vectors aren't directly useful for separation of variables, so this section is mainly optional. Although one indirect use follows from the fact that concircular vectors can be used to obtain CTs in spaces of constant curvature (see Corollary 6.1.4). Suppose $r \in \mathfrak{X}(M)$ is a *concircular vector* (CV), then Equation (6.0.1) must be satisfied, i.e

$$\nabla_x r = \phi x \tag{6.4.1}$$

for all $x \in \mathfrak{X}(M)$ and some fixed $\phi \in \mathcal{F}(M)$. The material we present here is mainly from [Cra07] where results on CTs were obtained by using the corresponding results on CVs as motivation. See also section 3.4 in [Ami03], which contain more references and other applications of these vectors.

We first explain where this object comes from, following Crampin in [Cra07]. A concircular transformation of a pseudo-Riemannian manifold (M, g) is a conformal transformation $(g \to \rho^2 g)$ which maps circles into circles (see Section 3.2) [Yan40]. It was shown by Yano in [Yan40] that in the Riemannian case, a necessary and sufficient condition for this is that $\nabla \rho$ satisfy Eq. (6.4.1). Concircular tensors are generalizations of concircular vectors to higher valence.

We give some examples as follows:

Example 6.4.1

Any covariantly constant vector field gives a trivial example of a CV with $\phi \equiv 0$.

Example 6.4.2 (The Dilatational vector field in \mathbb{E}^n_{μ})

A non-trivial prototypical example is given in pseudo-Euclidean space as follows. Set $M = \mathbb{E}_{\nu}^{n}$ and let (x^{i}) be Cartesian coordinates for M. Let $r := \sum_{i} x^{i} \partial_{i}$, then for $v \in \mathfrak{X}(M)$

$$(\nabla_v r)^j = v^i \partial_i x^j = v^j$$

Hence r is a concircular vector with $\phi \equiv 1$, which is known as the *dilitational vector* field (in \mathbb{E}^n_{ν}).

We can easily calculate the general CV in \mathbb{E}_{ν}^{n} :

Proposition 6.4.3 (Concircular vectors in \mathbb{E}^n_{ν} [Cra07])

A vector $v \in \mathfrak{X}(\mathbb{E}^n_{\nu})$ is a CV in \mathbb{E}^n_{ν} iff there exists $a \in C^0_0(\mathbb{E}^n_{\nu})$ and $b \in C^1_0(\mathbb{E}^n_{\nu})$ such that

$$v = ar + b$$

where r is the dilatational vector field.

PROOF In \mathbb{E}^n_{ν} with canonical Cartesian coordinates (x^i) , Eq. (6.4.1) becomes:

$$\frac{\partial v^i}{\partial x^j} = \phi \delta^i{}_j$$

This equation can be easily solved by observing the following:

$$\frac{\partial \phi}{\partial x^k} \delta^i{}_j = \frac{\partial^2 v^i}{\partial x^k \partial x^j} = \frac{\partial \phi}{\partial x^j} \delta^i{}_k$$

Thus taking $i = j \neq k$, we find that $\frac{\partial \phi}{\partial x^k} = 0$. Thus $\phi \in \mathbb{R}$ and we find that v must have the form given by $v^i = \phi x^i + b^i$ where each $b^i \in \mathbb{R}$.

Then using Corollary 6.1.4 we can deduce the general CT in \mathbb{E}^n_{μ} :

Proposition 6.4.4 (Concircular 2-tensors in \mathbb{E}^n_{μ})

L is a concircular 2-tensor in \mathbb{E}^n_{ν} iff there exists $A \in C^2_0(\mathbb{E}^n_{\nu})$, $w \in C^1_0(\mathbb{E}^n_{\nu})$ and $m \in C^0_0(\mathbb{E}^n_{\nu})$ such that:

$$L = A + 2w \odot r + mr \odot r$$

where r is the dilatational vector field. The tensors A, w and m are uniquely determined by L. \Box

The dilatational vector field in \mathbb{E}_{ν}^{n} gives us some intuition for CVs. The following proposition shows that many of the properties held by the dilatational vector field are shared by general CVs [Cra07].

Proposition 6.4.5 (Properties of Concircular vectors)

Suppose $r \in \mathfrak{X}(M)$ is a concircular vector. Let $E := r^{\perp}$ and $r^2 = \langle r, r \rangle$, then the following statements are true:

- 1. r is a conformal Killing vector with conformal factor ϕ .
- 2. $\nabla_r r = \phi r$, so the integral curves of r are affinely parameterized geodesics.
- 3. $dr^{\flat} = 0$, so r is of gradient-type.
- 4. If r is non-null, then E is a Killing distribution, i.e. E is spherical with geodesic orthogonal complement.

5.
$$dr^2(E) = 0$$

PROOF The first property follows since we have shown that concircular tensors are conformal Killing tensors. The second property follows by definition.

To verify the third property, suppose $x, y \in \mathfrak{X}(M)$ and [x, y] = 0, then

$$dr^{\flat}(x,y) = x(\langle y,r\rangle) - y(\langle x,r\rangle)$$

= $\langle \nabla_x y,r\rangle + \langle y,\nabla_x r\rangle - \langle \nabla_y x,r\rangle - \langle x,\nabla_y r\rangle$
= $\langle [x,y],r\rangle + \phi \langle y,x\rangle - \phi \langle x,y\rangle$
= 0

The last two properties follow from the fact that if r is non-null, then $r \odot r$ is an orthogonal concircular tensor. Although one can prove these properties directly, for example if $x \in \Gamma(E)$, then

$$\begin{aligned} x(r^2) &= 2 \left\langle \nabla_x r, r \right\rangle \\ &= 2 \left\langle \phi x, r \right\rangle \\ &= 0 \end{aligned}$$

thus property 5 holds even if r is null.

Note that property 3 implies that a non-null CV r naturally induces a warped product. In fact, if r is non-null, then $r \odot r$ is an OCT, hence from Proposition 6.3.6, we can choose an adapted warped product metric to be:

$$g = r^2 g' + r^2 \tilde{g}$$

6.5 KEM webs

Our main motivation for working with concircular tensors is because they are invariants of Kalnins-Eisenhart-Miller (KEM) webs. In this section we will define KEM webs and show that they are separable. We will see throughout this chapter that several questions concerning KEM webs can be answered using their defining concircular tensors.

Before we introduce the general notion of a KEM web, we first present the following simple motivating example:

Example 6.5.1 (KEM webs)

In this example we work in \mathbb{E}^3 with the CT $L = d \odot d$ where $d \neq 0$ is a constant vector. In this case, L has a simple eigenspace $S_1 := \operatorname{span}\{d\}$ and a multidimensional eigenspace $D_1 := d^{\perp}$. Clearly a warped product manifold adapted to the WP-net (S_1, D_1) is $\mathbb{E}^1 \times \mathbb{E}^2$.

Now in \mathbb{E}^2 we can specify a Cartesian coordinate system via the CT L = A where A is symmetric, constant and has simple eigenspaces. We can also specify polar coordinates via the CT $L = r \odot r$ where r is the dilatational vector field as in the previous section. In both cases it is well known that this defines a separable web \mathcal{E}_1 in \mathbb{E}^2 .

Back in \mathbb{E}^3 we can define an orthogonal web, \mathcal{E} , formed by S_1 together with the lift of \mathcal{E}_1 (which is obtained by translating \mathcal{E}_1 along d). In the first case we obtain a web defining Cartesian coordinates and in the second case we obtain a web defining cylindrical coordinates, both of which are separable.

We have shown two examples where an orthogonal (in fact separable) web was obtained recursively using concircular tensors. For low dimensions we define a KEM web as follows: When n = 1 the tangent bundle TM itself is trivially defined to be a KEM web. When n = 2 any non-trivial OCT has simple eigenfunctions, hence is a Benenti tensor and defines an orthogonal web. So when n = 2 we define a KEM web to be any orthogonal web associated with a Benenti tensor. In the general case we define recursively a KEM web as follows:

Definition 6.5.2 (KEM web)

Let L be a non-trivial OCT with simple eigenspaces $(S_i)_{i=1}^k$ and multidimensional eigenspaces $(D_i)_{i=1}^l$. For each i = 1, ..., l, let \mathcal{E}_i be a KEM web on an integral manifold of D_i . Then the web formed by $(S_i)_{i=1}^k$ together with the lifts of \mathcal{E}_i is called a *Kalnins-Eisenhart-Miller (KEM) web*.

Remark 6.5.3

One can check that the above definition is well-defined since each D_i is necessarily integrable and the lift of \mathcal{E}_i is necessarily an orthogonal web at least locally.

Theorem 6.5.4 (KEM webs)

A KEM web is a separable web.

PROOF Suppose inductively that this theorem holds for all KEM webs with dimension k < n and note that the statement trivially holds for k = 1 since the metric is always an OCT. Now we prove the proposition for KEM webs of dimension $n > k \ge 1$.

Let L be the OCT in the definition of the KEM web and let K be the KBDT associated with L (see Eq. (6.3.1)). Let $D_1, ..., D_l$ be the multidimensional eigenspaces of L. These are necessarily Killing distributions by Proposition 6.3.6. Then the net formed by $D_1, ..., D_l$ together with $D_0 := \bigcap_{i=1}^l D_i^{\perp}$ is a WP-net. So fix $\bar{p} \in M$ and let $N = \prod_{i=0}^l N_i$ be a connected product manifold adapted to this net and passing through \bar{p} . For each i = 1, ..., l, let K_i be a ChKT for \mathcal{E}_i on N_i which is given by Theorem 5.3.2. It follows from Proposition 4.5.2 that K_i can be extended to a KT on M (which we call K_i). After adding a constant

multiple of the induced metric on N_i to K_i if necessary, we can assume that $K + \sum_{i=1}^{l} K_i$ is a ChKT at least locally. Since $K + \sum_{i=1}^{l} K_i$ is a ChKT for this KEM web, it follows from Theorem 5.3.2 that this KEM web is a separable web.

Thus the result follows by induction.

KEM webs are associated to the separable coordinate systems originally discovered by Kalnins and Miller (see [Kal86]) for spaces of constant curvature. We will show later on that in spaces of constant curvature, the converse of the above theorem is true (see Theorem 7.1.1). We also note here that it's clear from the proof above, that (in principle) OCTs can be used to construct separable webs which are not necessarily KEM webs. Although KEM webs will be the most straightforward to analyze, so we will work exclusively with them.

Any coordinates adapted to a KEM web are called *KEM coordinates*. We now show how to reduce the problem of obtaining KEM coordinates for a KEM web to the special case of webs associated with Benenti tensors. Fix a non-trivial OCT L as in the definition of a KEM web and assume it has a multidimensional eigenspace. Let $\psi: N_0 \times_{\rho_1} N_1 \times \cdots \times_{\rho_k} N_k \to M$ be a local warped product decomposition adapted to the WP-net induced by the multidimensional eigenspaces of L. Since $L|_{N_0}$ is a Benenti tensor, there exist local coordinates (x_0) on N_0 which diagonalize $L|_{N_0}$. Inductively assume for each i > 0 that $(x_i) = (x_i^1, \ldots, x_i^{n_i})$ are separable coordinates for N_i adapted to the KEM web \mathcal{E}_i . Then the above theorem shows that the product coordinates $\psi(x_0, x_1, \ldots, x_k)$ are separable coordinates for M. Some examples of this construction are given in Sections 2.3.2 and 9.6.2.

Hence constructing KEM coordinates reduces to constructing coordinates adapted to a Benenti tensor. For Benenti tensors which are also ICTs, canonical coordinates can be constructed for the associated webs (see the discussion following Definition 6.3.3). The case for more general Benenti tensors can be reduced to ICTs using appropriate warped products (see Proposition 6.2.2 (2)).

We now give another property of KEM coordinates which will be important for their further study. First we need a definition. An orthogonal coordinate system is said to have *diagonal curvature* if the Riemann curvature tensor satisfies $R_{ijik} = 0$ for $j \neq k$ in the coordinate induced basis. This definition is equivalent to requiring the curvature operator (which is a $\binom{2}{2}$ -tensor associated with R which induces a map in $\text{End}(\wedge^2(M))$ [Pet06]) to be diagonal in the coordinate induced basis. One can also check that the diagonal curvature condition implies that the Ricci tensor is diagonalized. Now, we have the following result.

Proposition 6.5.5

KEM coordinates have diagonal curvature, and hence they diagonalize the Ricci tensor. \Box

PROOF Assume that (x^i) are KEM coordinates. First, observe that the metric necessarily has the following form:

$$g = \sum_{a \in M} e_a \rho_a^2 \mathrm{d} x_a^2 + \sum_{I \in P} \rho_I^2 g^I$$

where $\{1, \ldots, n\} = M \cup (\bigcup_{I \in P} I)$ is a partition (here *P* is an index set and each $I \in P$ is a subset of $\{1, \ldots, n\}$), each $e_a = \pm 1$ as the case may be and each $\rho_i(x_1, \ldots, x_m)$ is a positive valued function and each g^I has the following form:

$$g_{ij}^{I} = \begin{cases} f_{ij}^{I}(x^{I}) & i, j \in I \\ 0 & i \notin I \end{cases}$$

Define \tilde{g} as follows:

$$\tilde{g} = \sum_{a \in M} e_a \mathrm{d}x_a^2 + \sum_{I \in P} g^I$$

Let R (resp. \tilde{R}) denote the Riemann curvature tensor of g (resp. \tilde{g}). Then for $i \in I, j \in J, k \in K$ with i, j, k distinct, it follows from Eq. (3.5.2) that

$$\left\langle (R(\partial_i,\partial_k) - \tilde{R}(\partial_i,\partial_k))\partial_j,\partial_i \right\rangle = g(\partial_i,\partial_i)(\left\langle \nabla_{\partial_k}U_I - \left\langle \partial_k,U_I \right\rangle U_I,\partial_j \right\rangle)$$

where $U_I = -\nabla \log \rho_I$ is the negative gradient of $\log \rho_I$. By using Eq. (3.5.1), we get the following:

$$\langle \nabla_{\partial_k} U_I - \langle \partial_k, U_I \rangle U_I, \partial_j \rangle = (\partial_k \log \rho_J \partial_j + \partial_j \log \rho_K \partial_k) \log \rho_I - (\partial_k \log \rho_I) (\partial_j \log \rho_I)$$

The above vanishes if either $j \notin M$ or $k \notin M$. So we can assume further that I, J, K are distinct. Then from a direct calculation using the specific form of the twist functions (see Proposition 6.3.6), it follows that the above is identically zero in this case. Thus we have proven that if i, j, k are distinct, then

$$\langle R(\partial_i, \partial_k)\partial_j, \partial_i \rangle = \left\langle \tilde{R}(\partial_i, \partial_k)\partial_j, \partial_i \right\rangle$$

First observe that $R_{ijik} = \langle R(\partial_i, \partial_k) \partial_j, \partial_i \rangle$ and we can assume i, j, k are distinct to check the diagonal curvature condition. Also note that $\tilde{R}(\partial_i, \partial_k)\partial_j$ is not necessarily zero only if I = J = K or if $i, j, k \in M$ (see Proposition 3.5.6). In the later case, clearly $R_{ijik} = 0$. In the former case the result follows by induction from the above equation.

Remark 6.5.6

When the KEM coordinates are generated by an ICT, this fact was originally shown by Crampin in [Cra03]. We were motivated to generalize this fact after observing that all separable coordinates derived by Kalnins and Miller (see [Kal86]) are KEM coordinates and they have diagonal curvature.

Remark 6.5.7

When the coordinates are separable, the additional condition that the Ricci tensor is diagonal is known as the *Robertson condition* [Eis34]. It is known that if this condition is satisfied, then the free particle Schördinger equation (a.k.a Helmholtz equation) is separable [Ben02; Eis34]. Eisenhart first gave this characterization (in terms of the Ricci tensor) in [Eis34]. The above proposition shows that KEM coordinates satisfy the Robertson condition.

Remark 6.5.8

In a space of constant curvature any orthogonal coordinates have diagonal curvature (see Eq. (1.4.1)). This is a crucial observation that enabled Eisenhart [Eis34] and then Kalnins and Miller [KM86; KM82] to classify orthogonal separable coordinates in these spaces. We will use this observation to give an independent classification of these coordinates in Chapter 7.

The fact that KEM coordinates are orthogonal separable with diagonal curvature is almost sufficient to characterize them. We will observe this in Chapter 7 when we will solve for all such metrics. The solution from that chapter will show that the Schwarzschild metric is orthogonally separable and has diagonal curvature. Although one can deduce from Proposition 6.3.6 that the separable web associated with this metric is not a KEM web. Furthermore, not all orthogonal separable coordinates have diagonal curvature. A simple counter-example is given by the Liouville metric [Cra05]. The Liouville metric is conformally Euclidean with each $g_{ii} = \varphi_1(x^1) + \cdots + \varphi_n(x^n)$. This metric is a classic example of an orthogonally separable metric; one can verify this using the Levi-Civita equations (see Eq. (5.2.3a)). One can show that this metric has diagonal curvature iff $\varphi'_j \varphi'_k = 0$ for each $j \neq k$ (see Eq. (3.5.2)).

6.6 The Killing-Stackel space of KEM webs

In this section we show how one can obtain Killing tensors which pair-wise commute (algebraically as linear operators) from a concircular tensor. This was the original motivation for studying Benenti tensors [Ben92a]. In contrast with Benenti's approach in which elementary symmetric polynomials are used, we will make use of the coordinate-independent theory of the cofactor tensor summarized in Appendix A.

Theorem 6.6.1 (Killing tensors from Concircular tensors [Cra03])

If L is a non-singular concircular tensor then K := cof(L) is a Killing tensor satisfying the following equation:

$$\nabla_x K = (\nabla_x \mu) K - \nabla \mu \odot K x$$

for all $x \in \mathfrak{X}(M)$ where $\mu = \log |\det L|$.

PROOF We follow the proof of Theorem 3.1 in [Ben05]. Recall (Eq. (A.0.3)) that K = cof(L) satisfies

$$KL = LK = (\det L)I \tag{6.6.1}$$

Since K is a polynomial in L (see Eq. (A.0.4)), it follows that K is self-adjoint. Also note that Proposition B.1.1 implies that

$$Ld(\det L) = (\det L)d(\operatorname{Tr} L) \qquad \Rightarrow Kd(\operatorname{Tr} L) = d(\det L)$$

Differentiating Eq. (6.6.1) with respect to $x \in \mathfrak{X}(M)$ gives:

$$\nabla_x KL + K \nabla_x L = \nabla_x (\det L) I$$

Right multiplying by K gives:

$$(\det L)\nabla_x K = \nabla_x (\det L)K - K(\nabla_x L)K$$
$$= \nabla_x (\det L)K - \frac{1}{2}K(x \otimes \operatorname{d}\operatorname{Tr} L + \nabla\operatorname{Tr} L \otimes x^{\flat})K$$
$$= \nabla_x (\det L)K - \frac{1}{2}(Kx \otimes \operatorname{d}\det L + \nabla\det L \otimes (Kx)^{\flat})$$

which gives us the equation for $\nabla_x K$. Symmetrizing ∇K proves that K is a Killing tensor.

In our discussion, for a CT L, we will let $K_a := (\wedge^{n-1}L^a)^{\wedge *}$ which has been defined in Appendix A. We call the tensors K_0, \ldots, K_{n-1} the *L*-sequence generated by L. Note that these tensors satisfy a number of identities given in Appendix A. By working with an equivalent CT, L+cG, for some $c \in \mathbb{R}$, we can assume (locally) that the CT is non-singular. Then by Eq. (A.0.1) and the above proposition, we see that each tensor K_a is a Killing tensor. Furthermore we note that by Eq. (A.0.4), each of these tensors are polynomials in L. These observations allow us to obtain a generalization of Benenti's theorem [Ben04] for an arbitrary OCT:

Theorem 6.6.2 ([Ben04])

Suppose L is an OCT with k distinct eigenspaces. Then L generates a k dimensional space of orthogonal Killing tensors which are point-wise independent, which algebraically and Schouten commute. This space of Killing tensors is $\mathcal{K} = \text{span}\{K_0, \ldots, K_{n-1}\}$ where each K_a was defined above.

In particular, if L is a Benenti tensor then \mathcal{K} is the KS-space associated with the separable web induced by L.

PROOF Since each tensor K_a is a polynomial in L, it follows that they commute algebraically and are simultaneously diagonalized in any coordinate system which diagonalizes L. Thus it follows by Proposition 4.4.12 that any $K, J \in \mathcal{K}$ Schouten commute.

We now calculate the dimension of this space. First note that $\dim \mathcal{K}$ is the (abstract vector space) dimension of \mathcal{K} , i.e. not the point-wise dimension. Fix a point $p \in M$. Let $\lambda_1, \ldots, \lambda_k$ be the eigenvalues of L at p. Then by using Proposition A.0.12, we can obtain Killing tensors $\tilde{K}_1, \ldots, \tilde{K}_k$ such that at p, \tilde{K}_i has eigenspaces E_i and E_i^{\perp} . This implies that $\tilde{K}_1, \ldots, \tilde{K}_k$ must be independent KTs in a neighborhood of p. The fact that they are elements of \mathcal{K} follows from Lemma A.0.5. This result together with the fact that the elements of \mathcal{K} can be simultaneously diagonalized with L implies that the point-wise dimension of \mathcal{K} is k. It also implies that $\dim \mathcal{K} \geq k$. In particular when L is a Benenti tensor, $\dim \mathcal{K} = n$, and the tensors K_0, \ldots, K_{n-1} are independent.

We now consider the case L is a non-trivial OCT. For convenience, we assume that L has one multidimensional eigenspace D. Let $B \times_{\rho} F$ be a local connected warped product adapted to (D^{\perp}, D) with adapted contravariant metric $G = G' + \rho^{-2}\tilde{G}$. By Proposition 6.2.2 (5) it follows that $\tilde{L} := L|_B$ is a Benenti tensor. If \tilde{K}_a denote the Killing tensors in the associated L-sequence, it follows² by Proposition 5.3.10 that these tensors admit extensions \tilde{K}'_a to KTs on M which have the form:

$$\tilde{K}'_a = \tilde{K}_a + t_a \tilde{G}$$

where $dt_a = \tilde{K}_a d\kappa$. Let K_α denote the Killing tensors in the L-sequence generated by L. Since the tensors \tilde{K}_a form a basis for the KS-space generated by \tilde{L} , it follows by Proposition 4.5.2 that there exist constants $A_{\alpha}{}^a$ such that

$$K_{\alpha} = \sum_{a} A_{\alpha}{}^{a} (\tilde{K}_{a} + t_{a} \tilde{G}) + c_{\alpha} \tilde{G}$$

for some $c_{\alpha} \in \mathbb{R}$. This proves that $\dim \mathcal{K} \leq k$, hence $\dim \mathcal{K} = k$. The general case follows similarly.

²See the discussion following the proof for more details.

The last fact concerning the case when L is a Benenti tensor follows from Corollary 5.3.3.

As a corollary, we see that if L is a Benenti tensor, then the L-sequence forms a basis for the KS-space generated by L. The proof also shows us how to obtain a basis for \mathcal{K} in the more general case. Indeed, suppose $M = B \times_{\rho} F$ is a warped product with adapted contravariant metric $G = G' + \tilde{G}$ and L is a Benenti tensor on B satisfying:

$$L(\operatorname{d}\log\rho) = \operatorname{d}(c\log\rho + \frac{1}{2}\operatorname{tr}(L))$$
(6.6.2)

for some $c \in \mathbb{R}$. Then $\overline{L} := L + c\tilde{G}$ is a CT on M by Proposition 6.3.7. Hence by Remark 6.2.3 we see that $\overline{K} := \operatorname{tr}(\overline{L})G - \overline{L}$ is KT on M which pulls back to a ChKT on B. Thus by Proposition 5.3.10, the KS-space generated by L on B can be extended into M. The following proposition explicitly gives this extended KS-space.

Proposition 6.6.3 (Extension of the L-sequence)

Suppose $B \times_{\rho} F$ is a warped product and L is a Benenti tensor on B as above. Then the L-sequence can be extended into independent KTs on M given as follows in contravariant form:

$$\bar{K}_a := K_a + (\sum_{i=0}^a (-c)^i \sigma_{a-i}) \tilde{G}$$

for each $0 \le a \le m-1$ where $m = \dim B$ and $\sigma_a := (\wedge^m L^a)^{\wedge *}$. In terms of \overline{L} , they have the following form:

$$\bar{K}_a = \sum_{i=0}^a (-1)^i \sigma_{a-i} \bar{L}^i \qquad \diamondsuit$$

PROOF Throughout this proof we work exclusively on B. Denote by K_a the L-sequence generated by L. It follows by Eq. (A.0.2) that they satisfy the following equations:

$$K_0 = I, K_a = \sigma_a I - K_{a-1} L 1 < a < m (6.6.3)$$

Furthermore since L is torsionless, Eq. (B.1.1) from Proposition B.1.2 implies the following:

$$K_{a-1}\mathrm{d}\operatorname{tr}(L) = \mathrm{d}\sigma_a \tag{6.6.4}$$

We now proceed to calculate the function t in Eq. (4.5.1) for each K_a . For a > 1, by using Eqs. (6.6.2) and (6.6.3) we have

$$K_a \mathrm{d} \log \rho = \sigma_a \mathrm{d} \log \rho - K_{a-1} \mathrm{d} (c \log \rho + \frac{1}{2} \mathrm{tr}(L))$$
$$\stackrel{(6.6.4)}{=} \sigma_a \mathrm{d} \log \rho - c K_{a-1} \mathrm{d} \log \rho - \frac{1}{2} \mathrm{d} \sigma_a$$

Then if we let $\kappa := \rho^{-2}$, we have

$$K_{a}d\kappa = \kappa K_{a}d\log \kappa$$

= $-2\kappa K_{a}d\log \rho$
= $\sigma_{a}d\kappa - cK_{a-1}d\kappa + \kappa d\sigma_{a}$
= $d(\kappa\sigma_{a}) - cK_{a-1}d\kappa$ (6.6.5)

which gives us a recursive equation for $K_a d\kappa$. We solve it as follows:

$$K_1 d\kappa = d(\kappa \sigma_1) - c d\kappa$$

= d((\sigma_1 - c)\kappa)
$$K_2 d\kappa = d(\kappa \sigma_2) - c d((\sigma_1 - c)\kappa)$$

= d((\sigma_2 - c(\sigma_1 - c))\kappa)
$$\Rightarrow K_a d\kappa = d((\sum_{i=0}^{a} (-c)^i \sigma_{a-i})\kappa)$$

We check the above equation for $K_a d\kappa$ using induction:

$$K_a d\kappa \stackrel{(6.6.5)}{=} d(\kappa \sigma_a) - c K_{a-1} d\kappa$$
$$= d((\sigma_a - c(\sum_{i=0}^{a-1} (-c)^i \sigma_{a-1-i}))\kappa)$$
$$= d((\sigma_a + (\sum_{i=0}^{a-1} (-c)^{i+1} \sigma_{a-1-i}))\kappa)$$
$$= d((\sum_{i=0}^{a} (-c)^i \sigma_{a-i})\kappa)$$

Thus it follows by Proposition 4.5.2 that each \bar{K}_a is a KT on M. The second formula

for \overline{K}_a follows from Eq. (A.0.4).

Remark 6.6.4

The conclusions follow as long as L is a CT.

Thus the above two propositions together with Proposition 5.3.12 reduces the calculation of the KS-space for KEM webs to algebraic operations (provided the appropriate warped product decompositions are known). We also note that one can construct a ChKT for any KEM web using the KBDTs associated with the defining CTs. This construction is elaborated in the proof of Theorem 6.5.4.

We now end this section with the expected result that the KS-space generated by an ICT is not reducible. In order to do this, it is sufficient to first work with orthogonal CKTs. Suppose L is an orthogonal CKT with orthogonally integrable eigenspaces $E_1, ..., E_k$ and associated eigenfunctions $\lambda_1, ..., \lambda_k$. Let $D_c \subset \{1, \ldots, k\}$ with $|D_c| \geq 2$ and define $D = \bigoplus_{i \in D_c} E_i$. We will need the following lemma which gives the mean curvature normal of D:

Lemma 6.6.5

Suppose $\mathcal{E} = (E_i)_{i=1}^k$ is a TP-net and let $D = \bigoplus_{i \in D_c} E_i$ (similar to above), then the mean curvature normal H_D of D is given as follows:

$$H_D = \sum_{i \in D_c} \frac{m_i}{d} H_i^{\perp D}$$

where H_i is the mean curvature normal of E_i , $m_i = \dim E_i$ and $d = \dim D$. In particular D is umbilical iff $H_i^{\perp D} = H_D = H_j^{\perp D}$ for each $i, j \in D_c$.

PROOF By Eq. (3.5.1), the second fundamental form of D is

$$h(x,y) = \sum_{i \in D_c} \left\langle x^i, y^i \right\rangle U_i^{\perp L}$$

Hence the formula for H_D follows since $H_i^{\perp D} = U_i^{\perp D}$ for each $i \in D_c$. Then D is umbilical iff $H_D = H_i^{\perp D}$ for each $i \in D_c$, hence the result follows.

We now assume that D is umbilical. From the above lemma, D is umbilical iff $H_i^{\perp D} = H_j^{\perp D}$ for each $i, j \in D_c$. Thus from Corollary 4.4.6, the following equation must be satisfied for $i, j \in D_c$ and $k \notin D_c$.

$$(\nabla \log |\lambda_i - \lambda_k|)^k = (\nabla \log |\lambda_j - \lambda_k|)^k$$
$$\Leftrightarrow (\lambda_j - \lambda_k)(\nabla(\lambda_i - \lambda_k))^k = (\lambda_i - \lambda_k)(\nabla(\lambda_j - \lambda_k))^k$$

Thus $\log \left| \frac{\lambda_i - \lambda_k}{\lambda_j - \lambda_k} \right|$ is independent of E_k . Now if L is a torsionless orthogonal CKT this condition simplifies to

$$(\lambda_j - \lambda_k) \nabla \lambda_k = (\lambda_i - \lambda_k) \nabla \lambda_k$$
$$\Leftrightarrow (\lambda_j - \lambda_i) \nabla \lambda_k = 0$$
$$\Leftrightarrow \nabla \lambda_k = 0 \quad \text{for each } k \notin D_c$$

Hence we have the following:

Proposition 6.6.6 (Some Geometrical properties of torsionless CKTs)

Suppose L is a torsionless orthogonal CKT with orthogonally integrable eigenspaces. If a distribution D (constructed as above) containing at least two eigenspaces is umbilical, then λ_k is a constant for each $k \notin D_c$. Thus we can deduce the following:

- 1. Any such distribution D induces a warped product with D as the geodesic distribution and the eigenspaces complementary to D as the Killing distributions.
- 2. In particular if dim D = n 1, then D^{\perp} is tangent to a Killing vector field.
- 3. If the eigenfunctions of L are functionally independent then any L-invariant distribution with dimension greater than one is not umbilical.

Remark 6.6.7

This proposition contains a well known property of Benenti tensors stated in some articles [Ben05; CRA07]. Namely that a simple eigenspace of a torsionless orthogonal CKT cannot be tangent to proper CKV (a CKV which is not a KV) if its orthogonal complement contains more than one eigenspace. This follows directly from property 2 above and the fact that the orthogonal complement of normal non-null CKV is umbilical. \Box

Hence the above proposition shows that the KS-space generated by an ICT is not reducible. In fact, even more can be said:

Corollary 6.6.8 (The KS-space of an ICT)

Suppose L is an ICT. If K is in the Killing-Stäckel space generated by L, then K is either a constant multiple of the metric or characteristic. \diamond

PROOF By hypothesis K is diagonalized in any coordinate system adapted to the eigenspaces of L. If K has an eigenspace D which has dimension d satisfying 1 < d < n then D is umbilical by Corollary 4.4.6. Also D is a direct sum of at least two eigenspaces of L, hence by Proposition 6.6.6 at least one of the eigenfunctions of L must be a constant, a contradiction. Thus we conclude that either K is characteristic or has a single eigenspace in which case it must be a constant multiple of the metric.

Remark 6.6.9

This corollary highlights a weakness of the intrinsic characterization of separable webs using ChKTs (Theorem 5.3.2). The non-uniqueness of ChKTs makes it more difficult to classify the associated webs using them. This is in sharp contrast with CTs, see Proposition 6.2.5. Although the cost of working with CTs is that, in general, we have to work with multiple CTs. \Box

Proposition 6.6.6 can also be used to derive properties of the Killing-Stäckel space, \mathcal{K} , of a Benenti tensor. For example, suppose L is a Benenti tensor with only one constant eigenfunction and the rest of which are non-constant. In this case \mathcal{K} contains a rank 1 KT, say V, which is the tensor product of a KV with itself by Proposition 6.2.2. Then one can use Proposition 6.6.6 to deduce that all KTs in \mathcal{K} must be either a constant multiple of the metric, a KT sharing the same eigenspaces as V, or a ChKT.

6.7 Separation in KEM webs: The BEKM Separation Algorithm

In this section we will present the *Benenti-Eisenhart-Kalnins-Miller (BEKM) separation* algorithm, which is named after the researchers whose work anticipated this algorithm [Ben05; Eis34; KM86]. We fix a potential $V \in \mathcal{F}(M)$ and suppose $n = \dim M > 1$. We present a tractable intrinsically defined algorithm to determine separability of the natural Hamiltonian associated with V in a KEM web.

This algorithm is developed using the structure of KEM webs. In the proof of Theorem 6.5.4, we showed how to construct a ChKT for a KEM web using KBDTs associated with the defining CTs. We now observe that given a KEM web \mathcal{E} , the KBDT, K', associated with the first CT defining this web is in the KS-space associated with \mathcal{E} . Thus by Theorem 5.4.1 any potential separable in \mathcal{E} must satisfy the dKdV equation with K'. We use these observations and the theory of the separation of the Hamilton-Jacobi equation in warped products (see Section 5.4.1) to obtain a recursive algorithm to find separable coordinates for V.

Remark 6.7.1

The authors originally discovered the necessity of KBDTs for \mathbb{E}^n and \mathbb{S}^n implicitly through Corollary 5.4 in [WW03]. Indeed, according to the remarks following Equation 4.2 in [Ben04], the Bertrand-Darboux equations in [WW03] are the dKdV equations generated by a KBDT. Hence Corollary 5.4 in [WW03] implies the necessity of KBDTs for the special case of \mathbb{E}^n . Corollary 5.4 in [WW03] also implies a similar statement for \mathbb{S}^n . This explains the origin of the name Bertrand-Darboux in Killing-Bertrand-Darboux tensor and one of our initial reasons for working with CTs. Now we present the BEKM separation algorithm, so assume M is an arbitrary pseudo-Riemannian manifold. Let L denote the general concircular tensor on (M, g) and K := tr(L)G - L be the KBDT generated by L. Now impose the condition:

$$d(KdV) = 0 \tag{6.7.1}$$

which is called the *Killing Bertrand-Darboux (KBD)* equation. The above equation defines a system of linear equations in the unspecified parameters of L. Indeed, by Theorem 6.1.2, the C-tensors form a finite-dimensional vector space. Since the KBDT is linearly related to L, it follows that the above equation defines a linear system. Furthermore by Theorem 6.1.2 the maximum number of unknowns in the above equation is $\frac{1}{2}(n+1)(n+2)$.

Suppose now that K is a particular solution of the KBD equation and let L be the associated C-tensor. We make the assumption that L is an orthogonal tensor (which is always satisfied on a Riemannian manifold). Let $(E_i)_{i=1}^k$ be the eigenspaces of L and $(\lambda_i)_{i=1}^k$ the corresponding eigenfunctions. We now classify such a solution:

- **Case 1** (k = 1, i.e. all the eigenfunctions coincide) In this case L = cG where $c := \lambda_1 \in \mathbb{R}$, thus the associated KBDT, K = c(n-1)G is the trivial solution of Eq. (6.7.1) and so the algorithm yields no information.
- Case 2 (the eigenfunctions are simple) K is a characteristic Killing tensor, then by Benenti's theorem (Theorem 5.4.1), Vis separable in the web of the eigenspaces of L.
- **Case 3** (at least one eigenfunction is not simple)

In this case, we enumerate the eigenspaces $D_1, ..., D_l$ with dimension greater than one. Since each D_i is Killing by Proposition 6.3.6, the net formed by $D_1, ..., D_l$ together with $D_0 := \bigcap_{i=1}^l D_i^{\perp}$ is a WP-net. So fix $\bar{p} \in M$ and let $N = \prod_{i=0}^l N_i$ be a connected product manifold adapted to this net and passing through \bar{p} .

If $D_0 \neq 0$, then K restricted to D_0 is characteristic by construction. Let $V_i := \tau_i^* V \in \mathcal{F}(N_i)$ and suppose for each i = 1, ..., k there exists a ChKT \tilde{K}_i on N_i such that $d(\tilde{K}_i dV_i) = 0$.

Then by Theorem 5.4.4, V is separable in the web formed by the simple eigenspaces of L together with the lifts of the simple eigenspaces of $\tilde{K}_1, ..., \tilde{K}_l$.

The algorithm can be applied recursively in the case L has a non-simple eigenfunction. In the notation of case 3 one would have to apply the algorithm to each N_i equipped with the induced metric for i = 1, ..., l.

Now, some remarks are in order:

Remark 6.7.2

In case 3 even if there are no ChKTs on the submanifolds N_i which satisfy the dKdV equation with V_i , one should be able to prove that the Hamilton-Jacobi equation is partially separable. By Theorem 6.6.2, one can at least obtain first integrals for the Hamiltonian.

Remark 6.7.3

Since the metric is always a solution of the KBD equation and because the KBD equation is linear in K, we always consider a solution of the KBD equation modulo multiplies of the metric.

In the following example we will show how to use the theory just presented to show that the Calogero-Moser system is separable in cylindrical coordinates. It was originally shown to be separable in these coordinates by Calogero in [Cal69].

Example 6.7.4 (Calogero-Moser system)

The Calogero-Moser system is a natural Hamiltonian system with configuration manifold \mathbb{E}^3 given by the following potential in Cartesian coordinates (q_1, q_2, q_3) :

$$V = (q_1 - q_2)^{-2} + (q_2 - q_3)^{-2} + (q_1 - q_3)^{-2}$$

First note that the constant vector $d = \frac{1}{\sqrt{3}}[1, 1, 1]$ is a symmetry of V, i.e. $\mathcal{L}_d V = 0$. Hence we observe that the CT $L = d \odot d$ is a solution of the KBD equation associated with V. From Example 6.5.1 we know that a warped product manifold adapted to Lhas the form $\mathbb{E}^1 \times \mathbb{E}^2$. One can choose Cartesian coordinates (q'_1, q'_2, q'_3) adapted to this product manifold, such that V takes the form:

$$V = \frac{9(q_3'^2 + q_2'^2)^2}{2q_2'^2(3q_3'^2 - q_2'^2)^2}$$

In this case V naturally restricts to a potential on \mathbb{E}^2 with coordinates (q'_2, q'_3) . In \mathbb{E}^2 one can apply the BEKM separation algorithm to find that the only solution of the KBD equation (up to constant multiplies) is $L = r \odot r$ where r is the dilatational vector field. Hence we conclude that V is separable in cylindrical coordinates which are obtained by taking polar coordinates $(r \cos(\theta), r \sin(\theta))$ on \mathbb{E}^2 .

The above example will be worked out in greater detail in Section 10.2. When n = 3, we will show that the Calogero-Moser system is separable in four additional coordinate systems. The following example illustrates how one can obtain ChKTs when an ignorable coordinate is present.

Example 6.7.5 (Separation in Static space-times)

A static space time is the product manifold $M = B \times_{\rho} \mathbb{E}^{1}_{1}$ equipped with warped product metric $g = \tilde{g} - \rho^{2} dt^{2}$ where \tilde{g} is a Riemannian metric. By Proposition 5.3.12, M admits a ChKT K with timelike eigenvector field $\frac{\partial}{\partial t}$ iff there exists a ChKT $\tilde{K} \in S^{2}(B)$ satisfying:

$$d(\tilde{K}d\rho^{-2}) = 0$$

This observation is a special case of the connection between separation of potentials and extensions of KTs. We note here that in order to find \tilde{K} , the BEKM separation algorithm can be applied on B with $V := \rho^{-2}$. In particular if B is a space of constant curvature, we will observe immediately after this example that the BEKM separation algorithm gives a complete method for determining \tilde{K} satisfying the above equation if it exists.

Completeness of the algorithm It follows from the definition of the KEM web that if this algorithm is applied recursively then it will always test if the potential is separable in a KEM web. Since it will be proven in Chapter 7 that every separable web in a space of constant curvature is a KEM web, it follows that this algorithm gives a complete test for separability in spaces of constant curvature. Although if one uses a ChKT not associated with a KEM web in case 3 of the algorithm, then one can test for separability against more general separable webs.

Practical Implementation For spaces of constant curvature, we will work out sufficient details in Chapter 9 to concretely implement this algorithm in Section 10.3. To do this, the only problems that remain are the classification of OCTs modulo the action of the isometry group, then obtaining the transformation to Cartesian coordinates for their associated webs and classifying warped product decompositions on these spaces. These problems are solved in Chapter 9.

This algorithm has been implemented concretely in Euclidean and spherical space by Waksjo and Wojciechowski in their solution [WW03]. Their solution which was more classical, involved Stäckel theory and was based on the work of Kalnins-Miller [Kal86]. They made no use of Benenti's modern formulation of the separation of the Hamilton-Jacobi equation [Ben97] in terms of Killing tensors which is independent of Stäckel theory.

Like the algorithm in [WW03], in spaces of constant curvature the BEKM separation algorithm reduces to a series of problems in linear algebra. Although for hyperbolic space and Minkowski space-time, one will have to deal with finding the Jordan canonical form of non-diagonalizable (constant) matrices.

6.8 Notes

Reviews of the theory presented in this chapter are presented in Sections 2.2 to 2.4 and 9.1.1.

In this chapter we have not studied non-orthogonal CTs. Non-orthogonal CTs may be applicable to complex and non-orthogonal separation. In [BM13], a procedure is given to obtain the local canonical (normal) forms for CTs in pseudo-Riemannian manifolds. Hence this procedure may be of interest for those who wish to study non-orthogonal CTs.

Part II

Specialization to Spaces of Constant Curvature

Chapter 7

In Spaces of Constant Curvature: Separable webs are KEM webs

In this chapter we will prove the fundamental result: in spaces of constant curvature, separable webs are KEM webs. This fact can be deduced from the Kalnins-Miller classification [Kal86], but we will give an independent proof which somewhat generalizes their results. This chapter builds on Section 6.5 which we assume the reader is familiar with. The contents of this chapter are from the article [RM14a].

In order to prove this result, we will solve for all orthogonally separable metrics with diagonal curvature. This generalizes results by Eisenhart in [Eis34] and Kalnins and Miller in [Kal86]. Note that this is sufficient to solve for all orthogonal separable coordinates in spaces of constant curvature, since by Eq. (1.4.1), all orthogonal coordinates in these spaces have diagonal curvature.

We now provide an outline of this chapter. In the first section we will summarize the results. In Section 7.2 we present the first steps of the derivation of all orthogonal separable coordinates with diagonal curvature. In Section 7.3 we will finish off this derivation. Finally in Section 7.4, we will do additional calculations in order to prove that all separable coordinates in spaces of constant curvature are KEM coordinates.

7.1 Summary of Results

In this section we will present the results of this chapter in detail and combine them to prove the following fundamental result:

Theorem 7.1.1 (Separable Webs in Spaces of Constant Curvature)

In a space of constant curvature, every separable web is a KEM web.

First we need a preliminary characterization of orthogonal concircular tensors, which is the coordinate form of Proposition 6.3.6. Suppose (x_i) are local coordinates and L is a

tensor defined as follows:

$$L = \sum_{a \in M} \sigma_a \partial_a \otimes \mathrm{d}x_a + \sum_{I \in P} e_I \sum_{i \in I} \partial_i \otimes \mathrm{d}x_i \tag{7.1.1}$$

where $\{1, \ldots, n\} = M \cup (\bigcup_{I \in P} I)$ is a partition (here *P* is an index set and each $I \in P$ is a subset of $\{1, \ldots, n\}$), the $\sigma_a(x_a)$ are non-constant and the e_I are constants. Proposition 6.3.6 states that if *L* is a concircular tensor, then the metric has the following form:

$$g = \sum_{a \in M} \Phi_a \prod_{\substack{b \in M \\ b \neq a}} (\sigma_a - \sigma_b) dx_a^2 + \sum_{I \in P} \left(\prod_{a \in M} (e_I - \sigma_a) \right) g^I$$
$$g_{ij}^I = \begin{cases} f_{ij}^I(x^I) & i, j \in I \\ 0 & i \notin I \end{cases}$$
(7.1.2a)

where Φ_a is a function of x^a only. Conversely, it follows by Proposition 6.3.6, that if the metric has the above form, then L is a CT. It will follow by the proof of our main result (see Section 7.3), that given a metric with the above form, one can construct L such that its eigenspaces are uniquely determined from the metric.

We will see that most orthogonally separable metrics with diagonal curvature have a form given by the above equation, i.e. they admit a concircular tensor diagonalized in the coordinates. We now list the general form of orthogonally separable metrics with diagonal curvature.

The ones having a form given by Eq. (7.1.2) can be divided into the following three classes. The first class are the **irreducible metrics**

$$g = \sum_{a=1}^{n} \Phi_a \prod_{b \neq a} (\sigma_a - \sigma_b) \mathrm{d}x_a^2$$
(7.1.3)

which occur when the eigenfunctions of any associated concircular tensor are functionally independent. These metrics were first found by Eisenhart in his article [Eis34]. The remaining two classes of metrics are referred to as reducible metrics. The following are **product metrics**

$$g = \sum_{I=1}^{p} g^{I}$$
 (7.1.4)

where each g^{I} is given in Eq. (7.1.2a). The final class are the warped product metrics

$$g = \sum_{a=1}^{m} \Phi_a \prod_{\substack{b \le m \\ b \ne a}} (\sigma_a - \sigma_b) \mathrm{d}x_a^2 + \sum_{I=1}^{p} \left(\prod_{a \le m} (e_I - \sigma_a) \right) g^I$$
(7.1.5)

where each g^I is given in Eq. (7.1.2a).

There is one class of orthogonally separable metric with diagonal curvature which is not in general associated with a concircular tensor, it is given as follows:

$$g = \Phi_1 dx_1^2 + \sum_{I=1}^p \sigma_1^I g^I$$
(7.1.6)

where Φ_1, σ_1^I are functions of x_1 at most with each σ_1^I non-constant. In conclusion, every orthogonally separable metric with diagonal curvature has a form given by Eq. (7.1.2) or Eq. (7.1.6). We will show later that if g is an orthogonally separable metric with diagonal curvature, then each of the metrics g^I must also be an orthogonally separable metric with diagonal curvature. This shows why the classification is recursive: if |I| > 1 then our classification will tell us that each g^I must be of the form given by Eq. (7.1.2) or Eq. (7.1.6). Thus one must recursively apply this classification to obtain all orthogonally separable metrics with diagonal curvature for a given dimension.

Using the above classification, we will prove the following theorem concerning orthogonal separation in spaces of constant curvature:

Theorem 7.1.2 (KEM Separation Theorem)

Suppose (M,g) is a space of constant curvature. In orthogonal separable coordinates, g necessarily has the form given by Eq. (7.1.2).

In terms of tensors, suppose K is a characteristic Killing tensor defined on M. Then there is a non-trivial concircular tensor L defined on M such that each eigenspace of K is L-invariant, i.e. L is diagonalized in coordinates adapted to the eigenspaces of K. Furthermore, the eigenspaces of L are uniquely determined by the separable web defined by K.

The above theorem is a generalization of the results due to Kalnins and Miller from [Kal86]; it holds in Lorentzian spaces as well. For Riemannian spaces of constant curvature, this theorem can be proven by connecting the classification of separable metrics given by Kalnins and Miller in [Kal86] with Proposition 6.3.6. Indeed, by examining the separable metrics given in [Kal86], it can be shown that all separable metrics derived in [Kal86] have the form given by Eq. (7.1.2). Then the desired concircular tensor, L, is given by Eq. (7.1.1). For a space of constant curvature with arbitrary signature, we will generalize the classification given by Kalnins and Miller and show that all separable metrics still have the form given by Eq. (7.1.2).

We will apply the above theorem, shortly, to prove Theorem 7.1.1. But first, we note here that Theorem 7.1.1 together with the results presented in [Ben92b] (cf. [Kal86]) allow us to conclude the following:

Theorem 7.1.3 (KEM Separation Theorem II)

Suppose (M, g) is a space of constant curvature with Euclidean signature or Lorentzian¹ signature with positive curvature. Then every separable (not necessarily orthogonal) coordinate system has an orthogonal equivalent which is a KEM coordinate system.

In the above theorem, the term "equivalent" is in the sense of Definition 5.1.5. Precisely, it means that in the aforementioned spaces, every separable solution to the geodesic Hamilton-Jacobi equation induces the same Lagrangian foliation as a separable solution associated with some KEM coordinate system. Now, we will need the following lemma, which will be proven in Section 7.4.

Lemma 7.1.4

In a space of constant curvature, a Killing foliation is a foliation of homothetic² spaces of constant curvature. $\hfill\square$

In particular for \mathbb{E}^n one can show that a Killing foliation is foliation by subsets of (affine) spheres or planes of lesser dimension. We are now ready to prove Theorem 7.1.1. PROOF (THEOREM 7.1.1) Suppose inductively that this theorem holds for all separable webs in spaces of constant curvature of dimension k < n. The statement trivially holds when k = 1. We now show that the theorem holds when dim M = n.

Suppose K is a ChKT defined on a space of constant curvature M defining a separable web. Then let L be a concircular tensor guaranteed by the KEM separation theorem.

Case 1 If L has simple eigenfunctions (i.e. is a Benenti tensor), then it follows that the separable web determined by K is a KEM web.

Case 2 Suppose L has multidimensional eigenspaces D_1, \ldots, D_l ; these must be Killing by Proposition 6.3.6. Thus each D_i induces a foliation of spherical submanifolds of M. Then it follows by Lemma 7.1.4 that this is a foliation of spaces of constant curvature of lesser dimension. Suppose N_i is an integral manifold of D_i . Then it follows from Proposition 4.4.15 that K restricts to a ChKT \tilde{K}_i on N_i . Thus \tilde{K}_i is a ChKT on a space of constant curvature N_i which has dimension less than n. Hence by induction hypothesis, it follows that the separable web \mathcal{E}_i associated with \tilde{K}_i is a KEM web. Thus by definition it follows that the separable web associated with K is a KEM web.

The result then follows by induction on n.

¹We take Lorentzian signature to be $(-+\cdots+)$

²By homothetic pseudo-Riemannian manifolds, we mean a pair of pseudo-Riemannian manifolds whose metrics are related by the equation $h = \lambda^2 g$ where $\lambda \in \mathbb{R}^+$.

7.2 Preliminary results

In this section we will present some relevant calculations from the literature for completeness. In particular, we will partially solve for all orthogonally separable metrics with diagonal curvature. This calculation will be finished in the next section.

We now assume (x^i) are orthogonal separable coordinates with diagonal curvature. Assume the covariant metric $g = \text{diag}(e_1H_1^2, ..., e_nH_n^2)$ where each $e_i = \pm 1$ as the case may be. The assumption of orthogonal separability implies the metric satisfies the Levi-Civita equations (Eq. (5.2.3a)). They take the following form:

$$\frac{\partial^2 \log H_i^2}{\partial x_i \partial x_j} + \frac{\partial \log H_i^2}{\partial x_j} \frac{\partial \log H_j^2}{\partial x_i} = 0$$
(7.2.1a)

$$\frac{\partial^2 \log H_i^2}{\partial x_j \partial x_k} - \frac{\partial \log H_i^2}{\partial x_j} \frac{\partial \log H_i^2}{\partial x_k} + \frac{\partial \log H_i^2}{\partial x_j} \frac{\partial \log H_j^2}{\partial x_k} + \frac{\partial \log H_i^2}{\partial x_k} \frac{\partial \log H_i^2}{\partial x_k} = 0 \quad (7.2.1b)$$

where i, j, and k are all distinct. We will now proceed to solve the above equations in combination with the diagonal curvature condition.

The following calculation is from [Cra03, proposition 6] which is adapted from Kalnins' book [Kal86] which is from [Eis34]. First note that in orthogonal coordinates the Riemann curvature component R_{jiik} for i, j, k distinct has the following form [Eis34]:

$$\begin{split} R_{jiik} &= \frac{e_i H_i^2}{4} \bigg[2 \frac{\partial^2 \log H_i^2}{\partial x_j \partial x_k} + \frac{\partial \log H_i^2}{\partial x_j} \frac{\partial \log H_i^2}{\partial x_k} \\ &\quad - \frac{\partial \log H_i^2}{\partial x_j} \frac{\partial \log H_j^2}{\partial x_k} - \frac{\partial \log H_i^2}{\partial x_k} \frac{\partial \log H_i^2}{\partial x_j} \bigg] \end{split}$$

In consequence of the second integrability condition, Eq. (7.2.1b), we find that:

$$R_{jiik} = \frac{3}{4} e_i H_i^2 \frac{\partial^2 \log H_i^2}{\partial x_i \partial x_k}$$

Thus the diagonal curvature assumption implies that for i,j,k distinct:

$$\frac{\partial^2 \log H_i^2}{\partial x_i \partial x_k} = 0 \tag{7.2.2}$$

Solving the above equation we find that:

$$H_i^2 = \prod_{j \neq i} \Psi_{ij}(x_i, x_j)$$
(7.2.3)

Now the first integrability condition, Eq. (7.2.1a), applied twice to $i \neq j$ implies that:

$$\frac{\partial^2 \log H_i^2}{\partial x_i \partial x_j} = -\frac{\partial \log H_i^2}{\partial x_j} \frac{\partial \log H_j^2}{\partial x_i}$$
(7.2.4a)

$$\frac{\partial^2 \log H_i^2}{\partial x_i \partial x_j} = \frac{\partial^2 \log H_j^2}{\partial x_i \partial x_j}$$
(7.2.4b)

If we substitute the form of H from Eq. (7.2.3) into Eq. (7.2.4b) we have that:

$$\frac{\partial^2 \log \frac{\Psi_{ij}}{\Psi_{ji}}}{\partial x_i \partial x_j} = 0$$

Thus

$$\frac{\Psi_{ij}}{\Psi_{ji}} = \frac{\chi_{ij}(x_i)}{\chi_{ji}(x_j)}$$

If we let $\Phi_{ij} = \Phi_{ji} = \frac{\Psi_{ij}}{\chi_{ij}}$ and $\Phi_i = \prod_{j \neq i} \chi_{ij}$, Eq. (7.2.3) becomes: $H_i^2 = \Phi_i(x_i) \prod_{j \neq i} \Phi_{ij}(x_i, x_j)$

Now if we substitute the form of H above into Eq. (7.2.4a) we have that:

$$\frac{\partial^2 \Phi_{ij}}{\partial x_i \partial x_j} = 0$$

Thus

$$\Phi_{ij}(x_i, x_j) = \sigma_{ij}(x_i) + \sigma_{ij}(x_j)$$

This gives us the following general form of H satisfying Eq. (7.2.1a) and Eq. (7.2.2):

$$H_i^2 = \Phi_i(x_i) \prod_{j \neq i} (\sigma_{ij}(x_i) + \sigma_{ji}(x_j)) \quad (i = 1, ..., n)$$
(7.2.5)

The above equation was first derived by Eisenhart in his seminal paper [Eis34] and it was used resourcefully by Kalnins and Miller in their classification of separable coordinates systems in S^n , E^n and H^n [Kal86]. When n = 2, the above equation gives the general solution and it follows that the metric has the form given by Eq. (7.1.2). Thus for the remainder of this solution we assume n > 2. Now for i, j, k distinct we evaluate Eq. (7.2.1b) with all cyclic permutations of i, j, k using the form of H given above to get the following system of equations:
$$\sigma'_{ji}\sigma'_{ki}(\sigma_{jk}+\sigma_{kj})-\sigma'_{ji}\sigma'_{kj}(\sigma_{ki}+\sigma_{ik})-\sigma'_{ki}\sigma'_{jk}(\sigma_{ij}+\sigma_{ji})=0$$
(7.2.6a)

$$\sigma'_{kj}\sigma'_{ij}(\sigma_{ki}+\sigma_{ik}) - \sigma'_{kj}\sigma'_{ik}(\sigma_{ij}+\sigma_{ji}) - \sigma'_{ij}\sigma'_{ki}(\sigma_{jk}+\sigma_{kj}) = 0$$
(7.2.6b)

$$\sigma'_{ik}\sigma'_{jk}(\sigma_{ij}+\sigma_{ji}) - \sigma'_{ik}\sigma'_{ji}(\sigma_{jk}+\sigma_{kj}) - \sigma'_{jk}\sigma'_{ij}(\sigma_{ki}+\sigma_{ik}) = 0$$
(7.2.6c)

where the primes indicates differentiation. Now since each $\Phi_{ij} = \sigma_{ij} + \sigma_{ji}$ is non-zero, the determinant of the above equations must vanish, this gives us the following equation:

$$\sigma'_{ij}\sigma'_{jk}\sigma'_{ki} + \sigma'_{ji}\sigma'_{kj}\sigma'_{ik} = 0 \tag{7.2.7}$$

We will solve the remaining equations in the next section.

7.3 Classification of orthogonal separable coordinates with diagonal curvature

We continue the derivation started in the previous section. An important subset of coordinates are the coordinates i which satisfy $\sigma'_{ij} \neq 0 \quad \forall j \neq i$. These coordinates will be called **connecting coordinates** for reasons that will become apparent later on. The set of all connecting coordinates for a given separable metric will be denoted by M and we will assume the coordinates are chosen such that $M = \{1, ..., m\}$.

First we give a rough idea of how we will do this classification. When there are no connecting coordinates, we show that metric is necessarily a product metric. When there is at least one connecting coordinate we show that the metric is any one of the other metrics listed in the introduction. In order to prove that the metric is a product metric when it has no connecting coordinates we define a relation among the coordinates. We then prove that this relation is an equivalence relation. Then we use this equivalence relation to prove that the metric has at least one connecting coordinate or is a product metric.

We now define a relation among the coordinates to distinguish between the different possible metrics that can occur. The relation is designed so that if it gives multiple partitions then these partitions are associated with a product metric. Furthermore, we should be able to conclude that the metric is connected if there is only one partition. It's easiest to first define when two coordinates i and j are inequivalent. If I and J are distinct partitions from the product metric in Eq. (7.1.4) and $i \in I$ and $j \in J$, then the first thing to notice is that $\sigma'_{ij} = \sigma'_{ji} = 0$. But with this definition of in-equivalence, if there are multiple partitions, it's still possible that we're dealing with a warped product metric given by Eq. (7.1.5); we need to make sure that there is no third coordinate k such that $\sigma'_{ki}, \sigma'_{kj} \neq 0$. This gives us a definition of equivalence:

Definition 7.3.1

Two distinct variables i and j are said to be connected and denoted $i \sim j$ if one of the following conditions hold:

$$\sigma'_{ij} \neq 0$$
$$\sigma'_{ji} \neq 0$$
$$\exists \ k \neq i, j : \sigma'_{ki}, \sigma'_{kj} \neq 0$$

Also we define \sim such that $i \sim i$.

There are two special types of connectedness that arise, the first is when i,j satisfy $\sigma'_{ij} \neq 0$ or $\sigma'_{ji} \neq 0$, in this case we say that i and j are **strongly connected**. If i,j are connected but not strongly connected, we say that i and j are **weakly connected** by k or just that i and j are **weakly connected**.

Proposition 7.3.2

The relation \sim defined in Definition 7.3.1 is an equivalence relation.

PROOF We check that this relation is transitive, as reflexivity and symmetry are immediately verified. So suppose that $i \sim j$ and $j \sim k$ where i, j, k are mutually distinct.

Case 1 $(\sigma'_{ji} \neq 0 \text{ and } \sigma'_{jk} \neq 0)$ In this case *i* and *k* are weakly connected by *j*.

Case 2 $(\sigma'_{ij} \neq 0 \text{ and } \sigma'_{kj} \neq 0)$

Assume to the contrary that $\sigma'_{ik} = \sigma'_{ki} = 0$, then Eq. (7.2.6b) can't be satisfied. Thus *i* must be strongly connected to *k*.

Case 3 $(\sigma'_{ij} \neq 0 \text{ and } \sigma'_{jk} \neq 0 \text{ or } \sigma'_{ji} \neq 0 \text{ and } \sigma'_{kj} \neq 0)$

Assume first that $\sigma'_{ij} \neq 0$ and $\sigma'_{jk} \neq 0$ and to the contrary that $\sigma'_{ik} = 0$, then Eq. (7.2.6c) can't be satisfied. Also the case where $\sigma'_{ji} \neq 0$ and $\sigma'_{kj} \neq 0$ is just a permutation of the first, so the same argument applies. Thus in either case *i* must be strongly connected to *k*.

Case 4 ($\sigma'_{ij} \neq 0$ and j and k are weakly connected or $\sigma'_{kj} \neq 0$ and i and j are weakly connected)

Suppose first that $\sigma'_{ij} \neq 0$ and j and k are weakly connected by h. So we have that $\sigma'_{hj}, \sigma'_{hk} \neq 0$.

If h = i then $\sigma'_{ik} \neq 0$, so assume that $h \neq i$. If $\sigma'_{hi} \neq 0$ then *i* and *k* are weakly connected by *h*, so assume that $\sigma'_{hi} = 0$. If $\sigma'_{ih} \neq 0$ then by Case 3 we get that

 $\sigma'_{ik} \neq 0$, so assume further that $\sigma'_{ih} = 0$. Then after checking Eq. (7.2.6c) with the following coordinates, we get a contradiction.

$$\begin{array}{c} h \rightarrow i \\ i \rightarrow j \\ j \rightarrow k \end{array}$$

The case where $\sigma'_{kj} \neq 0$ and *i* and *j* are weakly connected is just a permutation of the first case. Thus we conclude that *i* is connected to *k*.

Case 5 $(\sigma'_{ji} \neq 0 \text{ and } j \text{ and } k \text{ are weakly connected or } \sigma'_{jk} \neq 0 \text{ and } i \text{ and } j \text{ are weakly connected})$

Suppose first that $\sigma'_{ji} \neq 0$ and j and k are weakly connected by h. So we have that $\sigma'_{hj}, \sigma'_{hk} \neq 0$.

If h = i then $\sigma'_{ik} \neq 0$, so assume that $h \neq i$. Since $\sigma'_{hj} \neq 0$ and $\sigma'_{ji} \neq 0$, by Case 3 we get that $\sigma'_{hi} \neq 0$. Thus *i* and *k* are weakly connected by *h*.

The case where $\sigma'_{jk} \neq 0$ and *i* and *j* are weakly connected is just a permutation of the first case. Thus we conclude that *i* is connected to *k*.

Case 6 (*i* and *j* are weakly connected and *j* and *k* are weakly connected) Suppose l satisfies $\sigma'_{li}, \sigma'_{lj} \neq 0$ and *h* satisfies $\sigma'_{hj}, \sigma'_{hk} \neq 0$.

If h = l then i and k are clearly weakly connected, so assume that $h \neq l$.

Note that l is strongly connected to j and $j \sim k$ then we can use one of the previous cases considered to find that $l \sim k$. Similarly because i is strongly connected to l and $l \sim k$ we find that $i \sim k$.

Thus we conclude that \sim is transitive and thus defines an equivalence relation.

Now suppose that \sim gives a single partition of the coordinates, i.e. the coordinates are connected. Our goal is to show that there must be at least one connecting coordinate. First we need a definition. We define S, called the set of strongly connected coordinates as follows:

 $S \equiv \{ i : i \text{ is strongly connected to every } j \}$

The reason to make this definition is because $M \subseteq S$ (this inclusion might be proper in some cases which can be observed by inspecting KEM metrics derived by Kalnins-Miller [Kal86]). So the idea is to first show that $S \neq \emptyset$ since this is easier to do using the hypothesis of connectedness. It turns out that this is possible.

Proposition 7.3.3

When the coordinates are connected, S has at least one coordinate.

PROOF Suppose to the contrary that $S = \emptyset$. Then $\forall i$, there exists j, k with i, j, k distinct such that

$$\sigma'_{ij} = \sigma'_{ji} = 0 \text{ and } \sigma'_{ki}, \sigma'_{kj} \neq 0 \tag{7.3.1}$$

So fix some *i*, and choose *j*, *k* satisfying Eq. (7.3.1). Then by Eq. (7.2.6a) we must have that $\sigma'_{jk} = 0$, similarly by Eq. (7.2.6b) we must have that $\sigma'_{ik} = 0$. Now let $A = \{i, j\}$ then note that $k \notin A$ and $\forall l \in A$, $\sigma'_{lk} = 0$.

Claim 7.3.3.1

Suppose we have a coordinate f and a set of coordinates $A \neq \emptyset$ such that $f \notin A$ and $\forall i \in A, \ \sigma'_{if} = 0$. Furthermore assume that $\{f\} \cup A \neq \{1, ..., n\}$. Then we can obtain a new set A' such that $A \cup \{f\} \subseteq A'$ and an $h \notin A'$ such that $\forall i \in A', \ \sigma'_{ih} = 0$.

PROOF (PROOF OF CLAIM) By assumption there exists g, h satisfying Eq. (7.3.1) with f in place of i and g in place of j. Since $\sigma'_{hf} \neq 0$, $h \notin A$. If $\{f\} \cup A = \{1, ..., n\}$, then we have reached a contradiction, so assume otherwise. As we observed earlier for a similar case, we must have $\sigma'_{gh} = \sigma'_{fh} = 0$. Also $\forall i \in A$ since $\sigma'_{if} = 0$ and $\sigma'_{hf} \neq 0$ by evaluating Eq. (7.2.6b) with $i \to i, f \to j, h \to k$ we find that $\sigma'_{ih} = 0$.

Thus if we let $A' = A \cup \{g, f\}$ then $\forall i \in A', \sigma'_{ih} = 0$. Also note that |A'| > |A| and $h \notin A'$.

Now we can inductively apply Claim 1 to get a set of coordinates $A \neq \emptyset$, an $f \notin A$ such that $\forall i \in A$, $\sigma'_{if} = 0$ and $\{f\} \cup A = \{1, ..., n\}$. Then by assumption there must exist a coordinate g such that f is weakly connected to g. So there is a coordinate h, with $h \neq f$, such that $\sigma'_{hf} \neq 0$. Since $\{f\} \cup A = \{1, ..., n\}$, $h \in A$, thus $\sigma'_{hf} = 0$, a contradiction.

Thus $S \neq \emptyset$.

Then assuming $S \neq \emptyset$ we try to prove that $M \neq \emptyset$. This is also possible.

Proposition 7.3.4

When the coordinates are connected and S has at least one coordinate then there must be at least one connecting coordinate. Thus due to the previous proposition we find that when the coordinates are connected there must be at least one connecting coordinate. \Box

PROOF Assume to the contrary that $M = \emptyset$. Then $\forall i \in S$ there exists j such that

$$\sigma'_{ij} = 0 \text{ and } \sigma'_{ji} \neq 0 \tag{7.3.2}$$

Since $S \neq \emptyset$ by hypothesis, we can choose some $i \in S$ and some $j \neq i$ such that Eq. (7.3.2) is satisfied. Let $B = \{i\}$.

-

Claim 7.3.4.1

Suppose $\emptyset \neq B \subseteq S$ and there is a $j \notin B$ such that $\forall i \in B, \sigma'_{ij} = 0$. Furthermore assume that $\{j\} \cup B \neq \{1, ..., n\}$. Then we can obtain a new set $B' = \{j\} \cup B \subseteq S$ and a $k \notin B'$ such that $\forall i \in B', \sigma'_{ik} = 0$.

PROOF (PROOF OF CLAIM) Fix an $i \in B$, then $\sigma'_{ij} = 0$ and $\sigma'_{ji} \neq 0$. Now pick $k \neq i, j$, then $\sigma'_{ik} \neq 0$ or $\sigma'_{ki} \neq 0$.

If $\sigma'_{ik} \neq 0$ then by Eq. (7.2.6c) we must have that $\sigma'_{jk} \neq 0$. If $\sigma'_{ki} \neq 0$ then by Eq. (7.2.6a) either $\sigma'_{jk} \neq 0$ or $\sigma'_{kj} \neq 0$. In either case we find that j is strongly connected to k. Since k was arbitrary and because j is also strongly connected to i we find that $j \in S$. Then by Eq. (7.3.2) there exists $\exists k \neq i, j$ such that $\sigma'_{jk} = 0$ and $\sigma'_{kj} \neq 0$, note that $k \notin B$.

Assume $i \in B$ is arbitrary, then $\sigma'_{ij} = 0$ and $\sigma'_{kj} \neq 0$, thus by Eq. (7.2.6b) we must have that $\sigma'_{ik} = 0$.

Let $B' = B \cup \{j\} \subseteq S$ then $\forall i \in B'$ we have $\sigma'_{ik} = 0$, also note that $k \notin B'$.

Now we can inductively apply Claim 1 to a get set B satisfying $\emptyset \neq B \subseteq S$ and a $j \notin B$ such that $\forall i \in B, \sigma'_{ij} = 0$ and $\{j\} \cup B = \{1, ..., n\}$. As in the proof of the Claim 1, we find that $j \in S$. Then by Eq. (7.3.2) $\exists k \neq j$ such that $\sigma'_{kj} \neq 0$, but since $\{j\} \cup B = \{1, ..., n\}$ we must have that $k \in B$, then $\sigma'_{kj} = 0$, a contradiction.

Thus $M \neq \emptyset$.

The following proposition classifies all metrics with at least one connecting coordinate.

Proposition 7.3.5

If the metric has at least one connecting coordinate then the following statements are true. For $a \in M$:

$$H_a^2 = \Phi_a \prod_{\substack{b \in M \\ b \neq a}} (\sigma_a - \sigma_b)$$

If m > 1 then one can partition the coordinates in³ M^c such that if I is an equivalence class of this partition and $\alpha \in I$, then

$$H_{\alpha}^{2} = \Phi_{\alpha} \prod_{\substack{\beta \in I \\ \beta \neq \alpha}} (\sigma_{\alpha\beta} + \sigma_{\beta\alpha}) \prod_{a=1}^{m} (e_{I} - \sigma_{a}) \quad (m \ge 2)$$

If m = 1 then one can partition the coordinates in M^c such that if I is an equivalence class of this partition and $\alpha \in I$, then

³If $Y \subseteq X$, then we denote the complement of Y in X (elements of X not in Y) as Y^c .

$$H_{\alpha}^{2} = \Phi_{\alpha}\sigma_{a}^{I}\prod_{\substack{\beta \in I\\ \beta \neq \alpha}} (\sigma_{\alpha\beta} + \sigma_{\beta\alpha}) \quad (m = 1)$$

Equations (7.2.6) are satisfied whenever at least one of i, j or k is in M if and only if the functions H_i^2 are of the form just described. Furthermore Equations (7.2.6) are satisfied whenever i, j, k are not all in the same partition.

PROOF By hypothesis we can assume $m \ge 1$. We use Latin letters such as a to denote the connecting coordinates and the remaining coordinates are denoted with Greek letters such as α . Although i and j are reserved for arbitrary coordinates. Furthermore we denote $N = \{1, ..., n\}$. Then by definition $\forall a \in M, i \in N$ we have that $\sigma'_{ai} \neq 0$.

Claim 7.3.5.1

For $a \in M$ and $\alpha \in M^c$, $\sigma'_{\alpha a} = 0$.

PROOF For any $\alpha \in M^c$, there exists $i \in N$ such that $\sigma'_{\alpha i} = 0$.

Suppose first that $i \in M$ and let a = i. Suppose to the contrary that there exists $b \in M \setminus \{a\}$ such that $\sigma'_{\alpha b} \neq 0$. Then Eq. (7.2.6b) can't hold with $\alpha \to i, a \to j, b \to k$. Thus the claim holds in this case.

If $i \in M^c$, let $\beta = i$. If the first case doesn't hold for α , then $\sigma'_{\alpha a} \neq 0 \ \forall a \in M$. Fix $a \in M$, then Eq. (7.2.6b) can't hold with $\alpha \to i, \beta \to j, a \to k$. Thus the first case must hold for some $a \in M$, thus the claim must hold.

The proof for the following claim is mainly from [Eis34, P. 292].

Claim 7.3.5.2

For $a \in M$, the following holds

$$H_a^2 = \Phi_a \prod_{\substack{b \in M \\ b \neq a}} (\sigma_a - \sigma_b)$$

where each $\sigma_a(x_a)$.

PROOF Suppose first that m = 1 and let $a \in M$. Then for $\alpha \in M^c$ by the above claim we know that $\sigma_{a\alpha} + \sigma_{\alpha a}$ only depends on the *a* coordinate and so these factors can be absorbed into Φ_a and Φ_{α} . Thus the claim holds in this case.

So for the remainder of the proof of this claim assume that m > 1. To prove this statement, for $a, b \in M$ our goal is to remove the *b* dependence from σ_{ab} . First assume m > 2 and let $a, b, c \in M$. From Eq. (7.2.7) evaluated with $a \to i, b \to j, c \to k$ we get:

$$\sigma'_{ab}\sigma'_{bc}\sigma'_{ca} + \sigma'_{ba}\sigma'_{cb}\sigma'_{ac} = 0$$

Since each term is non-zero, by separating variables it follows that $\frac{\sigma'_{ab}}{\sigma'_{ac}}$ is a constant. Thus we can set $\sigma_{ab} = a_{ab}\sigma_a$ where a_{ab} is a constant and σ_a involves x_a at most. The above equation implies that the constants must satisfy the following:

$$a_{ab}a_{bc}a_{ca} + a_{ba}a_{cb}a_{ac} = 0 (7.3.3)$$

Assuming the above equation holds, it follows that all three Equations (7.2.6) are satisfied for $a \to i, b \to j, c \to k$. Now set

$$\sigma_a = a_{bc} a_{ca} \bar{\sigma}_a \qquad \qquad \sigma_b = a_{cb} a_{ac} \bar{\sigma}_b$$

Then Eq. (7.3.3) implies that $a_{ab}\sigma_a + a_{ba}\sigma_b = a_{ab}a_{bc}a_{ca}(\bar{\sigma}_a - \bar{\sigma}_b)$; in which case the constant factor may be absorbed into Φ_a and Φ_b . Thus we can assume $a_{ab} = -a_{ba} = 1$, then Eq. (7.3.3) becomes:

$$a_{bc}a_{ca} - a_{cb}a_{ac} = 0$$

Now set $a_{ca}\sigma_c = -a_{ac}\bar{\sigma}_c$, then we have:

$$a_{ca}\sigma_c + a_{ac}\bar{\sigma}_a = a_{ac}(\bar{\sigma}_a - \bar{\sigma}_c)$$

Thus we can assume $a_{ac} = -a_{ca} = 1$. Then $a_{bc}\bar{\sigma}_b + a_{cb}\bar{\sigma}_c = a_{bc}(\bar{\sigma}_b - \bar{\sigma}_c)$ and so we can assume $a_{bc} = -a_{cb} = 1$. Inductively, this process can be continued so that each $\sigma_{ab} = \pm \sigma_a$, where the sign is positive if σ_{ab} appears in H_a^2 and is negative if σ_{ab} appears in H_b^2 .

If m = 2 then we can define $\sigma_a = \sigma_{ab}$ and $\sigma_b = -\sigma_{ba}$ without loss of consistency. For $\alpha \in M^c$, $\sigma'_{\alpha a} = 0$, so we can absorb terms of the form $\sigma_{\alpha a} + \sigma_{a\alpha}$ into Φ_a and Φ_{α} . Thus we have proven the following:

$$H_a^2 = \Phi_a \prod_{\substack{b \le m \\ b \ne a}} (\sigma_a - \sigma_b)$$

We can now assume that Equations (7.2.6) have been solved whenever $i, j, k \in M$. Thus if m = n the above claim proves that the metric has the form given by Eq. (7.1.3) and so we are finished. So assume for the remainder of the proof that m < n.

Now fix $a, b \in M$ and $\alpha \in M^c$. Let $a_{\alpha a} = \sigma_{\alpha a} \in \mathbb{R}$ and $a_{\alpha b} = \sigma_{\alpha b} \in \mathbb{R}$. Then Eq. (7.2.6c) evaluated with $a \to i, b \to j, \alpha \to k$ gives:

$$\sigma_{a\alpha}'\sigma_{b\alpha}'(\sigma_a - \sigma_b) + \sigma_{a\alpha}'\sigma_b'(\sigma_{b\alpha} + a_{\alpha b}) - \sigma_{b\alpha}'\sigma_a'(a_{\alpha a} + \sigma_{a\alpha}) = 0$$

We now proceed to solve the above equations. First we rearrange terms to separate the variables:

$$(\sigma_a - \sigma_b) + \frac{\sigma'_b}{\sigma'_{b\alpha}} (\sigma_{b\alpha} + a_{\alpha b}) - \frac{\sigma'_a}{\sigma'_{a\alpha}} (a_{\alpha a} + \sigma_{a\alpha}) = 0$$

$$\Rightarrow \sigma_a - \frac{\sigma'_a}{\sigma'_{a\alpha}} (a_{\alpha a} + \sigma_{a\alpha}) = \sigma_b - \frac{\sigma'_b}{\sigma'_{b\alpha}} (\sigma_{b\alpha} + a_{\alpha b}) = c \in \mathbb{R}$$
(7.3.4)

Then one can show that $\frac{\sigma'_a}{\sigma'_{a\alpha}} = -d \in \mathbb{R} \setminus \{0\}$ and similarly $\frac{\sigma'_b}{\sigma'_{b\alpha}} = -f \in \mathbb{R} \setminus \{0\}$. Thus the above equation implies:

$$\sigma_a = -d(a_{\alpha a} + \sigma_{a\alpha} - \frac{c}{d})$$

$$\sigma_b = -f(a_{\alpha b} + \sigma_{b\alpha} - \frac{c}{f})$$

Now let $e_{\alpha} = c$ then the two equations above implies the following:

$$a_{\alpha a} + \sigma_{a\alpha} = \frac{e_{\alpha} - \sigma_a}{d}$$
$$a_{\alpha b} + \sigma_{b\alpha} = \frac{e_{\alpha} - \sigma_b}{f}$$

Thus by absorbing the constants d, f into the Φ functions we can assume $a_{\alpha a} = a_{\alpha b} = e_{\alpha}, \ \sigma_{a\alpha} = -\sigma_a$ and $\sigma_{b\alpha} = -\sigma_b$. With these assumptions, it follows that all three Equations (7.2.6) are satisfied for $a \to i, b \to j, \alpha \to k$. Thus we conclude that Equations (7.2.6) hold whenever $i, j \in M$ and $k \in M^c$.

Suppose m > 1, we just observed that for $\alpha \in M^c$ and $a \in M$ that $\sigma_{\alpha a} = e_{\alpha}$. Thus we can partition the $\alpha \in M^c$ by the value e_{α} . We consider $\alpha, \beta \in M^c$ to be in the same equivalence class, say I, if $e_{\alpha} = e_{\beta}$. We define e_I such that $\alpha \in I$ implies that $e_{\alpha} = e_I$. We denote these equivalence classes by I and J.

Suppose $a \in M$ and $\alpha \in I, \beta \in J$. We now check Equations (7.2.6) for $a \to i, \alpha \to j, \beta \to k$. Since $\sigma'_{\alpha a} = \sigma'_{\beta a} = 0$, Eq. (7.2.6a) is satisfied. Equation (7.2.6b) and (7.2.6c) reduce to the following:

$$\sigma_{\beta\alpha}'(\sigma_{a\alpha}'(e_{\beta} + \sigma_{a\beta}) - \sigma_{a\beta}'(\sigma_{a\alpha} + e_{\alpha})) = 0$$
(7.3.5)

$$\sigma_{\alpha\beta}'(\sigma_{a\alpha}'(e_{\beta} + \sigma_{a\beta}) - \sigma_{a\beta}'(\sigma_{a\alpha} + e_{\alpha})) = 0$$
(7.3.6)

Now we can use the fact that $\sigma_{a\alpha} = \sigma_{a\beta} = -\sigma_a$ to get the following:

$$\sigma'_{\beta\alpha}(e_{\alpha} - e_{\beta}) = 0$$

$$\sigma'_{\alpha\beta}(e_{\alpha} - e_{\beta}) = 0$$

If I = J then $e_{\alpha} = e_{\beta}$ and the above equations are satisfied. If $I \neq J$ then we must have that $\sigma'_{\alpha\beta} = \sigma'_{\beta\alpha} = 0$; in which case $\sigma_{\alpha\beta} + \sigma_{\beta\alpha}$ can be absorbed into the Φ functions. Thus we have proven the following: if $\alpha \in I$ then

$$H_{\alpha}^{2} = \Phi_{\alpha} \prod_{\substack{\beta \in I \\ \beta \neq \alpha}} (\sigma_{\alpha\beta} + \sigma_{\beta\alpha}) \prod_{a=1}^{m} (e_{I} - \sigma_{a}) \quad (m \ge 2)$$

Now suppose m = 1, then we can pullback to a submanifold given by $x_1 = \text{constant}$ and then partition the coordinates in M^c into connected components. We denote these equivalence classes by I and J. Let $a \in M$ and $\alpha \in I, \beta \in J$. As for the case m > 1 one can see that Eq. (7.2.6a) is satisfied. Furthermore Equation (7.2.6b) and (7.2.6c) reduce to Equations (7.3.5) and (7.3.6) above. If $I \neq J$ then $\sigma'_{\beta\alpha} = \sigma'_{\alpha\beta} = 0$, thus Equations (7.3.5) and (7.3.6) are both satisfied. Otherwise assume I = J and $\alpha, \beta \in I$ satisfy $\sigma'_{\alpha\beta} \neq 0$ for the moment, then Eq. (7.3.6) implies

$$\frac{\sigma_{a\alpha}'}{\sigma_{a\beta}'}(e_{\beta}+\sigma_{a\beta})-\sigma_{a\alpha}+e_{\alpha}=0$$

As with Eq. (7.3.4) we can deduce that $\frac{\sigma'_{a\alpha}}{\sigma'_{a\beta}} = d \in \mathbb{R}$. Then the above equation implies that

$$e_{\beta} + \sigma_{a\beta} = \frac{e_{\alpha} + \sigma_{a\alpha}}{d}$$

Let $\sigma_a^I = e_{\alpha} + \sigma_{a\alpha}$, then after absorbing d into the Φ functions and relabelling, we can assume $\sigma_{a\alpha} = \sigma_{a\beta} = \sigma_a^I$ and $a_{\alpha a} = a_{\beta b} = 0$. Since the coordinates in I are connected, we can assume that for any $\alpha, \beta \in I$ with $\alpha \neq \beta$ that $\sigma_{a\alpha} = \sigma_{a\beta} = \sigma_a^I$ and $a_{\alpha a} = a_{\beta b} = 0$ and then Equations (7.3.5) and (7.3.6) are both satisfied. Then for $\alpha \in I$ and $a \in M$, we have proven the following:

$$H_{\alpha}^{2} = \Phi_{\alpha} \sigma_{a}^{I} \prod_{\substack{\beta \in I \\ \beta \neq \alpha}} (\sigma_{\alpha\beta} + \sigma_{\beta\alpha}) \quad (m = 1)$$

Also note that Equations (7.2.6) are satisfied whenever $i \in M$ and $j, k \in M^c$. Thus we can conclude that Equations (7.2.6) are satisfied whenever at least one of i, j or k is in M. Furthermore one can easily check that Equations (7.2.6) are satisfied whenever i, j, k are not all in the same partition.

When the coordinates are disconnected, i.e. \sim gives multiple partitions, one can easily show that the metric is a product metric.

Proposition 7.3.6

If the coordinates are disconnected, then the metric is a product metric which is given by Eq. (7.1.4). Furthermore Equations (7.2.6) are satisfied whenever i, j or k aren't in the same connected component.

PROOF Suppose $i \in I$ and $j \in J$ where I and J are a disconnected set of coordinates. Then it follows by definition that $\sigma_{ij'} = \sigma'_{ji} = 0$, thus we can absorb the factors $\sigma_{ij} + \sigma_{ji}$ into the Φ functions. Thus we have proven that for $i \in I$ the following holds:

$$H_i^2 = \Phi_i \prod_{\substack{j \in I \\ j \neq i}} (\sigma_{ij} + \sigma_{ji})$$

Thus the metric has the form given by Eq. (7.1.4). One can easily check that Equations (7.2.6) are satisfied whenever i, j or k aren't in the same connected component.

Proposition 7.3.7

If g is a reducible orthogonal separable metric with diagonal curvature given by Eqs. (7.1.4) to (7.1.6) and |I| > 1, the metric g^I pulls back to a orthogonal separable metric with diagonal curvature on the submanifold with metric proportional to g^I .

PROOF We know that an orthogonally separable web restricted to one of the integral manifolds of its n foliations is still separable Proposition 4.4.15, and g^I still has the form given by Eq. (7.2.5) thus its Riemann curvature tensor will still satisfy $R_{ijik} = 0$ for $j \neq k$ on the integral manifolds.

We can see how the classification works. If n = 2 then we've noted in the previous section that the general solution is given by Eq. (7.1.2). So suppose n > 2 and the general orthogonal separable metrics with diagonal curvature are known on manifolds with dimension $k \le n - 1$. If the coordinates are disconnected then Proposition 7.3.6 shows us that the metric must have the form given by Eq. (7.1.4) and the only equations that haven't been solved are Equations (7.2.6) when i, j, k are inside a connected component. If the coordinates are connected then Proposition 7.3.4 in conjunction with Proposition 7.3.5 tells us that the metric must have the form given by Eq. (7.1.2) or Eq. (7.1.6). Furthermore in this case the only equations that haven't been solved are Equations (7.2.6) when $i, j, k \in I$ where $I \subseteq M^c$ is an equivalence class as given in Proposition 7.3.5. Now fix some |I| > 1, then by Proposition 7.3.7, g^I pulls back to an orthogonal separable metric with diagonal curvature on the submanifold with coordinates (x_I) . Thus inductively we know the general form of g^I since the dimension of the submanifold is at most n-1. In particular if |I| > 2then the components of g^I satisfy Equations (7.2.6). Thus the solution g will satisfy Equations (7.2.6) for all i, j, k distinct and so g satisfies all relevant equations. Thus we have found all the orthogonal separable metrics with diagonal curvature.

7.4 Spaces of Constant Curvature

In this section our main goal is to show that the metric given Eq. (7.1.6) can be ruled out in spaces of constant curvature. In other words, all metrics in spaces of constant curvature are given by Eq. (7.1.2) and thus are invariantly characterized by concircular tensors.

Let $M = M_0 \times_{\rho_1} M_1 \times \cdots \times_{\rho_k} M_k$ be a warped product with M a space of constant curvature κ . We now use the formulas for the sectional curvature given in Corollary 3.5.8 for the following calculations.

After applying the polarization identity to Eq. (3.5.3), we get

$$S^{\rho_i}(X,Y) + \kappa \rho_i \left\langle X, Y \right\rangle = 0 \tag{7.4.1}$$

By Eq. (3.5.5), we have:

$$K_{VU}^{M_i} = \kappa \rho_i^2 + (\nabla \rho_i)^2$$

Now,

$$\nabla_X K_{VU}^{M_i} = 2\kappa \rho_i \nabla_X \rho_i + 2 \langle \nabla_X \nabla \rho_i, \nabla \rho_i \rangle$$
$$= 2(\kappa \rho_i \nabla_X \rho_i + S^{\rho_i} (X, \nabla \rho_i))$$
$$= 2(\kappa \rho_i \nabla_X \rho_i - \kappa \rho_i \langle X, \nabla \rho_i \rangle)$$
$$= 0$$

Hence for each i > 0, (M_i, g_i) necessarily has constant curvature, say κ_i ; this proves Lemma 7.1.4. Finally Eq. (3.5.4) gives us the following:

$$\langle \nabla \log \rho_i, \nabla \log \rho_k \rangle = -\kappa \quad (i \neq k)$$
 (7.4.2)

Now suppose dim $M_0 = 1$ and suppose coordinates on M_0 are chosen such that $\tilde{g} = \varepsilon dx_1^2$ where $\varepsilon = \pm 1$ as the case may be. Then Eqs. (7.4.1) and (7.4.2) imply the following:

$$\partial_1^2 \rho_i = -\kappa \varepsilon \rho_i$$
$$(\partial_1 \log \rho_i)(\partial_1 \log \rho_k) = -\varepsilon \kappa \quad (i \neq k)$$

Hereafter we denote $\omega = \kappa \varepsilon$ and let $\sigma_i = \rho_i^2$. We make exclusive use of the above two equations in the following calculations, suppose $i \neq k$, then:

$$\begin{split} \frac{\sigma'_i}{\sigma'_k} &= \frac{\rho_i \rho'_i}{\rho_k \rho'_k} \\ &= \frac{\rho_i \rho'_i}{\rho_k^2} (-\omega \frac{\rho'_i}{\rho_i}) \\ &= -\omega (\frac{\rho'_i}{\rho_k})^2 \end{split}$$

Thus

$$(\frac{\sigma'_i}{\sigma'_k})' = -2\omega(\frac{\rho'_i}{\rho_k})(\frac{\rho'_i}{\rho_k})'$$
$$= -2\omega(\frac{\rho'_i}{\rho_k})(\frac{\rho''_i\rho_k - \rho'_i\rho'_k}{\rho_k^2})$$
$$= -2\omega(\frac{\rho'_i}{\rho_k^2})(\rho''_i + \omega\rho_i)$$
$$= 0$$

Hence any warped product decomposition of a space of constant curvature with $\dim M_0 = 1$ has a metric given by Eq. (7.1.5). Thus we have proven Theorem 7.1.2.

7.5 Conclusion

In this chapter our main result is that in a space of constant curvature, every orthogonal separable coordinate system is a KEM coordinate system. This fact motivates the systematic study of concircular tensors in these spaces in the next section. The results in that chapter will allow us to apply the theory presented in Chapter 6.

7.6 Notes

The results of this chapter raise the following question: Are there any other spaces of interest in which one can prove that orthogonally separable coordinates have diagonal curvature, or at least admit such coordinates? This is a very nice property to have since if we can rule out the metric Eq. (7.1.6), then the separable coordinates are invariantly characterized by concircular tensors and hence highly amenable to analysis (see Chapter 6). It may also be of interest to find another condition in addition to the diagonal curvature condition in order to characterize KEM coordinates.

A related question: Given a pseudo-Riemannian manifold, what is a necessary and sufficient intrinsic condition that guarantees the existence of an orthogonal coordinate system having diagonal curvature? One can show that a necessary condition is the existence of an orthogonal coordinate system in which the Ricci tensor is diagonal.

We end with some notes on the KEM separation theorem (Theorem 7.1.2), which guarantees a non-trivial OCT associated with every ChKT in a space of constant curvature. For the Euclidean and spherical spaces, this theorem is implicitly applied in [WW03]; see Remark 6.7.1 for more details. This theorem was explicitly known for the special class of separable webs defined by Benenti tensors, see [Ben92a]. It was first stated in its present form in [RM14b], then proven in [RM14a].

Chapter 8

Preliminaries from Linear Algebra and Geometry

In this chapter we introduce the perquisite notation and results to read Chapter 9.

8.1 pseudo-Euclidean space

We will briefly review some definitions for a pseudo-Euclidean space from Section 1.4.1. First recall, that an *n*-dimensional vector space V equipped with metric g of signature¹ ν is denoted by \mathbb{E}_{ν}^{n} and called *pseudo-Euclidean space*. In some contexts the space is simply denoted V, and the metric $\langle \cdot, \cdot \rangle$ is called a *scalar product* (following [O'N83]). We also refer to ν as the *index* of the subspace V, denoted ind V. We obtain Euclidean space \mathbb{E}^{n} in the special case where $\nu = 0$. Also Minkowski space M^{n} is obtained by taking $\nu = 1$. Also note that since \mathbb{E}_{ν}^{n} is a vector space, for any $p \in \mathbb{E}_{\nu}^{n}$ we identify vectors in $T_{p}\mathbb{E}_{\nu}^{n}$ with points in \mathbb{E}_{ν}^{n} . This will be done tacitly.

Recall, a set $v_1, ..., v_n$ for V is said to be *orthonormal* if $\langle v_i, v_i \rangle = \pm 1$ and $\langle v_i, v_j \rangle = 0$ for $i \neq j$. Clearly an orthonormal set forms a basis for V and the metric in this basis is $g = \text{diag}(\pm 1, ..., \pm 1)$.

Now assume the scalar product $\langle \cdot, \cdot \rangle$ is (possibly) degenerate. We say a sequence of vectors v_1, \ldots, v_p is a *skew-normal sequence* of (length p) and (sign $\varepsilon = \pm 1$) if $\langle v_i, v_j \rangle = \varepsilon$ when i + j = p + 1 and $\langle v_i, v_j \rangle = 0$ otherwise. We will show shortly that these vectors are necessarily linearly independent, so let $H = \text{span}\{v_1, \ldots, v_p\}$. Then the bilinear form restricted to H is *skew-diagonal* and is given as follows:

¹The signature is equal to the number of negative diagonal entries in a basis which diagonalizes g.

$$S_k := \begin{pmatrix} 0 & \varepsilon \\ & \ddots & \\ \varepsilon & & 0 \end{pmatrix}$$

The following lemma shows that a skew-normal sequence forms a linearly independent set and it gives the index of the space spanned by such a set of vectors.

Lemma 8.1.1

Suppose V is a (possibly complex) scalar product space. Suppose $\varepsilon \in \{-1, 1\}$ and let z_1, \ldots, z_p be a skew-normal sequence of sign ε .

Then z_1, \ldots, z_p form a linearly independent set and the subspace H spanned by these vectors is non-degenerate and has index:

ind
$$H = \begin{cases} \lfloor \frac{p+1}{2} \rfloor & \text{if } \epsilon = -1 \\ p - \lfloor \frac{p+1}{2} \rfloor & \text{if } \epsilon = 1 \end{cases}$$

PROOF Given $1 \le i \le p$, denote the additive conjugate of i by i' = p + 1 - i. Note that $1 \le i' \le p$ and i + i' = p + 1. Suppose $i < \frac{p+1}{2}$ and $j > \frac{p+1}{2}$. Define vectors v_i and v_j as follows:

$$v_i = \frac{1}{\sqrt{2}}(z_i + z_{i'})$$

 $v_j = \frac{1}{\sqrt{2}}(z_j - z_{j'})$

If 2i = p + 1 then let $v_i = z_i$. Now for i < j suppose i + j = p + 1, then note that i' = j and j' = i, observe that:

$$\langle v_i, v_j \rangle = \frac{1}{2} (\langle z_i + z_j, z_j - z_i \rangle)$$

= $\frac{1}{2} (\langle z_j, z_j \rangle + \langle z_i, z_i \rangle)$
= 0

$$\langle v_i, v_i \rangle = \frac{1}{2} (\langle z_j + z_i, z_j + z_i \rangle)$$

= $\frac{1}{2} (2 \langle z_i, z_j \rangle)$
= ϵ

$$\langle v_j, v_j \rangle = \frac{1}{2} (\langle z_j - z_i, z_j - z_i \rangle)$$

= $-\frac{1}{2} (2 \langle z_i, z_j \rangle)$
= $-\epsilon$

Furthermore if 2i = p + 1 then $\langle v_i, v_i \rangle = \epsilon$. Now suppose $i + j \neq p + 1$, then the only way in which $\langle v_i, v_j \rangle \neq 0$ if i + j' = p + 1 or i' + j = p + 1, but one can see immediately that i + j' = p + 1 iff i' + j = p + 1. So suppose i + j' = p + 1. Then p + 1 = i + i' = i + p + 1 - j, hence i = j, in which case $\langle v_i, v_j \rangle$ reduces to the ones examined.

Thus we conclude that $\langle v_i, v_i \rangle = \epsilon$ if $i \leq \lfloor \frac{p+1}{2} \rfloor$, $\langle v_i, v_i \rangle = -\epsilon$ if $i > \lfloor \frac{p+1}{2} \rfloor$ and $\langle v_i, v_j \rangle = 0$ if $i \neq j$. Thus the conclusions follow.

8.2 Self-adjoint operators in pseudo-Euclidean space

In this section we review the metric-Jordan canonical form of a self-adjoint operator on a pseudo-Euclidean space. The details of the theory behind this canonical form is given in Appendix C; these are solutions to exercises 18-19 in [O'N83, P. 260-261]. Appendix C will be useful to those who want to calculate the metric-Jordan canonical form for a given self-adjoint operator. Now recall, that a linear operator T on a scalar product space V is called self-adjoint if $\langle Tx, y \rangle = \langle x, Ty \rangle$ for all $x, y \in V$.

A Jordan block of dimension k with eigenvalue $\lambda \in \mathbb{C}$ is a $k \times k$ matrix denoted by $J_k(\lambda)$, and defined as:

$$J_k(\lambda) := \begin{pmatrix} \lambda & 1 & & \\ & \lambda & \ddots & & 0 \\ & & \ddots & 1 & \\ & & & \lambda & 1 \\ & & & & \lambda \end{pmatrix}$$

Recall, the *skew-diagonal matrix* of dimension k is denoted by S_k , and defined as:

$$S_k := \begin{pmatrix} 0 & 1 \\ & \ddots & \\ 1 & & 0 \end{pmatrix}$$

In order to express the metric-Jordan canonical form of a self-adjoint operator on a pseudo-Euclidean space, we use the signed integer $\varepsilon k \in \mathbb{Z}$ where $k \in \mathbb{N}$ and $\varepsilon = \pm 1$. Then

the notation $J_{\varepsilon k}(\lambda)$ is short hand for the pair:

$$A = J_k(\lambda) \qquad \qquad g = \varepsilon S_k$$

Furthermore, given matrices A_1 and A_2 , we denote the following block diagonal matrix by $A_1 \oplus A_2$

$$A_1 \oplus A_2 := \begin{pmatrix} A_1 & 0\\ 0 & A_2 \end{pmatrix}$$

A key fact used to derive the metric-Jordan canonical form is the following:

Proposition 8.2.1

Suppose V is a scalar product space and T is a self-adjoint operator on V. Suppose $H \subseteq V$ is an invariant subspace of T. Then $T(H^{\perp}) \subseteq H^{\perp}$, i.e. H^{\perp} is an invariant subspace of T.

First we give the complex metric-Jordan canonical form of a self-adjoint operator.

Theorem 8.2.2 (Complex metric-Jordan canonical form [O'N83])

A real operator T on a pseudo-Euclidean space \mathbb{E}_{ν}^{n} is self-adjoint iff there exists a (possibly complex) basis β such that

$$T|_{\beta} = J_{\varepsilon_1 k_1}(\lambda_1) \oplus \cdots \oplus J_{\varepsilon_l k_l}(\lambda_l)$$

Furthermore there exists a canonical basis such that the unordered list

$$\{J_{\varepsilon_1k_1}(\lambda_1),\ldots,J_{\varepsilon_lk_l}(\lambda_l)\}$$

is uniquely determined by T and an invariant of T under the action of the orthogonal group $O(\mathbb{E}^n_{\nu})$. This unordered list is by definition, the complex metric-Jordan canonical form of T.

Remark 8.2.3

Since T is real, each Jordan block $J_{\varepsilon k}(\lambda)$ with $\lambda \in \mathbb{C} \setminus \mathbb{R}$ comes with a complex conjugate pair $J_{\varepsilon k}(\overline{\lambda})$. For complex eigenvalues, we can additionally assume that $\varepsilon = 1$.

In order to describe the real metric-Jordan canonical form, we need some additional notation. A real Jordan block of dimension k with parameters $a \pm ib \in \mathbb{C}$ is a $2k \times 2k$ matrix denoted by $J_{2k}(a \pm ib)$ defined as [LR05]:

$$J_{2k}(a \pm ib) := \begin{pmatrix} a & b & 1 & 0 & & & \\ -b & a & 0 & 1 & & 0 & \\ & & \ddots & & & \\ & & a & b & 1 & 0 \\ & & & -b & a & 0 & 1 \\ 0 & & & & a & b \\ & & & & -b & a \end{pmatrix}$$

If we denote the basis vectors by $\{u_1, v_1, ..., u_k, v_k\}$, then in addition to the above real Jordan block, we will assume the non-zero metric coefficients are given by the relations $\langle u_i, u_j \rangle = 1 = -\langle v_i, v_j \rangle$ where i + j = k + 1. Note that in contrast with the complex case, there is no sign associated with the real Jordan blocks. If we let $\lambda = a + ib$, then $J_{2k}(a \pm ib)$ is obtained from $J_k(\lambda) \oplus J_k(\overline{\lambda})$ by an appropriate change of basis (see the discussion preceding Lemma C.3.6).

The following crucial lemma allows one to understand and apply the (real) metric-Jordan canonical form, it will be proven in Appendix C (see Lemma C.3.6).

Lemma 8.2.4

Suppose V is a real scalar product space (where the scalar product is possibly degenerate) and T is a self-adjoint operator on V. Consider the case where $T = J_{\varepsilon k}(\lambda)$ for some $\lambda \in \mathbb{R}$, then

ind
$$V = \begin{cases} \lfloor \frac{k+1}{2} \rfloor & \text{if } \varepsilon = -1 \\ k - \lfloor \frac{k+1}{2} \rfloor & \text{if } \varepsilon = 1 \end{cases}$$

If $T = J_{2k}(a \pm ib)$ for $a, b \in \mathbb{R}$, then

 $\operatorname{ind} V = k$

In particular we observe that in both cases, the scalar product is necessarily non-degenerate. \Box

Thus we have:

Theorem 8.2.5 (Real metric-Jordan canonical form [O'N83])

A real operator T on a pseudo-Euclidean space \mathbb{E}^n_{ν} is self-adjoint iff there exists a real basis β such that

$$T|_{\beta} = J_{\varepsilon_1 k_1}(\lambda_1) \oplus \cdots \oplus J_{\varepsilon_l k_l}(\lambda_l) \oplus J_{2\tilde{k}_1}(a_1 \pm ib_1) \oplus \cdots \oplus J_{2\tilde{k}_r}(a_r \pm ib_r)$$

where each $\lambda_i, a_j, b_j \in \mathbb{R}$. Furthermore, the unordered list

$$\{J_{\varepsilon_1k_1}(\lambda_1),\ldots,J_{\varepsilon_lk_l}(\lambda_l),J_{2\tilde{k}_1}(a_1\pm ib_1),\ldots,J_{2\tilde{k}_r}(a_r\pm ib_r)\}$$

is uniquely determined by T and an invariant of T under the action of the orthogonal group $O(\mathbb{E}^n_{\nu})$. This unordered list is by definition, the (real) metric-Jordan canonical form of T.

We will apply the above results in Section 8.2.1 to enumerate the possible metric-Jordan canonical forms in Minkowski space. For now, we give an important example which clearly distinguishes the metric-Jordan canonical form the standard Jordan canonical form.

Example 8.2.6

Suppose V is Minkowski space equipped with the standard metric

$$g = \operatorname{diag}(-1, 1, \dots, 1)$$

For $\lambda_1 < \ldots < \lambda_n \in \mathbb{R}$ define two self-adjoint operators T_1 and T_2 as follows:

$$T_1 = \operatorname{diag}(\lambda_1, \lambda_2, \lambda_3, \dots, \lambda_n)$$
$$T_2 = \operatorname{diag}(\lambda_2, \lambda_1, \lambda_3, \dots, \lambda_n)$$

Now observe that even though T_1 and T_2 have the same eigenvalues, they have different metric-Jordan canonical forms. Hence the above theorem shows that these operators are isometrically inequivalent.

8.2.1 Minkowski Space

Fix a self-adjoint operator T in Minkowski space. We will use Theorem 8.2.5 to enumerate the possible Jordan canonical forms of T together with the metric in an adapted basis. As a consequence of Theorem 8.2.5, we simply have to determine which combination of Jordan blocks are possible in Minkowski space by imposing the dimension and signature restrictions. This can be done with the help of Lemma 8.2.4, since it gives us the index of a given subspace associated with a Jordan block. We denote by D_k a diagonal $k \times k$ matrix and I_k the identity $k \times k$ matrix. We have the following cases.

Case 1 T is diagonalizable with real spectrum

In this case T must have a time-like eigenvector. Indeed, since each eigenspace E_{λ} is non-degenerate, one eigenspace, say H, must have index 1. Then by obtaining

an orthonormal basis for H, we can obtain a time-like eigenvector. Thus T has the following form:

$$T = D_n \qquad \qquad g = \operatorname{diag}(-1, 1, \dots, 1)$$

Case 2 T has a complex eigenvalue $\lambda = a + ib$ with $b \neq 0$

By Lemma 8.2.4 the real subspace H spanned by a complex eigenvector with eigenvalue λ and its complex conjugate must have index 1. Since this subspace is T-invariant, by Proposition 8.2.1 H^{\perp} is a complementary invariant subspace, which must be Euclidean. Hence T must have the following form:

$$T = \begin{pmatrix} a & b & 0 \\ -b & a & \\ 0 & D_{n-2} \end{pmatrix} \qquad \qquad g = \begin{pmatrix} 1 & 0 & 0 \\ 0 & -1 & \\ 0 & I_{n-2} \end{pmatrix}$$

Case 3 T has real eigenvalues but is not diagonalizable

In this case we go through the possible multidimensional Jordan blocks associated to a real irreducible subspace, say H, of T. By Theorem 8.2.5, each basis for this subspace can be adapted to the scalar product, hence is non-degenerate. By Lemma 8.2.4 there are three types of Jordan blocks which have an associated subspace, H, with index one. For each of these subspaces, H^{\perp} is a complementary T-invariant Euclidean subspace. The first two cases occur when dim H = 2, and are given as follows:

$$T = \begin{pmatrix} \lambda & 1 & 0 \\ 0 & \lambda \\ 0 & D_{n-2} \end{pmatrix} \qquad g = \begin{pmatrix} 0 & \epsilon & 0 \\ \epsilon & 0 \\ 0 & I_{n-2} \end{pmatrix} \quad \epsilon = \pm 1$$

Note that the above form contains two metric-Jordan canonical forms depending on the sign of ϵ . The third occurs when dim H = 3:

T =	$\begin{pmatrix} \lambda \\ 0 \end{pmatrix}$	$1 \\ \lambda$	01	0	g =	/0 0	0 1	1 0	0
	0	0	λ	D_{n-3}		1	0 0	0	I_{n-3}

We also note that this case (T has real eigenvalues but is not diagonalizable) holds iff T has a unique lightlike eigenvector. This fact can be deduced by inspection of the above canonical forms.

Now, we collect some necessary and sufficient conditions concerning the diagonalizability of T in the following theorem. The second and third facts are from Theorem 4.1 in [Hal+96] while the last fact is from Section 9.5 in [Gre75]. All these facts can be readily deduced from the canonical forms listed above.

Theorem 8.2.7 (Properties of self-adjoint operators in Minkowski space) Let V be a Minkowski space and T a self-adjoint operator on V. Then the following statements are true:

- 1. T is diagonalizable with a real spectrum iff T has 1 timelike eigenvector or equivalently T has n 1 linearly independent spacelike eigenvectors.
- 2. If T has two linearly independent null eigenvectors then T is diagonalizable with a real spectrum and T has a time-like eigenspace of dimension at least 2 containing these eigenvectors.
- 3. If T has a real spectrum, then T is diagonalizable iff it has no null eigenvectors or at least two linearly independent null eigenvectors. In other words, T is not diagonalizable iff it has a unique null eigendirection.
- 4. If $n \ge 3$ and $\langle Tx, x \rangle \ne 0$ for all null vectors x then T is diagonalizable with a real spectrum.

8.3 Spaces of Constant Curvature in pseudo-Euclidean space

In this section we will briefly review the models of spaces of constant curvature as subsets of pseudo-Euclidean space. It is well known that \mathbb{E}_{ν}^{n} has constant zero curvature (flat space) and signature ν . There is another useful model of flat space which we will review towards the end.

Given an open subset $U \subseteq \mathbb{E}_{\nu}^{n}$ and $\kappa \in \mathbb{R} \setminus \{0\}$, we denote by $U(\kappa)$ the *central* hyperquadric of \mathbb{E}_{ν}^{n} contained in U, which is defined by:

$$U(\kappa) = \{ p \in U \mid \langle p, p \rangle = \kappa^{-1} \}$$

Usually $U = \mathbb{E}_{\nu}^{n}$ and this is denoted $\mathbb{E}_{\nu}^{n}(\kappa)$. The notation $U(\kappa)^{\circ}$ represents a maximal connected component of $U(\kappa)$. It is well known that $\mathbb{E}_{\nu}^{n}(\kappa)$ is a pseudo-Riemannian manifold of dimension n-1 with signature $\nu + \frac{(\operatorname{sgn} \kappa - 1)}{2}$ and constant curvature² κ [O'N83]. Since $\mathbb{E}_{\nu}^{n}(\kappa) \subset \mathbb{E}_{\nu}^{n}$, for any $p \in \mathbb{E}_{\nu}^{n}(\kappa)$ we identify vectors in $T_{p}\mathbb{E}_{\nu}^{n}(\kappa)$ with points in \mathbb{E}_{ν}^{n} . Occasionally we use the following conventions: If $\kappa = 0$, we set $\mathbb{E}_{\nu}^{n}(0) := \mathbb{E}_{\nu}^{n}$, if $\kappa = \infty$ we set $\mathbb{E}_{\nu}^{n}(\infty)$ to be the light cone, i.e. the set of non-zero null vectors. We also use the following notations: If $\kappa > 0$ then $S_{\nu}^{n}(\kappa) := \mathbb{E}_{\nu}^{n+1}(\kappa)^{\circ}$, if $\kappa < 0$ then $H_{\nu}^{n}(\kappa) := \mathbb{E}_{\nu+1}^{n+1}(\kappa)^{\circ}$.

We define the parabolic embedding of \mathbb{E}_{ν}^{n} in $\mathbb{E}_{\nu+1}^{n+2}$ with mean curvature vector $-a \in \mathbb{E}_{\nu}^{n}(\infty)$ by [Toj07]

$$\mathbb{P}^n_{\nu} := \{ p \in \mathbb{E}^{n+2}_{\nu+1}(\infty) : \langle p, a \rangle = 1 \}$$

An explicit isometry with \mathbb{E}_{ν}^{n} is obtained by choosing $b \in \mathbb{P}_{\nu}^{n}$, i.e. b is lightlike and $\langle a, b \rangle = 1$. We let $V := \operatorname{span}\{a, b\}^{\perp}$, note that $V \cong \mathbb{E}_{\nu}^{n}$, then for $x \in V$:

$$\psi(x) = b + x - \frac{1}{2}x^2 a \in \mathbb{P}^n_{\nu}$$
(8.3.1)

For the proofs of these properties of \mathbb{P}^n_{ν} , see Proposition D.2.2.

8.4 Warped products in Spaces of Constant Curvature

In this section we will briefly describe the warped product decompositions of spaces of constant curvature, generalizing results by Nolker in [Nol96]. This exposition will be sufficient for our applications. More information and proofs can be found in Appendix D.

We will use the notation $\mathbb{E}_{\nu}^{n}(\kappa)$ (where κ can be zero) to represent the general space of constant curvature. First we will need to know the spherical submanifolds of these spaces.

Theorem 8.4.1 (Spherical submanifolds of $\mathbb{E}^n_{\nu}(\kappa)$)

Let $\bar{p} \in \mathbb{E}_{\nu}^{n}(\kappa)$ be arbitrary, $V \subset T_{\bar{p}}\mathbb{E}_{\nu}^{n}(\kappa)$ a non-degenerate subspace with $m := \dim V \ge 1$, $\mu := \operatorname{ind} V$ and $z \in V^{\perp} \cap T_{\bar{p}}\mathbb{E}_{\nu}^{n}(\kappa)$. Let $a := \kappa \bar{p} - z$, $\tilde{\kappa} := a^{2}$ and $W := \mathbb{R}a \oplus V$. There is exactly one m-dimensional connected and geodesically complete spherical submanifold \tilde{N} with $\bar{p} \in \tilde{N}$, $T_{\bar{p}}\tilde{N} = V$ and having mean curvature vector at \bar{p} , z. \tilde{N} is an open submanifold

²See Appendix D.2.

of N; N is referred to as the spherical submanifold determined by (\bar{p}, V, a) , it is geodesic iff z = 0 and is given as follows (where \simeq means isometric to):

(a) a = 0, in this case $N \simeq \mathbb{E}^m_{\mu}$

$$N = \overline{p} + V$$

- (b) a is timelike, then $\mu \leq \nu 1$ and $N \simeq H^m_{\mu}(\tilde{\kappa})$
- (c) a is spacelike, then $N \simeq S^m_{\mu}(\tilde{\kappa})$ For cases (b) and (c), let $c = \overline{p} - \frac{a}{\tilde{\kappa}}$ be the center of N, then N is given as follows:

$$N = c + \{ p \in W \mid p^2 = \frac{1}{\tilde{\kappa}} \}$$

(d) a is lightlike, then $\mu \leq \nu - 1$ and $N \simeq \mathbb{E}^m_{\mu}$

$$N = \bar{p} + \{ p - \frac{1}{2} p^2 a \mid p \in V \}$$

Remark 8.4.2

If a is lightlike, then N is isometric to \mathbb{P}^m_{μ} with mean curvature vector -a. Furthermore, let $b \in V^{\perp}$ be a lightlike vector satisfying $\langle a, b \rangle = 1$. Then the orthogonal projector onto V, P, induces an isometry of $N - \bar{p} + b$ onto V.

Remark 8.4.3

One can find more details on when N is connected in the remarks following Theorems D.4.1 and D.6.2. $\hfill \Box$

PROOF See Theorems D.4.1 and D.6.2.

With the knowledge of these spherical submanifolds, we can now specify how to construct warped products in $\mathbb{E}_{\nu}^{n}(\kappa)$. This construction depends on the following data: A point $\bar{p} \in \mathbb{E}_{\nu}^{n}(\kappa)$, a decomposition $T_{\bar{p}}\mathbb{E}_{\nu}^{n}(\kappa) = \bigoplus_{i=0}^{k} V_{i}$ into non-trivial (hence non-degenerate) subspaces with $k \geq 1$, and vectors $z_{1}, \ldots, z_{k} \in V_{0}$ such that the vectors $a_{i} := \kappa \bar{p} - z_{i}$ are pair-wise orthogonal and independent. We call the data $(\bar{p}; \bigoplus_{i=0}^{k} V_{i}; a_{1}, \ldots, a_{k})$, *initial data* for a (proper) warped product decomposition of $\mathbb{E}_{\nu}^{n}(\kappa)$. If $\kappa = 0$, one can more generally let some of the a_{i} be zero, this results in Cartesian products as done in [Nol96]. Since we assume the a_{i} are non-zero, we sometimes use the additional qualifier "proper".

With this initial data, for i > 0 let N_i be the sphere in $\mathbb{E}_{\nu}^n(\kappa)$ determined by (\bar{p}, V_i, a_i) and $\rho_i(p_0) = 1 + \langle a_i, p_0 - \bar{p} \rangle$. Let N_0 be the subset of the sphere in $\mathbb{E}_{\nu}^n(\kappa)$ determined

by $(\bar{p}, V_0, \kappa \bar{p})$ where each $\rho_i > 0$. Then the data $(\bar{p}; \bigoplus_{i=0}^{\kappa} V_i; a_1, ..., a_k)$, induces a warped product decomposition (of $\mathbb{E}^n_{\nu}(\kappa)$) given as follows:

$$\psi: \begin{cases} N_0 \times_{\rho_1} N_1 \times \dots \times_{\rho_k} N_k & \to \mathbb{E}^n_{\nu}(\kappa) \\ (p_0, \dots, p_k) & \mapsto p_0 + \sum_{i=1}^k \rho_i(p_0)(p_i - \overline{p}) \end{cases}$$
(8.4.1)

We note that ψ has the property that $\psi(\bar{p}, \ldots, p_i, \bar{p}, \ldots, \bar{p}) = p_i$. Often the point \bar{p} doesn't enter calculations, hence we will usually omit it.

For actual calculations, it will be more convenient to work with canonical forms. The following definition will be particularly useful.

Definition 8.4.4 (Canonical form for Warped products of \mathbb{E}^n_{ν})

We say that a proper warped product decomposition of \mathbb{E}^n_{ν} determined by $(\bar{p}; \bigoplus_{i=0}^k V_i; a_1, ..., a_k)$ is in canonical form if: $\bar{p} \in V_0$ and $\langle \bar{p}, a_i \rangle = 1$.

Any proper warped product decomposition ψ of \mathbb{E}^n_{ν} can be brought into canonical form, see the discussion preceding Corollary D.4.10 for details.

We will now give more information on standard warped product decompositions of \mathbb{E}^n_{ν} in canonical form. Suppose the initial data $(\bar{p}; V_0 \oplus V_1; a)$ is in canonical form, and let ψ be the associated warped product decomposition given by Eq. (8.4.1). Denote $\kappa := a^2$ and $\epsilon := \operatorname{sgn} \kappa$. We have two types of warped products:

non-null warped decomposition If $\kappa \neq 0$, let $W_0 := V_0 \cap a^{\perp}$ and $W_1 := W_0^{\perp}$.

null warped decomposition If $\kappa = 0$, then *a* is lightlike, so fix another lightlike vector $b \in V_0$ such that $\langle a, b \rangle = 1$, let $W_0 := V_0 \cap \operatorname{span}\{a, b\}^{\perp}$ and $W_1 := V_1$.

For i = 0, 1, let $P_i : \mathbb{E}^n_{\nu} \to W_i$ be the orthogonal projection. Then the following holds:

Theorem 8.4.5 (Standard Warped Products in \mathbb{E}^{n}_{ν} [Nol96])

Let ψ be the warped product decomposition of \mathbb{E}^n_{ν} determined by the initial data $(\bar{p}; V_0 \oplus V_1; a)$ given above. Then N_0 has the following form:

$$N_0 = \{ p \in V_0 | \langle a, p \rangle > 0 \}$$

and

$$\rho: \begin{cases} N_0 & \to \mathbb{R}_+ \\ p_0 & \mapsto \langle a, p_0 \rangle \end{cases}$$

The map ψ is an isometry onto the following set:

$$\operatorname{Im}(\psi) = \begin{cases} \{p \in \mathbb{E}_{\nu}^{n} \mid \operatorname{sgn}(P_{1}p)^{2} = \epsilon\} & \text{non-null case} \\ \{p \in \mathbb{E}_{\nu}^{n} \mid \langle a, p \rangle > 0\} & \text{null case} \end{cases}$$

Furthermore, the following equation holds:

$$\psi(p_0, p_1)^2 = p_0^2 \tag{8.4.2}$$

PROOF See Corollary D.4.10.

In fact, for $(p_0, p_1) \in N_0 \times N_1$, ψ has one of the following forms, first if ψ is non-null:

$$\psi(p_0, p_1) = P_0 p_0 + \langle a, p_0 \rangle (p_1 - c)$$
(8.4.3)

where $c = \bar{p} - \frac{a}{a^2}$, and if ψ is null:

$$\psi(p_0, p_1) = P_0 p_0 + (\langle b, p_0 \rangle - \frac{1}{2} \langle a, p_0 \rangle (P_1 p_1)^2) a + \langle a, p_0 \rangle b + \langle a, p_0 \rangle P_1 p_1$$
(8.4.4)

The above forms are obtained from the equation for ψ from the above theorem by expanding p_0 in an appropriate basis. We note that the warped products with multiple spherical factors can be obtained using the standard ones described above. Indeed, suppose $\phi_1 : N'_0 \times_{\rho_1} N_1 \to \mathbb{E}^n_{\nu}$ is the warped product decomposition determined by $(\bar{p}; V_0 \oplus V_1; a_1)$ as above. Since V_0 is pseudo-Euclidean, consider a warped product decomposition, $\phi_2 : \tilde{N}_0 \times_{\rho_2} N_2 \to V_0$, determined by $(\bar{p}; \tilde{V}_0 \oplus \tilde{V}_1; a_2)$ with $V_0 \cap W_0^{\perp} \subset \tilde{W}_0$ (hence $a_1 \in \tilde{W}_0$). Note that \tilde{W}_0 is the subspace W_0 from the above construction for ϕ_2 . Let $N_0 := N'_0 \cap \tilde{N}_0$, then one can check that the map ψ defined by:

$$\psi: \begin{cases} N_0 \times_{\rho_1} N_1 \times_{\rho_2} N_2 & \to \mathbb{E}_{\nu}^n \\ (p_0, p_1, p_2) & \mapsto \phi_1(\phi_2(p_0, p_2), p_1) \end{cases}$$

is a warped product decomposition of \mathbb{E}^n_{ν} satisfying Eq. (8.4.1). We illustrate this construction with an example.

Example 8.4.6 (Constructing multiply warped products)

Suppose ϕ_1 and ϕ_2 are given as follows:

$$\phi_1(p'_0, p_1) = P'_0 p'_0 + \langle a_1, p'_0 \rangle (p_1 - c_1)$$

$$\phi_2(\tilde{p}_0, p_2) = \tilde{P}_0 \tilde{p}_0 + \langle a_2, \tilde{p}_0 \rangle (p_2 - c_2)$$

Now observe that $\rho_1(\phi_2(\tilde{p}_0, p_2)) = \rho_1(\tilde{p}_0)$, which follows from the above equation for ϕ_2 and the fact that $a_1 \in \tilde{W}_0$. Then,

$$\psi(p_0, p_1, p_2) = \phi_1(\phi_2(p_0, p_2), p_1)$$

= $P'_0\phi_2(p_0, p_2) + \langle a_1, \phi_2(p_0, p_2)\rangle(p_1 - c_1)$
= $P'_0\tilde{P}_0p_0 + \langle a_2, p_0\rangle(p_2 - c_2) + \langle a_1, p_0\rangle(p_1 - c_1)$

where $P'_0 \tilde{P}_0$ is the orthogonal projector onto $\tilde{W}_0 \cap W_0 = \tilde{V}_0 \cap \operatorname{span}\{a_1, a_2\}^{\perp}$. A similar calculation shows that ψ satisfies Eq. (8.4.1), since ϕ_1 and ϕ_2 each satisfy it.

This procedure can be repeated as many times as necessary to obtain the more general warped products given by Eq. (8.4.1). Hence the properties of the more general warped product decompositions of \mathbb{E}^n_{ν} can be deduced from Theorem 8.4.5.

The following proposition shows that any proper warped product decomposition of \mathbb{E}^n_{ν} in canonical form restricts to a warped product decomposition of $\mathbb{E}^n_{\nu}(\kappa)$ where $\kappa \neq 0$. Its proof is straightforward consequence of Eq. (8.4.2).

Theorem 8.4.7 (Restricting Warped products to $\mathbb{E}^n_{\nu}(\kappa)$)

Let ψ be a proper warped product decomposition of \mathbb{E}^n_{ν} associated with $(\bar{p}; \bigoplus_{i=0}^k V_i; a_1, ..., a_k)$ in canonical form. Suppose $\kappa^{-1} := \bar{p}^2 \neq 0$ and let $N' := N_0(\kappa) \times_{\rho_1} N_1 \times \cdots \times_{\rho_k} N_k$. Then $\phi: N' \to \mathbb{E}^n_{\nu}(\kappa)$ defined by $\phi := \psi|_{N'}$ is a warped product decomposition of $\mathbb{E}^n_{\nu}(\kappa)$ passing through \bar{p} .

PROOF See Theorem D.6.5.

Hence the details of warped product decompositions of $\mathbb{E}_{\nu}^{n}(\kappa)$ can be deduced from Theorem 8.4.5. More information on these decompositions can be found in Appendix D. The results presented here will be applied in Section 9.5, where examples can also be found.

Chapter 9

Concircular tensors in Spaces of Constant Curvature

It has been shown in Section 6.5 that any point-wise diagonalizable concircular tensor hereafter called a *OCT* can be used to recursively construct separable coordinates for the (geodesic) Hamilton-Jacobi equation. These coordinates were called Kalnins-Eisenhart-Miller (KEM) coordinates. It was shown in Chapter 7 that all orthogonal separable coordinates for the Hamilton-Jacobi equation in spaces of constant curvature occur this way. Hence the classification of OCTs in spaces of constant curvature is of fundamental importance for classifying separable coordinates in these spaces.

Specifically, OCTs have the following uses:

- 1. An algebraic classification of these tensors modulo the action of the isometry group can be used to obtain a notion of in-equivalence for KEM coordinate systems.
- 2. Crampin has shown in [Cra03] that one can obtain transformations to separable coordinates for OCTs with functionally independent eigenfunctions. It was shown in Section 6.5 that a knowledge of the warped product decompositions of the space is sufficient to obtain transformations to separable coordinates for any KEM coordinate system.
- 3. When concircular tensors have simple eigenfunctions, it was shown in [Ben05] (see also [Ben92a; Ben93; Ben04]) that a basis for the Killing-Stackel space can be obtained. These results have been generalized to arbitrary KEM webs in Section 6.6.
- 4. With a classification of concircular tensors, the BEKM separation algorithm (presented in Section 6.7), can be executed to solve the separation of variables problem for natural Hamiltonians.

In this chapter we will obtain a complete (local) classification of orthogonal concircular tensors in all spaces of constant curvature with Euclidean and Lorentzian signature¹. This will enable one to carry out the above tasks using only (linear) algebraic operations. We note that the contents of this chapter are from the article [RM14c].

More details on our classification and the way in which it's done is given in Section 9.1.3, after we have briefly reviewed the necessary material from Chapter 6 in Sections 9.1.1 and 9.1.2. We will assume the reader is familiar with results and notations introduced in Chapter 8. Some of our results are also summarized in Section 9.1.3.

9.1 Preliminaries and Summary

9.1.1 Concircular tensors

Recall from Chapter 6 that a tensor $L \in S^p(M)$ is called a *concircular tensor* (CT) of valence p if there exists $C \in S^{p-1}(M)$ (called the *conformal factor*) such that

$$\nabla_x L = C \odot x$$

for all $x \in \mathfrak{X}(M)$. Throughout this chapter, we will simply call L a concircular tensor when p = 2.

We now recall some properties of OCTs from Chapter 6. First, given a $\binom{1}{1}$ tensor L, let N_L be the Nijenhuis tensor (torsion) of L (see Definition B.0.13). We say that L is *torsionless* if its Nijenhuis tensor vanishes. Then if L is a concircular tensor, the following equations hold by Proposition 6.3.1

$$[L,G] = -2\nabla \operatorname{tr}(L) \odot G \quad ([L,G]_{abc} = -2\nabla_{(a}L_{bc)})$$
$$N_L = 0$$

Conversely, by Proposition 6.3.5, an orthogonal tensor satisfying the above equations is a C-tensor. The first of the above equations tells us that a C-tensor is a conformal Killing tensor of trace-type. The second equation can be interpreted if we assume L is an OC-tensor.

Suppose now that L is an OC-tensor with eigenspaces $(E_i)_{i=1}^k$ and corresponding eigenfunctions $\lambda^1, ..., \lambda^k$. Since an OC-tensor has Nijenhuis torsion zero, by Theorem B.0.20 the eigenspaces $(E_i)_{i=1}^k$ are orthogonally integrable and each eigenfunction λ^i depends only on E_i . Furthermore the trace-type condition implies that the eigenfunction corresponding to a multidimensional eigenspace of L is a constant (see Proposition 6.2.2 (3)).

¹The classification for other signatures can be obtained fairly easily if one wishes.

Suppose D is a multidimensional eigenspace of a non-trivial² OCT L. Denote by D^{\perp} the distribution orthogonal to D. Then one can show that (see Proposition 6.3.6):

- There is a local product manifold B × F of Riemannian manifolds (B, g_B) and (F, g_F) such that:
 {p} × F is an integral manifold of D for any p ∈ B and
 - $B \times \{q\}$ is an integral manifold of D^{\perp} for any $q \in F$.
- $B \times F$ equipped with the metric $\pi_B^* g_B + \rho^2 \pi_F^* g_F$ for a specific function $\rho : B \to \mathbb{R}^+$ is locally isometric to (M, g); where π_B (resp. π_F) is the canonical projection onto B (resp. F).

Such a product manifold is called a *warped product* and is denoted $B \times_{\rho} F$. We also say in this case that the warped product $B \times_{\rho} F$ is *adapted* to the splitting (D^{\perp}, D) . The manifold F is a *spherical submanifold* and B is *geodesic submanifold* of M (see Section 3.1). An important observation is that L restricted to B is an OCT (by Proposition 4.4.15); we will use this later to construct OCTs from Benenti tensors.

In general if L has multiple multidimensional eigenspaces, we will have to consider more general warped products. So suppose $M = \prod_{i=0}^{k} M_i$ is a product manifold of pseudo-Riemannian manifolds (M_i, g_i) where dim $M_i > 0$ for i > 0. Equip M with the metric $g = \sum_{i=0}^{k} \rho_i^2 \pi_i^* g_i$ where $\rho_i : M_0 \to \mathbb{R}^+$ are functions with $\rho_0 \equiv 1$ and $\pi_i : M \to M_i$ are the canonical projection maps. Additionally we assume either dim $M_0 > 0$ or k > 1. Then (M, g) is called a *warped product* and the metric g is called a *warped product metric*. If dim $M_0 = 0$ then (M, g) is called a *pseudo-Riemannian product*. The warped product is denoted by $M_0 \times_{\rho_1} M_1 \times \cdots \times_{\rho_k} M_k$. M_0 is called the geodesic factor of the warped product and the M_i for i > 0 are called spherical factors. See Section 3.5 and references therein for more on warped products.

The following class of OCTs are fundamental to the classification:

Definition 9.1.1 (Irreducible concircular tensors)

An OC-tensor with functionally independent eigenfunctions is referred to as an ICT or more succinctly an *IC-tensor*. To be precise, an IC-tensor has real eigenfunctions $u^1, ..., u^k$ (counted without multiplicity) satisfying:

$$\mathrm{d}u^1\wedge\cdots\wedge\mathrm{d}u^k\neq 0$$

Furthermore an OC-tensor which is not irreducible is called *reducible*.

Since we observed earlier that the eigenfunction associated with a multidimensional eigenspace of an OCT is constant, it follows that an ICT must have simple eigenfunctions,

²By a non-trivial concircular tensor, we mean one which is not a multiple of the metric when n > 1.

hence ICTs are Benenti tensors. The special property that ICTs have is that their eigenfunctions can be used as (local) coordinates for the separable web they induce [Cra03]. We will refer to these coordinates as the *canonical coordinates* induced by these tensors.

Away from singular points, locally, we can assume a reducible OC-tensor has eigenfunctions $u^1, ..., u^k$ which are functionally independent and the rest of which are constants. Indeed, in this thesis, this is what we will mean by a reducible OC-tensor. More generally we say a CT is *reducible* if it admits a non-degenerate eigenspace with constant eigenfunction. We will outline in Section 9.1.3 how we will break down the classification in terms of irreducible and reducible OCTs.

9.1.2 Properties of OCTs

We will now list some properties of OCTs that will be used later. The following proposition gives a necessary and sufficient condition to determine when two OCTs (one of which is not covariantly constant) share the same eigenspaces.

Proposition 9.1.2

Suppose M is a connected manifold and L is an OCT on M which is not covariantly constant (around any neighborhood). Then \tilde{L} is a CT sharing the same eigenspaces as Liff there exists $a \in \mathbb{R} \setminus \{0\}$ and $b \in \mathbb{R}$ such that

$$\tilde{L} = aL + bG$$
 \diamondsuit

PROOF See Proposition 6.2.5.

The above proposition no longer holds if we relax the assumption that L is not covariantly constant. One can easily see why by considering any non-trivial covariantly constant symmetric tensor in Euclidean space. We now define an important notion for classifying KEM webs.

Definition 9.1.3 (Geometric Equivalence of CTs)

We say two CTs L and L are geometrically equivalent if there exists $a \in \mathbb{R} \setminus \{0\}, b \in \mathbb{R}$ and $T \in I(M)$ such that

$$\tilde{L} = aT_*L + bG$$

An immediate corollary of the above proposition is the following:

Corollary 9.1.4 (Geometric Equivalence of OCTs)

Suppose M is a connected manifold. Suppose L and \tilde{L} are OCTs with respective eigenspaces $\mathcal{E} = (E_1, \ldots, E_k)$ and $\tilde{\mathcal{E}} = (\tilde{E}_1, \ldots, \tilde{E}_k)$. Suppose further that \mathcal{E} is not a Riemannian product net, equivalently one of the CTs is not covariantly constant. Then \mathcal{E} and $\tilde{\mathcal{E}}$ are related by $T \in I(M)$, i.e. $\tilde{E}_i = T_* E_{\sigma(i)}$ for each i (where σ is a permutation of $\{1, \ldots, k\}$) iff L and \tilde{L} are geometrically equivalent.

The above corollary implies that the classification of isometrically inequivalent KEM webs can be reduced to the classification of geometrically inequivalent OCTs. For the proof of the following theorem (which was already presented in Section 6.1), see [TCS05; Cra07].

Theorem 9.1.5 (The Vector Space of Concircular tensors [TCS05])

If n > 1, then the C-tensors of valence $r \leq 2$ form a finite dimensional real vector space with maximal dimension equal to the dimension of the space of constant symmetric r-tensors in \mathbb{R}^{n+1} . Furthermore the maximal dimension is achieved if and only if the space has constant curvature.

The above theorem implies the following:

Corollary 9.1.6 (Concircular tensors in spaces of constant curvature)

Suppose M^n is a space of constant curvature with n > 1 and let $r \leq 2$. Let $\beta = \{v_1, \ldots, v_{n+1}\}$ be a basis for the space of concircular vectors, then a given C-tensor of valence r can be written uniquely as a linear combination of r-fold symmetric products of the vectors in β .

9.1.3 Summary of Results

We first give an overview of the classification. The classification breaks down into three parts: obtaining canonical forms for C-tensors modulo the action of the isometry group (Sections 9.2 and 9.3), classifying the webs described by IC-tensors (Section 9.4) and obtaining warped product decompositions adapted to reducible OCTs (Section 9.5).

The webs formed by IC-tensors are the basic building blocks of all separable webs. Section 9.4 is devoted to obtaining information about these webs from the corresponding IC-tensors. In that section we obtain the transformation from the canonical coordinates (u^i) induced by these tensors to Cartesian coordinates (x^i) and we obtain the metric in canonical coordinates. This is done by first calculating the characteristic polynomial of all CTs in spaces of constant curvature in a Cartesian coordinate system. In examples, we will also show how to obtain the coordinate domains for coordinate systems induced by IC-tensors.

To obtain all orthogonal separable coordinates in spaces of constant curvature, we also have to consider reducible OCTs. Let L be a non-trivial reducible OCT and suppose $\psi: N_0 \times_{\rho_1} N_1 \times \cdots \times_{\rho_k} N_k \to M$ is a local warped product decomposition of M adapted to the eigenspaces of L such that $L_0 := L|_{N_0}$ is an ICT³. Let $(x_0) = (u^1, \ldots, u^{n_0})$ be the canonical coordinates induced by L_0 on some open subset of N_0 . For i > 0 suppose $(x_i) = (x_i^1, \ldots, x_i^{n_i})$ are separable coordinates for N_i , then it was shown in Section 6.5

³If L has only constant eigenfunctions, we can choose N_0 to be a point.

that the coordinates $\psi(x_0, x_1, \ldots, x_k)$ are separable coordinates for M. To construct the separable coordinates (x_i) on N_i where i > 0, one would apply this procedure again on N_i equipped with the induced metric. It was shown in Chapter 7 that all orthogonal separable coordinates for spaces of constant curvature arise this way. Hence a remaining problem is to develop a method to construct warped product decompositions which decompose a given reducible OCT as above; this is done in Section 9.5. Together with the results of Section 9.4, this gives a recursive procedure to construct the orthogonal separable coordinates of these spaces.

Finally in Section 9.6 we will show how to apply the theory developed in this chapter to solve motivating problems.

The classification generally breaks down into one for pseudo-Euclidean space \mathbb{E}_{ν}^{n} then one for its spherical submanifolds $\mathbb{E}_{\nu}^{n}(\kappa)$ (which usually reduces to a similar problem in \mathbb{E}_{ν}^{n}). We give more details in the following subsections.

pseudo-Euclidean space

First we define the *dilatational vector field*, r, to be the vector field given in Cartesian coordinates (x^i) by $r = \sum_i x^i \partial_i$. The general concircular contravariant tensor in \mathbb{E}^n_{ν} is given as follows (see Proposition 9.2.2):

$$L = A + 2w \odot r + mr \odot r \tag{9.1.1}$$

where $A \in C_0^2(\mathbb{E}^n_{\nu})$, $w \in C_0^1(\mathbb{E}^n_{\nu})$ and $m \in C_0^0(\mathbb{E}^n_{\nu})$. For $k \ge 0$, define constants ω_k as follows:

$$\omega_k = \begin{cases} m & \text{if } k = 0\\ \langle w, A^{k-1}w \rangle & \text{else} \end{cases}$$
(9.1.2)

The above constants aren't necessarily invariant under isometries. But invariants can be defined from them.

Definition 9.1.7

Suppose L is a CT in \mathbb{E}_{ν}^{n} as defined above. Then we define the *index* of L to be the first integer $k \geq 0$ for which $\omega_{k} \neq 0$; L is said to be *non-degenerate* if such an integer exists. Furthermore if L is non-degenerate, it has an associated sign (characteristic):

$$\varepsilon = \begin{cases} 1 & 1 \text{ if } k \text{ is even} \\ \operatorname{sgn} \omega_k & \operatorname{if } k \text{ is odd} \end{cases} \square$$

The following theorem which is proven in Section 9.2 summarizes our results on the canonical forms of concircular tensors; it classifies C-tensors into five disjoint classes.

Theorem 9.1.8 (Canonical forms for CTs in \mathbb{E}^n_{μ})

Let $\tilde{L} = \tilde{A} + mr \otimes r^{\flat} + w \otimes r^{\flat} + r \otimes w^{\flat}$ be a CT in \mathbb{E}^{n}_{ν} . Let k be the index and ε be the sign of \tilde{L} if \tilde{L} is non-degenerate. These quantities are geometric invariants of \tilde{L} . Furthermore, after a possible change of origin and after changing to a geometrically equivalent CT, $L = a\tilde{L}$ for some $a \in \mathbb{R} \setminus \{0\}$, \tilde{L} admits precisely one of the following canonical forms.

Central: If k = 0

$$L = A + r \otimes r^{\natural}$$

non-null Axial: If k = 1, *i.e.* m = 0, and $\langle w, w \rangle \neq 0$:

There exists a vector $e_1 \in \text{span}\{w\}$ such that L has the following form:

$$L = A + e_1 \otimes r^{\flat} + r \otimes e_1^{\flat} \qquad \qquad Ae_1 = 0, \quad \langle e_1, e_1 \rangle = \varepsilon$$

null Axial: If $k \ge 2$, hence m = 0 and $\langle w, w \rangle = 0$:

There exists a skew-normal sequence $\beta = \{e_1, ..., e_k\}$ with $\langle e_1, e_k \rangle = \varepsilon$ where $e_1 \in$ span $\{w\}$ which is A-invariant such that L has the following form:

$$L = A + e_1 \otimes r^{\flat} + r \otimes e_1^{\flat}$$
$$A|_{\beta} = J_k(0)^T = \begin{pmatrix} 0 & & & \\ 1 & 0 & & \\ & 1 & \ddots & \\ & & \ddots & 0 & \\ & & & 1 & 0 \end{pmatrix}$$

Cartesian: If k doesn't exist, m = 0 and w = 0

$$L = \tilde{A}$$

degenerate null Axial: If k doesn't exist and $w \neq 0$

Remark 9.1.9

The degenerate null axial concircular tensors will be of no concern to us. In Euclidean space they don't occur and it will be proven later (see Section 9.2.3) that in Minkowski space that they are never orthogonal concircular tensors. \Box

Remark 9.1.10

The precise classification for Euclidean and Minkowski space can be directly inferred from the above theorem by imposing the signature of the metric. The classification for Euclidean space is clear. In Minkowski space, $k \leq 3$ and when k = 3 the sign of the axial CT must be positive (see Lemma 8.1.1).

Remark 9.1.11

When k = 0 and 1 respectively, the translation vector v for the isometry $T : r \to r + v$ which sends \tilde{L} to canonical form is given as follows:

$$v = \frac{w}{\omega_0} \qquad \qquad \text{if } k = 0 \qquad (9.1.3)$$

$$v = \frac{1}{\omega_1} (Aw - \frac{1}{2} \frac{\omega_2}{\omega_1} w)$$
 if $k = 1$ (9.1.4)

For the general case, see Section 9.2.3.

One can easily deduce that in Euclidean or Minkowski space, any covariantly nonconstant OCT is non-degenerate. Hence we will only be interested in non-degenerate CTs throughout this chapter.

Some notation will be useful. The matrix A will be called the *parameter matrix* and the vector w the *axial vector* of the CT. When $k \ge 1$ in the above theorem, we will refer to the CT as an *axial concircular tensor*.

Suppose L is a non-degenerate CT in the canonical form given by Theorem 9.1.8. We denote by D the A-invariant subspace spanned by w, Aw, \ldots . This subspace is either zero (if w = 0) or metrically non-degenerate. We will let $A_c := A|_{D^{\perp}}, A_d := A|_D$ and the central CT in D^{\perp} with parameter matrix A_c by L_c . Furthermore we define the following functions:

$$p(z) := \det(zI - L)$$
$$B(z) := \det(zI - A_c)$$

where the second determinant is evaluated in D^{\perp} .

The canonical forms for non-degenerate CTs can be enumerated by choosing a nondegenerate CT from Theorem 9.1.8 then choosing a metric-Jordan canonical form for the pair $(A|_{D^{\perp}}, g|_{D^{\perp}})$. The proofs of these canonical forms, which are given in Section 9.2, can be omitted on first reading. Once these canonical forms are obtained, in Sections 9.4.1 and 9.4.2 we will calculate the characteristic polynomial for non-degenerate CTs in \mathbb{E}_{ν}^{n} . Using this, for ICTs we can calculate the transformation from their canonical coordinates

to Cartesian coordinates and the metric in canonical coordinates. Then in Section 9.5.1 we will show how to obtain the warped product decompositions induced by reducible OCTs.

Spherical submanifolds of pseudo-Euclidean space

First the orthogonal projection R onto the spherical distribution r^{\perp} is given as follows:

$$R = I - \frac{r \otimes r^{\flat}}{r^2} \qquad \qquad R^* = I - \frac{r^{\flat} \otimes r}{r^2}$$

Then the general CT in $\mathbb{E}^n_{\nu}(\kappa)$ is obtained by restricting $A \in C^2_0(\mathbb{E}^n_{\nu})$ to $\mathbb{E}^n_{\nu}(\kappa)$. It is given as follows in \mathbb{E}^n_{ν} in contravariant form (see Proposition 9.3.2):

$$L = RAR^* = A + \kappa^2 \langle r, Ar \rangle r \odot r - 2\kappa (Ar \odot r) \qquad L^{ij} = R^i_{\ l} A^{lk} R^j_{\ k} \tag{9.1.5}$$

The matrix A is called the *parameter matrix* of the CT. We denote by L_c the central CT in \mathbb{E}^n_{ν} with parameter matrix A. Note that $L = RL_cR^*$. We will see later that several questions concerning L can be related to similar ones concerning L_c .

The canonical forms for these CTs can be enumerated by choosing a metric-Jordan canonical form for the pair (A, g). The proofs of these canonical forms, which are given in Section 9.3, can be omitted on first reading. Once these canonical forms are obtained, in Section 9.4.3 we will calculate the characteristic polynomial for CTs in $\mathbb{E}^n_{\nu}(\kappa)$ by making use of the solution to the similar problem in \mathbb{E}^n_{ν} . Using this, for ICTs we can calculate the transformation from their canonical coordinates to Cartesian coordinates and the metric in canonical coordinates. Then in Section 9.5.2 we will show how to obtain the warped product decompositions induced by reducible OCTs by making use of the solution to the similar problem in \mathbb{E}^n_{ν} .

9.2 Canonical forms for Concircular tensors in pseudo-Euclidean space

9.2.1 Standard Model of pseudo-Euclidean space

In this section we recall the CVs and CTs in \mathbb{E}_{ν}^{n} in its standard vector space model, which were calculated in Section 6.4. These results are well known [Cra07; Ben05], but we include it here for completeness.

First we define the *dilatational vector field*, r, to be the vector field satisfying for any $p \in \mathbb{E}_{\nu}^{n}$, $r_{p} = p \in T_{p}\mathbb{E}_{\nu}^{n}$. In Cartesian coordinates (x^{i}) , we have

$$r = \sum_{i} x^{i} \partial_{i}$$

The general CV in \mathbb{E}^n_{ν} is given by the following proposition.

Proposition 9.2.1 (Concircular vectors in \mathbb{E}^n_{ν} [Cra07])

A vector $v \in \mathfrak{X}(\mathbb{E}^n_{\nu})$ is a CV in \mathbb{E}^n_{ν} iff there exists $a \in C^0_0(\mathbb{E}^n_{\nu})$ and $b \in C^1_0(\mathbb{E}^n_{\nu})$ such that

$$v = ar + b$$

where r is the dilatational vector field.

PROOF See Proposition 6.4.3.

Then using Corollary 9.1.6 we can deduce the general CT in \mathbb{E}^n_{ν} :

Proposition 9.2.2 (Concircular tensors in \mathbb{E}^n_{ν})

L is a concircular 2-tensor in \mathbb{E}^n_{ν} iff there exists $A \in C^2_0(\mathbb{E}^n_{\nu})$, $w \in C^1_0(\mathbb{E}^n_{\nu})$ and $m \in C^0_0(\mathbb{E}^n_{\nu})$ such that:

$$L = A + 2w \odot r + mr \odot r$$

where r is the dilatational vector field. The tensors A, w and m are uniquely determined by L. \Box

9.2.2 Parabolic Model of pseudo-Euclidean space

In order to obtain canonical forms for CTs it will be useful to work with a different model of \mathbb{E}_{ν}^{n} . We will refer to it as the parabolic model of \mathbb{E}_{ν}^{n} , to be introduced shortly. The main reason for working with this model is because it is a spherical submanifold of the ambient space in which the isometries of \mathbb{E}_{ν}^{n} are linearized, which we will elaborate on shortly.

Recall that \mathbb{P}^n_{ν} was defined in Section 8.3. We stated that an explicit isometry with \mathbb{E}^n_{ν} can be obtained by fixing $b \in \mathbb{P}^n_{\nu}$, i.e. b is lightlike and $\langle a, b \rangle = 1$. If we let $V := \operatorname{span}\{a, b\}^{\perp}$, note that $V \cong \mathbb{E}^n_{\nu}$, then for $x \in V$:

$$\psi(x) = b + x - \frac{1}{2}x^2 a \in \mathbb{P}^n_{\nu}$$
(9.2.1)

gives an explicit isometry between \mathbb{E}_{ν}^{n} and \mathbb{P}_{ν}^{n} . By definition of \mathbb{P}_{ν}^{n} , it follows that $T_{p}\mathbb{P}_{\nu}^{n} = p^{\perp} \cap a^{\perp} = \operatorname{span}\{p, a\}^{\perp}$. Also note that for $x \in \mathbb{P}_{\nu}^{n}$

$$\psi^{-1}(x) = x - \langle x, b \rangle a - \langle x, a \rangle b$$

An important reason for working with \mathbb{P}^n_{ν} is the following [Nol96]:
Proposition 9.2.3 (Isometry group of \mathbb{P}^n_{μ})

The isometry group of \mathbb{P}^n_{ν} is:

$$I(\mathbb{P}^{n}_{\nu}) = \{ T \in O_{\nu+1}(n+2) \mid Ta = a \}$$

Furthermore suppose we fix an isometry with \mathbb{E}_{ν}^{n} via Eq. (9.2.1) by fixing a subspace $V \subset a^{\perp}$ such that $V \simeq \mathbb{E}_{\nu}^{n}$, then for $p \in V$ and $\tilde{p} \in V^{\perp}$ we have the following Lie group isomorphism:

$$\phi: \begin{cases} O_{\nu}^{n}(V) \ltimes V & \to I(\mathbb{P}_{\nu}^{n}) \\ (B, v) & \mapsto \phi(B, v) \end{cases}$$

where

$$\phi(B,v)(p+\tilde{p}) = \tilde{p} + Bp + \langle a, \tilde{p} \rangle v - (\langle Bp, v \rangle + \frac{1}{2} \langle a, \tilde{p} \rangle v^2))a \qquad (9.2.2)$$

PROOF See Proposition D.5.2 or [Nol96, lemma 6] which covers the case when \mathbb{E}_{ν}^{n} is Euclidean.

Remark 9.2.4

If $\psi : \mathbb{E}_{\nu}^{n} \to \mathbb{P}_{\nu}^{n}$ is the standard embedding from Eq. (9.2.1), then ψ is equivariant. In other words, if we let Tp := Bp + v for $(B, v) \in O_{\nu}^{n}(V) \ltimes V$ as above, and $\hat{T} := \phi(B, v)$ then $\psi \circ T(p) = \hat{T} \circ \psi(p)$.

We also have the following:

Lemma 9.2.5

For $\bar{p} \in V$ and $X \in T_{\bar{p}}V$

$$\psi_* X = X - \langle X, \bar{p} \rangle a$$

For $Y \in T_{\psi(\bar{p})} \mathbb{P}^n_{\nu}$, the inverse of the above map is given by:

$$P_b: \begin{cases} T_{\psi(\bar{p})} \mathbb{P}^n_{\nu} & \to T_{\bar{p}} V \\ Y & \mapsto Y - \langle Y, b \rangle a \end{cases} \square$$

PROOF The first statement is clear. First observe that $P_b\psi_*X = X$. Now,

$$\psi_* P_b Y = Y - \langle Y, b \rangle a - \langle Y, \bar{p} \rangle a$$

Now $0 = \langle Y, \psi(\bar{p}\rangle) = \langle Y, b \rangle + \langle Y, \bar{p} \rangle$. Thus $\psi_* P_b Y = Y$.

Furthermore we denote by P_1 the orthogonal projector onto $T\mathbb{P}^n_{\nu}$. It is given as follows for $r \in \mathbb{E}^{n+2}_{\nu+1}$

$$P_1: \begin{cases} T_r \mathbb{E}_{\nu+1}^{n+2} & \to T_r \mathbb{E}_{\nu+1}^{n+2} \\ V & \mapsto V - \langle V, r \rangle \, a - \langle V, a \rangle \, r \end{cases}$$

We will now calculate the CT in $\mathbb{E}_{\nu+1}^{n+2}$ which restricts to the most general CT in \mathbb{P}_{ν}^{n} . Due to Corollary 9.1.6 we only need to examine how CVs restrict. By Proposition 9.2.1 and Theorem 9.1.5, the general CV in $\mathbb{E}_{\nu+1}^{n+2}$ can be written

$$v = c_0 r + \sum_{i=1}^{n} c_i a_i + c_{n+1} b + c_{n+2} a_i$$

where each $c_i \in \mathbb{R}$, a_1, \ldots, a_n is a basis for V and r is the dilatational vector field in $\mathbb{E}_{\nu+1}^{n+2}$. Then

$$P_b P_1 v = P_b (\sum_{i=1}^n c_i (a_i - \langle a_i, r \rangle \, a) + c_{n+1} (b - \langle b, r \rangle \, a - r))$$

= $\sum_{i=1}^n c_i a_i - c_{n+1} x$

where x is the dilatational vector field in V. Then using Corollary 9.1.6 we have proven the following:

Proposition 9.2.6

Suppose \mathbb{P}^n_{ν} is identified with \mathbb{E}^n_{ν} by the embedding in Eq. (9.2.1). Denote by $V = \text{span}\{a,b\}^{\perp}$, let $\tilde{A} \in C_0^2(V)$, $w \in C_0^1(V)$, and $m \in C_0^0(V)$. Define

$$A = \tilde{A} + mb \odot b - 2w \odot b \tag{9.2.3}$$

Then the restriction of A to V, denoted L, via the embedding in Eq. (9.2.1) is:

$$L = \tilde{A} + mr \odot r + 2w \odot r$$

Note that A is completely determined by the condition Ab = 0. Now for $A \in C_0^2(\mathbb{E}_{\nu+1}^{n+2})$, define A_b by

$$(A_b)^{ij} := (P_b)^i{}_l A^{lk} (P_b)^j{}_k$$

Note that b is an eigenvector of A_b with eigenvalue 0. Also observe that

$$P_1P_b = P_1 - w \otimes b^\flat + w \otimes b^\flat = P_1$$

The above equation shows that A and A_b induce the same CT on \mathbb{P}^n_{ν} . From the calculations proceeding Eq. (9.2.3) we see that

$$\{a_1 - \langle a_1, r \rangle a, \dots, a_n - \langle a_n, r \rangle a, b - \langle b, r \rangle a - r\}$$

is basis for the space of CVs on \mathbb{P}^n_{ν} . Thus it follows from Corollary 9.1.6 and the proceeding calculations that $A, B \in C^2_0(\mathbb{E}^{n+2}_{\nu+1})$ induce the same CT on \mathbb{P}^n_{ν} iff for some $b \in \mathbb{P}^n_{\nu}$ we have

$$A_b = B_b$$

Furthermore, one should note that if $b, c \in \mathbb{P}^n_{\nu}$, then $(A_c)_b = A_b$. Hence it follows that if $A_b = B_b$ for some $b \in \mathbb{P}^n_{\nu}$ then $A_c = B_c$ for all $c \in \mathbb{P}^n_{\nu}$.

9.2.3 Existence of Canonical forms

In this section $A \in C_0^2(\mathbb{E}_{\nu+1}^{n+2})$. We are interested in finding canonical forms for the CT on \mathbb{P}_{ν}^n induced by this tensor. As it was shown in the previous section, the induced CT depends only on A_b for some $b \in \mathbb{P}_{\nu}^n$. Hence our goal will be to find $\tilde{b} \in \mathbb{P}_{\nu}^n$ such that $A_{\tilde{b}}$ is in a canonical form. Since the isometry with \mathbb{E}_{ν}^n (see Eq. (9.2.1)) is fixed by a vector $b \in \mathbb{P}_{\nu}^n$, we will then choose $T \in I(\mathbb{P}_{\nu}^n)$ such that $T\tilde{b} = b$. This will transform $A_{\tilde{b}}$ to $(T_*A)_b$ which can be restricted to \mathbb{E}_{ν}^n using Proposition 9.2.6 to obtain a canonical form for the original CT in \mathbb{E}_{ν}^n .

To obtain the canonical choice of $b \in \mathbb{P}_{\nu}^{n}$, first note that A_{b} is completely determined by the fact that $A_{b}b = 0$. Secondly, note that since isometries of \mathbb{P}_{ν}^{n} fix a, it follows that for each $l \geq 0$, $\langle a, A^{l}a \rangle$ are invariants of A. Although these are in general not invariants of the CT induced by A, they will play a significant role in the classification. Thirdly, since a cannot be transformed by isometries, we will attempt to choose $b \in \mathbb{P}_{\nu}^{n}$ such that a is a basis vector in a metric-Jordan canonical basis for A_{b} . Since $\langle a, b \rangle = 1$, one can deduce that (using the metric-Jordan canonical form discussed in Section 8.2) in the simplest cases, a, b lie in the same eigenspace of A_{b} or a generates a Jordan cycle ending in a constant multiple of b. These observations motivate our search for b.

For the following calculations, $b \in \mathbb{P}^n_{\nu}$ is arbitrary and we let $\tilde{A} := A_b$. The following lemma will get us started:

Lemma 9.2.7

Suppose there is $k \in \mathbb{N}$ such that $\langle a, A^l a \rangle = 0$ for $0 \le l < k$. Then for each $0 \le l \le k$

$$\tilde{A}^{l}a = A^{l}a - \sum_{j=0}^{l-1} \left\langle b, A^{l-j}a \right\rangle \tilde{A}^{j}a \tag{9.2.4}$$

Furthermore, if $0 \leq l \leq k$ then

$$\left\langle a, \tilde{A}^{l}a \right\rangle = \left\langle a, A^{l}a \right\rangle$$
 (9.2.5)

So the constants $\langle a, A^l a \rangle$ are invariants of the CT on \mathbb{P}^n_{ν} induced by A.

PROOF We prove Eq. (9.2.4) by induction. It clearly holds for l = 0, 1. Now assume it holds for l - 1, then

$$\begin{split} \tilde{A}^{l}a &= \tilde{A}A^{l-1}a - \sum_{j=0}^{l-2} \left\langle b, A^{l-1-j}a \right\rangle \tilde{A}^{j+1}a \\ &= A^{l}a - a \left\langle b, A^{l}a \right\rangle - \sum_{j=0}^{l-2} \left\langle b, A^{l-1-j}a \right\rangle \tilde{A}^{j+1}a \\ &= A^{l}a - a \left\langle b, A^{l}a \right\rangle - \sum_{j=1}^{l-1} \left\langle b, A^{l-j}a \right\rangle \tilde{A}^{j}a \\ &= A^{l}a - \sum_{j=0}^{l-1} \left\langle b, A^{l-j}a \right\rangle \tilde{A}^{j}a \end{split}$$

Hence the first equation follows by induction. Suppose $0 \le l < k$, then

$$\left\langle a, \tilde{A}^{l}a \right\rangle = -\sum_{j=0}^{l-1} \left\langle b, A^{l-j}a \right\rangle \left\langle a, \tilde{A}^{j}a \right\rangle$$

Thus it follows by induction that $\langle a, \tilde{A}^l a \rangle = 0$. Thus $\langle a, \tilde{A}^k a \rangle = \langle a, A^k a \rangle$. Now, define ω_i by

$$\omega_i := \left\langle a, A^{i+1}a \right\rangle$$

We will also need the following lemma to calculate ω_i in \mathbb{E}^n_{ν} .

Lemma 9.2.8

Suppose A has the form given by Eq. (9.2.3), then

$$A^{l}a = \begin{cases} mb - w & l = 1\\ \left\langle w, \tilde{A}^{l-2}w \right\rangle b - \tilde{A}^{l-1}w & l > 1 \end{cases}$$

$$(9.2.6)$$

and ω_i is given by Eq. (9.1.2).

Using the above lemma we can also apply the definitions of index, sign and degeneracy of CTs in \mathbb{E}^n_{ν} from Definition 9.1.7 to CTs in \mathbb{P}^n_{ν} .

Non-degenerate cases

Now we consider the case where there exists a least $k \in \mathbb{N}$ such that $\langle a, A^k a \rangle \neq 0$. This will be the most important case for our interests. Motivated by special cases and the metric-Jordan canonical form of \tilde{A} discussed earlier, we will try to find b such that $a, \tilde{A}a, \ldots, \tilde{A}^k a$ forms a skew-normal sequence with $\langle a, A^k a \rangle b = \tilde{A}^k a$. The following lemma describes b provided it exists:

Lemma 9.2.9

Suppose there is $k \in \mathbb{N}$ such that $\langle a, A^{l}a \rangle = 0$ for $0 \leq l < k$ and $\langle a, A^{k}a \rangle \neq 0$. Assume there exists a b such that $\langle a, A^{k}a \rangle b = \tilde{A}^{k}a$ and $\langle \tilde{A}^{j}a, \tilde{A}^{k}a \rangle = 0$ for all $1 \leq j \leq k$. Then b must satisfy the following equations for each $l \in \{0, \ldots, k\}$

$$2\langle b, A^{l}a \rangle = \frac{\langle A^{l}a, A^{k}a \rangle}{\langle a, A^{k}a \rangle} - \sum_{j=1}^{l-1} \langle b, A^{l-j}a \rangle \langle b, A^{j}a \rangle$$

$$(9.2.7)$$

PROOF Suppose $0 < l \leq k$. Expanding $\tilde{A}^k a$ using Eq. (9.2.4), we have

$$\left\langle \tilde{A}^{l}a, \tilde{A}^{k}a \right\rangle = \left\langle \tilde{A}^{l}a, A^{k}a \right\rangle - \left\langle b, A^{l}a \right\rangle \left\langle \tilde{A}^{l}a, \tilde{A}^{k-l}a \right\rangle$$
$$\stackrel{(9.2.5)}{=} \left\langle \tilde{A}^{l}a, A^{k}a \right\rangle - \left\langle b, A^{l}a \right\rangle \left\langle a, A^{k}a \right\rangle$$

By imposing the condition $\left\langle \tilde{A}^{l}a, \tilde{A}^{k}a \right\rangle = 0$, the above equation implies that:

$$\left\langle \tilde{A}^{l}a, A^{k}a \right\rangle - \left\langle b, A^{l}a \right\rangle \left\langle a, A^{k}a \right\rangle = 0$$
 (9.2.8)

Now expanding $\tilde{A}^l a$ using Eq. (9.2.4), the above equation becomes

$$\begin{split} \left\langle \tilde{A}^{l}a, A^{k}a \right\rangle &= \left\langle A^{l}a - \sum_{j=0}^{l-1} \left\langle b, A^{l-j}a \right\rangle \tilde{A}^{j}a, A^{k}a \right\rangle \\ &= \left\langle A^{l}a, A^{k}a \right\rangle - \sum_{j=0}^{l-1} \left\langle b, A^{l-j}a \right\rangle \left\langle \tilde{A}^{j}a, A^{k}a \right\rangle \\ &= \left\langle A^{l}a, A^{k}a \right\rangle - \left\langle b, A^{l}a \right\rangle \left\langle a, A^{k}a \right\rangle - \sum_{j=1}^{l-1} \left\langle b, A^{l-j}a \right\rangle \left\langle \tilde{A}^{j}a, A^{k}a \right\rangle \\ &\stackrel{(9.2.8)}{=} \left\langle A^{l}a, A^{k}a \right\rangle - \left\langle b, A^{l}a \right\rangle \left\langle a, A^{k}a \right\rangle - \sum_{j=1}^{l-1} \left\langle b, A^{l-j}a \right\rangle \left\langle b, A^{j}a \right\rangle \left\langle a, A^{k}a \right\rangle \end{split}$$

Equating the above equation with Eq. (9.2.8) and solving for $\langle b, A^l a \rangle$ proves the result.

Now we will use the above lemma and Eq. (9.2.4) to construct a vector b such that \tilde{A} is in canonical form. First define a sequence b_1, \ldots, b_k of scalars recursively as follows:

$$2b_l := \frac{\left\langle A^l a, A^k a \right\rangle}{\left\langle a, A^k a \right\rangle} - \sum_{j=1}^{l-1} b_{l-j} b_j$$

Then define vectors s_0, s_1, \ldots, s_k as follows:

$$s_l := A^l a - \sum_{j=0}^{l-1} b_{l-j} s_j$$

Then define b by $b\langle a, A^k a \rangle := s_k$. The following lemma shows that this choice does work:

Proposition 9.2.10

The vectors s_0, s_1, \ldots, s_k form a skew-normal sequence with $\langle s_0, s_k \rangle = \langle a, A^k a \rangle$. If $\tilde{A}^l a$ are defined as in Eq. (9.2.4) with the above vector b then $\tilde{A}^l a = s_l$.

PROOF The fact that s_0, s_1, \ldots, s_k form a skew-normal sequence follows verbatim from Lemma 9.2.7 and the proceeding arguments by replacing $s_l \to \tilde{A}^l a$ and $b_l \to \langle b, A^l a \rangle$.

Suppose that s_0, s_1, \ldots, s_k form a skew-normal sequence where $\langle s_0, s_k \rangle = \langle a, A^k a \rangle$. By definition of s_l , it follows that each $A^l a$ can be expanded in this basis as:

$$A^{l}a = s_{l} + \sum_{j=0}^{l-1} b_{l-j}s_{j}$$

Thus

$$\langle A^k a, a \rangle \langle b, A^l a \rangle = \langle s_k, A^l a \rangle = b_l \langle A^k a, a \rangle$$

Hence $b_l = \langle b, A^l a \rangle$. Then it follows by definition of s_l and $\tilde{A}^l a$ in Eq. (9.2.4) that $\tilde{A}^l a = s_l$.

Now suppose A is in the canonical form stated above. Let $V = \text{span}\{a, b\}^{\perp}$ where b was chosen as above. Then $H = \text{span}\{a, Aa, \dots, A^ka\}$ is a non-degenerate A-invariant subspace (see Lemma 8.2.4). Hence H^{\perp} is a non-degenerate A-invariant subspace complementary to H. We now mention more precisely what we mean by "the" canonical form:

Definition 9.2.11

Suppose L is a CT in \mathbb{P}^n_{ν} with parameter matrix A as above and index $k' := k - 1 \ge 0$, i.e. L is non-degenerate. The iso-canonical form for L is the metric-Jordan canonical form for $(A|_{H^{\perp}}, g|_{H^{\perp}})$ together with the index k' and constant $\langle a, A^{k'+1}a \rangle \in \mathbb{R} \setminus \{0\}$.

We will prove later on that this canonical form is uniquely determined by L. But for now we will examine this canonical form further. Let $\tilde{A} := A|_{H^{\perp}}$, then we can write:

$$A = \tilde{A} + \omega_0 b \odot b - 2w \odot b$$

where $w = \omega_0 b - Aa$.

If $\omega_0 \neq 0$ then it follows that w = 0 and it follows by Proposition 9.2.6 that the induced CT on V is

$$\tilde{A} + \omega_0 r \odot r$$

Thus after dividing by ω_0 we get the central CT from Theorem 9.1.8. If $\omega_0 = 0$, one can check that $w, \tilde{A}w, \ldots, \tilde{A}^{k-2}w \in V$ form a skew-normal sequence with $\langle w, \tilde{A}^{k-2}w \rangle = \omega_{k-1}$. It follows by Proposition 9.2.6 that the induced CT on V is

$$\tilde{A} + 2w \odot r$$

This CT is a constant multiple of a (null) axial CT with the same index and sign from Theorem 9.1.8 (after an appropriate choice of basis).

Transformation to Canonical form: We now denote by b the vector b obtained above which puts A into a canonical form. The vector $b \in \mathbb{P}^n_{\nu}$ is fixed by an isometry with \mathbb{E}^n_{ν} (see Eq. (9.2.1)), furthermore we let $V = \operatorname{span}\{a, b\}^{\perp}$. We can assume A has the form given by Eq. (9.2.3). The last problem is to choose $T \in I(\mathbb{P}^n_{\nu})$ such that $T\tilde{b} = b$. We can obtain a unique transformation if we require T to induce a translation in V. Indeed, by Eq. (9.2.2) the most general transformation of this type is

$$T = I - a \otimes (\frac{1}{2}v^2 a^{\flat} + v^{\flat}) + v \otimes a^{\flat}$$

where $v \in V$ is arbitrary. The unique transformation with the above form satisfying $T\tilde{b} = b$ is obtained by taking

$$v = b - \tilde{b} + a\left\langle \tilde{b}, b \right\rangle$$

We now proceed to calculate v. First we can write

$$\tilde{b} = \frac{1}{\omega_{k-1}} \sum_{i=0}^{k} c_i A^i a_i$$

Since $\langle b, A^l a \rangle = 0$ for any l > 0, we see that

$$\tilde{b} - a\left\langle \tilde{b}, b \right\rangle = \frac{1}{\omega_{k-1}} \sum_{i=1}^{k} c_i A^i a$$

Since for 0 < l < k, $\langle a, A^l a \rangle = 0$ it follows by Eq. (9.2.6) that $A^l a = -\tilde{A}^{l-1} w$. Thus

$$v = -\frac{1}{\omega_{k-1}} \sum_{i=1}^{k} c_i A^i a + b$$
$$= \frac{1}{\omega_{k-1}} \sum_{i=1}^{k} c_i \tilde{A}^{i-1} w$$

where the last equation follows from the fact that $c_k = 1$. We have calculated the first four coefficients (which are sufficient for Euclidean and Minkowski space):

$$c_{k} = 1$$

$$c_{k-1} = -\frac{1}{2} \frac{\omega_{k}}{\omega_{k-1}}$$

$$c_{k-2} = \frac{1}{16} \frac{(-8\omega_{k-1}\omega_{k+1} + 6\omega_{k}^{2})}{\omega_{k-1}^{2}}$$

$$c_{k-3} = \frac{1}{16} \frac{(-8\omega_{k-1}^{2}\omega_{k+2} + 12\omega_{k-1}\omega_{k}\omega_{k+1} - 5\omega_{k}^{3})}{\omega_{k-1}^{3}}$$

In particular when k = 1 and 2 respectively we have the following:

$$v = \frac{w}{\omega_0}$$
$$v = \frac{1}{\omega_1} (\tilde{A}w - \frac{1}{2}\frac{\omega_2}{\omega_1}w)$$

Finally, we note that by equivariance of the map ψ (see remark after Proposition 9.2.3), one only needs to apply the isometry $T: V \to V$ given by $r \mapsto r + v$ to send the induced CT in V into canonical form. Hence in practice one does not need to work in \mathbb{P}^n_{ν} .

Degenerate cases

We now consider the case where $\langle a, A^l a \rangle = 0$ for every $l \in \mathbb{N}$. First note that the dimension of the subspace spanned by a, Aa, \ldots must be at most n-1 by non-degeneracy of the scalar product. So there exists a least $l \leq n-1$ such that $\{a, Aa, \ldots, A^l a\} \subseteq a^{\perp}$ is a linearly independent set but $A^{l+1}a \in \text{span}\{a, Aa, \ldots, A^l a\}$. Thus it follows that $A^m a \in \text{span}\{a, Aa, \ldots, A^l a\}$ for all m > l. Also note by Lemma 9.2.7 it follows that these properties are invariant under the transformation $A \to A_b$.

Case 1 l = 0

In this case *a* is an eigenvector of *A*. After transforming *A* to A_b (if necessary), we can assume that Aa = 0. Also Ab = 0, then since $\langle a, b \rangle = 1$ it follows that span $\{a, b\}$ is a non-degenerate *A*-invariant subspace. Hence after identifying $\mathbb{E}_{\nu}^n \simeq \text{span}\{a, b\}^{\perp}$, it follows by Proposition 9.2.6 that *A* restricts to a Cartesian CT on \mathbb{E}_{ν}^n .

Case 2 $l \ge 1$

Fix $b \in \mathbb{P}^n_{\nu}$, let $V = \operatorname{span}\{a, b\}^{\perp}$ and assume Ab = 0. Then we can write:

$$A = \tilde{A} + 2w \odot b$$

Now note that for any $j \in \mathbb{N}$, $\langle a, A^j a \rangle = 0$. Suppose inductively that for all $1 \leq j \leq i$ that $A^j a \in V$ then

$$AA^ia = \tilde{A}A^ia \in V$$

since $\langle A^i a, w \rangle = \langle A^i a, Aa \rangle = 0$ and $\langle A^i a, b \rangle = 0$. Hence by induction for any $j \in \mathbb{N}$, $\langle b, A^j a \rangle = 0$. Thus $Aa, \ldots, A^l a, A^{l+1} a \in V$.

In particular, when l = 1 we see that w is a lightlike eigenvector of A. Then by Proposition 9.2.6, A induces the following CT in \mathbb{E}^n_{ν}

$$L = \tilde{A} - 2w \odot r$$

Observe that w is a lightlike eigenvector of L with non-constant eigenfunction. Thus L is never an OC-tensor because lightlike eigenvectors of OC-tensors must have constant eigenfunctions.

If l > 1, we see that $Aa, A^2a \in V$ are linearly independent orthogonal lightlike vectors. Thus this case can't occur in Euclidean or Minkowski case, so we ignore it.

9.2.4 Uniqueness of Canonical Forms

In this section we will show that the canonical forms obtained in the previous section are uniquely determined by a given CT in \mathbb{P}^n_{ν} . As a consequence of this we will show that the different canonical forms divide the CTs into isometrically inequivalent classes. We will be working with the case when the CT is non-degenerate as the other cases are either straightforward or uninteresting.

Suppose L and M are CTs in \mathbb{P}^n_{ν} with parameter matrices A and B respectively. We observed at the end of Section 9.2.2 that L = M iff for one (hence all) $b \in \mathbb{P}^n_{\nu}$:

$$A_b = B_b$$

Thus it follows that $L = T_*M$ for some $T \in I(\mathbb{P}^n_{\nu})$ iff for one (hence all) $b \in \mathbb{P}^n_{\nu}$:

$$A_b = (T_*B)_b$$

Lemma 9.2.12

Suppose A_2 is a parameter matrix, and $A_1 = (A_2)_b$ for some $b \in \mathbb{P}^n_{\nu}$. Assume each A_i have the same index and admit a vector b_i which transforms it to canonical form according to Proposition 9.2.10. Then $b_1 = b_2$.

PROOF Let $A_0 = (A_2)_{b_2}$, then $A_1 = (A_0)_b$. Since A_0 is in canonical form,

 $a, A_0 a, \dots, A_0^k a$ forms an adapted cycle of generalized eigenvectors for A_0 with eigenvalue 0. In this case $\langle a, A_0^k a \rangle \in \mathbb{R} \setminus \{0\}$.

Let b_1 be the vector admitted by A_1 and let $A_3 := (A_1)_{b_1} = (A_0)_{b_1}$. Now by Proposition 9.2.10 and Lemma 9.2.7, b_1 satisfies:

$$\langle a, A_1^k a \rangle b_1 = A_3^k a = A_0^k a - \sum_{j=0}^{k-1} \langle b_1, A_0^{k-j} a \rangle A_3^j a$$
 (9.2.9)

Since A_3 is in canonical form, it follows for each $l \in \{1, \dots, k\}, \langle b_1, A_0^l a \rangle$ satisfies Eq. (9.2.7). Then since A_0 is in canonical form, we have $\langle b_1, A_0^l a \rangle = 0$ for $l \in \{1, \dots, k\}$. Thus Eq. (9.2.9) shows that

$$\left\langle a, A_1^k a \right\rangle b_1 = A_0^k a = \left\langle a, A_1^k a \right\rangle b_2$$

Hence $b_1 = b_2$.

In the following theorem we will show that the iso-canonical form defined in Definition 9.2.11 for non-degenerate CTs is uniquely determined by the CT.

Theorem 9.2.13 (Isometric Equivalence of CTs in \mathbb{E}^n_{μ})

Suppose L and M are CTs in \mathbb{P}^n_{ν} such that M has an index $k \geq 0$. Then $L = T_*M$ for some $T \in I(\mathbb{P}^n_{\nu})$ iff L and M have the same iso-canonical form.

PROOF Assume that $L = T_*M$ for some $T \in I(\mathbb{P}^n_{\nu})$. Then for some $b \in \mathbb{P}^n_{\nu}$:

$$A_b = (T_*B)_b$$

By the above equation and Lemma 9.2.7 it follows that the index of L is also k. Let b_2 be the vector which puts B in canonical form given by Proposition 9.2.10. Then Tb_2 sends T_*B to canonical form. By Lemma 9.2.12, Tb_2 is the vector obtained from Proposition 9.2.10 which puts A in canonical form. Let $\tilde{b} := Tb_2$ then

$$A_{\tilde{b}} = (T_*B)_{\tilde{b}} = T_*(B_{b_2})$$

Hence B_{b_2} is isometric to $A_{\tilde{b}}$. Then it follows from the uniqueness of the metric-Jordan canonical form (see Theorem 8.2.5) that $A_{\tilde{b}}$ and B_{b_2} have the same iso-canonical form.

Conversely suppose L and M have the same iso-canonical form. Then A (resp. B) each admit a vector $b_1 \in \mathbb{P}^n_{\nu}$ (resp. $b_2 \in \mathbb{P}^n_{\nu}$) such that A_{b_1} and B_{b_2} have the same iso-canonical form. Then one can easily construct $T \in I(\mathbb{P}^n_{\nu})$ which transforms a metric-Jordan canonical basis of B_{b_2} into A_{b_1} , so that $A_{b_1} = T_*(B_{b_2})$. Thus

$$\Rightarrow T(B_{b_2})^k a = (A_{b_1})^k a$$
$$\Rightarrow Tb_2 = b_1$$

Note that in the last equation we have used the fact that $\langle a, B^k a \rangle = \langle a, A^k a \rangle$. Then

$$A_{b_1} = T_*(B_{b_2}) = (T_*B)_{b_1}$$

Thus $L = T_*M$, which proves the converse.

Geo-Canonical forms We now give a geo-canonical form for non-degenerate CTs in \mathbb{P}^n_{ν} . Suppose L is such a CT with index k and parameter matrix A in iso-canonical form. Then for $c \in \mathbb{R}$, cL has parameter matrix cA and

$$\langle a, (cA)^{k+1} a \rangle = c^{k+1} \langle a, A^{k+1} a \rangle$$

Hence after an appropriate transformation $L \to cL$, we can assume

$$\langle a, A^{k+1}a \rangle = \begin{cases} 1 & 1 \text{ if } k \text{ is even} \\ \pm 1 & \text{if } k \text{ is odd} \end{cases}$$

Note that when k is odd, c is only determined up to sign. Hence there are two possible geo-canonical forms in this case. Now, if L is an axial CT, we can fix $d \in \mathbb{R}$ by requiring that $(A + dI)^k a \in \text{span}\{a, b\}$. This condition is satisfied in the iso-canonical form. If L is central, we choose d such that the real part of the smallest eigenvalue (see Definition E.0.9) of $A|_{H^{\perp}}$ is zero.

9.3 Canonical forms for Concircular tensors in Spherical submanifolds of pseudo-Euclidean space

9.3.1 Obtaining concircular tensors in umbilical submanifolds by restriction

Let \tilde{M} be a pseudo-Riemannian submanifold of M with Levi-Civita connections $\tilde{\nabla}$ and ∇ respectively. We say \tilde{M} is an umbilical submanifold (see Section 3.1 for more details) if there exists $H \in \mathfrak{X}(\tilde{M})^{\perp}$ (i.e. H is orthogonal to $T\tilde{M}$) called the *mean curvature normal* of \tilde{M} such that

$$\nabla_x y = \tilde{\nabla}_x y + \langle x, y \rangle H$$

for all $x, y \in \mathfrak{X}(\tilde{M})$. By generalizing an observation made in [Cra03] one can deduce the following:

Proposition 9.3.1 (Restriction of CTs to umbilical submanifolds [Cra03]) Suppose \tilde{M} is an umbilical submanifold of M with mean curvature normal H and L is a concircular r-tensor on M with conformal factor C in covariant form. Then the pullback of L to \tilde{M} is a concircular r-tensor with conformal factor equal to the pullback of C + rL(H), where in components, $L(H)_{i_1,\ldots,i_{r-1}} = L_{i_1,\ldots,i_{r-1}j}H^j$.

Since spherical submanifolds are umbilical submanifolds and $\mathbb{E}_{\nu}^{n}(\kappa)$ is a spherical submanifold (see Section 3.1), the above proposition allows us to obtain CTs on $\mathbb{E}_{\nu}^{n}(\kappa)$. We will do this in the following section.

9.3.2 Concircular tensors in Spherical submanifolds of pseudo-Euclidean space

In this section we study the CTs in $\mathbb{E}_{\nu}^{n}(\kappa)$ via the canonical embedding in \mathbb{E}_{ν}^{n} . Let r denote the dilatational vector field, we work on the subset of \mathbb{E}_{ν}^{n} for which $r^{2} \neq 0$. Let $E := r^{\perp}$ and let L be a CT on M. To obtain the CT on $\mathbb{E}_{\nu}^{n}(\frac{1}{r^{2}})$ (which is an integral manifold of E), we first let $R := I - \frac{r^{\flat} \otimes r}{r^{2}}$ where I is the identity endomorphism then $L_{E} := L|_{E}$ is given as follows:

$$(L_E)^{ij} = R^i_{\ l} L^{lk} R^j_{\ k}$$

Now we will calculate the general CT on $\mathbb{E}^n_{\nu}(\kappa)$.

Proposition 9.3.2 (Concircular tensors in $\mathbb{E}^n_{\nu}(\kappa)$)

 \tilde{L} is a concircular tensor in $\mathbb{E}^n_{\nu}(\frac{1}{r^2})$ where n > 2 iff there exists $A \in C^2_0(\mathbb{E}^n_{\nu})$ such that \tilde{L} has the following form embedded in \mathbb{E}^n_{ν} :

$$L = A_E = A + \frac{\langle r, Ar \rangle}{r^4} r \odot r - \frac{2}{r^2} (Ar \odot r)$$

A is uniquely determined by \tilde{L} . Furthermore \tilde{L} is covariantly constant iff its a constant multiple of the metric on $\mathbb{E}^n_{\nu}(\frac{1}{r^2})$, i.e. A = c G for some $c \in \mathbb{R}$ where G is the metric of \mathbb{E}^n_{ν} .

PROOF Fix $\tilde{L} \in S^2(\mathbb{E}^n_{\nu}(\frac{1}{r^2}))$. Choose an orthonormal basis a_1, \ldots, a_n for \mathbb{E}^n_{ν} . Let $R^* = I - \frac{r \otimes r^{\flat}}{r^2}$, then it follows from Proposition 9.3.1 that the vectors

$$R^*a_i = a_i - \frac{\langle r, a_i \rangle}{r^2}r \quad i = 1, \dots, n$$

are CVs on $\mathbb{E}_{\nu}^{n}(\frac{1}{r^{2}})$. Furthermore one can check that these vectors are linearly independent. Thus by Corollary 9.1.6 every CT can be written uniquely as a linear combination of symmetric products of the above CVs. Thus it follows that we can choose a unique $A \in C_{0}^{2}(\mathbb{E}_{\nu}^{n})$ such that $\tilde{L} = A_{E}$ on $\mathbb{E}_{\nu}^{n}(\frac{1}{r^{2}})$. In \mathbb{E}_{ν}^{n} , A_{E} is given as follows:

$$A_E = R^* A R$$

= $A + A(r^{\flat}, r^{\flat}) \frac{r \odot r}{r^4} - \frac{2}{r^2} A(r^{\flat}) \odot r$
= $A + \langle r, Ar \rangle \frac{r \odot r}{r^4} - \frac{2}{r^2} A r \odot r$

Conversely by Corollary 9.1.6 it follows that for any $A \in C_0^2(\mathbb{E}_{\nu}^n)$, A_E corresponds to CT on $\mathbb{E}_{\nu}^n(\frac{1}{r^2})$.

The last statement follows from Proposition 9.3.1.

Remark 9.3.3

The general CT in $\mathbb{E}^n_{\nu}(\kappa)$ has been obtained in [TCS05, Section 3] with respect to certain canonical coordinates for these spaces. They use a different method for obtaining these tensors based on the theory developed in their article.

For the remainder of this chapter we will always work with CTs in $\mathbb{E}_{\nu}^{n}(\kappa)$ via the tensor L defined in \mathbb{E}_{ν}^{n} in the above proposition.

Definition 9.3.4

Suppose L is a CT in $\mathbb{E}^n_{\nu}(\kappa)$ with parameter matrix $A \in S^2(\mathbb{E}^n_{\nu})$ as above. The *iso-canonical* form for L is the metric-Jordan canonical form for (A, g).

Except for hyperbolic space H_0^{n-1} and the space anti-isomorphic to it S_{n-1}^{n-1} , uniqueness of the iso-canonical form follows from the uniqueness of the metric-Jordan canonical form and the fact that $I(\mathbb{E}_{\nu}^{n}(\kappa)) = O(\mathbb{E}_{\nu}^{n})$ [O'N83]. For H_0^{n-1} , $I(H_0^{n-1})$ is the subset of $O(\mathbb{E}_1^{n})$ that preserves time orientation [O'N83]. In this case, minor modifications of the proof of the uniqueness of the metric-Jordan canonical form will show that it holds true with $I(H_0^{n-1})$ in place of of $O(\mathbb{E}_1^{n})$. A similar argument goes for S_{n-1}^{n-1} . Hence we have proven the following:

Theorem 9.3.5 (Isometric Equivalence of CTs in $\mathbb{E}^n_{\nu}(\kappa)$)

Suppose L and M are CTs in $\mathbb{E}^n_{\nu}(\kappa)$. Then $L = T_*M$ for some $T \in I(\mathbb{E}^n_{\nu}(\kappa))$ iff L and M have the same iso-canonical form.

Geo-Canonical forms By definition, the restriction of G to $\mathbb{E}_{\nu}^{n}(\kappa)$ is the metric on $\mathbb{E}_{\nu}^{n}(\kappa)$. Hence we see that if $a \in \mathbb{R} \setminus \{0\}, b \in \mathbb{R}$ and $A \in C_{0}^{2}(\mathbb{E}_{\nu}^{n})$, then A and aA + bG induce geometrically equivalent CTs on $\mathbb{E}_{\nu}^{n}(\kappa)$ (see Proposition 9.1.2). We now show how to obtain the geo-canonical forms. Suppose $\lambda_{1}, \ldots, \lambda_{k} \in \mathbb{C}$ are the distinct eigenvalues of A. Let $|\cdot|$ denote the modulus of a complex number, then define:

$$|a| := \min_{i,j} |\lambda_i - \lambda_j| > 0$$

Note that this quantity is invariant under geometric equivalence. By making the transformation $\lambda_i \to \frac{\lambda_i}{|a|}$, we can assume |a| = 1. Furthermore we choose $b \in \mathbb{R}$ such that the real part of the smallest eigenvalue (see Definition E.0.9) of A is zero. Since its not possible to specify the sign of a, we conclude that there are (in general) two geo-canonical forms for CTs in $\mathbb{E}^n_{\nu}(\kappa)$. Although in practice one can often use more information from the metric-Jordan canonical form of A to obtain a single geo-canonical form, as the following example shows:

Example 9.3.6 (Separable coordinates in hyperbolic space)

Consider $H^{n-1} = \mathbb{E}_1^n(-1)$ with the standard metric:

$$g = \operatorname{diag}(-1, 1, \dots, 1)$$

For $\lambda_1 < \cdots < \lambda_n \in \mathbb{R}$ define two linear operators A_1 and A_2 as follows:

$$A_1 = \operatorname{diag}(\lambda_1, \dots, \lambda_n)$$
$$A_2 = \operatorname{diag}(-\lambda_1, \dots, -\lambda_n)$$

These two operators are isometrically inequivalent since they have different metric-Jordan canonical forms. The timelike eigenvalue of the first is the smallest, while that of the second is the largest. Although $-A_2 = A_1$ and hence the CT on H^{n-1} induced by these operators are geometrically equivalent. So, in H^{n-1} we can work with inequivalent CTs (under change of sign) by working with those whose parameter matrix has a timelike eigenvalue which is less than or equal to $\lfloor \frac{n}{2} \rfloor$ spacelike eigenvalues.

Thus the set of eigenvalues $\lambda_1 < \cdots < \lambda_n \in \mathbb{R}$ induce $\lceil \frac{n}{2} \rceil$ inequivalent separable coordinates in H^{n-1} ; in contrast with the *n* inequivalent separable coordinates in \mathbb{E}_1^n induced by central CTs.

9.4 Properties of Concircular tensors in Spaces of Constant Curvature

In this section we will assume that each CT in \mathbb{E}_{ν}^{n} or $\mathbb{E}_{\nu}^{n}(\kappa)$ is in a canonical form listed in Section 9.1.3. Furthermore we will assume that the Cartesian coordinates are chosen such that the parameter matrix A_{c} is in the complex metric-Jordan canonical form stated in Theorem 8.2.2 (see Appendix C for details). We now describe how to transform to real Cartesian coordinates such that A_{c} obtains the real metric-Jordan canonical form given by Theorem 8.2.5. Suppose $\lambda \in \mathbb{C} \setminus \mathbb{R}$ and (A, g) is given as follows:

$$A = J_k(\lambda) \oplus J_k(\overline{\lambda}) \qquad \qquad g = S_k \oplus S_k$$

in coordinates $(x^1, \ldots, x^k, \overline{x}^1, \ldots, \overline{x}^k)$. Define real coordinates $(s^1, t^1, \ldots, s^k, t^k)$ implicitly as follows:

$$x^j = \frac{1}{\sqrt{2}}(s^j - it^j)$$

$$\overline{x}^j = \frac{1}{\sqrt{2}}(s^j + it^j)$$

These coordinates were chosen so that the pair (A, g) are in the real metric-Jordan canonical form in the real coordinates $(s^1, t^1, \ldots, s^k, t^k)$ after applying the appropriate tensor transformation law.

In Cartesian coordinates (x^i) , we will use the convention that $x_i := g_{ij}x^j$; this is the only case where the Einstein summation convention is used in this section.

We now list some generic facts about tensors and C-tensors that will be used. We first recall some facts about $\binom{1}{1}$ -tensors which were first stated in Section 1.4.2. In the following proposition, we use the notation C^p to denote the differentiability class of a geometric object, where $p \in \mathbb{N} \cup \{\infty, \omega\}$, and C^{ω} denotes the analytic class.

Proposition 9.4.1

Suppose T is a $\binom{1}{1}$ -tensor of class C^p and fix $q \in M$.

Let λ_0 be a simple eigenvalue of T_q . Then there exists a neighborhood of q in which T has a simple eigenfunction λ with a corresponding eigenvector field which are both of class C^p , and $\lambda(q) = \lambda_0$.

If T_q has simple eigenvalues, then there exists a neighborhood of q in which T has simple eigenfunctions of class C^p , and T admits a basis of eigenvector fields of class C^p .

The above proposition shows that Benenti tensors necessarily locally admit a smooth basis of eigenvectors with corresponding smooth eigenfunctions. The following proposition gives necessary and sufficient conditions to determine when a given Benenti tensor is an IC-tensor.

Proposition 9.4.2

Suppose L is a Benenti tensor in a neighbourhood U of a point p. If the eigenfunctions of L are not constant in U, then the eigenfunctions are functionally independent, i.e. L is an IC-tensor in a dense open subset of U. \Box

PROOF This is a direct consequence of the torsionless property of these tensors. Since in this case there are coordinates (q^i) such that L is diagonal and each eigenfunction $u^i(q^i)$. Then

$$du^{1} \wedge \dots \wedge du^{n} = \frac{du^{1}}{dq^{1}} dq^{1} \wedge \dots \wedge \frac{du^{n}}{dq^{n}} dq^{n}$$
$$= (\prod_{i=1}^{n} \frac{du^{i}}{dq^{i}}) dq^{1} \wedge \dots \wedge dq^{n}$$

Hence if $du^i \neq 0$ for each *i*, the eigenfunctions are functionally independent. If the u^i are analytic functions of q^i , then by assumption it follows that *L* is an IC-tensor in a dense open subset of *U*.

Proposition 9.4.3

Suppose L is an OCT and $p(z) = \det(zI - L)$ is its characteristic polynomial. Suppose u^i is a simple eigenfunction of L and $du^i \neq 0$, then the corresponding eigenform is given by:

$$du^i = -\frac{(dp)|_{z=u^i}}{p'(u^i)}$$

where dp is the exterior derivative of p with respect to the ambient coordinates and p' is the partial derivative of p with respect to z. Furthermore if L is an IC-tensor, then the metric in the coordinates induced by the eigenfunctions of L is:

$$g^{ij} = \begin{cases} (p'(u^i))^{-2} \langle (dp)|_{z=u^i}, (dp)|_{z=u^i} \rangle & \text{if } i = j \\ 0 & else \end{cases}$$

PROOF Since $p(z) = (z - u^i)f(z)$ for a smooth function f(z). By taking the exterior derivative, we get:

$$\mathrm{d}p = -f\mathrm{d}u^i + (z - u^i)\mathrm{d}f$$

Then by L'Hopital's rule, we find that:

$$(\mathrm{d}p)|_{z=u^i} = -p'(u^i)\mathrm{d}u^i$$

which can be solved for du^i since u^i is a simple eigenfunction. The fact that $Ldu^i = u^i du^i$ follows from the fact that L is torsionless.

To calculate the metric, first it follows that $g^{ij} = 0$ when $i \neq j$ since L is self-adjoint and has simple eigenfunctions. For the remaining component:

$$g^{ii} = \left\langle \mathrm{d}u^i, \mathrm{d}u^i \right\rangle$$
$$= (p'(u^i))^{-2} \left\langle (\mathrm{d}p)|_{z=u^i}, (\mathrm{d}p)|_{z=u^i} \right\rangle$$

Remark 9.4.4

The assumption that L is a concircular tensor can be replaced with any symmetric contravariant tensor whose associated endomorphism is torsionless.

The following lemma on determinants will be used several times.

Lemma 9.4.5

Suppose $T = A + v \otimes x$ where $A = [a_1, ..., a_n]$ is an $n \times n$ matrix, $v \in \mathbb{F}^n$ and $x \in \mathbb{F}^n$ (where \mathbb{F} is \mathbb{R} or \mathbb{C}). Then det T is given as follows:

$$\det T = \bigwedge_{i=1}^{n} (a_i + x_i v) = \bigwedge_{i=1}^{n} a_i + \sum_{i=1}^{n} a_1 \wedge \dots \wedge x_i v \wedge \dots \wedge a_n \qquad \Box$$

PROOF The formula clearly holds for n = 1, so inductively suppose the formula holds for k = n - 1, then:

$$\bigwedge_{i=1}^{n} (a_i + x_i v) = \bigwedge_{i=1}^{n-1} (a_i + x_i v) \wedge (a_n + x_n v)$$

$$= (\bigwedge_{i=1}^{n-1} a_i + \sum_{i=1}^{n-1} a_1 \wedge \dots \wedge x_i v \wedge \dots \wedge a_{n-1}) \wedge (a_n + x_n v)$$

$$= \bigwedge_{i=1}^{n} a_i + \sum_{i=1}^{n-1} a_1 \wedge \dots \wedge x_i v \wedge \dots \wedge a_n + \bigwedge_{i=1}^{n-1} a_i \wedge x_n v$$

$$= \bigwedge_{i=1}^{n} a_i + \sum_{i=1}^{n} a_1 \wedge \dots \wedge x_i v \wedge \dots \wedge a_n$$

In the following sections, we will obtain the following information. First we will calculate the characteristic polynomial for CTs in spaces of constant curvature. Using this, for ICTs we will calculate the transformation from the canonical coordinates they induce to Cartesian coordinates, and we will calculate the metric in canonical coordinates.

9.4.1 Central Concircular tensors

The following general lemma will be used to calculate the characteristic polynomial of central CTs.

Lemma 9.4.6 (Determinant of Central Concircular tensors)

Suppose $L = A + r \otimes r^{\flat}$ is a central Concircular tensor, where $r^{i} = x^{i}$. Then,

$$\det L = \bigwedge_{i=1}^{n} a_i + \sum_{i=1}^{n} a_1 \wedge \dots \wedge x_i r \wedge \dots \wedge a_n$$
(9.4.2)

Suppose U is a non-degenerate A-invariant subspace (hence U^{\perp} is A-invariant), let $L_u = L|_U$ and $L_{u^{\perp}} = L|_{U^{\perp}}$, then:

$$\det L = \det L_u \det A_{u^{\perp}} + \det A_u (\det L_{u^{\perp}} - \det A_{u^{\perp}})$$

$$(9.4.3)$$

PROOF The first statement follows from Lemma 9.4.5 by taking $A \to A$, $r \to v$ and $r^{\flat} \to x$.

Now for the second part, let $k = \dim U$, then in a basis adapted to the decomposition $V = U \oplus U^{\perp}$, we have:

$$A = \begin{pmatrix} B & 0\\ 0 & C \end{pmatrix}$$

where B is a $k \times k$ matrix and C is a $(n-k) \times (n-k)$ matrix. Furthermore $r = r_b + r_c$ where $r_b \in U$ and $r_c \in U^{\perp}$. The main fact we use is that for any square matrix, T, of the form:

$$\begin{pmatrix} A & B \\ 0 & C \end{pmatrix}$$

we have $\det T = \det A \det C$. Thus:

$$\det L = \bigwedge_{i=1}^{n} a_{i} + \sum_{i=1}^{n} a_{1} \wedge \dots \wedge x_{i}r \wedge \dots \wedge a_{n}$$

$$= \bigwedge_{i=1}^{k} b_{i} \wedge \bigwedge_{i=1}^{n-k} c_{i} + \left(\sum_{i=1}^{k} b_{1} \wedge \dots \wedge x_{i}r_{b} \wedge \dots \wedge b_{k}\right) \wedge \bigwedge_{i=1}^{n-k} c_{i}$$

$$+ \bigwedge_{i=1}^{k} b_{i} \wedge \left(\sum_{i=1}^{n-k} c_{1} \wedge \dots \wedge x_{i}r_{c} \wedge \dots \wedge c_{n-k}\right)$$

$$= \left(\bigwedge_{i=1}^{k} b_{i} + \sum_{i=1}^{k} b_{1} \wedge \dots \wedge x_{i}r_{b} \wedge \dots \wedge b_{k}\right) \wedge \bigwedge_{i=1}^{n-k} c_{i}$$

$$+ \bigwedge_{i=1}^{k} b_{i} \wedge \left(\sum_{i=1}^{n-k} c_{1} \wedge \dots \wedge x_{i}r_{c} \wedge \dots \wedge c_{n-k}\right)$$

$$= \det L_{u} \det A_{u^{\perp}} + \det A_{u} (\det L_{u^{\perp}} - \det A_{u^{\perp}})$$

Now consider the simplest case where $A = \text{diag}(\lambda_1, ..., \lambda_n)$. Then Eq. (9.4.2) can be used to get the characteristic polynomial of L, which is:

$$p(z) = \det(zI - L) = \prod_{i=1}^{n} (z - \lambda_i) - \sum_{i=1}^{n} x_i x^i \prod_{j \neq i} (z - \lambda_j)$$
(9.4.4)

Now suppose L is an ICT with eigenfunctions (u^1, \ldots, u^n) , then from the above equation we have:

$$\prod_{j=1}^{n} (u^j - \lambda_i) = p(\lambda_i) = -\varepsilon_i (x^i)^2 \prod_{j \neq i} (\lambda_i - \lambda_j)$$

One can check that by assumption we must have $\lambda_i \neq \lambda_j$ if $i \neq j$. This will eventually be proven later. Thus we deduce the transformation from the coordinates (u^1, \ldots, u^n) to Cartesian coordinates to be:

$$(x^{i})^{2} = \varepsilon_{i} \frac{\prod_{j=1}^{n} (u^{j} - \lambda_{i})}{\prod_{j \neq i} (\lambda_{j} - \lambda_{i})}$$

$$(9.4.5)$$

The derivation of the transformation to Cartesian coordinates follows that of [Cra03, section 5]. We will use this method for all other types of CTs as well. Now, it will be useful to write the characteristic polynomial in standard form:

Proposition 9.4.7

Suppose L is a central CT with parameter matrix $A = \text{diag}(\lambda_1, ..., \lambda_n)$ and arbitrary orthogonal metric. Write the characteristic polynomial of A as:

$$B(z) = \det(zI - A) = \sum_{l=0}^{n} a_l z^{l}$$

Then the characteristic polynomial of L is:

$$p(z) = \det(zI - L) = \sum_{l=0}^{n} (a_l - \sum_{j=0}^{n-1-l} a_{j+1+l} \langle r, A^j r \rangle) z^l$$
(9.4.6)

PROOF We will prove this formula by expanding Eq. (9.4.4). For the following calculations, if a(z) is a polynomial in z, then $[z^l]a(z)$ is the coefficient of z^l in this polynomial. First observe that

$$[z^{l}]\prod_{j}(z-\lambda_{j}) = [z^{l}][z\prod_{j\neq i}(z-\lambda_{j}) - \lambda_{i}\prod_{j\neq i}(z-\lambda_{j})]$$
$$= [z^{l-1}]\prod_{j\neq i}(z-\lambda_{j}) - \lambda_{i}[z^{l}]\prod_{j\neq i}(z-\lambda_{j})$$
$$\Rightarrow [z^{l-1}]\prod_{j\neq i}(z-\lambda_{j}) = [z^{l}]\prod_{j}(z-\lambda_{j}) + \lambda_{i}[z^{l}]\prod_{j\neq i}(z-\lambda_{j})$$

We also have

$$[z^{n-1}]\prod_{j\neq i}(z-\lambda_j)=1$$

We will prove inductively that

$$[z^l]\prod_{j\neq i}(z-\lambda_j) = \sum_{j=0}^{n-1-l}\lambda_i^j a_{j+1+l}$$

Then by inductive hypothesis, we have

$$[z^{l-1}] \prod_{j \neq i} (z - \lambda_j) = a_l + \lambda_i \sum_{j=0}^{n-1-l} \lambda_i^j a_{j+1+l}$$
$$= a_l + \sum_{j=1}^{n-l} \lambda_i^j a_{j+l}$$
$$= \sum_{j=0}^{n-l} \lambda_i^j a_{j+l}$$

Then

$$[z^{l}] \sum_{i=1}^{n} x_{i} x^{i} \prod_{j \neq i} (z - \lambda_{j}) = \sum_{i=1}^{n} g_{ii} (x^{i})^{2} [z^{l}] \prod_{j \neq i} (z - \lambda_{j})$$
$$= \sum_{i=1}^{n} g_{ii} (x^{i})^{2} \sum_{j=0}^{n-1-l} \lambda_{i}^{j} a_{j+1+l}$$
$$= \sum_{j=0}^{n-1-l} a_{j+1+l} \sum_{i=1}^{n} g_{ii} (x^{i})^{2} \lambda_{i}^{j}$$
$$= \sum_{j=0}^{n-1-l} a_{j+1+l} \langle r, A^{j} r \rangle$$

Which together with Eq. (9.4.4) proves the proposition.

In the following theorem we collect a useful limiting procedure for dealing with Jordan blocks. It has been proven by Kalnins, Miller, and Reid in [KMR84] for general dimensions. We have independently verified it only for dimensions less than three. The details of this verification are only partially included in the following proof, which can be omitted without loss of continuity.

Theorem 9.4.8 ([KMR84])

Let $A_0 := J_n^T(\lambda_1)$ and $g_0 := \varepsilon S_n$. For $n \leq 3$, there exists a sequence of diagonal matrices $A := \operatorname{diag}(\lambda_1, \ldots, \lambda_n), g := \operatorname{diag}(a_1, \ldots, a_n)$ and transformation matrices Λ such that

$$\Lambda^{-1}A\Lambda \to A_0$$
 $\Lambda^T g\Lambda \to g_0$

PROOF First consider the following definitions:

$$\begin{split} \Lambda^{i}{}_{j} &:= \epsilon^{j-1}_{i+1-j} = \prod_{l=2}^{j} (\epsilon^{1}_{i-1} - \epsilon^{1}_{l-2}) & \epsilon^{k}_{l} &:= \begin{cases} 0 & \text{if } l \leq 0 \\ 1 & \text{if } k \leq 0 \end{cases} \\ a_{i} &:= \frac{\varepsilon}{\prod_{k \neq i} (\epsilon^{1}_{i-1} - \epsilon^{1}_{k-1})} \end{split}$$

Note that ϵ_l^k is of order k if k, l > 0. Finally let $\lambda_i := \lambda_1 + \epsilon_{i-1}^1$. Then the conclusion follows by direct calculation if for each $i = 2, ..., n, \epsilon_i^1 \to 0$.

Now suppose L is a central CT with parameter matrix $A = J_k^T(0)$. We will use the above theorem to obtain this CT as a limit of central CTs with parameter matrix $A = \operatorname{diag}(0, \lambda_2, \ldots, \lambda_k)$. The characteristic polynomial of these CTs is given by Eq. (9.4.6). In order to obtain the characteristic polynomial for a CT with $A = J_k^T(0)$ we will use the fact that the characteristic polynomial of $J_k^T(0)$ is z^k . Then starting with $A = \operatorname{diag}(0, \lambda_2, \ldots, \lambda_k)$, by Eq. (9.4.6) we have:

$$p(z) = \sum_{l=0}^{k} (a_l - \sum_{j=0}^{k-1-l} a_{j+1+l} \langle r, A^j r \rangle) z^l$$

$$\rightarrow z^k - \sum_{l=0}^{k-1} \langle r, A^{k-1-l} r \rangle z^l$$

$$= z^k - \sum_{l=0}^{k-1} \langle r, A^{k-1-l} r \rangle z^l$$

$$= z^k - \varepsilon \sum_{l=0}^{k-1} \sum_{i=1}^{l+1} x^i x^{l+2-i} z^l$$

Thus we have proven part of the following:

Proposition 9.4.9

Suppose L is a central CT with parameter matrix $A = J_k^T(0)$ and metric $g = \varepsilon S_k$. Then

the characteristic polynomial of L is:

$$p(z) = \det(zI - L) = z^k - \varepsilon \sum_{l=0}^{k-1} \sum_{i=1}^{l+1} x^i x^{l+2-i} z^l$$

Furthermore the following are true:

- L has no constant eigenfunctions.
- If $T(z) = \frac{p(z)}{B(z)}$ and $k \le 3$, then $\langle dT, dT \rangle = 4 \frac{d}{dz} T(z)$

PROOF We first prove the case where A is a real Jordan block. To prove that L has no constant eigenfunctions, we differentiate an equation preceding this proposition to obtain:

$$\nabla p = -2\sum_{l=0}^{k-1} z^l A^{k-1-l} r$$

from which we see that $\langle e_k, \nabla p \rangle = -2\varepsilon z^{k-1}x^1$. Thus L cannot have a constant eigenfunction. The equation for $\langle dT, dT \rangle$ is proven as follows. When $A = \text{diag}(0, \lambda_2, \dots, \lambda_k)$ one can easily prove this formula using Eq. (9.4.4). Then the formula for $A = J_k^T(0)$ follows by applying the limiting technique in Theorem 9.4.8 used above. Finally, for the case of a complex Jordan block, i.e. $A = J_k^T(\lambda)$ where $\lambda \in \mathbb{C}$, note that these proofs hold by replacing $A \to A - \lambda I$ and $z \to z + \lambda$.

Now one can use the second part of Lemma 9.4.6 to obtain the characteristic polynomial of any central CT in \mathbb{E}^n_{ν} . Indeed, suppose L is a central CT with parameter matrix

$$A = J_k^T(0) \oplus \operatorname{diag}(\lambda_{k+1}, \dots, \lambda_n) \qquad g = \varepsilon_0 S_k \oplus \operatorname{diag}(\varepsilon_{k+1}, \dots, \varepsilon_n)$$

We can apply Lemma 9.4.6 with U equal to the subspace corresponding to $J_k^T(0)$, then

$$p(z) = \det(zI - L) = \left(\prod_{i=k+1}^{n} y_i\right) \left(z^k - \varepsilon_0 \sum_{l=0}^{k-1} \left(\sum_{i=1}^{l+1} x^i x^{l+2-i}\right) z^l\right) - z^k \left(\sum_{i=k+1}^{n} x_i x^i \prod_{j=k+1, j \neq i}^{n} y_j\right)$$

Now when L is an ICT, we can obtain a transformation from canonical coordinates to Cartesian coordinates. Our formula is motivated by one in [KMR84] and is given as follows:

$$\sum_{i=1}^{l+1} x^i x^{l+2-i} = \frac{-\varepsilon_0}{l!} \left(\frac{d}{dz}\right)^l \left(\frac{p(z)}{B_{u\perp}(z)}\right)\Big|_{z=0} \quad l = 0, \dots, k-1$$
(9.4.7a)

$$(x^{i})^{2} = -\varepsilon_{i} \frac{p(\lambda_{i})}{B'(\lambda_{i})} \qquad \qquad i = k+1, \dots, n \qquad (9.4.7b)$$

The following lemma will be used to obtain the metric in canonical coordinates adapted to an ICT defined in a space of constant curvature.

Lemma 9.4.10

Suppose L is a central CT with parameter matrix A. Let

$$T(z) = \frac{p(z)}{B(z)}$$
 Then $\langle dT, dT \rangle = 4 \frac{d}{dz} T(z).$

PROOF We prove this by induction. The base cases are given by Proposition 9.4.9. Suppose U is a non-degenerate invariant subspace of A such that L_u has the form given by Proposition 9.4.9 and U^{\perp} satisfies the induction hypothesis.

By Eq. (9.4.3) we can write:

$$p(z) = p_u(z)B_{u^{\perp}}(z) + B_u(z)(p_{u^{\perp}}(z) - B_{u^{\perp}}(z))$$

Then

$$\mathrm{d}p = B_{u^{\perp}}\mathrm{d}p_u + B_u\mathrm{d}p_{u^{\perp}}$$

Thus from the above equation, we have:

$$\begin{aligned} \frac{\mathrm{d}p}{B} &= \frac{\mathrm{d}p_u}{B_u} + \frac{\mathrm{d}p_{u^{\perp}}}{B_{u^{\perp}}} \\ \Rightarrow \mathrm{d}T &= \mathrm{d}T_u + \mathrm{d}T_{u^{\perp}} \\ \Rightarrow \langle \mathrm{d}T, \mathrm{d}T \rangle &= \langle \mathrm{d}T_u, \mathrm{d}T_u \rangle + \langle \mathrm{d}T_{u^{\perp}}, \mathrm{d}T_{u^{\perp}} \rangle \\ &= 4\frac{d}{dz}T_u(z) + 4\frac{d}{dz}T_{u^{\perp}}(z) \\ &= 4\frac{d}{dz}T(z) \end{aligned}$$

Examples We end this section with some separable coordinate systems induced by central ICTs which can be analyzed fairly easily. These examples are a natural generalization of those presented in [Cra03, section 5] by Crampin.

Example 9.4.11 (Generalization of elliptic coordinates to \mathbb{E}^n_{ν})

Our first example is the central CT in \mathbb{E}_{ν}^{n} with parameter matrix $A = \operatorname{diag}(\lambda_{1}, ..., \lambda_{n})$ and orthogonal metric g = (-1, ..., -1, 1, ..., 1). This CT is easiest to analyze if we assume $\lambda_{1} < \lambda_{2} < \cdots < \lambda_{n}$. Recall from Eq. (9.4.4), the characteristic polynomial of L is:

$$p(z) = \det(zI - L) = \prod_{i=1}^{n} y_i - \sum_{i=1}^{n} x_i x^i \prod_{j \neq i} y_j$$

Using the above formula, one can show that L has no constant eigenfunctions (e.g. see the proof of Proposition 9.4.9). Then by Proposition 9.4.2, this CT is an ICT near any point where the eigenfunctions of L are simple. We will now show that L is an ICT in a dense subset of \mathbb{E}^n_{ν} . First note that

$$p(\lambda_i) = -\varepsilon_i (x^i)^2 \prod_{j \neq i} (\lambda_i - \lambda_j)$$
(9.4.8)

Assume each $x^i \neq 0$, then from Equation 9.4.8, we find that $\operatorname{sgn} p(\lambda_i) = \varepsilon_i (-1)^{n+1-i}$. Also since the coefficient of leading degree of p(z) is z^n , we find that $\lim_{z\to\infty} p(z) = 1$ and $\lim_{z\to-\infty} p(z) = (-1)^n$. Since by assumption we have that $\varepsilon_n = 1$, we can use the intermediate value theorem to deduce the following about the roots of p(z). If $\nu = 0$ (i.e. in Euclidean space), there are n distinct roots u^1, \dots, u^n satisfying:

$$\lambda_1 < u^1 < \lambda_2 < u^2 \dots < \lambda_n < u^n$$

If $\nu > 0$ then there are n distinct roots $u^1, ..., u^n$ satisfying:

$$u^{1} < \lambda_{1} < u^{2} \cdots < u^{\nu} < \lambda_{\nu} < \lambda_{\nu+1} < u^{\nu+1} < \lambda_{\nu+2} < u^{\nu+2} \cdots < \lambda_{n} < u^{n}$$
(9.4.9)

Hence L is an IC-tensor on an open dense subset of \mathbb{E}_{ν}^{n} ; because of this property one could consider the induced separable coordinates to be a generalization of elliptic coordinates. Since $p(\lambda_{i}) = \prod_{j=1}^{n} (\lambda_{i} - u^{j})$, by Equation (9.4.8), we can obtain the Cartesian coordinates in terms of the separable coordinates $u^{1}, ..., u^{n}$

$$(x^{i})^{2} = \varepsilon_{i} \frac{\prod_{j=1}^{n} (u^{j} - \lambda_{i})}{\prod_{j \neq i} (\lambda_{j} - \lambda_{i})}$$

By using Eq. (9.4.9) and Proposition 9.4.15, one can check that in the separable coordinates (u^1, \ldots, u^n) , for $1 \le i \le \nu$, $\operatorname{sgn} g^{ii} = \frac{(-1)^{n-i+1}}{(-1)^{n-i}} = -1$. Hence $\partial_1, \ldots, \partial_{\nu}$ are

timelike vector fields and the remaining ones are spacelike.

We now show that if we relax the condition that $\lambda_1 < \cdots < \lambda_n$ in the above example then the coordinate system may no longer be defined on a dense subset of \mathbb{E}^n_{ν} . Although one should note that the in \mathbb{E}^n that condition was not restrictive. The simplest case occurs in \mathbb{E}^2_1 .

Example 9.4.12

Consider a central CT L in \mathbb{E}_1^2 with parameter matrix $A = \text{diag}(\lambda_1, \lambda_2)$ where $\lambda_1 > \lambda_2$ and orthogonal metric g = diag(-1, 1). Denote Cartesian coordinates by (t, x). In this case the characteristic polynomial of L, p(z), given by Eq. (9.4.6) reduces to:

$$p(z) = z^{2} + (2(t^{2} - x^{2}) - \lambda_{1} - \lambda_{2})z - 2t^{2}\lambda_{2} + 2x^{2}\lambda_{1} + \lambda_{1}\lambda_{2}$$

One can calculate the discriminant of this polynomial to be:

$$4\left((t-x)^2 + \frac{\lambda_2 - \lambda_1}{2}\right)\left((t+x)^2 + \frac{\lambda_2 - \lambda_1}{2}\right)$$

If we define new Cartesian coordinates (y^1, y^2) by:

$$y^1 := \sqrt{2}(t-x)$$
 $y^2 := \sqrt{2}(t+x)$

and we let $e := \sqrt{\lambda_1 - \lambda_2}$, then L is a Benenti tensor on the following connected regions:

Region	(u^1, u^2)
N	$y^1 > e, y^2 < -e$
Е	$y^1, y^2 > e$
S	$y^1 < -e, y^2 > e$
W	$y^1, y^2 > -e$
C	$ y^1 , y^2 < e$

Hence the regions are separated by the lightlike lines $|y^i| = e$. Thus as claimed the associated separable coordinate systems aren't defined on a dense subset.

One can also find the coordinate domains as follows. Suppose L is an ICT with eigenfunctions $u^1 < u^2$. Then by requiring that the metric in these coordinates given by Proposition 9.4.15 to be Lorentzian, one finds the following constraints:

$$u^{1} < u^{2} < \lambda_{2} < \lambda_{1}$$
$$\lambda_{2} < \lambda_{1} < u^{1} < u^{2}$$
$$\lambda_{2} < u^{1} < u^{2} < \lambda_{1}$$
$$u^{1} < \lambda_{2} < \lambda_{1} < u^{2}$$

The above inequalities shown that in the subset where L is a Benenti tensor, if the eigenfunctions transition from one coordinate domain to another then one of the eigenfunctions must take the value λ_1 or λ_2 . Hence the transition manifolds are solutions of $p(\lambda_i) = 0$, i.e. by Eq. (9.4.4) where $(x^i)^2 = 0$. In this case, the eigenfunctions of L can be readily calculated:

$$t = 0 \Rightarrow \lambda_1, \lambda_2 + x^2$$
$$x = 0 \Rightarrow \lambda_1 - t^2, \lambda_2$$

Using the values of the eigenfunctions on these subsets and their possible ranges given in Eq. (9.4.10) one can deduce the following:

(y^1,y^2)	(u^1, u^2)
E, W	$u^1 < u^2 < \lambda_2 < \lambda_1$
N,S	$\lambda_2 < \lambda_1 < u^1 < u^2$
С	$\lambda_2 < u^1 < u^2 < \lambda_1$

Together with Eq. (9.4.5), this completes the analysis of these coordinate systems. \Box

Even in three dimensions, the above analysis becomes much more difficult. This is because in three dimensions one can show that the discriminant is an eight degree polynomial in the coordinates with many terms. Although we note two simplifications that could be made for the general case. First by transferring to a geometrically equivalent CT, we could have assumed one of the eigenvalues of A were zero. Secondly since the characteristic polynomial of L, given by Eq. (9.4.4) only depends on the quantities $(x^i)^2$ and not x^i explicitly, one can restrict the analysis to the quadrant where each $x^i > 0$ while losing no generality. This symmetry is a consequence of the non-uniqueness of the chosen basis, in particular due to the fact that if v is an eigenvector of A then so is -v.

9.4.2 Axial Concircular tensors

Proposition 9.4.13

Let L be an axial CT with parameter matrix $A = J_k(0)^T$ and metric $g = \varepsilon S_k$. Then

$$p(z) = \det(zI - L) = z^{k} + \sum_{l=2}^{k} \sum_{i=1}^{l-1} x^{k+1+i-l} x^{k+1-i} z^{k-l} - 2\varepsilon \sum_{i=1}^{k} x^{k-i+1} z^{k-i}$$
(9.4.11)

Furthermore the following are true:

• L has no constant eigenfunctions.

• If
$$k \leq 3$$
, then $\langle dp, dp \rangle = 4\varepsilon \frac{d}{dz} p(z)$.

PROOF We first outline how one proves the above formula for p(z). It is sufficient to calculate det L when L has the parameter matrix $A = J_k(\lambda)^T$. Let $\tilde{A} = [\tilde{a_1}, ..., \tilde{a_n}] := A + \varepsilon r \otimes e_k$. Then applying Lemma 9.4.5 to $L = \tilde{A} + e_1 \otimes r^{\flat}$ gives:

$$\det L = \bigwedge_{i=1}^{n} \tilde{a_i} + \sum_{i=1}^{n} \tilde{a_1} \wedge \dots \wedge x_i e_1 \wedge \dots \wedge \tilde{a_n}$$

After expanding r and e_1 in the basis $\{a_1, \ldots, a_k\}$ and simplifying, the result then follows by a straightforward but tedious calculation.

Suppose the above formula for p(z) holds. We now show that L has no constant eigenfunctions. The constant term of dp is:

$$-2\varepsilon \sum_{i=1}^{k} z^{k-i} \mathrm{d} x^{k-i+1}$$

If $\lambda \in \mathbb{R}$ satisfies $p(\lambda) \equiv 0$, then the above form must be identically zero. A contradiction, hence L has no constant eigenfunctions.

The formula involving $\langle dp, dp \rangle$ can be checked manually for the cases $k \leq 3$.

The following proposition will reduce the calculation of the characteristic polynomial for general axial concircular tensors to cases already considered.

Proposition 9.4.14 (Determinant of Axial Concircular tensors) Suppose L is an axial CT in canonical form given as follows:

$$L = A + e_1 \otimes r^{\flat} + r \otimes e_1^{\flat}$$
$$A = A_d \oplus A_c$$

where $A_d = J_k^T(\lambda)$. Then $p(z) = \det(zI - L)$ is given as follows:

$$p(z) = p_d(z)B(z) + \varepsilon(p_c(z) - B(z))$$
(9.4.12)
$$\diamondsuit$$

PROOF First note that it is sufficient to calculate det L. Write $r = r_d + r_c$ adapted to the decomposition $\mathbb{E}^n_{\nu} = D \oplus D^{\perp}$ where D is the A-invariant subspace generated by e_1 . Then

$$L = L_d + A_c + e_1 \otimes (r_c)^{\flat} + r_c \otimes e_1^{\flat}$$

where L_d is L restricted to D and A_c is A restricted to D^{\perp} . Let $\tilde{L} = L_d + A_c + e_1 \otimes (r_c)^{\flat}$, then applying Lemma 9.4.5 to $L = \tilde{L} + \varepsilon r_c \otimes e_k$ gives:

$$\det L = \det \tilde{L} + \varepsilon \tilde{L}_1 \wedge \dots \wedge r_c \wedge \dots \wedge \tilde{L}_n \tag{9.4.13}$$

where r_c appears in the kth spot. Now note that in block diagonal form

$$\tilde{L} = \begin{pmatrix} L_d & e_1 \otimes (r_c)^\flat \\ 0 & A_c \end{pmatrix}$$

Then after applying Lemma 9.4.5 once more, we get

$$\begin{split} \tilde{L}_1 \wedge \dots \wedge r_c \wedge \dots \wedge \tilde{L}_n &= \bigwedge_{i=1}^{k-1} (L_d)_i \wedge r_c \wedge (\sum_{i=k+1}^n a_{k+1} \wedge \dots \wedge x_i e_1 \wedge \dots \wedge a_n) \\ &= -\bigwedge_{i=1}^{k-1} (L_d)_i \wedge e_1 \wedge (\sum_{i=k+1}^n a_{k+1} \wedge \dots \wedge x_i r_c \wedge \dots \wedge a_n) \\ &= -\bigwedge_{i=1}^{k-1} a_i \wedge e_1 \wedge (\sum_{i=k+1}^n a_{k+1} \wedge \dots \wedge x_i r_c \wedge \dots \wedge a_n) \\ &= (-1)^k e_1 \wedge \dots \wedge e_k \wedge (\sum_{i=k+1}^n a_{k+1} \wedge \dots \wedge x_i r_c \wedge \dots \wedge a_n) \\ &= (-1)^k (\det(L_c) - \det(A_c)) \end{split}$$

where the second last equation follows by expanding e_1 in the basis $\{a_1, \ldots, a_k\}$. The result then follows by Eq. (9.4.13).

Now one can use Proposition 9.4.14 to obtain the characteristic polynomial of any axial CT in \mathbb{E}_{ν}^{n} . This is done as in the example in the discussion following Proposition 9.4.9. For example, we will calculate the Cartesian coordinates for a non-null axial CT (i.e. k = 1). Indeed, suppose L is a non-null axial CT and an ICT with eigenfunctions (u^{1}, \ldots, u^{n}) . Let $A_{c} = \operatorname{diag}(\lambda_{2}, \ldots, \lambda_{n})$, then from Eq. (9.4.12) and Eq. (9.4.11), we see that

$$p(z) = \det(zI - L) = (\prod_{i=2}^{n} y_i)(z - 2\varepsilon x^1) - \varepsilon(\sum_{i=2}^{n} x_i x^i \prod_{j=2, j \neq i}^{n} y_j)$$

where $y_i = z - \lambda_i$. Since $p(z) = \prod_{i=1}^n (z - u^i)$, we can deduce the transformation from the coordinates (u^1, \ldots, u^n) to Cartesian coordinates as follows. By evaluating $p(\lambda_i)$, we get

$$(x^{i})^{2} = -\varepsilon_{i}\varepsilon \frac{\prod_{j=1}^{n} (u^{j} - \lambda_{i})}{\prod_{j\geq 2, j\neq i} (\lambda_{j} - \lambda_{i})} \qquad \qquad i = 2, ..., n \qquad (9.4.14)$$

By taking the coefficient of z^{n-1} of p(z), we get:

$$x^{1} = \frac{\varepsilon}{2}(u^{1} + \dots + u^{n} - \lambda_{2} - \dots - \lambda_{n})$$
(9.4.15)

In conclusion, we note that this procedure can be generalized for $k \geq 2$.

Observe that Eq. (9.4.12) holds for a central CT if we define $p_d(z) \equiv 1$ in this case. We will use Eq. (9.4.12) and Lemma 9.4.10 to obtain the metric in canonical coordinates for some ICTs in \mathbb{E}^n_{ν} . We have the following:

Proposition 9.4.15 (ICT metrics in \mathbb{E}^n_{ν})

Suppose L is an ICT in Euclidean or Minkowski space in canonical form with eigenfunctions (u^1, \ldots, u^n) . Then the metric in adapted coordinates is orthogonal and

$$g_{ii} = \frac{\varepsilon}{4} \frac{p'(u^i)}{B(u^i)} = \frac{\varepsilon}{4} \frac{\prod_{j \neq i} (u^i - u^j)}{\prod_{j=1}^{n-k} (u^i - \lambda_j)}$$

where ε is the sign associated with L and $\lambda_1, \ldots, \lambda_{n-k}$ are the roots of B(z).

Remark 9.4.16

The above formula likely holds in general (see [KMR84]) but we haven't verified it for null axial CTs when k > 3.

PROOF Let $T(z) := \frac{p(z)}{B(z)}$, $S(z) = p_d(z)$ and $\tilde{T}(z) := \frac{p_c(z)}{B(z)}$, then Eq. (9.4.12) implies:

$$\mathrm{d}T = \varepsilon \mathrm{d}\tilde{T} + \mathrm{d}S$$

Also recall that in these spaces, the index $k \leq 3$. Hence

$$\begin{split} \langle \mathrm{d}T, \mathrm{d}T \rangle &= \mathrm{d}T(\nabla T) \\ &= \left\langle \mathrm{d}\tilde{T}, \mathrm{d}\tilde{T} \right\rangle + \left\langle \mathrm{d}S, \mathrm{d}S \right\rangle \\ &= 4 \frac{d}{dz} \tilde{T}(z) + 4\varepsilon \frac{d}{dz} S(z) \quad \text{by Lemma 9.4.10 and Proposition 9.4.13} \\ &= 4\varepsilon \frac{d}{dz} (\varepsilon \tilde{T}(z) + S(z)) \\ &\stackrel{(9.4.12)}{=} 4\varepsilon \frac{d}{dz} \frac{p(z)}{B(z)} \end{split}$$

Thus we have the following:

$$\frac{\langle (\mathrm{d}p)|_{z=u^{i}}, (\mathrm{d}p)|_{z=u^{i}}\rangle}{B(u^{i})^{2}} = 4\varepsilon \frac{d}{dz} \frac{p(z)}{B(z)}\Big|_{z=u^{i}}$$
$$= 4\varepsilon \frac{p'(u^{i})}{B(u^{i})}$$

From Proposition 9.4.3 we have:

$$g^{ii} = \frac{\langle (dp)|_{z=u^i}, (dp)|_{z=u^i} \rangle}{p'(u^i)^2}$$

= $4\varepsilon \frac{B(u^i)}{p'(u^i)}$
= $4\varepsilon \frac{\prod_{j=k+1}^n (u^i - \lambda_j)}{\prod_{j\neq i} (u^i - u^j)}$

Remark 9.4.17

The above trick for calculating the metric is based on Moser's calculation of the metric for sphere-elliptic coordinates in [Mos11, P. 179-180]. $\hfill \Box$

Corollary 9.4.18

Suppose L is a non-degenerate CT in Euclidean or Minkowski space in canonical form. Then the points at which a real eigenvalue of A_c is an eigenvalue of L are singular, i.e. L cannot be an ICT in any neighborhood of these points.

9.4.3 Concircular tensors in Spherical Submanifolds of pseudo-Euclidean space

In this section we treat the case of CTs defined on $\mathbb{E}_{\nu}^{n}(\kappa)$. We will be able to reduce most calculations to similar ones involving central CTs. The following proposition will allow us to do this.

Proposition 9.4.19 (Determinant of Spherical CTs)

Suppose $L = RL_c R^*$ is a CT in $\mathbb{E}^n_{\nu}(\frac{1}{r^2})$, the following holds:

$$p(z) = \det(zR - L + \frac{r \otimes r^{\flat}}{r^2}) = r^{-2}(B(z) - p_c(z))$$
(9.4.16)

PROOF It is sufficient to prove that:

$$\det(L + \frac{r \otimes r^{\flat}}{r^2}) = r^{-2}(\det L_c - \det A)$$

Observe that:

$$L + \frac{r \otimes r^{\flat}}{r^2} = AR + \frac{[(r \cdot A \cdot r) + r^2]}{r^4} r \otimes r^{\flat} - \frac{1}{r^2} r \otimes r^{\flat} \cdot A$$
$$= AR + r \otimes d$$

for some vector d and

$$AR = A - \frac{1}{r^2} Ar \otimes r^\flat$$

Let b_i be the columns of AR, then by Lemma 9.4.5 we have

$$\det(L + \frac{r \otimes r^{\flat}}{r^2}) = \bigwedge_{i=1}^n b_i + \sum_{i=1}^n b_1 \wedge \dots \wedge d_i r \wedge \dots \wedge b_n$$

Now observe that

$$0 = \det L = \bigwedge_{i=1}^{n} b_i + \sum_{i=1}^{n} b_1 \wedge \dots \wedge (d_i - \frac{x_i}{r^2}) r \wedge \dots \wedge b_n$$

Thus

$$\det(L + \frac{r \otimes r^{\flat}}{r^2}) = \frac{1}{r^2} \sum_{i=1}^n b_1 \wedge \dots \wedge x_i r \wedge \dots \wedge b_n$$
(9.4.17)

Now, again using Lemma 9.4.5, we have:

$$b_1 \wedge \dots \wedge r \wedge \dots \wedge b_n = (-1)^{i-1} r \wedge b_1 \wedge \dots \wedge \hat{b}_i \wedge \dots \wedge b_n$$

= $(-1)^{i-1} r \wedge (a_1 \wedge \dots \wedge \hat{a}_i \wedge \dots \wedge a_n - r^{-2} \sum_{j \neq i} a_1 \wedge \dots \wedge x_j A r \wedge \dots \wedge a_n)$

Note that the term \hat{b}_i , means b_i is missing from the product. Now note that for $i \neq j$

$$(-1)^{i-1}x_ir \wedge a_1 \wedge \dots \wedge x_jAr \wedge \dots \wedge a_n = -(-1)^{j-1}x_jr \wedge a_1 \wedge \dots \wedge x_iAr \wedge \dots \wedge a_n$$

Thus

$$\det(L + \frac{r \otimes r^{\flat}}{r^2}) \stackrel{(9.4.17)}{=} r^{-2} \sum_{i=1}^n b_1 \wedge \dots \wedge x_i r \wedge \dots \wedge b_n$$
$$= r^{-2} \sum_{i=1}^n a_1 \wedge \dots \wedge x_i r \wedge \dots \wedge a_n$$
$$= r^{-2} (\det(A + r \otimes r^{\flat}) - \det A)$$

Using Eq. (9.4.16), for ICTs, the transformation from canonical coordinates to Cartesian coordinates can be calculated using the standard method. Indeed, if L is an ICT in $\mathbb{E}_{\nu}^{n}(\frac{1}{r^{2}})$ with parameter matrix:

$$A = J_k^T(0) \oplus \operatorname{diag}(\lambda_{k+1}, \dots, \lambda_n) \qquad g = \varepsilon_0 S_k \oplus \operatorname{diag}(\varepsilon_{k+1}, \dots, \varepsilon_n)$$

Then by a calculation almost identical to the one used to derive Eqs. (9.4.7a) and (9.4.7b), one obtains the following now using Eq. (9.4.16):

$$\sum_{i=1}^{l+1} x^i x^{l+2-i} = \frac{r^2 \varepsilon_0}{l!} \left(\frac{d}{dz}\right)^l \left(\frac{p(z)}{B_{u^{\perp}}(z)}\right)\Big|_{z=0} \quad l = 0, \dots, k-1$$
(9.4.18a)

$$(x^i)^2 = r^2 \varepsilon_i \frac{p(\lambda_i)}{B'(\lambda_i)} \qquad \qquad i = k+1, \dots, n \qquad (9.4.18b)$$

The transformation from canonical coordinates (u^1, \ldots, u^{n-1}) to Cartesian coordinates are obtained by noting that $p(z) = \prod_{i=1}^{n-1} (z - u^i)$.

Example 9.4.20 (Circular coordinates)

Let $M = \mathbb{E}^2_{\nu}(\kappa)$ where $\kappa = \pm 1$. Consider the CT in M with parameter matrix:

$$A = \operatorname{diag}(0, 1) \qquad g = \operatorname{diag}(\kappa_1, \varepsilon) \quad \kappa_1, \varepsilon \in \{-1, 1\}$$

Then by Eqs. (9.4.18a) and (9.4.18b), Cartesian coordinates (x, y) are given by:

$$x^{2} = \kappa \kappa_{1} u$$
$$y^{2} = \kappa \varepsilon (1 - u)$$

We now show how to obtain the standard parameterizations of these coordinates. First note that by the metric-Jordan canonical form theory, there are three isometrically inequivalent cases⁴:

Case 1 $\kappa_1 = \kappa$ and $\varepsilon = \kappa$, thus $g = \text{diag}(\kappa, \kappa)$ If we take $u = \cos^2(t)$, then we obtain:

$$x^{2} = \cos^{2}(t)$$
$$y^{2} = \sin^{2}(t)$$

Case 2 $\kappa_1 = \kappa$ and $\varepsilon = -\kappa$, thus $g = \text{diag}(\kappa, -\kappa)$ If we take $u = \cosh^2(t)$, then we obtain:

$$x^{2} = \cosh^{2}(t)$$
$$y^{2} = \sinh^{2}(t)$$

Case 3 $\kappa_1 = -\kappa$ and $\varepsilon = \kappa$, thus $g = \text{diag}(-\kappa, \kappa)$

⁴Note that these cases additionally depend on ν .

If we take $u = -\sinh^2(t)$, then we obtain:

$$x^{2} = \sinh^{2}(t)$$
$$y^{2} = \cosh^{2}(t)$$

Although the last two cases are geometrically equivalent, it will be useful to distinguish them when we move on to reducible CTs. $\hfill \Box$

Also using Eq. (9.4.16), one can obtain the metric in ICT induced coordinates.

Proposition 9.4.21 (ICT metrics in $\mathbb{E}^n_{\nu}(\kappa)$)

Suppose L is an ICT in $\mathbb{E}^n_{\nu}(\frac{1}{r^2})$ with eigenfunctions (u^1, \ldots, u^{n-1}) . Then the metric in adapted coordinates is orthogonal and

$$g_{ii} = \frac{-r^2}{4} \frac{p'(u^i)}{B(u^i)} = \frac{-r^2}{4} \frac{\prod_{j \neq i} (u^i - u^j)}{\prod_{j=1}^n (u^i - \lambda_j)}$$

where $\lambda_1, \ldots, \lambda_n$ are the roots of B(z).

PROOF We will reduce this calculation to the corresponding one for L_c using Eq. (9.4.16). We will assume that L is an ICT with eigenfunctions (u^1, \ldots, u^{n-1}) in some neighborhood in $\mathbb{E}^n_{\nu}(\frac{1}{r^2})$.

Now if we let \tilde{d} denote the exterior derivative on the sphere, note that

$$\tilde{\mathrm{d}}p = R^*\mathrm{d}p$$

Now we make the following observation.

$$\left\langle \mathrm{d}p, r^{\flat} \right\rangle = \nabla_r p = 0$$

This can be proven, for example, by using Eq. (9.4.2) and the fact that r is a CV. Note that the above equation also implies that $\langle dp_c, r^{\flat} \rangle = -2r^2p$.

Hence we see that

$$\left\langle \tilde{\mathrm{d}}p, \tilde{\mathrm{d}}p \right\rangle = \left\langle \mathrm{d}p, \mathrm{d}p \right\rangle$$

Thus at a root $z = u^i$, we have

$$\left\langle \tilde{\mathrm{d}}p, \tilde{\mathrm{d}}p \right\rangle = r^{-4} \left\langle \mathrm{d}p_c, \mathrm{d}p_c \right\rangle$$

Then at $z = u^i$ we have

$$\frac{\left\langle \tilde{d}p, \tilde{d}p \right\rangle}{B^2} = \frac{r^{-4} \left\langle dp_c, dp_c \right\rangle}{B^2}$$
$$\stackrel{9.4.10}{=} 4r^{-4} \frac{d}{dz} \frac{p_c(z)}{B(z)} \Big|_{z=u^i}$$
$$= -4r^{-2} \frac{d}{dz} \frac{p(z)}{B(z)} \Big|_{z=u^i}$$
$$= -4r^{-2} \frac{p'(u^i)}{B(u^i)}$$

Thus Proposition 9.4.21 follows from the above equation and Proposition 9.4.3.

9.5 Classification of reducible concircular tensors

In this section, we will show how to find a warped product which "decomposes"⁵ a given reducible OCT defined in a space of constant curvature. To do this, we will use the knowledge of warped product decompositions of these spaces summarized in Section 8.4 and Proposition 6.3.7 which gives us a method to construct reducible OCTs.

The following definition will be useful.

Definition 9.5.1

Suppose L is a CT in M and let $N = N_0 \times_{\rho_1} N_1 \times \cdots \times_{\rho_k} N_k$ be a local warped product decomposition of M passing through $\bar{p} \in N \subseteq M$. We say L is *decomposable* in this warped product if for each $p \in N$ and i > 0, $T_p N_i$ is an invariant subspace for L.

9.5.1 In pseudo-Euclidean space

Suppose $N = N_0 \times_{\rho_1} N_1 \times \cdots \times_{\rho_k} N_k$ is a warped product and \tilde{L} is a CT in N_0 . We say \tilde{L} can be extended to a CT in N if \tilde{L} satisfies Eq. (6.3.3) for each *i* with some $\lambda_i \in \mathbb{R}$. Assuming \tilde{L} is an OCT, then Proposition 6.3.7 allows one to define a CT on N which restricts to \tilde{L} on N_0 . The following lemma will be our main tool for classifying reducible concircular tensors.

Lemma 9.5.2

Fix a proper warped product decomposition $(V_0 \oplus V_1; a)$ of \mathbb{E}^n_{ν} and let $L^i_{\ j} = A^i_{\ j} + mx^ix_j + w^ix_j + x^iw_j$ be a concircular tensor in N_0 . Then L can be extended to concircular tensor in \mathbb{E}^n_{ν} decomposable in this warped product iff a is an eigenvector of A orthogonal to w_{\square}

PROOF First observe

⁵This amounts to partially diagonalizing these CTs.
$$v^k \nabla_k \operatorname{tr}(L) = v^k \nabla_k (m x_i x^i + 2x^i w_i)$$

= $m[(v^k \nabla_k x_i) x^i + x_i (v^k \nabla_k x^i)] + 2[(v^k \nabla_k w_i) x^i + w_i (v^k \nabla_k x^i)]$
= $m(v_i x^i + x_i v^i) + 2v^i w_i$
= $2m v^i x_i + 2v^i w_i$

Hence $\nabla^i \operatorname{tr}(L) = 2(mx^i + w^i)$. Now let $\rho = a^i x_i = \langle a, x \rangle > 0$, then one can similarly show that

$$\nabla^i \log \rho = \frac{a^i}{\rho}$$

Then,

$$\begin{split} L^{i}_{\ j} \nabla^{j} \log \rho &- \frac{1}{2} \nabla^{i} \operatorname{tr}(L) = \frac{1}{\rho} (A^{i}_{\ j} a^{j} + mx^{i} x_{j} a^{j} + w^{i} x_{j} a^{j} + x^{i} w_{j} a^{j}) - mx^{i} - w^{i} \\ &= \frac{1}{\rho} (A^{i}_{\ j} a^{j} + x^{i} w_{j} a^{j}) + \frac{1}{\rho} (mx^{i} \rho + w^{i} \rho) - mx^{i} - w^{i} \\ &= \frac{1}{\rho} (A^{i}_{\ j} a^{j} + x^{i} w_{j} a^{j}) \end{split}$$

By definition, L can be extended to a CT decomposable in this warped product iff $L^i_{\ j} \nabla^j \log \rho - \frac{1}{2} \nabla^i \operatorname{tr}(L) \in \operatorname{span} \{\nabla^i \log \rho\}$. The above equation implies that this happens iff a is an eigenvector of A and $a \in w^{\perp}$.

We now use the above lemma to construct reducible CTs in \mathbb{E}^n_{ν} .

Proposition 9.5.3 (Constructing Reducible CTs in \mathbb{E}^n_{ν})

Fix a proper warped product decomposition $(V_0 \oplus V_1; a)$ of \mathbb{E}^n_{ν} and let $\tilde{L} = \tilde{A} + m\tilde{r} \odot \tilde{r} + 2\tilde{r} \odot \tilde{w}$ be a concircular tensor in N_0 (in contravariant form) which can be extended to a concircular tensor L in \mathbb{E}^n_{ν} via the above lemma. Since $N_0 \subset V_0 \subset \mathbb{E}^n_{\nu}$, we can consider \tilde{L} to be a tensor in \mathbb{E}^n_{ν} . Then L is given as follows:

$$L = A + mr \odot r + 2r \odot \tilde{w}$$

where as a linear operator, $A = \tilde{A} + \lambda I_{V_1}$, where λ is the eigenvalue of \tilde{A} associated with a and I_{V_1} is the identity on V_1 .

PROOF Throughout the proof, G is the contravariant metric for \mathbb{E}_{ν}^{n} and this metric adapted to the warped product is given as follows:

$$G = G' + \frac{1}{\rho^2}G_1$$

The non-null case: In this case $\kappa_1 := a^2 = \pm 1$. Let $m := \dim V_0$ and choose an orthonormal basis for V_0 , $\{a_1, ..., a_m\}$ with $a_m = a$.

First note that for $p = (p_0, p_1) \in N_0 \times N_1$ and $v = (v_0, v_1) \in T_p(N_0 \times N_1)$, Eq. (8.4.3) implies that

$$\psi_* v = P_0 v_0 + \langle a, v_0 \rangle (p_1 - c) + \langle a, p_0 \rangle v_1$$

Hence we observe the following:

$$\psi_* p_0 = P_0 p_0 + \langle a, p_0 \rangle (p_1 - c)$$

$$= \psi(p_0, p_1)$$
(9.5.1)

and

$$\psi_* a_i = a_i \quad \text{for } i = 1, ..., m - 1$$

$$(9.5.2)$$

Now let $\tilde{L} = \tilde{A} + m\tilde{r} \odot \tilde{r} + 2\tilde{w} \odot \tilde{r}$ be a concircular tensor in N_0 satisfying $\tilde{A}a = \lambda a$ for some λ and $\langle a, \tilde{w} \rangle = 0$. Then from Lemma 9.5.2 we know that $\psi_*(\tilde{L} + \frac{\lambda}{\rho^2}G_1)$ is a concircular tensor in \mathbb{E}^n_{ν} . We now calculate $\psi_*(\tilde{L} + \frac{\lambda}{\rho^2}G_1)$ explicitly.

First note that

$$\tilde{A} = A_0 + \lambda \kappa_1 a \odot a$$

where $A_0 a = 0$ and so $\psi_* A_0 = A_0$ by Eq. (9.5.2). Let G be the contravariant metric for \mathbb{E}^n_{ν} and G_0 be the restriction of G to W_0 , then

$$G = G' + \frac{1}{\rho^2} G_1$$
$$= G_0 + \kappa_1 a \odot a + \frac{1}{\rho^2} G_1$$

Thus

$$\frac{1}{\rho^2}G_1 = G - G_0 - \kappa_1 a \odot a$$

Let G_{V_1} be the restriction of G to V_1 , then

$$\psi_*(\tilde{A} + \frac{\lambda}{\rho^2}G_1) = \psi_*(A_0 + \lambda\kappa_1 a \odot a + \lambda(G - G_0 - \kappa_1 a \odot a))$$
$$= \psi_*(A_0 + \lambda(G - G_0))$$
$$= A_0 + \lambda(G - G_0)$$
$$= \tilde{A} + \lambda G_{V_1}$$

where the second last equality follows from Eq. (9.5.2) and the fact that ψ is an isometry.

Eq. (9.5.1) implies that $\psi_* \tilde{r} = r$, also Eq. (9.5.2) together with the fact that $\langle a, \tilde{w} \rangle = 0$ implies that $\psi_* \tilde{w} = \tilde{w}$. Thus we conclude that

$$\psi_*(\tilde{L} + \frac{\lambda}{\rho^2}G_1) = A + mr \odot r + 2r \odot \tilde{w}$$

where as a linear operator, $A = \tilde{A} + \lambda I_{V_1}$ where I_{V_1} is the identity on V_1 .

The null case: In this case a is a lightlike vector. Let $m := \dim V_0$ and choose a basis $\{a_1, ..., a_{m-2}, a, b\}$ for V_0 where $\{a_1, ..., a_{m-2}\}$ is an orthonormal basis for W_0 and a, b are as in the null warped product decomposition.

First note that for $p = (p_0, p_1) \in N_0 \times N_1$ and $v = (v_0, v_1) \in T_p(N_0 \times N_1)$, Eq. (8.4.4) implies that

$$\psi_* v = P_0 v_0 + (\langle b, v_0 \rangle - \frac{1}{2} \langle a, v_0 \rangle (P_1 p_1)^2 - \langle a, p_0 \rangle \langle P_1 p_1, P_1 v_1 \rangle) a + \langle a, v_0 \rangle b + \langle a, v_0 \rangle P_1 p_1 + \langle a, p_0 \rangle P_1 v_1$$

Hence we observe the following:

$$\psi_* p_0 = P_0 p_0 + (\langle b, p_0 \rangle - \frac{1}{2} \langle a, p_0 \rangle (P_1 p_1)^2) a + \langle a, p_0 \rangle b + \langle a, p_0 \rangle P_1 p_1$$
(9.5.3)
= $\psi(p_0, p_1)$

and

$$\psi_* a_i = a_i \quad i = 1, ..., m - 2$$
(9.5.4)
 $\psi_* a = a$

Now let $\tilde{L} = \tilde{A} + m\tilde{r} \odot \tilde{r} + 2\tilde{w} \odot \tilde{r}$ be a concircular tensor on N_0 satisfying $\tilde{A}a = \lambda a$ for some λ and $\langle a, \tilde{w} \rangle = 0$. Then from Lemma 9.5.2 we know that $\psi_*(\tilde{L} + \frac{\lambda}{\rho^2}G_1)$ is a concircular tensor in \mathbb{E}^n_{ν} . We now calculate $\psi_*(\tilde{L} + \frac{\lambda}{\rho^2}G_1)$ explicitly.

Since $\tilde{A}a = \lambda a$, \tilde{A} can be decomposed in contravariant form as follows:

$$\tilde{A} = A_0 + 2\lambda a \odot b$$

where $A_0 a = 0$ and so $\psi_* A_0 = A_0$ by Eq. (9.5.4). Let G be the contravariant metric for \mathbb{E}^n_{ν} and G_0 be the restriction of G to W_0 , then we see that

$$\frac{1}{\rho^2}G_1 = G - G_0 - 2a \odot b$$

Let G_{V_1} be the restriction of G to V_1 , then

$$\psi_*(\tilde{A} + \frac{\lambda}{\rho^2}G_1) = \psi_*(A_0 + 2\lambda a \odot b + \lambda(G - G_0 - 2a \odot b))$$
$$= \psi_*(A_0 + \lambda(G - G_0))$$
$$= A_0 + \lambda(G - G_0)$$
$$= A_0 + 2\lambda a \odot b + \lambda G_{V_1}$$
$$= \tilde{A} + \lambda G_{V_1}$$

where the third equality follows from Eq. (9.5.4) and the fact that ψ is an isometry.

Eq. (9.5.3) implies that $\psi_* \tilde{r} = r$, also Eq. (9.5.4) together with the fact that $\langle a, \tilde{w} \rangle = 0$ implies that $\psi_* \tilde{w} = \tilde{w}$. Thus we conclude that

$$\psi_*(\tilde{L} + \frac{\lambda}{\rho^2}G_1) = A + mr \odot r + 2r \odot \tilde{w}$$

where as a linear operator, $A = \tilde{A} + \lambda I_{V_1}$ where I_{V_1} is the identity on V_1 .

Remark 9.5.4

Note that even though the extended CT, L, can be naturally extended to all of \mathbb{E}_{ν}^{n} . It is the extension of \tilde{L} only for the subset $\operatorname{Im}(\psi)$ of \mathbb{E}_{ν}^{n} given by Theorem 8.4.5, which is in general not a dense subset of \mathbb{E}_{ν}^{n} .

The following corollary will be useful later on.

Corollary 9.5.5

Fix a proper warped product decomposition ψ determined by the data $(V_0 \oplus V_1; a)$ with $\kappa_1 := a^2 = \pm 1$. Let $\tilde{r} = P_1 r$ be the dilatational vector in W_1 and G_1 be the metric in W_1 . Write the metric adapted to the warped product as $G = G' + \frac{1}{a^2}\tilde{G}$, then:

$$\psi_* \tilde{G} = \kappa_1 \tilde{r}^2 (G_1 - \frac{1}{\tilde{r}^2} \tilde{r} \odot \tilde{r})$$

PROOF Let G be the contravariant metric for \mathbb{E}_{ν}^{n} and G_{0} (resp. G_{1}) be the restriction of G to W_{0} (resp. W_{1}), then recall that

$$\frac{1}{\rho^2}\tilde{G} = G - G_0 - \kappa_1 a \odot a$$

Hence the above equation together with Eq. (9.5.2) implies that

$$\psi_* \tilde{G} = \rho^2 (G - G_0 - \kappa_1 \psi_* (a \odot a))$$
$$= \rho^2 (G_1 - \kappa_1 \psi_* (a \odot a))$$

Let $\tilde{p}_1 = p_1 - c \in W_1(\kappa_1)$ then $\tilde{r} = P_1 r = \langle a, p_0 \rangle \tilde{p}_1$. Then by Eq. (9.5.1)

$$\psi_* a = \kappa_1 \tilde{p}_1$$
$$= \kappa_1 \frac{\tilde{r}}{\langle a, p_0 \rangle}$$
$$= \kappa_1 \frac{\tilde{r}}{\rho}$$

Thus since $\tilde{r}^2 = \frac{\rho^2}{\kappa_1}$, we have:

$$\psi_* \tilde{G} = \rho^2 (G_1 - \kappa_1 \psi_* (a \odot a))$$

= $\rho^2 (G_1 - \kappa_1 \frac{1}{\rho^2} \tilde{r} \odot \tilde{r})$
= $\kappa_1 \tilde{r}^2 (G_1 - \frac{1}{\tilde{r}^2} \tilde{r} \odot \tilde{r})$

We now present some examples which show how to use the above proposition (Proposition 9.5.3) to construct warped products which decompose a given reducible CT.

Example 9.5.6

Let $M = \mathbb{E}_{\nu}^{n}$ where $n \geq 3$. Consider the central CT L with parameter matrix $A = \varepsilon e \odot e$ with $\varepsilon := e^{2} = \pm 1$.

Let $W := e^{\perp}$ and P be the orthogonal projection onto W. Choose $\bar{p} \in \mathbb{E}^n_{\nu}$ such that $(P\bar{p})^2 \neq 0$, WLOG we assume $(P\bar{p})^2 = \pm 1$. We now construct a warped product passing through \bar{p} which decomposes L.

Let $\kappa_1 := \operatorname{sgn}(P\bar{p})^2$ and take $a := \kappa_1 P\bar{p} \in W$. Let $V_1 = W \cap a^{\perp}$ and $V_0 = V_1^{\perp} = \mathbb{R}e \oplus \mathbb{R}a$. Note that a was chosen so that the initial data $(\bar{p}; V_0 \oplus V_1; a)$ is in canonical form and also note $\kappa_1 = a^2$. Let $\psi : N_0 \times_{\rho} N_1 \to \mathbb{E}^n_{\nu}$ be the warped product in Theorem 8.4.5 determined by this initial data.

Now let $\tilde{A} := \varepsilon e \odot e + 0a \odot a \in C_0^2(N_0)$, then by construction we have that:

$$A = \tilde{A} + 0I_{V_1}$$

Let \hat{L} be the central CT in N_0 with parameter matrix \hat{A} and suppose the contravariant metric in the warped product decomposes as $G = G' + \frac{1}{\rho^2}G_1$. The above proposition shows that:

$$\psi_*(\tilde{L}+0\frac{1}{\rho^2}G_1)=L$$

for all points in the image of ψ , which includes \bar{p} . Hence this warped product decomposition decomposes L. Note that this warped product was constructed so that \tilde{A} has simple eigenvalues and so \tilde{L} is no longer reducible.

In the following we replace N_1 with $N_1 - c_1$ so that N_1 is a central hyperquadric. Then by Eq. (8.4.3), we have for $(p_0, p) = (\kappa_1 x a + y e, p) \in N_0 \times N_1$

$$\psi(p_0, p) = xp + ye \qquad \Box$$

The above example will be applied to construct separable coordinates in Section 9.6.2, see Example 9.6.4. We now give a non-Euclidean variation of the above example.

Example 9.5.7

Let $M = \mathbb{E}_{\nu}^{n}$ where $n \geq 3$. Consider the central CT L with parameter matrix $A = a \odot a$ with $a^{2} = 0$ and $a \neq 0$.

Let $W = a^{\perp}$. Choose $\bar{p} \notin W$, WLOG we assume $\langle \bar{p}, a \rangle = \pm 1$. We now construct a warped product passing through \bar{p} which decomposes L.

If $\langle \bar{p}, a \rangle = -1$, then set a := -a, so we can assume $\langle \bar{p}, a \rangle = 1$. Define b as follows:

$$b := \bar{p} - \frac{\bar{p}^2}{2}a \tag{9.5.5}$$

Note that b is a lightlike vector satisfying $\langle a, b \rangle = 1$. Define $V_1 = a^{\perp} \cap b^{\perp}$ and $V_0 = \operatorname{span}\{a, b\}$. Note that b was chosen so that the initial data $(\bar{p}; V_0 \oplus V_1; a)$ is in canonical form. Let $\psi : N_0 \times_{\rho} N_1 \to \mathbb{E}^n_{\nu}$ be the warped product in Theorem 8.4.5 determined by this initial data.

Note that $\{b, a\}$ forms a cycle of generalized eigenvectors for A and $A|_{V_1} = 0I_{V_1}$. Hence by the above proposition, $(\psi^{-1})_*L$ is decomposable in this warped product. Also by Theorem 8.4.5, $\bar{p} \in \text{Im}(\psi)$. Also, the restriction of $(\psi^{-1})_*L$ to N_0 , \tilde{L} , is a central CT with 2D parameter matrix $a \odot a$. In the following we replace N_1 with $P_1(N_1 - \bar{p})$ so that $N_1 = V_1$ is a vector space. Then by Eq. (8.4.4), we have for $(p_0, p) = (xb + ya, p) \in N_0 \times N_1$

$$\psi(p_0, p) = x(b + p - \frac{1}{2}p^2a) + ya$$

General Construction We will now show how to use Proposition 9.5.3 to construct a warped product which decomposes an interesting class⁶ of non-degenerate reducible CTs. This construction generalizes the above examples. First we need a preliminary definition. Suppose A is a linear operator on a vector space. We say that a vector v is a proper generalized eigenvector of A if $(A - \lambda I)^k v = 0$ for some $\lambda \in \mathbb{C}$ and k > 1.

Let $L = A + mr \odot r + 2r \odot w$ be a non-degenerate CT in \mathbb{E}_{ν}^{n} in the canonical form given by Theorem 9.1.8. We let the subspace D and the matrix A_{c} be as in the remarks following that theorem. We will assume that each real generalized eigenspace of A_{c} admits at most one proper generalized eigenvector. We lose no generality when working in Euclidean or Minkowski space (see Section 8.2.1).

Now let W_1, \ldots, W_k be the multidimensional (real) eigenspaces of A_c with corresponding eigenvalues $\lambda_1, \ldots, \lambda_k$. The following construction is based on the metric-Jordan canonical form of A_c , see Theorem 8.2.5.

Case 1 W_i is a non-degenerate subspace Choose a unit vector $a_i \in W_i$ and define $V_i := W_i \cap a_i^{\perp}$. The pair (V_i, a_i) determine a sphere.

Case 2 W_i is a degenerate subspace

Consider the metric-Jordan canonical form for A_c . By assumption there must be a single cycle v_1, \ldots, v_r of generalized eigenvectors with $v_r \in W_i$ being a lightlike eigenvector. Let $a_i := v_r$ and $V_i := W_i \cap v_1^{\perp}$, note that V_i is non-degenerate.

Now let $V_0 := \bigcap_{i=1}^k V_i^{\perp}$ and $\tilde{A} := A|_{V_0}$. By construction, the data $(\bigoplus_{i=0}^k V_i; a_1, ..., a_k)$, determines a warped product decomposition $\psi : N_0 \times_{\rho_1} N_1 \cdots \times_{\rho_k} N_k \to \mathbb{E}_{\nu}^n$ in canonical form. By repeatedly applying Proposition 9.5.3 we see that L is decomposable in the warped product decomposition induced by ψ , with the following properties:

- $((\psi^{-1})_*L)|_{N_0} = \tilde{A} + m\tilde{r} \odot \tilde{r} + 2\tilde{r} \odot w$ where \tilde{r} is the dilatational vector field in N_0
- $\tilde{A}|_{D^{\perp}}$ only has eigenspaces of dimension one, i.e. each Jordan block of $\tilde{A}|_{D^{\perp}}$ has a distinct eigenvalue.
- For each i > 0, TN_i is an eigenspace of $(\psi^{-1})_*L$ with constant eigenfunction λ_i

 $^{^6\}mathrm{This}$ class includes all reducible OCTs in Euclidean and Minkowski space.

On Completeness We will end this section by showing that the above construction is complete, meaning that the restriction of $(\psi^{-1})_*L$ to the geodesic factor N_0 no longer has constant eigenfunctions.

We also note here that with an appropriate choice of a_1, \ldots, a_k we can choose warped product decompositions to cover all of \mathbb{E}^n_{ν} except for a union of closed submanifolds with dimension strictly less than n. Examples 9.5.6 and 9.5.7 give more details on how to do this. In other words, for the non-degenerate CTs considered above, there exists a warped product decomposition $\psi : N_0 \times_{\rho_1} N_1 \cdots \times_{\rho_k} N_k \to \mathbb{E}^n_{\nu}$ such that $\operatorname{Im}(\psi)$ is a dense subset of \mathbb{E}^n_{ν} . Although the cost of this is that the factors N_i may no longer be connected subsets.

The following lemma shows that the classification of reducible CTs given above is complete for central CTs.

Lemma 9.5.8 (Reducible central CTs)

Let L be a central CT with parameter matrix A. Suppose that each real generalized eigenspace of A has at most one proper generalized eigenvector. Then A has a real eigenspace \tilde{E}_{λ} with dimension m > 1 iff L has a non-degenerate eigenspace E_{λ} (defined on a dense subset of \mathbb{E}_{ν}^{n}) with constant eigenfunction λ and dimension m - 1.

PROOF It was proven above that under the hypothesis, if A has a real eigenspace with dimension m > 1 then L has a non-degenerate eigenspace E_{λ} with dimension m - 1. We will now prove the converse.

To prove the converse, we simply have to prove that if all real eigenspaces of A are at most one dimensional then L has no non-degenerate eigenspaces with constant eigenfunctions defined on open subsets of \mathbb{E}^n_{ν} . It is sufficient to show that L has no constant eigenfunctions defined on open subsets of \mathbb{E}^n_{ν} .

We prove this by induction. The base cases are given by Proposition 9.4.9. Suppose U is a non-degenerate invariant subspace of A such that L_u has the form given by Proposition 9.4.9 and U^{\perp} satisfies the induction hypothesis. By Eq. (9.4.3) we can write:

$$p(z) = p_u(z)B_{u^{\perp}}(z) + B_u(z)(p_{u^{\perp}}(z) - B_{u^{\perp}}(z))$$

Then

$$\mathrm{d}p = B_{u^{\perp}}\mathrm{d}p_u + B_u\mathrm{d}p_{u^{\perp}}$$

By the induction hypothesis, $L_{u^{\perp}}$ has no constant eigenfunctions. Suppose λ is a constant eigenfunction of p, then by Proposition 9.4.9 and the above equation, it follows that

$$B_{u^{\perp}}(\lambda) = B_u(\lambda) = 0$$

If B_u has no real roots, we reach a contradiction. Otherwise, by construction A must have a real eigenspace with dimension m > 1, a contradiction. Hence we conclude that L has no constant eigenfunctions which proves the claim by induction.

Since a multidimensional eigenspace of an OCT has a constant eigenfunction, the above proposition allows us to classify these eigenspaces when the CTs considered induce an OCT on some subset of \mathbb{E}_{ν}^{n} . For completeness sake, we will now show that the hypothesis of the above proposition is the most general when it comes to classifying OCTs.

Proposition 9.5.9

Let L be a central CT with parameter matrix A. Suppose A has a real generalized eigenspace with multiple proper generalized eigenvectors, then L is not an OCT. \diamond

PROOF WLOG we can assume that this generalized eigenspace of A is associated with the eigenvalue zero. First we have

$$L = A + r \odot r$$
$$L^{2} = A^{2} + Ar \odot r + r^{2}r \odot r$$

By hypothesis, dim $N(L) \ge 1$. We also have that dim $N(A^2) \ge 4$. The above equation shows that the range of L^2 is spanned by $\{r, Ar\}$ and the range of A^2 (on a dense subset of \mathbb{E}^n_{ν}), hence we see that dim $N(L^2) \ge 1 + \dim N(L)$. This implies that L is not point-wise diagonalizable on some dense subset of \mathbb{E}^n_{ν} (see for example [FIS03]).

In fact one can show that if $A = J_2(0) \oplus J_2(0)$, then the associated central CT has a 2-cycle of generalized eigenvectors associated with eigenvalue zero.

The following lemma is the analogue of Lemma 9.5.8 for axial CTs. Its proof is also analogous and reduces to Lemma 9.5.8 with the help of Eq. (9.4.12) and Proposition 9.4.13.

Lemma 9.5.10 (Reducible axial CTs)

Let L be an axial CT with parameter matrix A. Suppose that each real generalized eigenspace of A_c has at most one proper generalized eigenvector. Then A_c has a real eigenspace \tilde{E}_{λ} with dimension m > 1 iff L has a non-degenerate eigenspace E_{λ} (defined on a dense subset of \mathbb{E}_{ν}^{n}) with constant eigenfunction λ and dimension m - 1.

In conclusion we have the following theorem which summarizes our classification:

Theorem 9.5.11 (Classification of Reducible CTs in \mathbb{E}^n_{ν})

Let L be a non-degenerate CT in \mathbb{E}_{ν}^{n} such that each real generalized eigenspace of A_{c} has at most one proper generalized eigenvector. Then L is reducible iff A_{c} has a multidimensional real eigenspace. If L is reducible, then there exists an explicitly constructible warped product decomposition $\psi : N_{0} \times_{\rho_{1}} N_{1} \cdots \times_{\rho_{k}} N_{k} \to \mathbb{E}_{\nu}^{n}$ such that the following hold:

- L is decomposable in the warped product $N_0 \times_{\rho_1} N_1 \cdots \times_{\rho_k} N_k$.
- The restriction of $(\psi^{-1})_*L$ to N_0 has no constant eigenfunctions.
- $\operatorname{Im}(\psi)$ is an open dense subset of \mathbb{E}^n_{ν} .

9.5.2 In Spherical submanifolds of pseudo-Euclidean space

In this section we will show how the problem of classifying reducible CTs in $\mathbb{E}_{\nu}^{n}(\kappa)$ can be reduced to the same problem in \mathbb{E}_{ν}^{n} . By using Theorem 8.4.7, we will show how to restrict a reducible CT in \mathbb{E}_{ν}^{n} to one in $\mathbb{E}_{\nu}^{n}(\kappa)$.

Proposition 9.5.12 (Restricting Reducible CTs to $\mathbb{E}^n_{\nu}(\kappa)$)

Let $\psi : N_0 \times_{\rho_1} N_1 \cdots \times_{\rho_k} N_k \to \mathbb{E}^n_{\nu}$ be a proper warped product decomposition in canonical form and let $\bar{p} \in \text{Im}(\psi)$ as in Theorem 8.4.7. Suppose L_c is a reducible central CT in \mathbb{E}^n_{ν} satisfying

$$L_c = \psi_* (\tilde{L}_c + \sum_{i=1}^k \lambda_i G_i)$$

where G_i is the restriction of G to TN_i , $\lambda_i \in \mathbb{R}$ and L_c is a CT in N_0 . Let $\phi := \psi|_{N'}$ be the induced warped product decomposition of $\mathbb{E}^n_{\nu}(\kappa)$ as in Theorem 8.4.7. Then if we let L(resp. \tilde{L}) be the restriction of L_c (resp. \tilde{L}_c) to $\mathbb{E}^n_{\nu}(\kappa)$ (resp. $N_0(\kappa)$), then

$$L = \phi_*(\tilde{L} + \sum_{i=1}^k \lambda_i G_i) \qquad \diamondsuit$$

PROOF Let \tilde{r} (resp. r) be the dilatational vector field in N_0 (resp. \mathbb{E}^n_{ν}). We will use the fact that $\psi_*\tilde{r} = r$; this can be deduced from the proof of Proposition 9.5.3 or Eq. (8.4.1). We let $R^* = I - \frac{r \otimes r^\flat}{r^2}$ be the orthogonal projection onto $T\mathbb{E}^n_{\nu}(\kappa)$ with a similar definition for \tilde{R}^* with respect to $TN_0(\kappa)$. In the following, given $L \in S^2(\mathbb{E}^n_{\nu})$, we denote by R^*L the restricted tensor given by $(R^*L)^{ij} = R^i_{\ l}L^{lk}R^j_{\ k}$.

Using the fact that ψ is an isometry and $\psi_* \tilde{r} = r$, one can show that $R^* \circ \psi_* = \psi_* \circ \tilde{R}^*$. Also note that $\tilde{R}^* G_i = G_i$. Thus

$$R^*L_c = R^*\psi_*(\tilde{L}_c + \sum_{i=1}^k \lambda_i G_i)$$
$$= \psi_*(\tilde{R}^*\tilde{L}_c + \sum_{i=1}^k \lambda_i \tilde{R}^*G_i)$$
$$= \psi_*(\tilde{R}^*\tilde{L}_c + \sum_{i=1}^k \lambda_i G_i)$$

 \diamond

By evaluating the above equation in $N_0(\kappa) \times_{\rho_1} N_1 \cdots \times_{\rho_k} N_k$, one obtains the desired result.

Now we show how to apply the above results to obtain a warped product decomposition in which a given CT in $\mathbb{E}_{\nu}^{n}(\kappa)$ is decomposable. Let L be a CT in $\mathbb{E}_{\nu}^{n}(\kappa)$, then there is a unique central CT, L_{c} , such that $L = R^{*}L_{c}$. As described in the previous section, provided L_{c} is reducible, we can choose a warped product decomposition of \mathbb{E}_{ν}^{n} , ψ , such that $L_{c} = \psi_{*}(\tilde{L}_{c} + \sum_{i=1}^{k} \lambda_{i}G_{i})$ satisfying the hypothesis of the above proposition. Thus the above proposition gives a warped product decomposition ϕ which decomposes L, and is obtained by an appropriate restriction of ψ . We now give some examples of this procedure to obtain the standard spherical coordinates.

Example 9.5.13 (Spherical Coordinates I)

Let $M = \mathbb{E}_{\nu}^{n}(\kappa)$ where $\kappa = \pm 1$ and $n \geq 3$. Consider the CT L in $\mathbb{E}_{\nu}^{n}(\kappa)$ induced by $A = \varepsilon e \odot e$ with $\varepsilon := e^{2} = \pm 1$. Let P be the orthogonal projector onto e^{\perp} and choose $\bar{p} \in \mathbb{E}_{\nu}^{n}(\kappa)$ such that $(P\bar{p})^{2} = \pm 1$. By Example 9.5.6 there is a warped product decomposition $\psi : N_{0} \times_{\rho} N_{1} \to \mathbb{E}_{\nu}^{n}$ passing through \bar{p} which decomposes $L_{c} := A + r \odot r$. For $(p_{0}, p) = (x\kappa_{1}a + ye, p) \in N_{0} \times N_{1}$, we have

$$\psi(p_0, p) = xp + ye$$

To obtain a warped product decomposition of $\mathbb{E}_{\nu}^{n}(\kappa)$, by Theorem D.6.5 we need to restrict ψ to $N_{0}(\kappa) \times N_{1}$. Let ϕ be the induced warped product decomposition of $\mathbb{E}_{\nu}^{n}(\kappa)$, then it follows by Proposition 9.5.12 that L is decomposable in this warped product. We now give the standard forms of this warped product by parameterizing (x, y) as in Example 9.4.20 while enforcing $x = \langle a, p_{0} \rangle > 0$ and $N_{0}(\kappa)$ to be connected. We have three different cases:

Case 1 $\kappa_1 = \kappa$ and $\varepsilon = \kappa$

$$\phi: \begin{cases} (0,\pi) \times_{\sin} N_1 & \to \mathbb{E}^n_{\nu}(\kappa) \\ (t,p) & \mapsto \sin(t)p + \cos(t)e \end{cases}$$

Case 2 $\kappa_1 = \kappa$ and $\varepsilon = -\kappa$

$$\phi: \begin{cases} \mathbb{R} \times_{\cosh} N_1 & \to \mathbb{E}^n_{\nu}(\kappa) \\ (t,p) & \mapsto \cosh(t)p + \sinh(t)e \end{cases}$$

Case 3 $\kappa_1 = -\kappa$ and $\varepsilon = \kappa$

$$\phi: \begin{cases} \mathbb{R}^+ \times_{\sinh} N_1 & \to \mathbb{E}^n_{\nu}(\kappa) \\ (t,p) & \mapsto \sinh(t)p + \cosh(t)e \end{cases}$$

Note that even though there is only one inequivalent coordinate system on $\mathbb{E}^2_{\nu}(\kappa)$, the last two warped products are inequivalent. This is due to the fact that $a^2 = \kappa_1$ is different in those cases and $N_0 = \{p \in V_0 | \langle a, p \rangle > 0\}$.

The following example is on spherical coordinates that only occur in non-Euclidean spheres.

Example 9.5.14 (Spherical Coordinates II)

Let $M = \mathbb{E}_{\nu}^{n}(\kappa)$ where $\kappa = \pm 1$ and $n \geq 3$. We now consider the CT L in $\mathbb{E}_{\nu}^{n}(\kappa)$ induced by $A = a \odot a$ with $a^{2} = 0$ and $a \neq 0$. This example proceeds similarly to the first. Fix $\bar{p} \in \mathbb{E}_{\nu}^{n}(\kappa)$ such that $\langle a, \bar{p} \rangle = 1$. By Example 9.5.7 there is a warped product decomposition $\psi : N_{0} \times_{\rho} N_{1} \to \mathbb{E}_{\nu}^{n}$ passing through \bar{p} which decomposes $L_{c} := A + r \odot r$. For $(p_{0}, p) = (xb + ya, p) \in N_{0} \times N_{1}$, we have

$$\psi(p_0, p) = x(b + p - \frac{1}{2}p^2a) + ya$$

Restricting ψ to $N_0(\kappa) \times N_1$ forces:

$$\kappa = p_0^2 = 2xy$$

Let ϕ be the warped product decomposition of $\mathbb{E}_{\nu}^{n}(\kappa)$ induced by ψ as in Theorem 8.4.7. Again, it follows by Proposition 9.5.12 that L is decomposable in this warped product. We now give ϕ with the standard parameterization of $N_{0}(\kappa)$, by enforcing $x = \langle a, p_{0} \rangle > 0$ and $N_{0}(\kappa)$ to be connected. These conditions are all satisfied if we take $x = \frac{1}{\sqrt{2}} \exp(t)$. Then we have the following:

$$\phi: \begin{cases} \mathbb{R} \times_{\frac{1}{\sqrt{2}} \exp} \mathbb{E}_{\nu-1}^{n-2} & \to \mathbb{E}_{\nu}^{n}(\kappa) \\ (t,p) & \mapsto \frac{1}{\sqrt{2}} \exp(t)(b+p-\frac{1}{2}p^{2}a) + \frac{\kappa}{\sqrt{2}} \exp(-t)a \end{cases}$$

Also note that if $\nu = -\kappa = 1$, then ϕ is an isometry onto a connected component of $\mathbb{E}_1^n(-1) \simeq H^{n-1}$.

In conclusion we have the following theorem which summarizes our classification:

Theorem 9.5.15 (Classification of Reducible CTs in $\mathbb{E}^n_{\nu}(\kappa)$)

Let L be a CT in $\mathbb{E}_{\nu}^{n}(\kappa)$ such that each real generalized eigenspace of A has at most one proper generalized eigenvector. Then L is reducible iff A has a multidimensional real eigenspace. If L is reducible, then there exists an explicitly constructible warped product decomposition $\psi : N_0 \times_{\rho_1} N_1 \cdots \times_{\rho_k} N_k \to \mathbb{E}_{\nu}^{n}(\kappa)$ such that the following hold:

- 1. L is decomposable in the warped product $N_0 \times_{\rho_1} N_1 \cdots \times_{\rho_k} N_k$.
- 2. The restriction of $(\psi^{-1})_*L$ to N_0 has no constant eigenfunctions.
- 3. Im(ψ) is an open dense subset of $\mathbb{E}^n_{\nu}(\kappa)$.

PROOF We give the proof of Item 2. First suppose λ is a constant eigenfunction of L, then one can naturally lift λ to a constant function on \mathbb{E}^n_{ν} . Let p(z) be the characteristic polynomial of L having the form given by Eq. (9.4.16). Then since $\mathcal{L}_r p = 0$ (see the proof of Proposition 9.4.21), we must have $p(\lambda) = 0$ on some open subset of \mathbb{E}^n_{ν} . Then the proof of Lemma 9.5.8 holds verbatim by Eq. (9.4.16), which proves the result.

 \diamond

Item 3 follows from the construction of ψ (see Proposition 9.5.12) and Theorem 9.5.11.

9.6 Applications and Examples

In this section we will show how to apply the theory developed in this chapter to solve some of the motivating problems stated in the introduction. First, in Section 9.6.1 we will show how to enumerate the isometrically inequivalent separable coordinates in a given space of constant curvature. Then in Section 9.6.2 we will show how to construct separable coordinate systems by way of examples. Finally, in Section 10.3 we will show how to explicitly execute the BEKM separation algorithm in general. We also give the details of executing the BEKM separation algorithm for the Calogero-Moser system.

9.6.1 Enumerating inequivalent separable coordinates

In this section we show how one can use the theory developed in this chapter to enumerate the isometrically inequivalent separable coordinate systems on a given space of constant curvature. For dimensions greater than two, this problem is recursive as described in Section 6.5. This recursive nature was originally discovered by Kalnins et al. and is discussed more concretely in [Kal86]. So one will also have to enumerate the separable coordinate systems on spherical submanifolds of the underlying space and then construct the separable coordinates systems using warped products (see the beginning of Section 9.1.3 and also Section 6.5).

The main step is to enumerate the geometrically inequivalent CTs, so we will focus on this. To do this, one has to enumerate the canonical forms summarized in Section 9.1.3 together with the metric-Jordan canonical forms for A_c and take into account geometric equivalence. We illustrate this idea with a few examples.

Example 9.6.1 (Central CTs)

Let L be a central CT with parameter matrix A. In this case, we essentially have to enumerate the different metric-Jordan canonical forms for A. Fix $\lambda_1 < \cdots < \lambda_n \in \mathbb{R}$.

In Euclidean space there is only one central CT we can build from these parameters, it is given by the parameter matrix $A = \text{diag}(\lambda_1, \ldots, \lambda_n)$ and it induces the well known elliptic coordinate system (see Example 9.4.11).

In Minkowski space there are n (geometrically inequivalent) central CTs we can build from these parameters, they are given as follows:

$$A = J_{-1}(\lambda_1) \oplus J_1(\lambda_2) \oplus \cdots J_1(\lambda_n)$$

$$\vdots$$

$$A = J_1(\lambda_1) \oplus J_1(\lambda_2) \oplus \cdots J_{-1}(\lambda_n)$$

They differ by the eigenvalue of A which is timelike. Similarly there are n-1 central CTs built only using $\lambda_2 < \cdots < \lambda_n$ with parameter matrix of the form:

$$A = J_{\pm 2}(\lambda_2) \oplus J_1(\lambda_3) \oplus \cdots J_1(\lambda_n)$$

Now consider the case where A has a two dimensional eigenspace, the rest being simple. Using $\lambda_2 < \cdots < \lambda_n$, in Euclidean space there are n-1 central CTs depending on which λ_i corresponds to the two dimensional eigenspace⁷. Each of these cases in Euclidean space induce n-1 different cases in Minkowski space depending on which λ_i becomes timelike, hence there are a total of $(n-1)^2$ cases in Minkowski space.

Finally we note that in Minkowski space A can have two complex conjugate eigenvalues, then since the corresponding real Jordan block is distinguishable from the other real eigenvalues of A, a similar analysis applies. In general one would have to order the complex eigenvalues (see Definition E.0.9).

Enumerating inequivalent axial CTs can largely be reduced to the same problem for central CTs. For example, in Euclidean space there is only one type of axial CT if all the eigenvalues of A_c are distinct. We end with CTs in spherical submanifolds of pseudo-Euclidean space as these are somewhat different.

Example 9.6.2 (CTs in $\mathbb{E}^n_{\nu}(\kappa)$)

Let *L* be the CT in $\mathbb{E}_{\nu}^{n}(\kappa)$ with parameter matrix *A*. Fix $\lambda_{1} < \cdots < \lambda_{n} \in \mathbb{R}$. In this case there are sometimes less geometrically inequivalent CTs then isometrically inequivalent ones.

⁷When n = 3 the two different cases induce the oblate and prolate spheroidal coordinate systems.

In the Euclidean sphere there is only one CT we can build from these parameters, it is given by the parameter matrix $A = \text{diag}(\lambda_1, \ldots, \lambda_n)$ and it induces the sphere-elliptic coordinate system.

Now suppose the ambient space is Minkowski space. Then we only need to consider $\left\lceil \frac{n}{2} \right\rceil$ cases given by (see Example 9.3.6):

$$A = J_{-1}(\lambda_1) \oplus J_1(\lambda_2) \oplus \cdots J_1(\lambda_n)$$

$$\vdots$$

$$A = J_1(\lambda_1) \oplus J_1(\lambda_2) \oplus \cdots \oplus J_{-1}(\lambda_{\lceil \frac{n}{2} \rceil}) \oplus \cdots J_1(\lambda_n)$$

Note that only the first $\left\lceil \frac{n}{2} \right\rceil$ eigenvalues of A are made timelike.

Most of the other cases can be deduced from the first example if one desires. Although we illustrate one difference with an example. For the Euclidean sphere $\mathbb{E}^3(1)$, fix $\lambda_1 < \lambda_2 \in \mathbb{R}$ and consider the CT induced by the following parameter matrices:

$$A_1 = \operatorname{diag}(\lambda_1, \lambda_1, \lambda_2)$$
$$A_2 = \operatorname{diag}(\lambda_1, \lambda_2, \lambda_2)$$

Note that $-A_2$ has the same form as A_1 , specifically the smallest eigenvalue of $-A_2$ is repeated. Hence in considering parameter matrices with two dimensional eigenspaces, we only need to enumerate those with the form given by A_1 , where the smaller eigenvalue is repeated.

We have described how to enumerate the geometrically inequivalent CTs in spaces of constant curvature. One should note though, that in non-Euclidean spaces a given CT could induce different coordinate systems on disjoint connected subsets of the space (see Example 9.4.12). Hence in these cases, more work has to be done to enumerate the isometrically inequivalent separable coordinate systems.

9.6.2 Constructing separable coordinates

In a two dimensional Riemannian manifold, all non-trivial CTs are Benenti tensors. Hence in this case, one can enumerate all isometrically inequivalent separable coordinates simply by enumerating the geometrically inequivalent CTs. The latter problem can be solved in pseudo-Euclidean space using Theorem 9.1.8. In Table 9.1 we have done this for \mathbb{E}^2 and included the standard transformations from separable to Cartesian coordinates.

1. Cartesian coordinates	$L = d \odot d$	x d + y e
2. Polar coordinates	$L = r \odot r$	$\rho\cos\theta d + \rho\sin\theta e$
3. Elliptic coordinates	$L = d \odot d + a^{-2}r \odot r$	$a\cos\phi\cosh\etad + a\sin\phi\sinh\etae$
4. Parabolic coordinates	$L = 2r \odot d$	$\frac{1}{2}(\mu^2 - \nu^2) d + \mu\nu e$
\square		

Table 9.1: Separable Coordinate Systems in \mathbb{E}^2

The vectors d, e form an orthonormal basis for \mathbb{E}^2 and a > 0.

We now show how one obtains the coordinate formula in Table 9.1 from formulas we have already calculated. For elliptic coordinates, take Cartesian coordinates (x, y) on \mathbb{E}^2 and let L be the central CT with parameter matrix $A = \text{diag}(\lambda_1, \lambda_2)$ where $\lambda_2 > \lambda_1$. Then the transformation from canonical coordinates (u^1, u^2) to Cartesian coordinates (x, y) read (see Eq. (9.4.5)):

$$x^{2} = \frac{(\lambda_{1} - u^{1})(\lambda_{1} - u^{2})}{(\lambda_{2} - \lambda_{1})} \qquad \qquad y^{2} = \frac{(\lambda_{2} - u^{1})(\lambda_{2} - u^{2})}{(\lambda_{1} - \lambda_{2})}$$

We can obtain the standard parameterization of elliptic coordinates as follows. Note that $L = \lambda_1 G + (\lambda_2 - \lambda_1) \tilde{L}$ where $\tilde{L} = e \odot e + (\lambda_2 - \lambda_1)^{-1} r \odot r$ is geometrically equivalent to L. The eigenfunctions of \tilde{L} , $(\tilde{u}^1, \tilde{u}^2)$, are related to those of L by $u^i = \lambda_1 + (\lambda_2 - \lambda_1) \tilde{u}^i$. Letting $a^2 := \lambda_2 - \lambda_1$ and substituting this expression for u^i in the above equation gives:

$$x^{2} = a^{2}\tilde{u}^{1}\tilde{u}^{2} \qquad \qquad y^{2} = a^{2}(1 - \tilde{u}^{1})(\tilde{u}^{2} - 1)$$

Then making the transformation $\tilde{u}^1 = \cos^2 \phi$ and $\tilde{u}^2 = \cosh^2 \eta$, we obtain the formula in Table 9.1.

The formula for parabolic coordinates follow similarly from Eqs. (9.4.14) and (9.4.15), after taking $u^1 = -\nu^2$ and $u^2 = \mu^2$ assuming $u^1 < u^2$.

We end with a few more examples to further illustrate the theory. The first example shows how to obtain coordinates which diagonalize a Benenti tensor which is not an ICT.

Example 9.6.3 (Spherical coordinates in \mathbb{S}^2)

Fix $d \in \mathbb{S}^2$ and let L be the CT induced in \mathbb{S}^2 by restricting $d \odot d$. As we observed earlier, L is necessarily a Benenti tensor. In Example 9.5.13 it was shown that a warped product which decomposes L is given by:

$$\psi(\phi, p) = \cos \phi \, d + \sin \phi \, p$$

where $p \in d^{\perp}(1)$, i.e. $p \in \mathbb{S}^2 \cap d^{\perp}$ and $\phi \in (0, \pi)$. Since $d^{\perp}(1)$ is the unit circle we obtain coordinates on it by taking $p = \cos \theta \, e + \sin \theta \, f$ where e, f is an orthonormal basis for d^{\perp} . Then the above equation becomes:

$$\psi(\phi, p) = \cos \phi \, d + \sin \phi(\cos \theta \, e + \sin \theta \, f)$$

Furthermore, since ψ is a warped product decomposition with warping function $\sin \phi$, it follows from Example 9.5.13 that the metric is:

$$g = (\mathrm{d}\phi)^2 + \sin^2\phi(\mathrm{d}\theta)^2$$

Example 9.6.4 (Oblate/Prolate spheroidal coordinates in \mathbb{E}^3) Fix a unit vector $d \in \mathbb{E}^n$, $c \neq 0$ and consider the following CT in \mathbb{E}^n :

$$L = c \ d \odot d + r \odot r \tag{9.6.1}$$

It follows from Example 9.5.6 that a warped product ψ which decomposes L is given as follows: Let $e \in d^{\perp}$ be a unit vector, then for $(p_0, p) = (xd + ye, p) \in N_0 \times N_1$

$$\psi(p_0, p) = xd + yp$$

Observe that $N_0 \simeq \mathbb{E}^2$ and L induces a Benenti tensor, \tilde{L} , on N_0 which has the form given by Eq. (9.6.1). If we let $a := \sqrt{|c|}$, then using Table 9.1 we can take coordinates on N_0 which diagonalize \tilde{L} yielding the following maps.

$$\psi(p_0, p) = \begin{cases} c > 0 & a \cos \phi \cosh \eta \, d + a \sin \phi \sinh \eta \, p \\ c < 0 & a \sin \phi \sinh \eta \, d + a \cos \phi \cosh \eta \, p \end{cases}$$

Also N_1 is the unit sphere in d^{\perp} , hence $N_1 \simeq \mathbb{S}^{n-2}$. We can obtain separable coordinates for \mathbb{E}^n by taking any separable coordinates for \mathbb{S}^{n-2} on N_1 (see Section 6.5). For example, if c > 0 and n = 3, we obtain prolate spheroidal coordinates:

$$\psi(p_0, p) = a \cos \phi \cosh \eta \, d + a \sin \phi \sinh \eta \, (\cos \theta \, e + \sin \theta \, f)$$

where e, f is any orthonormal basis for d^{\perp} . Also note that using Proposition 9.4.15 and the fact that ψ is a warped product decomposition with warping function $a \sin \phi \sinh \eta$, one can obtain the following expression for the metric:

$$g = a^{2}(\sinh^{2}\eta + \sin^{2}\phi)((d\phi)^{2} + (d\eta)^{2}) + a^{2}\sin^{2}\phi\sinh^{2}\eta(d\theta)^{2}$$

Finally note that oblate spheroidal coordinates can be obtained by taking c < 0. Example 9.6.5 (Product coordinates in \mathbb{E}^4)

Consider the decomposition $\mathbb{E}^n = V \oplus W$ into non-trivial subspaces. Let \tilde{G} denote the induced contravariant metric in V and consider the following CT in \mathbb{E}^n :

$$L = \tilde{G}$$

Observe that the warped product $\psi: V \times_1 W \to \mathbb{E}^n$ given by $(q, p) \to q + p$ is adapted to the eigenspaces of L. We can construct separable coordinates by parameterizing q(resp. p) with separable coordinates on V (resp. W). For example, if dim $V = \dim W = 2$, by taking polar (resp. elliptic) coordinates on V (resp. W) from Table 9.1, we have the following separable coordinates on \mathbb{E}^4 :

$$\psi(q, p) = \rho \cos \theta \, b + \rho \sin \theta \, c + a \cos \phi \cosh \eta \, d + a \sin \phi \sinh \eta \, e^{-\frac{1}{2}}$$

where b, c (resp. d, e) is an orthonormal basis for V (resp. W).

In conclusion, as an exercise, we recommend the reader prove that there are eleven classes of isometrically inequivalent separable coordinate systems in \mathbb{E}^3 .

9.7 Notes

A classification of CTs modulo the action of the isometry group in Euclidean space can be found in [Lun03] (cf. [Ben05]). A complete classification of these tensors for Euclidean space and the Euclidean sphere is implicit in [WW03].

Different parts of this problem have been solved for special cases by different researchers over the past few decades. A classification of separable coordinate systems in Riemannian spaces of constant curvature was originally done by Kalnins and Miller in [KM86; KM82], see also [Kal86] which is a book containing their results. Their solution primarily involves degenerating a general coordinate system (elliptic coordinates for \mathbb{E}^n) by taking limits of certain parameters appearing in the coordinate system. The insight provided by their classification was crucial for the development of the theory presented in this thesis. They have extended this work to spaces of constant curvature with arbitrary signature in [KMR84] to obtain a partial classification.

In [Kal75] orthogonal separable coordinates in two dimensional Minkowski space have been classified and those in three dimensional Minkowski space have been partially classified. A more detailed classification of a more general class of orthogonal separable coordinates in three dimensional Minkowski space has been given in [KM76]. This classification has been further refined in [Hin98] (cf. [HM08]). A classification of orthogonal separable coordinates for four dimensional Minkowski space has been given in [KM78] and references therein. Finally, building on results in [Kal86], a version of the BEKM separation algorithm has been given in [WW03] for Euclidean space and the Euclidean sphere.

Our approach to this problem has several advantages over previous approaches. First we gave a unified theory applicable to spaces of constant curvature with both Euclidean and Lorentzian signatures. This approach allows one to solve the different but related problems listed in the introduction. We gave a precise notion of in-equivalence for orthogonal separable coordinate systems in Minkowski space and thereby gave a clear, rigorous and complete classification in this space. The main drawback of our approach is that it is theoretical and not as easy to apply for those who wish to.

Chapter 10

Separation of Natural Hamiltonians

In this chapter we will use the theory developed so far to answer the following questions: Are there any special separable potentials one can construct for a given separable web, especially a KEM web? We will also answer the converse question for KEM webs (equivalently separable webs) in spaces of constant curvature, i.e. what are the separable webs (if any) in which a given potential separates? In other words, we will solve problems (2) and (3) (from the introduction) for spaces of constant curvature. Note that the answer to the first question is known in general (see Eq. (5.4.2)) in separable coordinates. The difference is that we give special potentials which have coordinate-independent formulas. The answer to the second question will involve working out the details of the BEKM separation algorithm which was introduced in Section 6.7.

An answer to the first question will be given in the first section. In the following section, we will answer the second question for a specific potential, namely the Calogero-Moser system. Finally, in the last section, we will answer the second question in the general case.

In order to execute the BEKM separation algorithm in \mathbb{E}_{ν}^{n} , we will need the KBD equation in \mathbb{E}_{ν}^{n} and in $\mathbb{E}_{\nu}^{n}(\kappa)$. Fix a function $V \in \mathcal{F}(\mathbb{E}_{\nu}^{n})$ and suppose n > 1. Then if L is the general CT in \mathbb{E}_{ν}^{n} given by Eq. (9.1.1) and $K_{e} := \operatorname{tr}(L)G - L$ is its KBDT, then the KBD equation in \mathbb{E}_{ν}^{n} is:

$$\mathrm{d}(K_e\mathrm{d}V) = 0$$

We will often refer to the above equation as just the *KBD equation*. It will be convenient to evaluate the KBD equation in $\mathbb{E}_{\nu}^{n}(\kappa)$ via its embedding in \mathbb{E}_{ν}^{n} . Then if \tilde{L} is the general CT in $\mathbb{E}_{\nu}^{n}(\kappa)$ given in \mathbb{E}_{ν}^{n} by Eq. (9.1.5), let $L := r^{2}\tilde{L}$ and $K_{s} := \operatorname{tr}(L)R - L$, then the KBD equation in $\mathbb{E}_{\nu}^{n}(\kappa)$ (embedded in \mathbb{E}_{ν}^{n}) is:

$$\mathbf{d}(K_s \mathbf{d}V) = 0 \tag{10.0.1}$$

We will often refer to the above equation as the spherical KBD equation. We will show

how this equation is derived in Section 10.3.1.

We should also mention here that we carry out the BEKM separation algorithm slightly differently than described in Section 6.7. We construct warped products which decompose reducible OCTs such that the induced CT on the geodesic factor is an ICT as opposed to a Benenti tensor. This allows one to simultaneously construct separable coordinates while carrying out the algorithm, as illustrated by Section 10.2.

10.1 Known Separable Potentials

In this section we will use some notation introduced in Appendix A. Our first method for constructing separable potentials comes from a reinterpretation of Proposition 6.3.7 in terms of separating potentials. Suppose $L \in S^2(M)$ is an OCT and $M \times_{\rho} F$ is a warped product with $\kappa := \rho^{-2}$ and adpated contravariant metric $G = G' + \tilde{G}$. From Proposition 6.3.7, it follows that we can extend L to an OCT on $M \times_{\rho} F$ iff there exists $t \in \mathbb{R}$ such that

$$L(d\kappa) + \kappa d \operatorname{tr}(L) = d(t\kappa)$$

It then follows from Proposition 5.3.10 that the L-sequence generated by L can be extended to KTs on $M \times_{\rho} F$. This has already been done in Proposition 6.6.3. It was shown that if K_a are the elements of the L-sequence generated by L, then they admit the following extensions

$$\bar{K}_a := K_a + \left(\sum_{i=0}^a (-t)^i \sigma_{a-i}\right) \tilde{G}$$

where $\sigma_a := (\wedge^n L^a)^{\wedge *}$ (see Appendix A). If we let $V_a := (\sum_{i=0}^a (-t)^i \sigma_{a-i})\kappa$, then it follows from Proposition 4.5.2 that they satisfy:

$$\mathrm{d}V_a = K_a \mathrm{d}\kappa$$

If L is a Benenti tensor, then the above equation implies that the potential κ is separable in the web associated with L. Generalizing these observations, we have the following:

Proposition 10.1.1 (Constructing separable potentials I)

Let $L \in S^2(M)$ is CT. Suppose $\kappa \in \mathcal{F}(M)$ satisfies the following equation

$$L(d\kappa) + \kappa d\operatorname{tr}(L) = d(t\kappa) \tag{10.1.1}$$

for some $t \in \mathcal{F}(M)$. Then the functions

$$V_a := \left(\sum_{i=0}^{a} (-t)^i \sigma_{a-i}\right) \kappa$$

for a = 0, ..., n - 1 satisfy $dV_a = K_a d\kappa$ iff t satisfies

$$L dt = t dt$$
 \diamondsuit

PROOF The proof is a straightforward generalization of Proposition 6.6.3 together with the above observations.

We note that the left hand side of Eq. (10.1.1) is a closed form iff κ satisfies the KBD equation with L. Hence one can check if a given potential satisfies the hypothesis of the above proposition while executing the BEKM separation algorithm.

As a corollary of the proof, we observe the following [Ben04]:

Corollary 10.1.2

Suppose $L \in S^2(M)$ is CT. If K_a are elements of the L-sequence for a = 0, ..., n - 1, then the following hold

$$d\sigma_{a+1} = K_a d\operatorname{tr}(L)$$

In particular, if L is an OCT then the potential tr(L) is separable in any KEM web defined by L.

PROOF See Proposition B.1.2.

We have another corollary, which is an application of the above proposition to spaces of constant curvature.

Corollary 10.1.3

Suppose $L = A + mr \odot r + 2w \odot r$ is a CT in \mathbb{E}_{ν}^{n} and let \tilde{L} be the restriction of L to $\mathbb{E}_{\nu}^{n}(\kappa)$. Let a be a covariantly constant vector and let $V := \langle r, a \rangle^{-2}$. If a is an eigenvector of A orthogonal to w then V satisfies the KBD equation with L in \mathbb{E}_{ν}^{n} . If a is an eigenvector of A then the restriction of V to $\mathbb{E}_{\nu}^{n}(\kappa)$ satisfies the KBD equation with \tilde{L} in $\mathbb{E}_{\nu}^{n}(\kappa)$.

PROOF We first consider the case in \mathbb{E}^n_{ν} . Under these hypothesis it follows by Lemma 9.5.2 that if $\rho := |\langle r, a \rangle|$, then we have:

$$L(d \log \rho) = d(\lambda \log \rho + \frac{1}{2} \operatorname{tr}(L))$$

for some $\lambda \in \mathbb{R}$. The result then follows from Proposition 10.1.1. A similar proof holds for the case in $\mathbb{E}_{\nu}^{n}(\kappa)$, but now the above equation with \tilde{L} follows either by restriction of the one in the ambient space or by Proposition 9.5.12 together with Eq. (6.3.3) from Proposition 6.3.7. One can naturally construct separable potentials from the above proposition, as the following example shows.

Example 10.1.4 (Constructing separable potentials I)

Suppose a_1, \ldots, a_n is an orthonormal basis for \mathbb{E}^n_{ν} , then the above corollary implies that the following potential is separable in generalized elliptic coordinates (see Example 9.4.11):

$$V = \sum_{i} k_i \langle r, a_i \rangle^{-2}$$

for some $k_i \in \mathbb{R}$. In fact this potential is clearly multi-separable. Furthermore we can also obtain a multi-separable potential on $\mathbb{E}^n_{\nu}(\kappa)$ by restriction. For these potentials the commuting first integrals guaranteed by the Jacobi theorem (see Corollary 5.4.2) can be explicitly calculated. Indeed, this follows by Theorem 6.6.2 together with Proposition 10.1.1.

There are some additional separable potentials that can be constructed for KEM webs. These potentials arise from a different approach to the theory of separation, see [Bla05, Section 3.3] for the original derivation. They have also been derived using yet another approach in [Lun03]. See [Lun01] for a review.

Let L be a Benenti tensor. Fix a potential $V_1 \in \mathcal{F}(M)$. By Theorem 5.4.1 and Theorem 6.6.2, it follows that V_1 is separable in the web induced by L iff there exist functions $V_1, \ldots, V_n \in \mathcal{F}(M)$ satisfying

$$dV_{a+1} = K_a dV_1 \quad a = 0, \dots, n-1$$
(10.1.2)

Expanding K_a in terms of its recursive definition (see Eq. (A.0.2)), we get:

$$dV_{a+1} = K_a dV_1$$

= $\sigma_a dV_1 - K_{a-1}L dV_1$
= $\sigma_a dV_1 - LK_{a-1}dV_1$
= $\sigma_a dV_1 - L dV_a$

where we have used the fact that each K_a commutes with L since the K_a are polynomials in L. Hence we have that

$$dV_{a+1} = \sigma_a dV_1 - L dV_a \quad a = 0, \dots, n-1$$
(10.1.3)

A straightforward calculation shows that the above equation is equivalent to Eq. (10.1.2). Now suppose we have functions $V_1, \ldots, V_n \in \mathcal{F}(M)$ satisfying the above equation, i.e. they form a *separable chain*. We are interested in creating new functions $\bar{V}_1, \ldots, \bar{V}_n \in \mathcal{F}(M)$ which form a separable chain. The following proposition does just this.

Proposition 10.1.5 (Constructing separable potentials II [Bla03; Bla05])

Suppose L is an OCT and V_a form a separable chain with respect to L, i.e. they satisfy Eq. (10.1.3). Then the chain \bar{V}_a defined as follows:

$$\bar{V}_a := V_{a+1} - \sigma_a V_1 \tag{10.1.4}$$

is separable. Furthermore the following chain, which is an "inverse" of the first is also separable:

$$\bar{V}_a := V_{a-1} - \frac{\sigma_{a-1}}{\sigma_n} V_n \qquad \bar{V}_1 := -\frac{V_n}{\sigma_n} \qquad (10.1.5)$$

where $\sigma_a := (\wedge^n L^a)^{\wedge *}$.

PROOF We will only prove that the first set of functions form a separable chain. Suppose V_a form a separable chain and define \bar{V}_a by Eq. (10.1.4). Following Blaszak in [Bla03, section 4.3], we show that \bar{V}_a form a separable chain:

$$\begin{split} L\mathrm{d}\bar{V}_{a} &- \sigma_{a}\mathrm{d}\bar{V}_{1} = L\mathrm{d}(V_{a+1} - \sigma_{a}V_{1}) - \sigma_{a}\mathrm{d}(V_{2} - \sigma_{1}V_{1}) \\ &= L\mathrm{d}V_{a+1} - \sigma_{a}L\mathrm{d}V_{1} - V_{1}L\mathrm{d}\sigma_{a} - \sigma_{a}\mathrm{d}V_{2} + \sigma_{a}\sigma_{1}\mathrm{d}V_{1} + \sigma_{a}V_{1}\mathrm{d}\sigma_{1} \\ &= L\mathrm{d}V_{a+1} - \sigma_{a}L\mathrm{d}V_{1} - V_{1}(L\mathrm{d}\sigma_{a} - \sigma_{a}\mathrm{d}\sigma_{1}) + \sigma_{a}(\sigma_{1}\mathrm{d}V_{1} - \mathrm{d}V_{2}) \\ &= L\mathrm{d}V_{a+1} - \sigma_{a}L\mathrm{d}V_{1} + V_{1}\mathrm{d}\sigma_{a+1} + \sigma_{a}L\mathrm{d}V_{1} \\ &= L\mathrm{d}V_{a+1} + V_{1}\mathrm{d}\sigma_{a+1} \\ &= -\mathrm{d}V_{a+2} + \sigma_{a+1}\mathrm{d}V_{1} + V_{1}\mathrm{d}\sigma_{a+1} \\ &= -\mathrm{d}(V_{a+2} - \sigma_{a+1}V_{1}) \\ &= -\mathrm{d}(\bar{V}_{a+1}) \end{split}$$

Which proves the result.

We give some applications of the above proposition to spaces of constant curvature in the following example.

Example 10.1.6 (Constructing separable potentials II)

Consider the central CT, $L = A + r \odot r$, in \mathbb{E}^n_{ν} where $A = \text{diag}(\lambda_1, \ldots, \lambda_n)$. Fix Cartesian coordinates (x^i) . Assuming each $\lambda_i \neq 0$, by Lemma 9.4.6, we see that

$$\det L = \sigma_n = (\prod_{i=1}^n \lambda_i)(1 + \sum_{i=1}^n \frac{x_i x^i}{\lambda_i})$$

The second chain given by Eq. (10.1.5) applied to the constant potential gives the following separable potential:

$$V = \frac{1}{1 + \sum_{i=1}^{n} \frac{x_i x^i}{\lambda_i}}$$

Similarly if we let \hat{L} be the induced CT in $\mathbb{E}^n_{\nu}(\kappa)$, then by Eq. (9.4.16) we obtain the following separable potential:

$$V = \frac{1}{\sum_{i=1}^{n} \frac{x_i x^i}{\lambda_i}}$$

10.2 Example: Calogero-Moser system

We first present an example which separates in several different coordinate systems and hence provides a good example for the BEKM separation algorithm. Our example is the Calogero-Moser system, which will be defined shortly. Another advantage of this example is that its separability properties have been studied by several different authors [HMS05; WW05; WW03; BCR00; Cal69], hence it allows one to compare and contrast different methods. Finally we mention that we obtained this example from [WW03] where an algorithm equivalent to the BEKM separation algorithm was used to study this example.

The *n*-dimensional Calogero-Moser system is given by the following natural Hamiltonian [Cal08]:

$$H(p,q) = \frac{1}{2} \sum_{i=1}^{n} \left(p_i^2 + \omega^2 q_i^2 \right) + \sum_{1 \le i < j \le n} \frac{g^2}{(q_i - q_j)^2}$$
(CM)

We will take $\omega = 0, g = 1$ for convenience. In this case this Hamiltonian models n point particles moving on a line acted on by forces depending on their relative distances. We can write the potential V as follows:

$$V = \sum_{i} \left\langle r, a_i \right\rangle^{-2}$$

where $a_i = e_k - e_l$ for some $k, l \in \{1, \ldots, n\}$ with $e_i := \partial_i$. Furthermore we let

$$d = \frac{1}{\sqrt{n}} \sum_{i=1}^{n} e_i$$

The following proposition gives the general solution to the KBD equation for the Calogero-Moser system.

Proposition 10.2.1

If V is the potential of the Calogero-Moser system given by Eq. (CM), then the general solution to the KBD equation is:

$$L = c d \odot d + 2w d \odot r + m r \odot r \tag{10.2.1}$$

where $c, w, m \in \mathbb{R}$. Furthermore the restriction of the above CT to \mathbb{S}^{n-1} is a solution of the spherical KBD equation.

PROOF We will first apply Corollary 10.1.3 to show that the above CT is in fact a solution to the KBD equation. Consider the vectors $b_i := e_1 - e_i$ for $i \neq 1$. We will construct the most general CT, $L = A + mr \odot r + 2w \odot r$, for which each vector b_i is an eigenvector of A and orthogonal to w. Observe that none of the b_i are pair-wise orthogonal, they span an n-1 dimensional subspace, and

$$\bigcap_i b_i^{\perp} = (\bigoplus_i \operatorname{span}\{b_i\})^{\perp} = \operatorname{span}\{d\}$$

Now suppose A is a self-adjoint operator such that each b_i is an eigenvector of A. Then it follows that A must have d^{\perp} as an eigenspace, hence $A = kI + cd \odot d$ for some $k, c \in \mathbb{R}$. Thus up to equivalence the above form of L (Eq. (10.2.1)) satisfies our requirements, and it follows by Corollary 10.1.3 that L satisfies the KBD equation with V.

The second statement on the spherical KBD equation follows by a similar argument using Corollary 10.1.3.

We will now show that Eq. (10.2.1) is in fact the most general solution to the KBD equation. Suppose L is the general CT in \mathbb{E}^n given by Eq. (9.1.1). For $\kappa := \rho^{-2}$ where $\rho := (\langle r, a \rangle)^{-2}$, we have from the proof of Lemma 9.5.2 that

$$L(\mathrm{d}\kappa) + \kappa \mathrm{d}\operatorname{tr}(L) = -2\rho^{-3}(Aa + \langle a, w \rangle r)$$

Also recall that:

$$d \log \rho = \frac{a}{\rho}$$
$$\Rightarrow d\rho^{-3} = -3\rho^{-4}a$$

Hence

$$d(L(d\kappa) + \kappa d \operatorname{tr}(L)) = -2d\rho^{-3} \wedge (Aa + \langle a, w \rangle r)$$
$$= 6\rho^{-4}[a \wedge (Aa + \langle a, w \rangle r)]$$

Thus,

$$\frac{1}{6} \sum_{i=1}^{n} c_i d(L(d\kappa_i) + \kappa_i d\operatorname{tr}(L)) = \sum_{i=1}^{n} c_i \rho_i^{-4} [a_i \wedge (Aa_i + \langle a_i, w \rangle r)]$$
$$= \sum_{i=1}^{n} c_i \rho_i^{-4} a_i \wedge Aa_i + \sum_{i=1}^{n} c_i \rho_i^{-4} a_i \wedge \langle a_i, w \rangle r$$
$$= \sum_{i=1}^{n} c_i \rho_i^{-4} a_i \wedge Aa_i + \sum_{i=1}^{n} (c_i \rho_i^{-4} \langle a_i, w \rangle a_i) \wedge r$$

Denote the above 2-form by ω , then note that for $x \in \mathfrak{X}(\mathbb{E}^n)$ that $\mathcal{L}_x \omega = d\omega(x)$. Let $\alpha = 6\rho^{-4}[a \wedge (Aa + \langle a, w \rangle r)]$, then

$$\alpha(x) = 6\rho^{-4}[\langle a, x \rangle (Aa + \langle a, w \rangle r) - \langle x, Aa + \langle a, w \rangle r \rangle a]$$

Suppose x is constant, then

$$d\alpha(x) = 6 \langle a, x \rangle d\rho^{-4} \wedge (Aa + \langle a, w \rangle r) - 6\rho^{-4} \langle a, w \rangle x \wedge a$$
$$= 6 \langle a, x \rangle (-4\rho^{-5})a \wedge (Aa + \langle a, w \rangle r) - 6\rho^{-4} \langle a, w \rangle x \wedge a$$

If we take x = d, we see that

$$\mathcal{L}_{d}\omega = -6\sum_{i} c_{i}\rho_{i}^{-4} \langle a_{i}, w \rangle d \wedge a_{i}$$
$$= 6(\sum_{i} c_{i}\rho_{i}^{-4} \langle a_{i}, w \rangle a_{i}) \wedge d$$

Hence $\omega \equiv 0$ iff $(\sum_{i} c_i \rho_i^{-4} \langle a_i, w \rangle a_i) \in \text{span}\{d\}$ iff

$$\sum_{i} c_i \rho_i^{-4} \langle a_i, w \rangle \, a_i = 0$$

If we differentiate the above equation with respect to to e_j , we get

$$\sum_{i} c_{i} \rho_{i}^{-5} \langle a_{i}, e_{j} \rangle \langle a_{i}, w \rangle a_{i} = 0$$

Since the a_i satisfying $\langle a_i, e_j \rangle \neq 0$ are linearly independent, the above equation implies that $\langle a_i, w \rangle = 0$ for each of these a_i . Thus we see that $\langle a_i, w \rangle = 0$ for each *i* is a necessary condition for separability. Now we are left to solve the following equation:

$$\sum_{i} c_i \rho_i^{-4} a_i \wedge A a_i = 0$$

By differentiating the above equation with respect to to e_j , we get

$$\sum_{i} c_i \rho_i^{-5} \langle a_i, e_j \rangle \, a_i \wedge A a_i = 0$$

Now differentiate the above with respect to to e_k to get the following:

$$\sum_{i} c_{i} \rho_{i}^{-6} \langle a_{i}, e_{k} \rangle \langle a_{i}, e_{j} \rangle a_{i} \wedge A a_{i} = 0$$

The above sum has precisely one term, it shows that $a_i \wedge Aa_i = 0$. Thus a necessary condition for separability is that for each i, a_i is an eigenvector of A.

In conclusion we see that a necessary and sufficient condition for separability is that for each i, a_i is an eigenvector of A satisfying $\langle a_i, w \rangle = 0$. This confirms that Eq. (10.2.1) is in fact the most general solution of the KBD equation by the preceding calculations.

Remark 10.2.2

When n = 3 one can check that the solution to the spherical KBD equation given in the above proposition is the most general.

Canonical forms We now obtain the canonical forms according to Theorem 9.1.8 for the CTs given by Eq. (10.2.1). First the constants ω_i from Eq. (9.1.2) are given as follows:

$$\omega_0 = m$$
$$\omega_1 = w^2$$

Note that in Euclidean space, one only needs to calculate ω_0 and ω_1 to carry out the classification. We now break into the cases given by Theorem 9.1.8:

Case 1 Elliptic: $\omega_0 \neq 0$

By applying the translation given by Eq. (9.1.3) and changing to a geometrically equivalent CT one obtains:

$$L = cd \odot d + r \odot r \tag{10.2.2}$$

for some $c \in \mathbb{R}$.

Case 2 Parabolic: $\omega_0 = 0, \omega_1 \neq 0$

By applying the translation given by Eq. (9.1.4) and changing to a geometrically equivalent CT one obtains:

$$L = 2d \odot r \tag{10.2.3}$$

Case 3 Cartesian: $\omega_0 = 0, \omega_1 = 0, c \neq 0$

In this case after changing to a geometrically equivalent CT, we have:

$$L = d \odot d \tag{10.2.4}$$

Hence the three geometrically inequivalent solutions of the KBD equation for the Calogero-Moser potential are given by Eqs. (10.2.2) to (10.2.4). Note that we can obtain these CTs from Eq. (10.2.1) with an appropriate choice of parameters, hence there is no need to apply any isometries.

Determining Separability We now analyze these solutions further to find separable coordinates. We will obtain a compete analysis for the case $n \leq 3$ for convenience. For the following analysis, we fix unit vectors $a \in d^{\perp}$ and $e \in d^{\perp} \cap a^{\perp}$.

We define N_1 to be the unit sphere in d^{\perp} :

$$N_1 = \{ p \in d^\perp \mid p^2 = 1 \}$$

Note if $d^{\perp} = \mathbb{R}a$, then we take $N_1 = \{a\}$. When dim $N_1 = 1$, we take coordinates on it as follows:

$$\sigma(\theta) = \cos(\theta)a + \sin(\theta)e$$

Case 1 Elliptic with $c \neq 0$

When n > 2, this CT is reducible and a warped product decomposition ψ which decomposes this CT is given by Example 9.5.6. First define N_0 as follows:

$$N_0 = \{ p \in \mathbb{R}d \oplus \mathbb{R}a \mid \langle a, p \rangle > 0 \}$$

For $(p_0, p) = (xa + yd, p) \in N_0 \times N_1$, ψ is given as follows (see Example 9.5.6):

$$\psi(p_0, p) = xp + yd$$

Note that this equation also holds when n = 2, but in this case ψ is not a warped product decomposition. Now, to separate V, we have to apply the BEKM separation algorithm with V restricted to N_1 on N_1 . Although it will be more convenient to use the spherical KBD equation in d^{\perp} , see the next section for more details.

When $n \leq 3$, no additional steps are needed since in this case dim $N_1 \leq 1$. Indeed, by Example 9.4.11 *L* restricted to N_0 is an ICT (in a dense subset) hence *L* has simple eigenfunctions (locally), and so one obtains separable coordinates for *V* by taking elliptic coordinates on N_0 Section 6.7. When c < 0 we obtain oblate spheroidal coordinates and when c > 0 we obtain prolate spheroidal coordinates; see Example 9.6.4 for more details.

Case 2 Parabolic

When n > 2, then proceeding as in Example 9.5.6 (see also Section 9.5.1), one observes that the same warped product ψ as in the above case decomposes this CT. When $n \leq 3$, with similar arguments as in the above case, one finds that L locally has simple eigenfunctions, and one obtains separable coordinates for V by taking parabolic coordinates on N_0 Section 6.7. The resulting coordinate system is often called rotationally symmetric parabolic coordinates.

Case 3 Spherical: Elliptic with c = 0

In this case, one can check that the following warped product, ψ , decomposes L. For $(p_0, p) = (\rho a, p) \in \mathbb{R}^+ a \times \mathbb{S}^{n-1}$, ψ is given as follows:

$$\psi(p_0, p) = \rho p$$

Now observe that even when n = 3, L does not have simple eigenfunctions; in contrast with the previous two cases. To fill the multidimensional eigenspace of Lcorresponding to r^{\perp} , we have to solve the spherical KBD equation (see the next section for more details). Although when n = 3, we can fill this degeneracy by using the solution to the spherical KBD equation given by Proposition 10.2.1. Indeed, that proposition shows that the CT on \mathbb{S}^{n-1} induced by $d \odot d$ is a solution of the spherical KBD equation. Hence by Example 9.6.3, this induced CT is diagonalized in spherical coordinates, and we see that V separates in the following coordinates Section 6.7.

$$\psi(\rho a, p) = \rho(\sin(\phi)(\cos(\theta)a + \sin(\theta)e) + \cos(\phi)d)$$

Case 4 Cartesian

In this case we obtain a product which decomposes L as follows. First let $N_0 = \mathbb{R}d$ and $N_1 = d^{\perp}$, then for $(p_0, p) = (xd, p) \in N_0 \times N_1$, we have:

$$\psi(p_0, p) = xd + p$$

As in the above case, even when n = 3, L does not have simple eigenfunctions. Hence we have to apply the BEKM separation algorithm with V restricted to N_1 on N_1 . When n = 3 one finds that the general solution to the KBD equation is $\tilde{r} \odot \tilde{r}$ where \tilde{r} is the dilatational vector field in N_1 . Thus if we take polar coordinates in N_1 , we obtain separable coordinates for V. For $(p_0, p) = (xd, y\sigma(\theta)) \in N_0 \times N_1$ with y > 0, we have:

$$\psi(p_0, y\sigma(\theta)) = xd + y(\cos(\theta)a + \sin(\theta)e)$$

We conclude with some remarks. First the analysis given above is complete when $n \leq 3$. Although when n > 3 the warped product decompositions obtained may allow for partial separation of the Hamilton-Jacobi equation. When n = 4 it was shown in [WW05] that no additional solutions to the (spherical) KBD equation could be obtained. Hence our analysis above is complete when n = 4.

Furthermore the above analysis holds verbatim for the weighted Calogero-Moser system with unequal masses, which can be modeled using the natural Hamiltonian in \mathbb{E}^n associated with the following potential (see e.g. [WW05, Section 3.3]):

$$V = \sum_{1 \le i < j \le n} \frac{g_{ij}}{(m_i q_i - m_j q_j)^2}$$

The only difference is that in this case:

$$d = \frac{1}{\sqrt{M}} \sum_{i=1}^{n} \frac{e_i}{m_i}, \quad M = \sum_{i=1}^{n} \frac{1}{m_i^2}$$

More examples can be found in [WW03, section 7], where an algorithm equivalent to the BEKM separation algorithm is used to determine separability of some natural Hamiltonians defined in \mathbb{E}^3 . See also [Ben93] where some Kepler type potentials are tested for separability in elliptic coordinates in \mathbb{E}^2 .

10.3 The BEKM separation algorithm

In this section we show how to execute the BEKM separation algorithm (see Section 6.7 for the general theory) in spaces of constant curvature using the classification of CTs given in the previous chapter. This generalizes the example given in the previous section to arbitrary potentials.

10.3.1 Spherical KBD Equation

We first show how to derive the spherical KBD equation. Suppose $V \in \mathcal{F}(\mathbb{E}^n_{\nu})$ is a potential in \mathbb{E}^n_{ν} which satisfies the KBD equation with $r \odot r$. Choose $a \in \mathbb{E}^n_{\nu}$ with $\kappa := a^2 = \pm 1$ and let $\rho := \langle a, r \rangle$. Then we can easily construct a warped product $\psi : \mathbb{R}^+ a \times_{\rho} \mathbb{E}^n_{\nu}(\kappa) \to \mathbb{E}^n_{\nu}$ which decomposes this CT. Let $\tau : \mathbb{E}^n_{\nu}(\kappa) \to \mathbb{E}^n_{\nu}$ be the standard embedding of this sphere. Hence to find separable coordinates for V, we have to apply the BEKM separation algorithm with $\tilde{V} := \tau^* V$ in $\mathbb{E}^n_{\nu}(\kappa)$.

If \tilde{L} is the general CT in $\mathbb{E}_{\nu}^{n}(\kappa)$ and $\tilde{K} := \operatorname{tr}(\tilde{L})R - \tilde{L}$ is the KBDT where R is the metric in $\mathbb{E}_{\nu}^{n}(\kappa)$, then we have to solve the equation (see Section 6.7):

$$\mathrm{d}(\tilde{K}\mathrm{d}\tilde{V}) = 0$$

Now let K be the lift of \tilde{K} (as a contravariant tensor) to \mathbb{E}_{ν}^{n} via the warped product ψ . Then Proposition 5.4.3 shows that the above equation is locally satisfied iff

$$d(KdV) = 0$$

Hence if we calculate this lift of K, we only need to solve the above equation in \mathbb{E}_{ν}^{n} . We now proceed to calculate this lift. Note that it is sufficient to find a contravariant tensor in \mathbb{E}_{ν}^{n} which restricts to the KBDT in $\mathbb{E}_{\nu}^{n}(\kappa)$ and satisfies $\mathcal{L}_{r}K = 0$. It will be sufficient to do this for the CT then calculate the KBDT using its defining equation. Also noting that r is a CV, we will do the following calculations in a more general context just using this fact.

Let r be a non-null CV, since $r \odot r$ is an OCT, it follows that any integral manifold of r^{\perp} is a spherical submanifold. Hence Proposition 9.3.1 shows that any CT on M induces one on any leaf of the foliation induced by r^{\perp} . The following proposition shows how to solve the problem described earlier in this more general context.

Proposition 10.3.1

Suppose L is a CT on M and r is a non-null CV. Let $E := r^{\perp}$, and $L_E := L|_E$. Then $\tilde{L} := r^2 L_E$ restricts to a CT on any integral manifold of E and it satisfies $\mathcal{L}_r \tilde{L} = 0$ on M where \tilde{L} is in contravariant form. \diamondsuit

PROOF The proof of this fact is a straightforward calculation. We first note that since r is a CV with conformal factor ϕ , we have that

$$\nabla_{(i}r_{j)} = \phi g_{ij}$$

Suppose $u, v \in \Gamma(E)$, then

$$(\mathcal{L}_r L_{ij})u^i v^j = (\nabla_r L_{ij})u^i v^j + L_{ij} (\nabla_u r^i) v^j + L_{ij} (\nabla_v r^i) u^j$$
$$= \alpha_{(i} r_{j)} u^i v^j + 2\phi L_{ij} u^i v^j$$
$$= 2\phi L_{ij} u^i v^j$$

Thus

$$(\mathcal{L}_r L^{ij})u_i v_j = \mathcal{L}_r (G^{ik} L_{kj} G^{lj})u_i v_j$$

= $-2\phi L_{ij} u^i v^j + (\mathcal{L}_r L_{ij})u^i v^j - 2\phi L_{ij} u^i v^j$
= $-2\phi L_{ij} u^i v^j$

Finally

$$(\mathcal{L}_r(r^2L^{ij}))u_iv_j = r^2(\mathcal{L}_rL^{ij})u_iv_j + (\nabla_r r^2)L^{ij}u_iv_j$$
$$= -2r^2\phi L^{ij}u_iv_j + 2r^2\phi L^{ij}u_iv_j$$
$$= 0$$

Thus since r^{\flat} is closed, we conclude that $\mathcal{L}_r \tilde{L} = 0$. Also, as we noted earlier, Proposition 9.3.1 implies that \tilde{L} induces a CT on any integral manifold of E.

Remark 10.3.2

The above ansatz for \tilde{L} was deduced by studying results obtained by Benenti in [Ben08]. Although one can also obtain \tilde{L} by solving a certain differential equation.

Now back in \mathbb{E}^n_{ν} , let r be the dilatational vector field and $L = r^2 L_E$ as in the above proposition. Note that L_E is given in general by Eq. (9.1.5). Let G be the metric of \mathbb{E}^n_{ν} , then $R = G_E$ is the induced metric on $\mathbb{E}^n_{\nu}(\frac{1}{r^2})$ and the above proposition shows that $\mathcal{L}_r(r^2R) = 0$. Hence r^2R is the r-lift of the metric of $\mathbb{E}^n_{\nu}(\kappa)$ (up to sign). Hence if $\operatorname{tr}(L)$ is obtained by using the metric of \mathbb{E}^n_{ν} , the lifted KBDT is given as follows:

$$K_s = (\operatorname{tr}(L)\frac{1}{r^2})(r^2R) - L = \operatorname{tr}(L)R - L$$

which is the KBDT in $\mathbb{E}_{\nu}^{n}(\kappa)$ embedded in \mathbb{E}_{ν}^{n} . Also note that it follows from Proposition 4.5.2 that K_{s} is a KT in \mathbb{E}_{ν}^{n} . Also, using Eq. (9.1.5), one can calculate K_{s} explicitly:

$$K_s = \operatorname{tr}(A)r^2R - \langle r, Ar \rangle G - r^2A + 2Ar \odot r$$

Note that since the term $\operatorname{tr}(A)r^2R$ is a multiple of the metric of $\mathbb{E}^n_{\nu}(\kappa)$, that term can be removed. We summarize our results in the following statement:

Proposition 10.3.3 (Spherical KBD equation)

Suppose $V \in \mathcal{F}(\mathbb{E}^n_{\nu})$ is a potential in \mathbb{E}^n_{ν} which satisfies the KBD equation with $r \odot r$. Let L be a CT in $\mathbb{E}^n_{\nu}(\kappa)$ with parameter matrix A. Then V satisfies the KBD equation induced by L in $\mathbb{E}^n_{\nu}(\kappa)$ iff it satisfies the spherical KBD equation (Eq. (10.0.1)) with L in \mathbb{E}^n_{ν} .

10.3.2 In pseudo-Euclidean space

We show how to execute the BEKM separation algorithm in pseudo-Euclidean space. Fix a non-trivial solution L of the KBD equation in \mathbb{E}_{ν}^{n} . First apply the classification given by Theorem 9.1.8 to L. We will now assume that L is in one of the canonical forms listed in that theorem. If L is a Cartesian CT then the analysis is straightforward, see Section 10.2 for example. So we now assume L is non-degenerate and each generalized eigenspace of A_c has at most one proper generalized eigenvector¹.

First if A_c has no multidimensional (real) eigenspaces, then it is not reducible by Theorem 9.5.11. Hence one obtains separable coordinates for the natural Hamiltonian on the subset where L is an ICT.

Now suppose A_c has multidimensional (real) eigenspaces W_1, \ldots, W_k . It was shown in Section 9.5.1 that one can obtain data $(\bar{p}; \bigoplus_{i=0}^k V_i; a_1, ..., a_k)$ which determines a warped product decomposition $\psi : N_0 \times_{\rho_1} N_1 \cdots \times_{\rho_k} N_k \to \mathbb{E}^n_{\nu}$ in canonical form. Note that ψ decomposes the KBDT, K, associated with L. We now work with K.

We

case later. Suppose K is an orthogonal KT in \mathbb{E}^n_{ν} which is decomposed by the warped product ψ just constructed. Furthermore assume that each N_i corresponds to a distinct eigenspace of K. Now we show how to apply the BEKM separation algorithm on the spheres N_i by working only in a pseudo-Euclidean space.

Case 1 N_i is a non-null sphere, i.e. $a_i^2 \neq 0$

Let $W_{i\perp} := W_i^{\perp}$ and $c_i := \bar{p} - \frac{a_i}{\kappa_i}$. Define $\phi : W_{i\perp} \times W_i \to \mathbb{E}^n_{\nu}$ to be the standard product decomposition. Embed W_i in \mathbb{E}^n_{ν} as follows:

 $^{^1\}mathrm{It}$ follows from the classification in Section 8.2.1 that we lose no generality with this assumption in Euclidean or Minkowski space.

$$\tau_i : \begin{cases} W_i & \to \mathbb{E}_{\nu}^n \\ p_i & \mapsto \phi(c_i, p_i) = c_i + p_i \end{cases}$$

Note that $N_i = W_i(\kappa_i)$ via the above affine embedding of W_i . Let r_i be the dilatational vector field in W_i . By Corollary 9.5.5 and Proposition 5.4.3, it follows that $\tau_i^* V$ satisfies the KBD equation with $r_i \odot r_i$. Hence by Proposition 10.3.3 it is necessary and sufficient to solve the spherical KBD equation on W_i with $\tau_i^* V$.

Case 2 N_i is a null sphere, i.e. $a_i^2 = 0$

Embed N_i in \mathbb{E}^n_{ν} as follows (see Eq. (8.4.1)):

$$\tau_i : \begin{cases} N_i & \to \mathbb{E}_{\nu}^n \\ p_i & \mapsto \psi(\bar{p}, \dots, \bar{p}, p_i, \bar{p}, \dots, \bar{p}) = p_i \end{cases}$$

In this case N_i is isometric to V_i which is a pseudo-Euclidean space. Hence the BEKM separation algorithm can be applied on V_i .

In the following section we will show how to apply the BEKM separation algorithm on $\mathbb{E}^n_{\nu}(\kappa)$.

10.3.3 In Spherical submanifolds of pseudo-Euclidean space

We show how to execute the BEKM separation algorithm in $\mathbb{E}_{\nu}^{n}(\kappa)$. First we show to change this to a problem in \mathbb{E}_{ν}^{n} . Let \tilde{V} be a potential in $\mathbb{E}_{\nu}^{n}(\kappa)$. Note that \tilde{V} can be naturally lifted to a potential in \mathbb{E}_{ν}^{n} satisfying $\mathcal{L}_{r}\tilde{V} = 0$ using an appropriate coordinate system. Then, one can check that the potential

$$V := \frac{\tilde{V}}{\kappa r^2}$$

in \mathbb{E}^n_{ν} satisfies the KBD equation with $r \odot r$ in \mathbb{E}^n_{ν} and restricts to \tilde{V} when restricted to $\mathbb{E}^n_{\nu}(\kappa)$. So we lose no generality in assuming $V \in \mathcal{F}(\mathbb{E}^n_{\nu})$ and satisfies the KBD equation with $r \odot r$.

First note that by Proposition 10.3.3, we only need to consider solutions of the spherical KBD equation in \mathbb{E}_{ν}^{n} . So let *L* be a non-trivial solution of the spherical KBD equation (Eq. (10.0.1)). As in the pseudo-Euclidean case, we assume each generalized eigenspace of *A* has at most one proper generalized eigenvector. In order to execute the BEKM separation algorithm in \mathbb{E}_{ν}^{n} , we will need the following lemma:

Lemma 10.3.4

Let L_c be the central CT associated with L and $K_s = tr(L)R - L$ be the KBDT associated with L. Suppose L_c is reducible and let $\psi : N_0 \times_{\rho_1} N_1 \cdots \times_{\rho_k} N_k \to \mathbb{E}^n_{\nu}$ be a warped product which decomposes L_c . Then ψ decomposes K_s .

PROOF This follows from the proof of Proposition 9.5.12. In that proof we obtained the following equation:

$$R^*L_c = \psi_*(\tilde{R}^*\tilde{L}_c + \sum_{i=1}^k \lambda_i G_i)$$

Then we have:

$$L = r^2 R^* L_c = \psi_* (\tilde{r}^2 \tilde{R}^* \tilde{L}_c + \sum_{i=1}^k \lambda_i \tilde{r}^2 G_i)$$
$$R = \psi_* (\tilde{R} + \sum_{i=1}^k G_i)$$

Hence the result follows.

Now by Proposition 9.5.12 it follows that L is reducible iff L_c is reducible. Hence if L_c is not reducible, one obtains separable coordinates for the natural Hamiltonian on the subset (of $\mathbb{E}^n_{\nu}(\kappa)$) where L is an ICT.

If L_c is reducible, then by the above lemma, one can follow the arguments given in the previous section using the warped product decomposition induced by L_c which decomposes the KT K_s . We now give some crucial remarks. Let $\psi : N_0 \times_{\rho_1} N_1 \cdots \times_{\rho_k} N_k \to \mathbb{E}_{\nu}^n$ be a warped product decomposition which decomposes L_c and let $\phi : N_0(\kappa) \times_{\rho_1} N_1 \cdots \times_{\rho_k} N_k \to \mathbb{E}_{\nu}^n(\kappa)$ be an induced warped product decomposition of $\mathbb{E}_{\nu}^n(\kappa)$ as in Theorem 8.4.7. First note that the separable coordinates are constructed using the warped product ϕ . Also because the spherical factors N_i (where i > 0) are simultaneously spherical factors of ψ and ϕ (see Theorem D.6.5), there is no difference coming from working in the ambient space.
Chapter 11

Conclusion

In this thesis we have shown how to solve problems (1), (2), and (3) in spaces of constant curvature using concircular tensors. In doing so, we have given a covariant theory of a special class of separable coordinates called Kalnins-Eisenhart-Miller (KEM) coordinates. An overview of this theory can be found in Chapter 2.

In our solution, there is one important problem that has been unresolved. In Minkowski space, M^n , with $n \ge 3$, it is still computationally difficult to find the subset on which a given concircular tensor (CT) is a Benenti tensor. This implies that we still don't have a complete understanding of the separable coordinate systems for these spaces.

In Minkowski space it is well known that non-orthogonal separation can occur [Ben92b; KM79]. Hence it would be interesting to see if the present theory could be generalized to this case. A natural question is if the non-orthogonal separable coordinates in these spaces could be intrinsically characterized using concircular tensors, and conversely if non-orthogonal CTs could be used to define non-orthogonal separable coordinates. See Section 6.8 for more on this.

In Minkowski space it is also known that complex separation can occur [DR07]. Hence similar questions to those stated above for non-orthogonal separation also apply to this case.

One can also try to apply this theory to spaces with non-constant curvature. Some general results related to this idea are given in Section 6.3.3. Applications to general relativity have been studied in [Gro11].

APPENDICES

Appendix A

Cofactor Operators via Exterior Algebra

In this appendix we summarize some results from the book [Win10] by Winitzki. We mainly show how to obtain the cofactor operator (matrix) coordinate-independently and list some related results in this notation. In addition, we present some results on the derivative of a characteristic polynomial and projectors onto eigenspaces. These results are heavily used only in Section 6.6. Throughout this appendix V is an n-dimensional vector space.

Given $A \in \text{End}(V)$ and a basis $\{v_1, \ldots, v_n\}$ for V, define $\wedge^k A^k \in \text{End}(\wedge^k V)$ as follows:

$$\wedge^k A^k (v_1 \wedge \dots \wedge v_k) = (Av_1 \wedge \dots \wedge Av_k)$$

We define $\wedge^k A^1 \in \operatorname{End}(\wedge^k V)$ by

$$\wedge^k A^1(v_1 \wedge \dots \wedge v_k) = \sum_{i=1}^k (v_1 \wedge \dots \wedge Av_i \wedge \dots \wedge v_k)$$

For $1 \leq l \leq k$, the operator $\wedge^k A^l \in \text{End}(\wedge^k V)$ is the most natural generalization of the above definitions (see [Win10, section 3.7]). Also if l > k then $\wedge^k A^l := 0$.

Now for $A \in \text{End}(\wedge^k V)$, we can define the exterior adjoint (transpose) $A^{\wedge *} \in \text{End}(\wedge^{n-k} V)$ as follows:

$$A^{\wedge *}v \wedge w = v \wedge Aw$$

where $v \in \wedge^{n-k}V$ and $w \in \wedge^k V$. One can prove that $A^{\wedge *}$ is well defined either by direct calculation or using a proof analogous to the case of adjoints coming from inner products. The main use of the exterior adjoint is to give a basis independent definition of the determinant and the cofactor operator. Indeed, for any $A \in \text{End}(V)$ we define det $A \in \mathbb{R}$ by:

$$\det A := (\wedge^n A^n)^{\wedge *}$$

and the *cofactor operator*, $cof(A) \in End(V)$ by:

$$\operatorname{cof}(A) := (\wedge^{n-1} A^{n-1})^{\wedge *}$$

One can show that these definitions agree with the conventional ones [Win10]. Our first result is the following combinatorial lemma [Win10, lemma 1, P.138].

Lemma A.0.5

For $A \in \text{End}(V)$, and $1 \leq j \leq k \leq n$ the following holds:

$$\wedge^k (A+I)^j = \sum_{i=0}^j \binom{k-i}{k-j} \wedge^k A^i \qquad \Box$$

We will only make use of the cases (k, j) = (n, n), (n - 1, j) from the above lemma. We have the following corollary:

Corollary A.0.6

For $A \in \text{End}(V)$, the following holds:

$$\det(zI - A) = (\wedge^n (zI - A)^n)^{\wedge *} = \sum_{i=0}^n z^{n-i} (-1)^i (\wedge^n A^i)^{\wedge *}$$
$$\operatorname{cof}(zI - A) = (\wedge^{n-1} (zI - A)^{n-1})^{\wedge *} = \sum_{i=0}^{n-1} z^{n-1-i} (-1)^i (\wedge^{n-1} A^i)^{\wedge *}$$
(A.0.1)

Note that the above formula for the determinant implies that $(\wedge^n A^1)^{\wedge *} = \text{Tr} A$. Another useful formula is the following [Win10, lemma 1, P.152]:

Proposition A.0.7

For $A \in \text{End}(V)$ the following equation holds:

$$(\wedge^{n-1}A^{k-1})^{\wedge*}A + (\wedge^{n-1}A^k)^{\wedge*} = (\wedge^n A^k)^{\wedge*}I$$
(A.0.2)

PROOF First we consider the case k = n, then have to prove that:

$$(\wedge^{n-1}A^{n-1})^{\wedge*}A = (\wedge^n A^n)^{\wedge*}I$$

For $\omega = \omega_1 \wedge \cdots \wedge \omega_n$, let $\alpha = \omega_1 \wedge \cdots \wedge \omega_{n-1}$ and $\beta = \omega_n$, then

$$\wedge^{n} A^{n} \omega = A \omega_{1} \wedge \dots \wedge A \omega_{n}$$
$$= A \omega_{1} \wedge \dots \wedge A \omega_{n-1} \wedge A \omega_{n}$$
$$= (\wedge^{n-1} A^{n-1}) \alpha \wedge A \beta$$
$$= \alpha \wedge (\wedge^{n-1} A^{n-1})^{\wedge *} A \beta$$

which proves the result. The general case is a consequence of the identity [Win10]:

$$(\wedge^n A^k)\alpha \wedge \beta = \wedge^{n-1} A^{k-1}\alpha \wedge A\beta + \wedge^{n-1} A^k \alpha \wedge \beta$$

Note that the above proposition implies the following well known formula:

$$cof(A)A = det(A)I \tag{A.0.3}$$

As a corollary of the above proposition, we see that $(\wedge^{n-1}A^k)^{\wedge *}$ can be expressed as a polynomial in A [Win10, exercise 2, P.152]. Corollary A.0.8

For $A \in \text{End}(V)$ the following equation holds:

$$(\wedge^{n-1}A^k)^{\wedge *} = \sum_{i=0}^k (\wedge^n A^{k-i})(-1)^i A^i$$
(A.0.4)

The operators $(\wedge^{n-1}A^k)^{\wedge*}$ also admit a recursive formula [Win10, statement 3, P.159]: **Proposition A.0.9 (Leverrier sequence)**

For $A \in \text{End}(V)$, let $A_k := (\wedge^{n-1}A^k)^{\wedge *}$ where $1 \leq k \leq n-1$ and $A_0 := I$. Then the following formulas hold:

$$A_k = \frac{1}{k} \operatorname{Tr}(A_{k-1}A)I - A_{k-1}A \qquad \Box$$

PROOF Note that these formulas follow from Eq. (A.0.2) if we can prove that:

$$\frac{1}{k}\operatorname{Tr}(A_{k-1}A) = (\wedge^n A^k)^{\wedge *}$$

To prove this, we use the fact that $\operatorname{Tr} A_k = (n-k)(\wedge^n A^k)^{\wedge *}$ (see [Win10, statement 2, P.158]). Then taking the trace of Eq. (A.0.2) shows that:

$$\operatorname{Tr}((\wedge^{n-1}A^{k-1})^{\wedge*}A) = n(\wedge^n A^k)^{\wedge*} - \operatorname{Tr}((\wedge^{n-1}A^k)^{\wedge*})$$
$$= n(\wedge^n A^k)^{\wedge*} - (n-k)(\wedge^n A^k)^{\wedge*}$$
$$= k(\wedge^n A^k)^{\wedge*}$$

which proves the result.

For convenience, we let $\sigma_k := (\wedge^n A^k)^{\wedge *}$ and $A_k := (\wedge^{n-1} A^k)^{\wedge *}$. The following is from [Win10, statement, P.179].

Proposition A.0.10 (Derivative of the Characteristic Polynomial)

Suppose the coefficients of A in some basis are a function t, then for $1 \le k \le n$ we have the following:

$$\frac{d\sigma_k}{dt} = \operatorname{Tr}(A_{k-1}\frac{dA}{dt})$$

Remark A.0.11

When k = n the above formula expressing the derivative of the determinant is called *Jacobi's formula*.

PROOF The proof is given on P. 179 in [Win10]. We will prove the special case when k = n following the proof of lemma 4 in [Win10, P. 177]. Let v_1, \ldots, v_n be a basis for V.

$$\partial_t (\wedge^n A^n) v_1 \wedge \dots \wedge v_n = \partial_t ((\wedge^n A^n) v_1 \wedge \dots \wedge v_n)$$

= $\partial_t (A v_1 \wedge \dots \wedge A v_n)$
= $\sum_{k=1}^n A v_1 \wedge \dots \wedge (\partial_t A) v_k \wedge \dots \wedge A v_n$
= $\sum_{k=1}^n v_1 \wedge \dots \wedge (\wedge^{n-1} A^{n-1})^{\wedge *} (\partial_t A) v_k \wedge \dots \wedge v_n$
= $\wedge^n (A_{n-1} \partial_t A)^1 v_1 \wedge \dots \wedge v_n$
= $\operatorname{Tr}(A_{n-1} \partial_t A) v_1 \wedge \dots \wedge v_n$

The last equality follows from Corollary A.0.6.

The final fact we will need is the following [Win10, statement 2, P.195]:

Proposition A.0.12 (Projectors onto Eigenspaces)

Suppose $A \in \text{End}(V)$ and λ is an eigenvalue of A with geometric and algebraic multiplicity k, then the operator

$$P_{\lambda}^{k} = \frac{(-1)^{n-k}}{\sigma_{n-k}} (\wedge^{n-1} (\lambda I - A)^{n-k})^{\wedge *}$$

is a projector onto the eigenspace corresponding to λ .

Appendix B

Nijenhuis tensors, Haantjes tensors and Integrability of Eigenspaces

In this appendix M is a manifold of dimension n and T is an endomorphism of TM (i.e. a $\binom{1}{1}$ -tensor). We assume the reader is familiar with Section 3.4. The theory presented in this appendix is motivated by the following question: Let $\mathcal{E} = (E_i)_{i=1}^k$ where E_i are the eigenspaces of T (which are assumed to be distributions), suppose $TM = \bigoplus_{i=1}^k E_i$. Then are there tensorial conditions depending only on T (and its derivatives) that characterize when \mathcal{E} is an integrable net? Nijenhuis, Haantjes and colleagues have answered this question in the affirmative. They have provided conditions in terms of the Haantjes tensor (of T) which is defined in terms of the Nijenhuis tensor (of T). We shall see that the Nijenhuis tensor (which is a byproduct of the solution to this problem) is a useful tool in integrable systems theory in its own right.

We give original references throughout this appendix. For contemporary references mainly related to integrable systems theory see [GVY08; Bog06].

The primary definition is the following:

Definition B.0.13 (Nijenhuis tensor [Nij51])

If T is an endomorphism of TM, then the Nijenhuis tensor (torsion) of T is a $\binom{1}{2}$ -tensor skew-symmetric in its covariant components, denoted by N_T and is defined as follows:

$$N_T(u,v) := T^2[u,v] + [Tu,Tv] - T([Tu,v] + [u,Tv])$$

Furthermore T is called *torsionless* if its Nijenhuis tensor vanishes.

The following proposition will make it clear that N_T is actually a tensor. It gives equivalent definitions of the Nijenhuis tensor.

Proposition B.0.14 (Equivalent Definitions of The Nijenhuis tensor)

Suppose T is an endomorphism of TM, then the following are equivalent definitions of N_T :

- 1. $N_T(u,v) = (\mathcal{L}_{Tu}T T\mathcal{L}_uT)v$ for all $u, v \in \mathfrak{X}(M)$ where \mathcal{L} is the Lie derivative
- 2. $N_T(u,v) = (\nabla_{Tu}T)v (\nabla_{Tv}T)u T((\nabla_uT)v (\nabla_vT)u)$ for all $u, v \in \mathfrak{X}(M)$ where ∇ is a torsion-free connection

3. $N_T = \frac{1}{2}[T,T]$ where $[\cdot,\cdot]$ is the Frölicher-Nijenhuis bracket [FN56]

PROOF We prove the first equation as follows

$$T^{2}[u,v] + [Tu,Tv] - T([Tu,v] + [u,Tv]) = T^{2}[u,v] - \mathcal{L}_{Tv}Tu + T(\mathcal{L}_{v}Tu + \mathcal{L}_{Tv}u)$$
$$= T^{2}[u,v] - T\mathcal{L}_{Tv}u - (\mathcal{L}_{Tv}T)u$$
$$+ T(T\mathcal{L}_{v}u + (\mathcal{L}_{v}T)u + \mathcal{L}_{Tv}u)$$
$$= -(\mathcal{L}_{Tv}T)u + T(\mathcal{L}_{v}T)u$$
$$= (T(\mathcal{L}_{v}T) - (\mathcal{L}_{Tv}T))u$$

Thus $N_T(u, v) = (T\mathcal{L}_v T - \mathcal{L}_{Tv}T)u$, hence $N_T(u, v) = (\mathcal{L}_{Tu}T - T\mathcal{L}_uT)v$.

Now suppose ∇ is a torsion-free connection, then $[u, v] = \nabla_u v - \nabla_v u$ for all $u, v \in \mathfrak{X}(M)$, thus

$$T^{2}[u, v] + [Tu, Tv] - T([Tu, v] + [u, Tv])$$

$$= T^{2}[u, v] + [Tu, Tv] - T(\nabla_{Tu}v - \nabla_{v}Tu + \nabla_{u}Tv - \nabla_{Tv}u)$$

$$= T^{2}[u, v] + [Tu, Tv] - T(\nabla_{Tu}v - (\nabla_{v}T)u - T\nabla_{v}u + (\nabla_{u}T)v + T\nabla_{u}v - \nabla_{Tv}u)$$

$$= T^{2}[u, v] + T^{2}(\nabla_{v}u - \nabla_{u}v) + [Tu, Tv] - T(\nabla_{Tu}v - (\nabla_{v}T)u + (\nabla_{u}T)v - \nabla_{Tv}u)$$

$$= \nabla_{Tu}Tv - \nabla_{Tv}Tu - T(\nabla_{Tu}v - (\nabla_{v}T)u + (\nabla_{u}T)v - \nabla_{Tv}u)$$

$$= (\nabla_{Tu}T)v - (\nabla_{Tv}T)u - T(-(\nabla_{v}T)u + (\nabla_{u}T)v)$$

The equation in terms of the Frölicher-Nijenhuis bracket is included for completeness, see for example [GVY08, Equation 13.37] for more details.

Remark B.0.15

Note in coordinates the equation for N_T in terms of a torsion-free connection ∇ can be written as follows¹:

$$(N_T)^{k}_{\ ij} = 2(T^{l}_{\ [i}\nabla_{|l|}T^{k}_{\ j]} - T^{k}_{\ l}\nabla_{[i}T^{l}_{\ j]})$$

¹The bar around the index l means that it's excluded from the anti-symmetrization.

Proof

$$N_{T}(u,v) = (\nabla_{Tu}T)v - (\nabla_{Tv}T)u - T((\nabla_{u}T)v - (\nabla_{v}T)u)$$

$$(N_{T})^{k}{}_{ij}u^{i}v^{j} = T^{l}{}_{i}u^{i}(\nabla_{l}T^{k}{}_{j})v^{j} - T^{l}{}_{j}v^{j}(\nabla_{l}T^{k}{}_{i})u^{i} - T^{k}{}_{l}(u^{i}(\nabla_{i}T^{l}{}_{j})v^{j} - v^{j}(\nabla_{j}T^{l}{}_{i})u^{i})$$

$$= (T^{l}{}_{i}\nabla_{l}T^{k}{}_{j} - T^{l}{}_{j}\nabla_{l}T^{k}{}_{i} - T^{k}{}_{l}(\nabla_{i}T^{l}{}_{j} - \nabla_{j}T^{l}{}_{i}))u^{i}v^{j}$$

$$= 2(T^{l}{}_{[i}\nabla_{|l|}T^{k}{}_{j]} - T^{k}{}_{l}\nabla_{[i}T^{l}{}_{j]})u^{i}v^{j}$$

Remark B.0.16

Note that the first characterization implies that T is torsionless iff for every $v \in \mathfrak{X}(M)$ we have:

$$\mathcal{L}_{Tv}T = T\mathcal{L}_v T \tag{B.0.1}$$

We say that a vector field v is a symmetry of T if $\mathcal{L}_v T = 0$. The above equation shows the remarkable property that if T is torsionless, then T maps symmetries to symmetries.

The Nijenhuis tensor is fundamental in this theory, but it alone is not enough to answer the question posed at the beginning of this appendix. For this we need to introduce the Haantjes tensor:

Definition B.0.17 (Haantjes tensor [Haa55])

If T is an endomorphism of TM, then the Haantjes tensor (torsion) of T is a $\binom{1}{2}$ -tensor skew-symmetric in its covariant components, denoted by H_T and is defined as follows:

$$H_T(u,v) := T^2 N_T(u,v) + N_T(Tu,Tv) - T(N_T(Tu,v) + N_T(u,Tv))$$

The fact that H_T is a tensor follows from the fact that N_T is a tensor. Now we need the following lemma, the proof of which follows by a direct calculation.

Lemma B.0.18

Suppose T is an endomorphism of TM. Let X, Y be eigenvector fields of T with eigenfunctions λ, μ respectively. Then N_T satisfies the following:

$$N_T(X,Y) = (T-\lambda)(T-\mu)[X,Y] + (\lambda-\mu)((Y\lambda)X + (X\mu)Y)$$
(B.0.2)

Now we can prove the main theorem in this theory:

Theorem B.0.19 (Haantjes [Haa55; FN56])

Suppose T is an endomorphism of TM with eigenspaces $\mathcal{E} = (E_i)_{i=1}^k$. Assume each eigenspace is a distribution and together they satisfy:

$$TM = \bigoplus_{i=1}^{k} E_i$$

In other words T is point-wise diagonalizable. Then \mathcal{E} is an integrable net iff the Haantjes tensor of T vanishes.

PROOF We follow the proof given in [GVY08] which is originally from [FN56]. Suppose X, Y are eigenvector fields of T with eigenfunctions λ, μ respectively. Then using the above lemma one can calculate the following:

$$H_T(X,Y) = (N - \lambda)^2 (N - \mu)^2 [X,Y]$$

Now we note that if $\lambda \neq \mu$, since T is point-wise diagonalizable $H_T(X,Y) = 0$ iff

$$(N - \lambda)(N - \mu)[X, Y] = 0$$

This equation holds iff $[X, Y] \in E_{\lambda} \oplus E_{\mu}$; it holds for arbitrary X, Y iff $E_{\lambda} \oplus E_{\mu}$ is Frobenius integrable. Similarly if $\lambda = \mu$, then $H_T(X, Y) = 0$ iff

$$(N - \lambda)[X, Y] = 0$$

This equation holds iff $[X, Y] \in E_{\lambda}$; it holds for arbitrary X, Y iff E_{λ} is Frobenius integrable. The theorem then follows from Theorem 3.4.3.

Now we prove another important theorem which follows from the above lemma.

Theorem B.0.20 (Nijenhuis [Nij51])

Suppose T is an endomorphism of TM with eigenspaces $\mathcal{E} = (E_i)_{i=1}^k$ and corresponding eigenfunctions $(\lambda_i)_{i=1}^k$. Assume each eigenspace is a distribution and together they satisfy:

$$TM = \bigoplus_{i=1}^{k} E_i$$

In other words T is point-wise diagonalizable. Then \mathcal{E} is an integrable net with each eigenfunction λ_i depending only on² E_i iff the Nijenhuis tensor of T vanishes.

PROOF First assume that the Nijenhuis tensor of T vanishes. Then from the Haantjes theorem above, \mathcal{E} is an integrable net. Furthermore by Eq. (B.0.2) in the above lemma it follows that each eigenfunction λ_i depends only on E_i .

Conversely if \mathcal{E} is an integrable net and each eigenfunction λ_i depends only on E_i then by Eq. (B.0.2) in the above lemma it follows that the Nijenhuis tensor of T vanishes identically.

The following optional result is a straightforward consequence of the Nijenhuis theorem.

²By this we mean that $Y(\lambda_i) = 0$ for every $Y \in E_i$ and $j \neq i$.

Corollary B.0.21 (Integrability of almost product structures)

Suppose T is an endomorphism of TM defining an almost product structure, i.e. $T^2 = I$. Then the almost product structure is integrable iff $N_T = 0$.

The above result can be generalized to almost complex structures (i.e. $T^2 = -I$) as well; this is known as the Newlander-Nirenberg theorem.

B.1 Properties of Torsionless Tensors

In this section we will list some identities satisfied by torsionless tensors which are used in the thesis. The contemporary references given earlier have more results on these tensors. We will make use of some notations and results from Appendix A; we will mainly be applying Proposition A.0.10.

Proposition B.1.1 (Jacobi's formula for Torsionless Tensors)

Suppose T is a torsionless tensor. Then for every $v \in \mathfrak{X}(M)$ we have:

$$\mathcal{L}_{Tv} \det T = \det T \mathcal{L}_v(\operatorname{Tr} T)$$

In terms of the adjoint T^* , this equation can be written:

$$T^* d(\det T) = \det T d(\operatorname{Tr} T)$$

PROOF This is a consequence of Proposition A.0.10 when k = 1. Indeed that proposition implies that:

$$\mathcal{L}_v \det T = \operatorname{Tr}((\wedge^{n-1}T^{n-1})^{\wedge *}\mathcal{L}_v T)$$

for any $v \in \mathfrak{X}(M)$. Using the fact that T is torsionless, we have the following:

$$\mathcal{L}_{Tv} \det T = \operatorname{Tr}((\wedge^{n-1}T^{n-1})^{\wedge*}\mathcal{L}_{Tv}T)$$

$$\stackrel{(\mathbf{B}.0.1)}{=} \operatorname{Tr}((\wedge^{n-1}T^{n-1})^{\wedge*}T\mathcal{L}_{v}T)$$

$$= (\det T)\mathcal{L}_{v}(\operatorname{Tr} T)$$

where in the last line we used the fact that $(\wedge^{n-1}T^{n-1})^{\wedge*}T = (\det T)I.$

More generally, we have the following formulas:

Proposition B.1.2

Suppose T is a torsionless tensor. For $1 \leq k \leq n$, let $\sigma_k := (\wedge^n T^k)^{\wedge *}$, then for every $v \in \mathfrak{X}(M)$ we have:

$$\mathcal{L}_{Tv}\sigma_k = \sigma_k \mathcal{L}_v \operatorname{Tr} T - \mathcal{L}_v \sigma_{k+1}$$

Equivalently, if for $1 \le k \le n-1$ we let $S_k := (\wedge^{n-1}T^k)^{\wedge *}$, then for $1 \le k \le n$

$$d\sigma_k = S_{k-1}^* d(\operatorname{Tr} T) \tag{B.1.1}$$

PROOF Proposition A.0.10 implies that for $v \in \mathfrak{X}(M)$ we have:

$$\mathcal{L}_v \sigma_k = \operatorname{Tr}((\wedge^{n-1} T^{k-1})^{\wedge *} \mathcal{L}_v T)$$

Thus

$$\mathcal{L}_{Tv}\sigma_{k} = \operatorname{Tr}((\wedge^{n-1}T^{k-1})^{\wedge*}\mathcal{L}_{Tv}T)$$

$$\stackrel{(\mathrm{B.0.1})}{=} \operatorname{Tr}((\wedge^{n-1}T^{k-1})^{\wedge*}T\mathcal{L}_{v}T)$$

$$\stackrel{(\mathrm{A.0.2})}{=} \sigma_{k}\operatorname{Tr}\mathcal{L}_{v}T - \operatorname{Tr}((\wedge^{n-1}T^{k})^{\wedge*}\mathcal{L}_{v}T)$$

$$= \sigma_{k}\operatorname{Tr}\mathcal{L}_{v}T - \mathcal{L}_{v}\sigma_{k+1}$$

If T^* denotes the adjoint of T, then we have:

$$\mathrm{d}\sigma_k = \sigma_{k-1}\mathrm{d}(\mathrm{Tr}\,T) - T^*\mathrm{d}\sigma_{k-1}$$

Thus Eq. (B.1.1) follows by induction on k from the above equation and the recursive equation Eq. (A.0.2) for S_k^* .

244

Appendix C

Self-adjoint operators in pseudo-Euclidean space

Self-adjoint operators are ubiquitous in pseudo-Riemannian geometry and hence in general relativity as well. Any linear operator metrically equivalent to a symmetric contravariant tensor is self-adjoint. The Ricci tensor, the Hessian of a smooth function, the shape operator associated with a pseudo-Riemannian hypersurface or an umbilical pseudo-Riemannian submanifold [O'N83, Definition 4.18], and Killing and conformal Killing tensors are all examples of such tensors. In general relativity the energy-momentum tensor is one as well. Hence their algebraic classification is an important problem.

In this section we call a pseudo-Euclidean space V a scalar product space, following [O'N83] (we will give more details shortly). Recall, a linear operator T on a scalar product space V is said to be self-adjoint if $\langle Tx, y \rangle = \langle x, Ty \rangle$ for all $x, y \in V$. Following O'Neil's solution given in exercises 18-19 in [O'N83, P. 260-261], we will obtain a canonical form for self-adjoint operators. Specifically, motivated by the Jordan canonical form, we develop an algorithm to find a Jordan canonical basis for a self-adjoint operator which also gives a canonical form for the scalar product. Our derivation has some advantages: it only depends on some results from Jordan form theory, we are able to prove existence and uniqueness of the canonical form independently of the corresponding results from Jordan form theory (i.e. ours results have few dependencies), and we obtain a simple algorithm to calculate the canonical forms for self-adjoint operators. The draw back is that our solution is less general than others (see for example [GL05]).

Finally, we note that the contents of this appendix are from [Raj14a]. They are included here for completeness.

C.1 Preliminaries on Scalar product spaces

We will assume the reader is familiar with Sections 1.4.1 and 8.1. In this appendix, we will also work with the complexification of a real vector space V denoted $V^{\mathbb{C}}$. We define the complexified bilinear form to be the symmetric bilinear form in $V^{\mathbb{C}}$ obtained from the real one by a linear extension. Note that the complexified bilinear form is symmetric, in contrast with the usual Hermitian form which is not symmetric. It follows immediately from the definition that the complexified bilinear form is non-degenerate iff the real bilinear form is non-degenerate. Thus a real scalar product space, V, can be canonically complexified to a complex scalar product space, hereafter denoted $V^{\mathbb{C}}$.

C.2 Preliminaries from Operator Theory

Given a complex scalar λ , a non-zero vector $x \in V$ is called a generalized eigenvector for T corresponding to λ if $(T - \lambda I)^p x = 0$ for some positive integer p.

Definition C.2.1 (Generalized Eigenspaces)

Let T be a linear operator on finite-dimensional complex vector space V and let λ be an eigenvalue of T. The generalized eigenspace (g-space) corresponding to λ , denoted K_{λ} , is the subset of V defined by:

$$K_{\lambda} = \{ x \in V : (T - \lambda I)^{p}(x) = 0 \text{ for some positive integer p} \}$$

We say a set of distinct scalars $\lambda_1, ..., \lambda_k$ are the spectrum of T if they constitute all eigenvalues of T. Furthermore the kernel of an operator T is denoted by kerT or N(T).

The following results concerning the g-spaces of a linear operator are proven in [FIS03, Section 7.1].

Theorem C.2.2

Let T be a linear operator on finite-dimensional complex vector space V. Suppose λ and μ are distinct eigenvalues of T, then the following statements are true:

- 1. K_{λ} is a non-zero T invariant subspace of V
- 2. $K_{\lambda} \cap K_{\mu} = \{0\}$
- 3. Let $U = (T \lambda I)|_{K_{\mu}}$, then K_{μ} is $(T \lambda I)$ -invariant and U is a bijection.

4. If m is the algebraic multiplicity of λ then $K_{\lambda} = N(T - \lambda I)^m$ and dim $K_{\lambda} \leq m$.

5. If $\lambda_1, ..., \lambda_k$ is the spectrum of T, then $V = \bigoplus_{i=1}^k K_{\lambda_i}$

Hence the above theorem implies T is block diagonal in a basis adapted to the g-spaces.

Definition C.2.3

Let T be a linear operator on finite-dimensional complex vector space E and let x be a generalized eigenvector of T corresponding to the eigenvalue λ . Suppose that p is the smallest positive integer such that $(T - \lambda I)^p(x) = 0$. Then the ordered set

$$\{(T-\lambda I)^{p-1}(x), (T-\lambda I)^{p-2}(x), \dots, (T-\lambda I)(x), x\}$$

is called a cycle (*p*-cycle) of generalized eigenvectors for T with eigenvalue λ . $(T - \lambda I)^{p-1}(x)$ and x are called the initial vector and the end vector of the cycle, respectively. We say that the cycle has length p and x generates a cycle (*p*-cycle) of generalized eigenvectors.

We first note that the subspace spanned by a *p*-cycle has dimension *p*. We also observe that a given cycle of generalized eigenvectors generated by *x* with eigenvalue λ lie in K_{λ} . Also *T* restricted to this cycle has the following matrix representation:

$$egin{pmatrix} \lambda & 1 & & \ \lambda & \ddots & 0 \ & \ddots & 1 \ & & \lambda & 1 \ 0 & & \lambda \end{pmatrix}$$

We denote $U_{\lambda} = T - \lambda I$ and if λ is fixed we remove the subscript and refer to $U_{\lambda} = U$.

Suppose T is a real linear operator and let λ be an eigenvalue with non-zero imaginary part. Suppose x generates a cycle of generalized eigenvectors of length p with eigenvalue λ whose end vector has linearly independent real and imaginary parts. Then it follows that \overline{x} generates a cycle of generalized eigenvectors of length p with eigenvalue $\overline{\lambda}$ which is linearly independent of the cycle generated by x. We denote the real subspace generated by these vectors as $K_{\lambda \oplus \overline{\lambda}}$ and call this the *real subspace* spanned by the cycle generated by x. If $\lambda \in \mathbb{R}$, then this real subspace is just K_{λ} .

Knowledge of the Jordan canonical form is unnecessary for our derivation. Although for readers familiar with it, note that $K_{\lambda} \simeq N_1 \oplus \frac{N_2}{N_1} \oplus \ldots \oplus \frac{N_p}{N_{p-1}}$ where $N_i = \ker U_{\lambda}^i$. This shows the non-uniqueness of a given Jordan canonical basis. We will use this fact to find a Jordan canonical basis for a self-adjoint operator adapted to the scalar product.

In order to prove the uniqueness of the metric-Jordan canonical form of a self-adjoint operator we will need some theory on symmetric bilinear forms. First a *diagonal representation* of a symmetric bilinear form is a basis in which the matrix representation of the form is diagonal.

Theorem C.2.4 (Sylvester's Law of Inertia)

For any symmetric bilinear form defined over a real vector space, the number of positive diagonal entries and negative diagonal entries in a diagonal representation is independent of the diagonal representation.

For any symmetric bilinear form defined over a complex vector space, the number of non-zero diagonal entries in a diagonal representation is independent of the diagonal representation.

PROOF For the real case, see Theorem 6.38 in [FIS03] or Theorem 6.8 in [Jac12]. For the complex case, see Theorem 6.6 in [Jac12]

C.3 Existence of the metric-Jordan canonical form

In this section we will show how to obtain the canonical form, culminating in Theorem C.3.7. First we need some properties of self-adjoint operators.

Theorem C.3.1 (Fundamental Properties of Self-Adjoint Operators)

Suppose V is a scalar product space and T is a self-adjoint operator on V. Suppose $H \subseteq V$ is an invariant subspace of T. Then

- 1. $T(H^{\perp}) \subseteq H^{\perp}$, i.e. H^{\perp} is an invariant subspace of T.
- 2. $(\ker T)^{\perp} = \operatorname{range} T$ and $V = \ker T \oplus \operatorname{range} T$ iff either $\ker T$ or $\operatorname{range} T$ is a nondegenerate subspace

3. Any polynomial in T is self-adjoint.

PROOF The proofs are immediate.

Remark C.3.2

The first statement of the above theorem also holds for unitary operators on V, as noted by O'Neil in [O'N83, Section 9.4].

The idea behind obtaining the canonical forms is as follows. First suppose that T is a self-adjoint operator on a scalar product space. When E is a Euclidean space, one can easily diagonalize T using property 1 and the fact that self-adjoint operators in Euclidean space have real eigenvalues. Indeed, after one finds a single eigenvector v, one can use property 1 to deduce that the subspace orthogonal to v must be T-invariant. Since in Euclidean space the subspace orthogonal to v must be complementary to v, one can repeat this procedure to find a basis of eigenvectors for T.

For general indefinite scalar products, our goal will be to find a cycle of generalized eigenvectors for T such that they span a non-degenerate subspace. Then as in the

Euclidean case, we can use property 1 to inductively build a Jordan canonical basis for T. We will now develop a series of lemmas to show that any self-adjoint operator admits a cycle of generalized eigenvectors whose span is a non-degenerate subspace. Then we will combine these lemmas in Theorem C.3.7 which shows how to obtain a Jordan canonical basis for T which also puts the scalar product in a canonical form.

The following theorem starts us off by showing that the g-spaces of a self-adjoint operator are always non-degenerate, in fact it says even more:

Lemma C.3.3

Suppose $V^{\mathbb{C}}$ is a scalar product space and T is a real self-adjoint operator on $V^{\mathbb{C}}$. Let λ and μ be distinct eigenvalues of T, then $K_{\lambda} \perp K_{\mu}$, hence if $\lambda_1, ..., \lambda_k$ is the spectrum of Tthen by Theorem C.2.2, $V^{\mathbb{C}} = \bigoplus_{i=1}^{k} K_{\lambda_i}$.

As an immediate corollary we find that each generalized eigenspace is a non-degenerate subspace. \Box

PROOF Suppose $x \in K_{\lambda}$ and $y \in K_{\mu}$. Suppose $U_{\lambda}^{p}(x) = 0$, since $\mu \neq \lambda$ Theorem C.2.2 says that U_{λ} is a bijection when restricted to K_{μ} , hence there exists a $z \in K_{\mu}$ such that $y = U_{\lambda}^{p}(z)$. Since U_{λ}^{p} is self-adjoint, property 2 implies that $\langle x, y \rangle = 0$.

Thus $K_{\lambda} \perp K_{\mu}$. As a consequence of this and Theorem C.2.2 we see that $E = K_{\lambda} \oplus K_{\lambda}^{\perp}$, hence K_{λ} is non-degenerate.

Suppose V is a scalar product space and T is a self-adjoint operator on V. Suppose λ is an eigenvalue of T and $x \in K_{\lambda}$ generates a cycle of generalized eigenvectors of T of length p. Let $U = (T - \lambda I)$ and $v_i = U^{p-i}x$ for $i \in \{1, ..., p\}$. Then observe that

$$\langle v_i, v_j \rangle = \left\langle U^{p-i}x, U^{p-j}x \right\rangle$$

$$= \left\langle U^{2p-i-j}x, x \right\rangle$$
(C.3.1)

If $i + j \leq p$ then by property 2 and the fact that $U^p x = 0$ the above equation implies that $\langle v_i, v_j \rangle = 0$. If i + j > p then the above equation implies that $\langle v_i, v_j \rangle$ only depends on the sum i + j. Thus in a cycle of length p there are only p scalar products that are variable and the above equation shows us that we only need to deal with the products $\langle v_i, v_p \rangle$. The following lemma will show that for every g-space we can always find a generator of a cycle such that $\langle v_1, v_p \rangle \neq 0$.

Lemma C.3.4

Suppose V is a scalar product space and T is a self-adjoint operator on V. Fix an eigenvalue λ of T and let $U = (T - \lambda I)|_{K_{\lambda}}$. Suppose $k \ge 0$ satisfies $U^k \ne 0$, then there exists an $x \in K_{\lambda}$ such that $\langle U^k x, x \rangle \ne 0$.

PROOF Suppose to the contrary that $\langle U^k(x), x \rangle = 0$ for all $x \in K_{\lambda}$. Define a bilinear form $[\cdot, \cdot] : K_{\lambda} \times K_{\lambda} \to \mathbb{F}$ by $[x, y] = \langle U^k(x), y \rangle$. Since U^k is self-adjoint, $[\cdot, \cdot]$ is a symmetric bilinear form. Thus by the polarization identity, it follows that for any $x, y \in K_{\lambda}$

$$0 = [x, y] = \left\langle U^k(x), y \right\rangle$$

Now, since $U^k \neq 0$, there exists an $x \in K_\lambda$ such that $U^k x \neq 0$. But by Lemma C.3.3 the scalar product is non-degenerate, hence the above equation implies that $U^k x = 0$, a contradiction. Hence the conclusion holds.

Assuming $\langle v_1, v_p \rangle \neq 0$, the following proposition shows how to adapt the cycle so that any other remaining scalar products are zero.

Lemma C.3.5

Suppose the v_i are as defined as above for a cycle of generalized eigenvectors of T generated by $x \in K_{\lambda}$. Let $H \subseteq K_{\lambda}$ be the subspace corresponding to the cycle generated by x. If $\langle v_1, v_p \rangle \neq 0$, then we can choose an $x' \in H$ such that x' generates a cycle of generalized eigenvectors $v'_i = U^{p-i}x'$ of length p spanning H, such that v_1, \ldots, v_p forms a skew-normal sequence of sign sgn $\langle v_1, v_p \rangle$ if $\lambda \in \mathbb{R}$ or 1 if $\lambda \in \mathbb{C} \setminus \mathbb{R}$.

PROOF Suppose first that $\lambda \in \mathbb{C} \setminus \mathbb{R}$, then let $v'_p = \langle v_1, v_p \rangle^{-\frac{1}{2}} v_p$ where any square root is fine. Then observe that:

$$\left\langle v_1', v_p' \right\rangle = \left\langle v_1, v_p \right\rangle^{-1} \left\langle v_1, v_p \right\rangle$$

= 1

If $\lambda \in \mathbb{R}$ then let $v'_p = |\langle v_1, v_p \rangle|^{-\frac{1}{2}} v_p$. Then observe that:

$$\langle v_1', v_p' \rangle = |\langle v_1, v_p \rangle|^{-1} \langle v_1, v_p \rangle$$

= ± 1

Thus we can assume that $|\langle v_1, v_p \rangle| = 1$. Inductively suppose that $|\langle v_1, v_p \rangle| = 1$ and that $\langle v_i, v_p \rangle = 0$ for $2 \le i \le k - 1$ for some $k \ge 2$.

Let $v'_p = v_p + av_{p-k+1}$ where a is to be determined. Now for $i \in \{1, ..., p\}$

$$v'_{i} = U^{p-i}v'_{p}$$
$$= U^{p-i}v_{p} + aU^{p-i}v_{p-k+1}$$
$$= v_{i} + av_{i-k+1}$$

Observe that $v'_i = v_i$ if $i - k + 1 \le 0$, i.e. $i \le k - 1$. The above equation also shows that each $v'_i \in H$ and since $v'_1 = v_1 \ne 0$ the cycle generated by v'_p has length p and thus forms a basis for H. Now using the fact that $\langle v_i, v_j \rangle$ only depends on i + j, we find that:

$$\langle v'_k, v'_p \rangle = \langle v_k + av_1, v_p + av_{p-k+1} \rangle$$

= $\langle v_k, v_p \rangle + a \langle v_k, v_{p-k+1} \rangle + a \langle v_1, v_p \rangle + a^2 \langle v_1, v_{p-k+1} \rangle$
= $\langle v_k, v_p \rangle + 2a \langle v_1, v_p \rangle$

where $\langle v_1, v_{p-k+1} \rangle = 0$ since $p - k + 2 \leq p$. Thus let $a = -\frac{\langle v_k, v_p \rangle}{2 \langle v_1, v_p \rangle}$ which forces $\langle v'_k, v'_p \rangle = 0$.

Now suppose $1 \leq i < k$, then note that $v'_i = v_i$, thus

$$\langle v'_i, v'_p \rangle = \langle v_i, v_p + av_{p-k+1} \rangle$$

= $\langle v_i, v_p \rangle + a \langle v_i, v_{p-k+1} \rangle$
= $\langle v_i, v_p \rangle$

where $\langle v_i, v_{p-k+1} \rangle = 0$ follows from the induction hypothesis in conjunction with the fact that because $k \geq 2$, we have that $p + i - k + 1 \leq p + k - 1$ and $k \leq p$ implies $p + i - k + 1 \neq 1$. Thus v'_p satisfies the induction hypothesis and after relabeling v'_p as v_p we can apply the induction hypothesis again until k = p in which case we will have proven the statement.

Suppose x generates a cycle of generalized eigenvectors satisfying the conclusions of the above proposition and let $z_i = U^{p-i}x$. Then by Eq. (C.3.1) we find that the only non-zero scalar products are $\langle z_i, z_j \rangle = \langle z_1, z_p \rangle$ where i + j = p + 1. Thus we say a given cycle of generalized eigenvectors with eigenvalue λ for a self-adjoint operator are adapted to the scalar product, if they form a skew-normal sequence of sign ± 1 if $\lambda \in \mathbb{R}$ or sign 1 if $\lambda \in \mathbb{C} \setminus \mathbb{R}$. If $\lambda \in \mathbb{R}$, then $\{z_1, \ldots, z_p\}$ form a real basis for the Jordan canonical form of T. If $\lambda \in \mathbb{C} \setminus \mathbb{R}$ (WLOG we can assume $\operatorname{Im}(\lambda) > 0$), then we choose a canonical real basis $\{u_1, v_1, \ldots, u_p, v_p\}$ for T as follows. Let

$$u_i = \frac{1}{\sqrt{2}}(z_i + \overline{z_i}) \tag{C.3.2a}$$

$$v_i = \frac{1}{i\sqrt{2}}(z_i - \overline{z_i}) \tag{C.3.2b}$$

Note that

It then follows that $\langle u_i, u_j \rangle = 1 = -\langle v_i, v_j \rangle$ if i + j = p + 1 with all other scalar products zero. Hence $\{u_i\}$ (resp. $\{v_i\}$) form a skew-normal sequence of sign 1 (resp. -1). Now if we set $u_{p+1} = v_{p+1} = 0$, T acts on this basis as follows:

$$Tu_i = \frac{1}{\sqrt{2}} (\lambda z_i + z_{i+1} + \overline{\lambda}\overline{z}_i + \overline{z}_{i+1})$$

$$= \frac{1}{\sqrt{2}} ((a+ib)z_i + (a-ib)\overline{z}_i) + u_{i+1}$$

$$= au_i + \frac{b}{i\sqrt{2}} (\overline{z}_i - z_i) + u_{i+1}$$

$$= au_i - bv_i + u_{i+1}$$

Similarly

$$Tv_i = \frac{1}{i\sqrt{2}} (\lambda z_i + z_{i+1} - \overline{\lambda}\overline{z}_i - \overline{z}_{i+1})$$

= $\frac{1}{i\sqrt{2}} ((a+ib)z_i - (a-ib)\overline{z}_i) + v_{i+1}$
= $av_i + \frac{b}{\sqrt{2}} (z_i + \overline{z}_i) + v_{i+1}$
= $av_i + bu_i + v_{i+1}$

In the following proposition we use these basis to show that the real subspace spanned by an adapted p-cycle is non-degenerate.

Lemma C.3.6

Suppose V is a real scalar product space and T is a self-adjoint operator on V. Let x be a generator for a p-cycle of generalized eigenvectors for T with eigenvalue λ adapted to the scalar product. Let $z_i = U^{p-i}x$, H be the real subspace spanned by this cycle and $\epsilon = \langle z_1, z_p \rangle = \pm 1$. Then H is non-degenerate.

Furthermore if $\lambda \in \mathbb{R}$, then

$$\dim H = p$$

$$\inf H = \begin{cases} \lfloor \frac{p+1}{2} \rfloor & \text{if } \epsilon = -1 \\ p - \lfloor \frac{p+1}{2} \rfloor & \text{if } \epsilon = 1 \end{cases}$$

If $\lambda \in \mathbb{C} \setminus \mathbb{R}$, then

$$\dim H = 2p$$

ind $H = p$

PROOF If $\lambda \in \mathbb{R}$, then the result follows by Lemma 8.1.1 applied to z_1, \ldots, z_p . If $\lambda \in \mathbb{C} \setminus \mathbb{R}$, consider the real vectors $\{u_1, v_1, \ldots, u_p, v_p\}$ defined in Eqs. (C.3.2a) and (C.3.2b). The result follows by Lemma 8.1.1 applied to the sequence u_1, \ldots, u_p and then to v_1, \ldots, v_p .

The following theorem is from [O'N83, P. 260-261].

Theorem C.3.7 (Existence of the metric-Jordan canonical form [O'N83]) A linear operator T on a scalar product space V is self adjoint if and only if $V = \bigoplus_{i=1}^{k} V_i$ (hence each V_i is non-degenerate) where each subspace V_i is T-invariant and $T|_{V_i}$ has one of the following forms:

$$\begin{pmatrix} \lambda & 1 & & \\ & \lambda & \ddots & & 0 \\ & & \ddots & 1 & \\ & & & \lambda & 1 \\ & & & & \lambda \end{pmatrix}$$

relative to a skew-normal sequence $\{v_1, ..., v_p\}$ with all scalar products zero except $\langle v_i, v_j \rangle = \varepsilon = \pm 1$ when i + j = p + 1, or

$$\begin{pmatrix} a & b & 1 & 0 \\ -b & a & 0 & 1 & 0 \\ & \ddots & & & \\ & & a & b & 1 & 0 \\ & & & -b & a & 0 & 1 \\ 0 & & & & a & b \\ & & & & -b & a \end{pmatrix}$$

relative to a basis $\{u_1, v_1, ..., u_p, v_p\}$ with all scalar products zero except $\langle u_i, u_j \rangle = 1 = -\langle v_i, v_j \rangle$ if i + j = p + 1.

The index and dimension of V_i is determined by the blocks $T|_{V_i}$ due to Lemma C.3.6, hence we must have $\operatorname{ind} V = \sum_{i=1}^k \operatorname{ind} V_i$ and $n = \sum_{i=1}^k \dim V_i$.

PROOF We proceed by induction. If n = 1 then this result trivially holds. So suppose $n \ge 2$ and this result is true for all self-adjoint operators on scalar product spaces of dimension strictly less than n. Now we show that this holds when dim V = n.

Fix an eigenvalue λ for T (which exists after complexification of V if necessary). Let $U = (T - \lambda I)$. Let p be the smallest integer such that dim $N(U^p) = \dim N(U^{p+1})$, thus $K_{\lambda} = N(U^p)$. Then dim $N(U^{p-1}) < \dim N(U^p)$, hence $U^{p-1}|_{K_{\lambda}} \neq 0$, thus by Lemma C.3.4 there exists an $x \in K_{\lambda}$ such that $\langle U^{p-1}x, x \rangle \neq 0$. Note that by construction p is the smallest integer such that $U^p x = 0$, hence x generates a p-cycle of generalized eigenvectors with eigenvalue λ .

Hence by Lemma C.3.5, the *p*-cycle of generalized eigenvectors generated by x can be modified into another such *p*-cycle spanning the same subspace as the original and adapted to the scalar product. Thus we now assume that the *p*-cycle of generalized eigenvectors generated by x is adapted to the scalar product. Note that it follows by Lemma 8.1.1 that the set of p vectors in this cycle are linearly independent. Let H be the real subspace spanned by the *p*-cycle(s) generated by x if $\lambda \in \mathbb{R}$ or by x and its conjugate if $\lambda \in \mathbb{C} \setminus \mathbb{R}$. By Lemma C.3.6, H is non-degenerate and by construction H is T-invariant. If H = Vthen we are done, so assume $H \subsetneq V$. Then by property 1, H^{\perp} is an invariant subspace of T, and is complementary to H by non-degeneracy of H. Let $T' = T|_{H^{\perp}}$, then H^{\perp} is a scalar product space with $0 < \dim H^{\perp} < n$ and T' is a self-adjoint operator on H^{\perp} . Hence the induction hypothesis applies to T', in which case we conclude that the result holds for T.

The converse is also easily checked.

C.4 Uniqueness of the metric-Jordan canonical form

In this section T is self adjoint operator on a scalar product space V. We will show in what sense each self-adjoint operator T admits a "unique" metric-Jordan canonical form. We will do this by showing that the parameters appearing in any two canonical forms derived by Theorem C.3.7 must be the same. Then we will show how this result can be used to determine if two self-adjoint operators are isometrically equivalent.

Lemma C.4.1

Let $U = (T - \lambda I)$ for some eigenvalue λ , suppose x generates an adapted cycle of length p and sign ε and denote by $v_i = U^{p-i}x$. Also let H be the subspace spanned by this cycle. For any $0 \leq k \leq p-1$ define a symmetric bilinear form $[\cdot, \cdot]_k$ on H by

$$[x,y]_k = \left\langle U^k x, y \right\rangle$$

for $x, y \in H$. Then the number of zeros in any diagonal representation for $[\cdot, \cdot]_k$ is k. If the $\lambda \in \mathbb{R}$ then the number of negative entries in any diagonal representation for $[\cdot, \cdot]_k$ is

$$\begin{cases} \lfloor \frac{(p-k)+1}{2} \rfloor & \text{if } \epsilon = -1\\ (p-k) - \lfloor \frac{(p-k)+1}{2} \rfloor & \text{if } \epsilon = 1 \end{cases}$$

In conclusion, we see that the of invariants of $[\cdot, \cdot]_k$ depends only on p, k, ε .

PROOF We prove this by exhibiting a diagonal representation for $[\cdot, \cdot]$ restricted to H. First observe that for $i, j \in \{1, \dots, p\}$

$$[v_i, v_j]_k = \left\langle U^k U^{p-i} x, U^{p-j} x \right\rangle = \left\langle U^{2p+k-i-j} x, x \right\rangle$$

The above equation is non-zero iff

$$2p + k - i - j = p - 1$$

$$\Leftrightarrow p + k - i - j = -1$$

$$\Leftrightarrow i + j = p + k + 1$$

It follows that if i < k + 1, then $[v_i, v_j] = 0$ for any j. Now define vectors $v'_i = v_{i+k}$ for $i \in \{1, \ldots, p-k\}$. Then $\langle v'_i, v'_j \rangle \neq 0$ iff

$$i + k + j + k = p + k + 1$$
$$\Leftrightarrow i + j = p - k + 1$$

Hence v'_1, \ldots, v'_{p-k} (or equivalently v_{k+1}, \ldots, v_p) form a pseudo-orthonormal set of vectors with sign ε . Thus the formula for the number of negative entries when $\lambda \in \mathbb{R}$ follows from Lemma 8.1.1. Also observe that the number of zeros is k. Then by Sylvester's law of inertia it follows that the invariants of $[\cdot, \cdot]_k$ are given as above and hence depend only on p, k, ε .

For a real eigenvalue λ , an adapted cycle $x, Ux, \ldots, U^{p-1}x$ is called *positive* if $\langle U^{p-1}x, x \rangle = 1$ or negative if $\langle U^{p-1}x, x \rangle = -1$. By a metric-Jordan canonical basis, we mean one that is obtained from Theorem C.3.7.

Theorem C.4.2 (Uniqueness of the metric-Jordan canonical form)

Suppose λ is an eigenvalue of T. If $\lambda \in \mathbb{R}$, then the number of positive (negative) cycles in K_{λ} of a given length is independent of any metric-Jordan canonical basis. If $\lambda \in \mathbb{C} \setminus \mathbb{R}$, then the number of cycles in K_{λ} of a given length is independent of any metric-Jordan canonical basis.

PROOF Fix an eigenvalue λ of T and let $U = (T - \lambda I)$. Restrict the argument to the vector space K_{λ} , i.e set $V = K_{\lambda}$. Denote by $[\cdot, \cdot]_i$ the symmetric bilinear form given by:

$$[x,y]_i = \left\langle U^i x, y \right\rangle$$

for $x, y \in K_{\lambda}$. We will prove that the number of positive (negative) cycles of a given length depend only on the number of positive (negative) entries in a diagonal representation for $[\cdot, \cdot]_0, [\cdot, \cdot]_1, \ldots, [\cdot, \cdot]_n$. It is understood that the complex representations are chosen so that there are only positive or zero entries in it. It will follow by Sylvester's law of inertia that these signs are independent of any basis.

Fix a metric-Jordan canonical basis for $T|_{K_{\lambda}}$. It's known that $U^l = 0$ for any l > n, hence it follows that the number of cycles of length larger than n are determined by invariants of $[\cdot, \cdot]_l$ for l > n. Suppose inductively that the statement holds for all cycles of length strictly larger than p. We will now prove the statement for cycles of length p.

Denote by H the T-invariant non-degenerate (possibly zero) subspace spanned by all cycles of length strictly larger than p in this canonical basis. Observe that since H is T-invariant, it follows for any $l \ge 0$ that $[x, y]_l = 0$ for $x \in H$ and $y \in H^{\perp}$.

Case 1 There are no cycles of length p in this canonical basis.

Then note that $[x, y]_{p-1} \equiv 0$ for any $x, y \in H^{\perp}$ and if $H \neq 0$ the invariants of $[\cdot, \cdot]_{p-1}$ on H are uniquely determined by invariants of $[\cdot, \cdot]_l$ for $l \geq p$ by Lemma C.4.1. Also the invariants of $[\cdot, \cdot]_{p-1}$ over K_{λ} are determined by Sylvester's law of inertia, hence it follows that the number of cycles of length p are uniquely determined.

Case 2 Let x_1, \ldots, x_m be generators for cycles of length p in this canonical basis. For vectors from H^{\perp} in this canonical basis the only non-zero diagonal entries of $[\cdot, \cdot]_{p-1}$ are

$$[x_i, x_i]_{p-1} = \langle U^{p-1} x_i, x_i \rangle = \pm 1 \quad i = 1, \dots, m$$

Again, if $H \neq 0$ the invariants of $[\cdot, \cdot]_{p-1}$ on H are uniquely determined by invariants of $[\cdot, \cdot]_l$ for $l \geq p$ by Lemma C.4.1. The invariants of $[\cdot, \cdot]_{p-1}$ over K_{λ} are determined by Sylvester's law of inertia, hence it follows that the number of positive (and negative) cycles of length p are uniquely determined. Thus the result follows by induction on p.

We can now state what we mean by "the" metric-Jordan canonical form:

Definition C.4.3

Let T be a self-adjoint operator on a scalar product space V. To each adapted p-cycle of sign ε with eigenvalue $\lambda \in \mathbb{C}$ we associate a 3-tuple $(\lambda, p, \varepsilon)$. A canonical form given by Theorem C.3.7 gives an un-ordered list of such 3-tuples counting multiplicities. We call this list the metric-Jordan canonical form.

By the above theorem, it follows that the above definition is well defined, i.e. each self-adjoint operator T admits precisely one metric-Jordan canonical form. The following example shows that the signs appearing in these canonical forms add some subtleties:

Example C.4.4

Suppose V is Minkowski space equipped with the standard metric

$$g = \operatorname{diag}(-1, 1, \dots, 1)$$

For $\lambda_1 < \ldots < \lambda_n \in \mathbb{R}$ define two self-adjoint operators T_1 and T_2 as follows:

$$T_1 = \operatorname{diag}(\lambda_1, \lambda_2, \lambda_3, \dots, \lambda_n)$$
$$T_2 = \operatorname{diag}(\lambda_2, \lambda_1, \lambda_3, \dots, \lambda_n)$$

Now observe that even though T_1 and T_2 have the same eigenvalues, they have different metric-Jordan canonical forms. We will show shortly that T_1 and T_2 are isometrically inequivalent, in the sense that there is no $R \in O(V)$ which relates T_1 and T_2 by a similarity transformation.

Note that the above example is in sharp contrast with the Euclidean case where T_1 and T_2 as defined above would be isometrically equivalent.

Theorem C.4.5 (Isometric Equivalence of self-adjoint operators)

Suppose S and T are self-adjoint operators on a scalar product space V. Then S and T differ by an isometry $R \in O(V)$ iff they have the same metric-Jordan canonical form. \Box

PROOF It's clear that if S and T have the same metric-Jordan canonical form then there is an isometry $R \in O(V)$ which relates the two operators, namely the transformation that relates a metric-Jordan canonical basis of S to a metric-Jordan canonical basis of T.

Suppose T is given as follows relative to S:

$$T = RSR^{-1}$$

Let $\beta = \{v_1, \ldots, v_n\}$ be a canonical basis for S. Then consider the basis $\tilde{\beta} = \{Rv_1, \ldots, Rv_n\}$ for T. Since R is an isometry, we have

$$g|_{\tilde{\beta}} = g|_{\beta}$$

The equation relating T to S implies that

$$T|_{\tilde{\beta}} = S|_{\beta}$$

Hence S and T have the same metric-Jordan canonical form.

Appendix D

Warped products in Spaces of Constant Curvature

We will obtain the warped product decompositions of spaces of constant curvature (with arbitrary signature) in their natural models as subsets of pseudo-Euclidean space (see Section 8.3). This generalizes the corresponding result by Nolker in [Nol96] to arbitrary signatures, and has a similar level of detail. Although our derivation is complete in some sense, none is proven.

Our solution can fairly easily be deduced from that in [Nol96], and it is. Thus the goal of this appendix is to expose the results for reference purposes. We also note that the contents of this appendix are from [Raj14b].

Our primary motivation comes from Section 6.5, where it was shown that one can use the warped product decompositions of a given space to try to construct coordinates which separate the Hamilton-Jacobi equation. Thus these decompositions can be used to construct KEM coordinates.

Another motivation is because warped products are ubiquitous in applications of pseudo-Riemannian geometry, particularly in general relativity [DU05]. Hence it may be of some general interest to pursue this problem.

Our work is mainly self-contained, so it can be used as a reference. The material covered in Section 3.1 will be assumed throughout. We use the notation from Section 8.3 and assume knowledge of warped products from Section 3.5. We also use some results from the theory of pseudo-Riemannian submanifolds in [Che11], which is only necessary to understand certain proofs. We also assume the reader is familiar with [O'N83]. Familiarity with the article [Nol96] is useful but not necessary.

This appendix is organized as follows. In Appendices D.1 to D.3 we review preliminary theory on the spherical submanifolds and warped products in spaces of constant curvature. We give the warped product decompositions of pseudo-Euclidean space in Appendix D.4 and of spherical submanifolds of pseudo-Euclidean space in Appendix D.5.

is an optional section which gives the isometry groups of spherical submanifolds of pseudo-Euclidean space, which builds on results from [O'N83].

D.1 Spherical Submanifolds of Spaces of Constant Curvature

In this section κ is allowed to be zero. The following optional proposition relates umbilical submanifolds to spherical ones in spaces of constant curvature.

Proposition D.1.1

Any umbilical submanifold of $\mathbb{E}^n_{\nu}(\kappa)$ with dimension greater than one is necessarily spherical.

PROOF This follows from Lemma 3.2 (a) in [Che11].

Here we state some properties of spherical submanifolds in spaces of constant curvature.

Proposition D.1.2 (Spherical Submanifolds in Spaces of Constant Curvature) Let $\phi : N \to \mathbb{E}^n_{\nu}(\kappa)^{\circ}$ be an isometric immersion of a pseudo-Riemannian manifold N. If N is a spherical submanifold, then

- (a) $\langle H, H \rangle$ is constant.
- (b) N is of constant curvature $\kappa + \langle H, H \rangle$

PROOF Lemma 3.2 from [Che11].

D.2 Standard spherical submanifolds of pseudo-Euclidean space

We collect some properties of $\mathbb{E}^n_{\nu}(\kappa)$ in the following proposition.

Proposition D.2.1

Let r denote the dilatational vector field and $r^2 = \langle r, r \rangle$. Fix $r^2 \in \mathbb{R}$, the following are true about $\mathbb{E}^n_{\nu}(\frac{1}{r^2})$

(a) It is a spherical submanifold with mean curvature normal

$$H = -\frac{r}{r^2} \tag{D.2.1}$$

(b) It has constant curvature $\frac{1}{r^2}$ and is geodesically complete.

PROOF The first follows from [O'N83, Lemma 4.27]. When dim $\mathbb{E}_{\nu}^{n}(\frac{1}{r^{2}}) > 1$, the first result together with Proposition D.1.1 shows that $\mathbb{E}_{\nu}^{n}(\frac{1}{r^{2}})$ is a spherical submanifold. In any case, it follows from Eq. (D.2.1) that $\mathbb{E}_{\nu}^{n}(\frac{1}{r^{2}})$ is a spherical submanifold. Hence the second result follows from Proposition D.1.2 (b). It follows from lemma 4.29 in [O'N83] that $\mathbb{E}_{\nu}^{n}(\frac{1}{r^{2}})$ is geodesically complete.

We collect similar properties of \mathbb{P}^n_{ν} .

Proposition D.2.2

The following are true about \mathbb{P}^n_{ν} with mean curvature vector -a:

(a) It is a spherical submanifold with mean curvature normal

$$H = -a$$

(b) It is globally isometric to \mathbb{E}^n_{ν} .

PROOF Consider the map ψ given by Eq. (8.3.1). It then follows that for $v \in TV$,

$$\psi_* v = v - \langle v, x \rangle a$$

The above equation shows that the induced metric at each point is the induced metric on V. Hence $\mathbb{P}^n_{\nu} \simeq \mathbb{E}^n_{\nu}$. Now to calculate the second fundamental form, we have for $w, v \in TV$:

$$\nabla_{\psi_* w} \psi_* v = \nabla_w v - \langle \nabla_w v, x \rangle a - \langle v, w \rangle a$$
$$= \psi_* \nabla_w v - \langle v, w \rangle a$$

Hence it follows that \mathbb{P}^n_{ν} is umbilical with mean curvature vector -a. Since -a is covariantly constant, it follows that \mathbb{P}^n_{ν} is spherical.

D.3 Warped product decompositions of Spaces of Constant Curvature

In this section we study warped product decompositions of $\mathbb{E}^n_{\nu}(\kappa)$ where κ may equal zero. Recall that warped products were introduced in Section 3.5. A warped product decomposition of a given pseudo-Riemannian manifold M is a warped product which is

(locally) isometric to M. Let $M = M_0 \times_{\rho_1} M_1 \times \cdots \times_{\rho_k} M_k$ be a warped product and $\psi: M \to \mathbb{E}^n_{\nu}(\kappa)$ a warped product decomposition of $\mathbb{E}^n_{\nu}(\kappa)$. Recall that M_0 is a geodesic submanifold and each M_i for i > 0 is a spherical submanifold (see Theorem 3.5.10). Fix $\bar{p} \in \psi(M)$. Let $H_i = -\nabla(\log \rho_i)$ be the mean curvature vector field associated to the canonical foliation L_i generated by M_i (see Proposition 3.5.9). Let $V_i := T_{\bar{p}_i}M_i$ for each i and $z_i := H_i|_{\bar{p}} \in V_0$ for i > 0. Then note that

$$T_{\overline{p}}M = \bigoplus_{i=0}^k V_i$$

Equation (3.5.4) implies that the mean curvature vectors satisfy the following equation for $i \neq j$:

$$\langle z_i, z_j \rangle = -\kappa \tag{D.3.1}$$

In this case we say that ψ is a warped product decomposition of $\mathbb{E}_{\nu}^{n}(\kappa)$ associated with the initial data $(\bar{p}; \bigoplus_{i=0}^{k} V_{i}; a_{1}, ..., a_{k})$ where $a_{i} := \kappa \bar{p} - z_{i}$. Conversely, let $\bar{p} \in \mathbb{E}_{\nu}^{n}(\kappa)$ where $n \geq 2$ and consider the following decomposition of

Conversely, let $\overline{p} \in \mathbb{E}_{\nu}^{n}(\kappa)$ where $n \geq 2$ and consider the following decomposition of $T_{\overline{p}}\mathbb{E}_{\nu}^{n}(\kappa)$, $T_{\overline{p}}\mathbb{E}_{\nu}^{n} = \bigoplus_{i=0}^{k} V_{i}$ into non-trivial subspaces (hence non-degenerate) with $k \geq 1$. Suppose $z_{1}, ..., z_{k} \in V_{0}$ satisfy Eq. (D.3.1). Let $a_{i} := \kappa \overline{p} - z_{i}$ and assume additionally that the subset of non-zero a_{i} are linearly independent. In this case, we say that $(\overline{p}; \bigoplus_{i=0}^{k} V_{i}; a_{1}, ..., a_{k})$ are *initial data* for a warped product decomposition of $\mathbb{E}_{\nu}^{n}(\kappa)$. We will show later on that in a space of constant curvature there always exists a warped product decomposition with a given initial data. It follows from Theorem 3.2.6 that in the category of Riemannian manifolds with n > 2, this property characterizes spaces of constant curvature.

The additional condition requiring the a_i to be linearly independent trivially holds in Euclidean space and in motivating applications. The reason we make this assumption will become more apparent later. Here is an optional lemma, which is given for completeness, and hints at why we make this assumption.

Lemma D.3.1

Suppose a_1, \ldots, a_k are linearly independent pair-wise orthogonal lightlike vectors. Then there exist vectors b_1, \ldots, b_k such that $\langle a_i, b_j \rangle = \delta_{ij}$ and $\langle b_i, b_j \rangle = 0$.

PROOF Suppose to the contrary that for any b_1 satisfying $\langle b_1, a_i \rangle = 0$ for i > 1 we have $\langle b_1, a_1 \rangle = 0$. Thus

$$\bigcap_{i=2}^k a_i^{\perp} \subseteq a_1^{\perp}$$

Define $T: V \to \mathbb{R}^k$ by:

$$T(v) = (\langle a_1, v \rangle, \dots, \langle a_k, v \rangle)$$

By hypothesis we have dim ker $T \ge n - (k - 1)$, hence dim Im $T \le k - 1$ by the rank-nullity theorem. Thus $a_1^{\flat}, \ldots, a_k^{\flat}$ are linearly dependent, a contradiction.

Thus there exists $b_1 \in \bigcap_{i=2}^k a_i^{\perp}$ with $\langle a_1, b_1 \rangle = 1$. The result then follows by induction. Indeed the next step is to find b_2 by applying the above result to $\{a_2, \ldots, a_k\} \subset$ span $\{a_1, b_1\}^{\perp}$ making use of the fact that span $\{a_1, b_1\}$ is non-degenerate by construction.

It has been shown by Nolker in [Nol96] that given any initial data for Riemannian spaces of constant curvature, there exists a unique warped product decomposition associated with the initial data. In this appendix we will show that given any initial data for a WP-decomposition of $\mathbb{E}_{\nu}^{n}(\kappa)$, there exists a WP-decomposition associated with the initial data. This WP-decomposition is probably uniquely determined but we don't use or prove this supposition.

Equation (3.5.3) implies that the Hessian H of each warping function ρ of a space of constant curvature satisfies the following equation on the geodesic factor:

$$H(X,Y) = -\kappa\rho \left\langle X,Y \right\rangle$$

This proves the following fact:

Lemma D.3.2

A space of constant non-zero curvature does not admit product decompositions.

D.4 Warped product decompositions of pseudo-Euclidean space

D.4.1 Spherical submanifolds of pseudo-Euclidean space

We first describe the spherical submanifolds of pseudo-Euclidean space. The following theorem is a generalization of Lemma 5 in [Nol96] to pseudo-Euclidean space.

Theorem D.4.1 (Spherical submanifolds of \mathbb{E}^n_{ν})

Let $\overline{p} \in$ be arbitrary, $V \subseteq \mathbb{E}_{\nu}^{n}$ a non-degenerate subspace with $m := \dim V \ge 1$, $\mu := \operatorname{ind} V$ and $z \in V^{\perp}$. Let $\tilde{\kappa} := z^{2}$, a := -z and $W = \mathbb{R}a \oplus V$. There is exactly one m-dimensional connected and geodesically complete spherical submanifold \tilde{N} with $\overline{p} \in \tilde{N}$, $T_{\overline{p}}\tilde{N} = V$ and having mean curvature vector at \overline{p} , z. \tilde{N} is an open submanifold of N; N is referred to as the spherical submanifold determined by (\overline{p}, V, a) and is given as follows (where \simeq means isometric to): (a) a = 0 iff N is geodesic, in this case $N \simeq \mathbb{E}^m_\mu$

$$N = \overline{p} + V$$

- (b) a is timelike, then $\mu \leq \nu 1$ and $N \simeq H^m_{\mu}(\tilde{\kappa})$
- (c) a is spacelike, then $N \simeq S^m_{\mu}(\tilde{\kappa})$ For cases (b) and (c), let $c = \overline{p} - \frac{a}{\tilde{\kappa}}$ be the center of N, then N is given as follows:

$$N = c + \{ p \in W \mid p^2 = \frac{1}{\tilde{\kappa}} \}$$

(d) a is lightlike, then $\mu \leq \nu - 1$ and $N \simeq E_{\mu}^{m}$

$$N = \overline{p} + \{ p - \frac{1}{2}p^2 a \mid p \in V \}$$

Remark D.4.2

 $\tilde{N} = N$ except in the following two cases (which are anti-isometric): When $N \simeq H_0^m(\tilde{\kappa})$ or $N \simeq S_m^m(\tilde{\kappa})$, N is disconnected [O'N83, Section 4.6] and so \tilde{N} is given as follows:

$$\tilde{N} = N \cap (c + \{p \in W \mid \langle a, p \rangle > 0\})$$

PROOF First we note that it suffices to show that there exists a single connected and geodesically complete sphere satisfying the initial conditions. By Lemma 3.2.5, it must be unique.

Item (a) is clear. For Items (b) and (c), it follows from Proposition D.2.1 that N is a sphere and the initial conditions are easily checked. The connectedness properties follow from lemma 4.25 in [O'N83]. It follows from lemma 4.29 in [O'N83] that N is geodesically complete.

Item (d) follows from Proposition D.2.2.

Remark D.4.3

See [Che11] for a different proof.

Since circles are one dimensional spherical submanifolds, we can use the above theorem to describe the circles in pseudo-Euclidean space.

Example D.4.4 (Proper Circles in pseudo-Euclidean space)

Suppose $(\bar{p}, \bar{V}, k\bar{Y})$ are initial conditions for a proper circle as in Lemma 3.2.1 with $\varepsilon_0 := \bar{V}^2 = \pm 1$, $\varepsilon_1 := \bar{Y}^2 = \pm 1$ and $||k\bar{Y}|| \neq 0$. We now describe the circle determined by this data.

By Example 3.2.2 the proper circle determined by these initial conditions determine a spherical submanifold of \mathbb{E}^n_{ν} characterized by $(\bar{p}, \mathbb{R}\bar{V}, \varepsilon_0 k \bar{Y})$. Now let $H := \varepsilon_0 k \bar{Y}$, $\kappa := \langle H, H \rangle = \varepsilon_1 k^2$ and $c := \bar{p} + \frac{H}{\kappa} = \bar{p} - \frac{\varepsilon_0 \varepsilon_1 \bar{Y}}{k}$.

Case 1 Euclidean circle, $\gamma = \mathbb{S}^1$: $\varepsilon_0 = \varepsilon_1 = \pm 1$

$$\gamma(t) = c + \frac{1}{k}(\sin(kt)\bar{V} - \cos(kt)\bar{Y})$$

Case 2 Hyperbolic circle, $\gamma = H^1$: $\varepsilon_0 = 1, \varepsilon_1 = -1$

Case 3 de Sitter circle, $\gamma = \mathbb{S}_1^1$: $\varepsilon_0 = -1, \varepsilon_1 = 1$

In the last two cases (which are anti-isometric), γ is given as follows:

$$\gamma(t) = c + \frac{1}{k} (\sinh(kt)\bar{V} - \varepsilon_0\varepsilon_1\cosh(kt)\bar{Y})$$

One can give a similar example for geodesics and null circles.

D.4.2 Warped product decompositions of pseudo-Euclidean space

Our classification of the warped product decompositions of \mathbb{E}_{ν}^{n} is based on the fact that a specification of the tangent spaces and mean curvature normals of the spherical foliations of a warped product at one point \overline{p} , uniquely determines a warped product decomposition in a neighborhood of \overline{p} . We now carry out this classification as follows. Suppose $\psi : N_0 \times_{\rho_1} N_1 \times \cdots \times_{\rho_k} N_k \to \mathbb{E}_{\nu}^{n}$ is a warped product decomposition of \mathbb{E}_{ν}^{n} associated with initial data $(\overline{p}; \bigoplus_{i=0}^{k} V_i; -z_1, ..., -z_k)$. By Eq. (D.3.1), the mean curvature vectors at \overline{p} satisfy the following equation:

$$\langle z_i, z_j \rangle = 0 \quad i \neq j$$

We now only consider the case $\nu \leq 1$ as the other signatures are straightforward generalizations of these standard ones. In this case, we will use Theorem D.4.1 to classify N_i up to homothety as follows. Say $z_1, ..., z_l = 0$ and the remaining are non-zero, then for i = 1, ..., l the N_i are pair-wise orthogonal planes passing through \overline{p} . We now consider the remaining possibilities:

Case 1 Since the z_i are orthogonal, there is at most one lightlike direction, say z_{l+1} . The remaining lightlike z_i are proportional to z_{l+1} , but since we assume the non-zero z_i are linearly independent, we will work with only one lightlike vector z_{l+1} . Then N_{l+1} a paraboloid isometric to Euclidean space. The orthogonality relations force the remaining z_i to be space-like and hence the remaining N_i are Euclidean spheres.

- **Case 2** Similarly, at most one of the z_i can be timelike, say z_{l+1} . Then N_{l+1} is isometric to hyperbolic space. The orthogonality relations force the remaining z_i to be space-like and hence the remaining N_i are Euclidean spheres.
- **Case 3** The remaining z_i are spacelike. If $\operatorname{ind} V_0 = 1$ or $\operatorname{ind} V_0 = 0$ in Euclidean space, then the remaining N_i are Euclidean spheres. If $\operatorname{ind} V_0 = 0$ in Minkowski space, then $\operatorname{ind} V_j = 1$ for precisely one $j \ge 1$, then N_j is de Sitter space while the remaining N_i are Euclidean spheres.
- **Case 4** All z_i are zero. Then each N_i is an affine plane and the warped product is a product of planes.

We summarize our findings in the following theorem.

Theorem D.4.5 (Warped products in \mathbb{E}^n and M^n)

Suppose $N = N_0 \times_{\rho_1} N_1 \times \cdots \times_{\rho_k} N_k$ is a proper warped product decomposition of an open subset of \mathbb{E}^n_{ν} . If at most one of the N_i are intrinsically flat, then N is isometric to one of the following warped products:

If \mathbb{E}^n_{ν} is Euclidean space:

$$\mathbb{E}^m \times_{\rho_1} S^{n_1} \times \cdots \times_{\rho_n} S^n$$

If \mathbb{E}^n_{ν} is Minkowski space:

 $M^{m} \times_{\lambda_{1}} \mathbb{E}^{n_{1}} \times_{\rho_{2}} S^{n_{2}} \times \cdots \times_{\rho_{s}} S^{n_{s}}$ $M^{m} \times_{\tau_{1}} H^{n_{1}} \times_{\rho_{2}} S^{n_{2}} \times \cdots \times_{\rho_{s}} S^{n_{s}}$ $\mathbb{E}^{m} \times_{\rho_{1}} dS^{n_{1}} \times_{\rho_{2}} S^{n_{2}} \times \cdots \times_{\rho_{s}} S^{n_{s}}$ $M^{m} \times_{\rho_{1}} S^{n_{1}} \times_{\rho_{2}} S^{n_{2}} \times \cdots \times_{\rho_{s}} S^{n_{s}}$

where $\nabla \rho_i, \nabla \tau_i, \nabla \lambda_i$ is a spacelike, timelike, lightlike vector field respectively.

The above theorem shows that there are at 1 and 4 distinct types of proper singly warped products in Euclidean and Minkowski space respectively. One can show that the multiply warped products can be built up from the singly warped products by iteratively decomposing the geodesic factor of the warped product into another warped product which is "compatible" with the original. Thus we only describe a special subset of warped products for simplicity.

The following theorem describes this interesting class of warped products. Its proof can be deduced from Theorem 7 in [Nol96]. It is a generalization of that theorem to pseudo-Euclidean space. Theorem D.4.6 (Standard Warped Products in \mathbb{E}^n_{ν} [Nol96])

Fix $\overline{p} \in \mathbb{E}_{\nu}^{n}$ where $n \geq 2$ and the following decomposition of $T_{\overline{p}}\mathbb{E}_{\nu}^{n}$, $T_{\overline{p}}\mathbb{E}_{\nu}^{n} = \bigoplus_{i=0}^{k} V_{i}$ into non-trivial subspaces (hence non-degenerate) with $k \geq 1$. Suppose $a_{1}, ..., a_{k} \in V_{0}$ are pair-wise orthogonal. Let $\kappa_{i} := a_{i}^{2}$ and $\epsilon_{i} := \operatorname{sgn} \kappa_{i}$. We consider the following warped decompositions:

non-null warped decomposition $Let \ \mu \geq 0$

$$\kappa_1 \leq \cdots \leq \kappa_\mu < 0 < \kappa_{\mu+1} \leq \cdots \leq \kappa_k$$

In this case, let $c = \overline{p} - \sum_{i=1}^{k} \frac{a_i}{\kappa_i}$ and $c_i = \overline{p} - \frac{a_i}{\kappa_i}$ for each i = 1, ..., k.

null warped decomposition k = 1, $a_1 := a$, $\kappa_1 = a^2 = 0$ but $a \neq 0$, *i.e.* a is lightlike.

In this case, fix a lightlike vector $b \in V_0$ such that $\langle a, b \rangle = 1$ and let $c = \overline{p} - b$.

Now, define N_0 as follows:

$$N_0 := c + \{ p \in V_0 | \langle a_i, p \rangle > 0 \text{ for all } i \}$$

Note that N_0 is an open subset of the plane determined by $(\overline{p}, V_0, 0)$. For i = 1, ..., k, let N_i be the spherical submanifold of \mathbb{E}^n_{ν} determined by (\overline{p}, V_i, a_i) . Define

$$\rho_i : \begin{cases} N_0 & \to \mathbb{R}_+ \\ p_0 & \mapsto \langle a_i, p_0 - c \rangle = 1 + \langle a_i, p_0 - \overline{p} \rangle \end{cases}$$

For i = 1, ..., k, let $W_i := \mathbb{R}a_i \oplus V_i$ and $P : \mathbb{E}^n_{\nu} \to W_i$ be the orthogonal projection. Then the map

$$\psi: \begin{cases} N_0 \times_{\rho_1} N_1 \times \dots \times_{\rho_k} N_k & \to \mathbb{E}_{\nu}^n \\ (p_0, \dots, p_k) & \mapsto p_0 + \sum_{i=1}^k \rho_i(p_0)(p_i - \overline{p}) \end{cases}$$
(D.4.1)

is an isometry onto the following set¹:

$$\operatorname{Im}(\psi) := \begin{cases} c + \{p \in \mathbb{E}_{\nu}^{n} \mid \operatorname{sgn}(P_{i}(p))^{2} = \epsilon_{i}, \text{ for each } i = 1, ..., k\} & \text{non-null case} \\ c + \{p \in \mathbb{E}_{\nu}^{n} \mid \langle a, p \rangle > 0\} & \text{null case} \end{cases}$$

Im (ψ) is dense in \mathbb{E}^n_{ν} only for a non-null warped decomposition when each W_i for i = 1, ..., k is Euclidean or anti-isometric to a Euclidean space.

¹Note that sgn 0 = 0, otherwise for $a \neq 0$, sgn a is the sign of a.

Remark D.4.7

Note that $\rho_i(\bar{p}) = 1$ for i = 1, ..., k. Also for each $p_i \in N_i$ we have $\psi(\bar{p}, ..., p_i, ..., \bar{p}) = p_i$, hence $\psi(\bar{p}, ..., \bar{p}) = \bar{p}$.

If the N_i are required to be connected, then $\operatorname{Im}(\psi)$ has to be modified slightly. For each N_i that is disconnected (see the remark following Theorem D.4.1), in addition to the restriction that $\operatorname{sgn}(P_i(p))^2 = \epsilon_i$ in the definition of $\operatorname{Im}(\psi)$, add the restriction that $\langle a_i, P_i(p) \rangle > 0$.

PROOF The idea of this proof is to assume Eq. (D.4.1) holds and then expand it by choosing an appropriate basis for V_0 . In the expanded form we will be able to prove all the claims made in the theorem. We have the following two cases.

The non-null case: Let W_0 be the orthogonal complement of $\bigoplus_{i=1}^{n} \mathbb{R}a_i$ in V_0 ; which is well defined since $a_i^2 \neq 0$ for each *i*. Thus we have that

$$V_0 = W_0 \oplus \bigoplus_{i=1}^k \mathbb{R}a_i \tag{D.4.2}$$

which implies:

$$\mathbb{E}_{\nu}^{n} = \bigoplus_{i=0}^{k} V_{i}$$
$$= W_{0} \oplus \bigoplus_{i=1}^{k} \mathbb{R}a_{i} \oplus \bigoplus_{i=1}^{k} V_{i}$$
$$= W_{0} \oplus \bigoplus_{i=1}^{k} (\mathbb{R}a_{i} \oplus V_{i})$$
$$= W_{0} \oplus \bigoplus_{i=1}^{k} W_{i}$$

Now let $P_i : \mathbb{E}_{\nu}^n \to W_i$ denote the orthogonal projection for i = 0, ..., k. Then from Eq. (D.4.2), we get the following orthogonal decomposition of V_0 which will be used extensively:

$$p = P_0 p + \sum_{i=1}^k \frac{1}{\kappa_i} \langle a_i, p \rangle a_i \quad \text{for all } p \in V_0$$
 (D.4.3)

Now we use the above decomposition of $p \in V_0$ to write $\psi(p_0, ..., p_k)$ adapted to the following affine decomposition of \mathbb{E}^n_{ν}

$$\mathbb{E}_{\nu}^{n} = c + \bigoplus_{i=0}^{k} W_{i}$$
We get the following for $(p_0, ..., p_k) \in N_0 \times \cdots \times N_k$

$$\psi(p_0, ..., p_k) = c + P_0(p_0 - c) + \sum_{i=1}^k \langle a_i, p_0 - c \rangle (p_i - c_i)$$
(D.4.4)

Now we prove that ψ is injective: Let $(p_0, ..., p_k), (q_0, ..., q_k) \in N_0 \times \cdots \times N_k$ and suppose that $\psi(p_0, ..., p_k) = \psi(q_0, ..., q_k)$. From Eq. (D.4.4), we deduce the following:

$$P_0(p_0 - c) = P_0(q_0 - c)$$

 $\langle a_i, p_0 - c \rangle (p_i - c_i) = \langle a_i, q_0 - c \rangle (q_i - c_i)$

Since for each i = 1, ..., k, $(p_i - c_i)^2 = (q_i - c_i)^2 = \frac{1}{\kappa_i}$ and $\langle a_i, p_0 - c \rangle$, $\langle a_i, q_0 - c \rangle \in \mathbb{R}^+$, we deduce that $p_i = q_i$. Then Eq. (D.4.3) shows $p_0 = q_0$.

Now for surjectivity: From Eq. (D.4.4) it's clear that $\psi(N_0 \times \cdots \times N_k) \subseteq \text{Im}(\psi)$. Given $p \in \text{Im}(\psi)$, using Eq. (D.4.4) in conjunction with Eq. (D.4.3) we can readily calculate the inverse $q = \psi^{-1}(p)$ given in components as follows:

$$\begin{aligned} q_0 &= c + P_0(p-c) + \sum_{i=1}^k \frac{\epsilon_i}{\sqrt{|\kappa_i|}} \left\| P_i(p-c) \right\| a_i \\ q_i &= c_i + \frac{1}{\sqrt{|\kappa_i|}} \frac{P_i(p-c)}{\|P_i(p-c)\|} \quad i = 1, ..., k \end{aligned}$$

Now we show that ψ is an isometry. Note first that for $p = (p_0, ..., p_k) \in N_0 \times \cdots \times N_k$ and $v = (v_0, ..., v_k) \in T_p(N_0 \times \cdots \times N_k)$, Eq. (D.4.4) implies that

$$\psi_* v = P_0 v_0 + \sum_{i=1}^k \langle a_i, v_0 \rangle (p_i - c_i) + \sum_{i=1}^k \langle a_i, p_0 - c \rangle v_i$$

Hence also using the fact that:

$$\langle p_i - c_i, v_i \rangle = 0$$
 for $i = 1, ..., k$

we get:

$$\begin{aligned} (\psi_* v)^2 &= (P_0 v_0)^2 + \sum_{i=1}^k (\langle a_i, v_0 \rangle (p_i - c_i))^2 + \sum_{i=1}^k (\langle a_i, p_0 - c \rangle v_i)^2 \\ &= (P_0 v_0)^2 + \sum_{i=1}^k \frac{\langle a_i, v_0 \rangle^2}{\kappa_i} + \sum_{i=1}^k \rho_i (p_0)^2 v_i^2 \\ &= (P_0 v_0 + \sum_{i=1}^k \frac{\langle a_i, v_0 \rangle}{\kappa_i} a_i)^2 + \sum_{i=1}^k \rho_i (p_0)^2 v_i^2 \\ &= v_0^2 + \sum_{i=1}^k \rho_i (p_0)^2 v_i^2 \end{aligned}$$

where the last two lines follow from the fact that $v_0 \in V_0$ and Eq. (D.4.3).

The null case: We have the following decomposition of V_0 :

 $V_0 = W_0 \oplus \operatorname{span}\{a, b\}$

where W_0 is the orthogonal complement of span $\{a, b\}$ relative to V_0 . Let P_i denote the orthogonal projection onto W_0 for i = 0 and onto V_1 for i = 1. Then for $p \in \mathbb{E}^n_{\nu}$:

$$p = P_0 p + \langle b, p \rangle a + \langle a, p \rangle b + P_1 p \tag{D.4.5}$$

and

$$p^{2} = (P_{0}p)^{2} + 2\langle b, p \rangle \langle a, p \rangle + (P_{1}p)^{2}$$
(D.4.6)

Let $c = \overline{p} - b$, $\tilde{p}_0 = p_0 - c$ and $\tilde{p}_1 = p_1 - \overline{p}$, then for $(p_0, p_1) \in N_0 \times N_1$

$$\psi(p_0, p_1) = c + P_0(\tilde{p}_0) + (\langle b, \tilde{p}_0 \rangle - \frac{1}{2} \langle a, \tilde{p}_0 \rangle (P_1(\tilde{p}_1))^2) a + \langle a, \tilde{p}_0 \rangle b + \langle a, \tilde{p}_0 \rangle P_1(\tilde{p}_1)$$
(D.4.7)

where the last two lines follow from Eq. (D.4.6).

Injectivity of ψ follows readily from Eq. (D.4.7).

Now for surjectivity: From Eq. (D.4.7) it's clear that $\psi(N_0 \times N_1) \subseteq \text{Im}(\psi)$. Given $p \in \text{Im}(\psi)$, let $\tilde{p} = p - c$, then using Eq. (D.4.4) in conjunction with Eq. (D.4.3) we can readily calculate the inverse $q = \psi^{-1}(p)$ given in components as follows:

$$q_{0} = c + P_{0}(\tilde{p}) + (\langle b, \tilde{p} \rangle + \frac{1}{2 \langle a, \tilde{p} \rangle} (P_{1}(\tilde{p}))^{2})a + \langle a, \tilde{p} \rangle b$$

$$q_{1} = \overline{p} + \frac{1}{\langle a, \tilde{p} \rangle} P_{1}(\tilde{p}) - \frac{1}{2 \langle a, \tilde{p} \rangle^{2}} (P_{1}(\tilde{p}))^{2}a$$
(D.4.8)

Now we show that ψ is an isometry. Note first that for $p = (p_0, p_1) \in N_0 \times N_1$ and $v = (v_0, v_1) \in T_p(N_0 \times N_1)$, Eq. (D.4.7) implies that

$$\psi_* v = P_0 v_0 + (\langle b, v_0 \rangle - \frac{1}{2} \langle a, v_0 \rangle (P_1(\tilde{p}_1))^2 - \langle a, \tilde{p}_0 \rangle \langle P_1 \tilde{p}_1, P_1 v_1 \rangle) a + \langle a, v_0 \rangle b + \langle a, v_0 \rangle P_1(\tilde{p}_1) + \langle a, \tilde{p}_0 \rangle P_1(v_1)$$

Hence we get that:

$$(\psi_* v)^2 = (P_0 v_0)^2 + 2 \langle b, v_0 \rangle \langle a, v_0 \rangle + \langle a, \tilde{p}_0 \rangle^2 (P_1 v_1)^2$$

= $(\langle b, v_0 \rangle a + \langle a, v_0 \rangle b + P_0 v_0)^2 + \rho(p_0)^2 (P_1 v_1)^2$
= $v_0^2 + \rho(p_0)^2 v_1^2$

where the last two lines follow from the fact that $v_0 \in V_0$, Eq. (D.4.6) and since P_1 : $T_{p_1}N_1 \to V_1$ is an isometry for each $p_1 \in N_1$.

Definition D.4.8

We call ψ the warped product decomposition of \mathbb{E}^n_{ν} determined by $(\overline{p}; N_1, ..., N_k)$ or by $(\overline{p}; \bigoplus_{i=0}^{\kappa} V_i; a_1, ..., a_k)$ as in the hypothesis of the above theorem.

Note that in the context of the above definition, the warped product decomposition is proper if each $a_i \neq 0$. For actual calculations we wish to work with canonical forms. The following definition will be particularly convenient.

Definition D.4.9 (Canonical form for Warped products of \mathbb{E}^n_{ν}) We say that a proper warped product decomposition of \mathbb{E}^n_{ν} determined by $(\bar{p}; \bigoplus_{i=0}^k V_i; a_1, ..., a_k)$ is in canonical form if: $\bar{p} \in V_0$ and $\langle \bar{p}, a_i \rangle = 1$.

We note here that any proper warped product decomposition ψ of \mathbb{E}_{ν}^{n} can be brought into canonical form by the translation $\psi \to \psi - c$. This follows from the above theorem by observing that $\langle \bar{p} - c, a_i \rangle = 1$ for each i > 0. The following corollary gives the standard warped product decompositions of \mathbb{E}^n_{ν} in canonical form.

Corollary D.4.10 (Canonical form for Warped products of \mathbb{E}_{ν}^{n}) Let ψ be a proper warped product decomposition of \mathbb{E}_{ν}^{n} determined by $(\bar{p}; \bigoplus_{i=0}^{k} V_{i}; a_{1}, ..., a_{k})$ which is in canonical form.

Then the conclusions of Theorem D.4.6 simplify as follows:

$$N_0 = \{ p \in V_0 | \langle a_i, p \rangle > 0 \text{ for all } i \}$$

$$\rho_i = \langle a_i, p_0 \rangle$$

$$\operatorname{Im}(\psi) = \begin{cases} \{p \in \mathbb{E}_{\nu}^{n} \mid \operatorname{sgn}(P_{i}(p))^{2} = \epsilon_{i}, \text{ for each } i = 1, ..., k\} & \text{non-null case} \\ \{p \in \mathbb{E}_{\nu}^{n} \mid \langle a, p \rangle > 0\} & \text{null case} \end{cases}$$

For $(p_0, ..., p_k) \in N_0 \times \cdots \times N_k$, ψ has the following form:

$$\psi(p_0, ..., p_k) = \begin{cases} P_0 p_0 + \sum_{i=1}^k \langle a_i, p_0 \rangle (p_i - c_i) & \text{non-null case} \\ P_0 p_0 + (\langle b, p_0 \rangle - \frac{1}{2} \langle a, p_0 \rangle (P_1(p_1))^2) a + \langle a, p_0 \rangle b + \langle a, p_0 \rangle P_1 p_1 & \text{null case} \end{cases}$$

Furthermore, the following equation holds:

$$\psi(p_0, \dots, p_k)^2 = p_0^2 \tag{D.4.9}$$

PROOF First note that for the non-null case:

$$\langle a_i, c \rangle = \left\langle a_i, \bar{p} - \frac{a_i}{\kappa_i} \right\rangle$$

= $1 - \frac{\langle a_i, a_i \rangle}{\kappa_i}$
= 0

Similarly for the null-case:

$$\langle c, a \rangle = \langle \bar{p} - b, a \rangle$$

= 0

Thus we see that

$$N_0 = c + \{ p \in V_0 | \langle a_i, p \rangle > 0 \text{ for all i } \}$$
$$= \{ p \in V_0 | \langle a_i, p \rangle > 0 \text{ for all i } \}$$

The formula for $\text{Im}(\psi)$ follows similarly. Clearly $\rho_i(p_0) = \langle a_i, p_0 - c \rangle = \langle a_i, p_0 \rangle$. Now we break into cases.

The non-null case:

Note that $c \in W_0$, so $P_0c = c$, hence

$$\psi(p_0, ..., p_k) = c + P_0(p_0 - c) + \sum_{i=1}^k \langle a_i, p_0 - c \rangle (p_i - c_i)$$

= $c + P_0(p_0 - c) + \sum_{i=1}^k \langle a_i, p_0 \rangle (p_i - c_i) - \sum_{i=1}^k \langle a_i, c \rangle (p_i - c_i)$
= $P_0 p_0 + \sum_{i=1}^k \langle a_i, p_0 \rangle (p_i - c_i)$

It follows from the above equation that $\psi(p_0, ..., p_k)^2 = p_0^2$. The null case:

By Eq. (D.4.5), c can be written as follows:

$$c = P_0 c + \langle b, c \rangle a$$

Thus Eq. (D.4.7) reduces to

$$\psi(p_0, p_1) = c + P_0(p_0) - P_0 c - \langle b, c \rangle a + (\langle b, p_0 \rangle - \frac{1}{2} \langle a, p_0 \rangle (P_1(\tilde{p}_1))^2) a + \langle a, p_0 \rangle b + \langle a, p_0 \rangle P_1(\tilde{p}_1) = P_0(p_0) + (\langle b, p_0 \rangle - \frac{1}{2} \langle a, p_0 \rangle (P_1(p_1))^2) a + \langle a, p_0 \rangle b + \langle a, p_0 \rangle P_1(p_1)$$

In the last equation we used the fact that $P_1\tilde{p}_1 = P_1p_1$ since $\bar{p} \in V_0$.

Finally, it follows from the above equation that $\psi(p_0, p_1)^2 = p_0^2$.

D.5 Isometry groups of Spherical submanifolds of pseudo-Euclidean space*

Warped products of spaces of constant curvature are closely related to certain integrable subgroups of the isometry group due to the following fact [Zeg11]:

Proposition D.5.1 (Lifting isometries from Killing distributions)

Let $M = B \times_{\rho} F$ be a warped product and suppose $\tilde{f} : F \to F$ is an isometry of F. Then the lift f defined by

$$f(x,y) := (x, \tilde{f}(y)), \quad (x,y) \in B \times F$$

is an isometry of M.

Theorem 5.1 in [Zeg11] shows conversely that given a certain integrable group action on a pseudo-Riemannian manifold M, one can obtain a warped product whose spherical foliation is invariant under the action of the group. Hence in spaces of constant curvature one can show that the above property characterizes warped products. In view of this, in this section we state the isometry groups which preserve the spherical submanifolds of pseudo-Euclidean space.

The isometry groups of H^n_{ν} and S^n_{ν} are well documented, see for example [O'N83, section 9.2]. In this section we will describe the isometry group of \mathbb{P}^n_{ν} . This is given in [Nol96, lemma 6] for the case when $\nu = 0$; that proof should generalize easily. Although, we will give a different proof (motivated by Nolker's results) using our knowledge of warped product decompositions and Proposition D.5.1.

We denote the homogeneous isometry group (i.e. orthogonal group) of $\mathbb{E}_{\nu+1}^{n+2}$ by $O_{\nu+1}(n+2)$ (see [O'N83]). Then we have the following:

Proposition D.5.2

Let -a be the mean curvature vector of \mathbb{P}^n_{ν} . The isometry group of \mathbb{P}^n_{ν} is:

$$I(\mathbb{P}^{n}_{\nu}) = \{ T \in O_{\nu+1}(n+2) \mid Ta = a \}$$

Furthermore suppose we fix an embedding of \mathbb{E}_{ν}^{n} by fixing a subspace $V \simeq \mathbb{E}_{\nu}^{n}$, then for $p \in V$ and $\tilde{p} \in V^{\perp}$ we have the following Lie group isomorphism:

$$\phi: \begin{cases} O(V) \ltimes V & \to I(\mathbb{P}^n_{\nu}) \\ (B, v) & \mapsto \phi(B, v) \end{cases}$$

where

$$\phi(B,v)(p+\tilde{p}) = \tilde{p} + Bp + \langle a, \tilde{p} \rangle v - (\langle Bp, v \rangle + \frac{1}{2} \langle a, \tilde{p} \rangle v^2))a$$

PROOF Consider the warped product decomposition:

$$\psi(p_0, p) = \langle a, p_0 \rangle b + \langle a, p_0 \rangle p + (\langle b, p_0 \rangle - \frac{1}{2} \langle a, p_0 \rangle p^2) a$$

for $p_0 \in N_0$ and $p \in V$. Note that

$$\psi(b,p) = b + p - \frac{1}{2}p^2a$$

is a map onto \mathbb{P}^n_{ν} . As in Eq. (D.4.8), one can deduce that the inverse of ψ is

$$q_0 = (\langle b, p \rangle + \frac{1}{2 \langle a, p \rangle} (Pp)^2) a + \langle a, p \rangle b$$
$$q_1 = \frac{1}{\langle a, p \rangle} Pp$$

Let $B \in O(V)$, $v \in V$ and define Tp = Bp + v for $p \in V$. Now define \hat{T} by:

$$\hat{T}: \begin{cases} \mathbb{E}_{\nu+1}^{n+2} & \to \mathbb{E}_{\nu+1}^{n+2} \\ p & \mapsto \psi(p_0, Tp_1) \end{cases}$$

Since ψ is a warped product decomposition, it follows by Proposition D.5.1 that \hat{T} induces an isometry of some open subset of $\mathbb{E}_{\nu+1}^{n+2}$ onto itself. We will now calculate \hat{T} explicitly.

For arbitrary $x \in \mathbb{E}_{\nu+1}^{n+2}$ write $x = p + \tilde{p}$ where $p \in V$ and $\tilde{p} \in V^{\perp}$.

$$(Tq_1)^2 = \left\| \frac{1}{\langle a, x \rangle} Bp + v \right\|$$
$$= \left(\frac{1}{\langle a, x \rangle^2} (Px)^2 + \frac{2}{\langle a, x \rangle} \langle Bp, v \rangle + v^2 \right)$$

$$\begin{split} \psi(q_0, Tq_1) &= \langle a, q_0 \rangle \, b + \langle a, q_0 \rangle \, Tq_1 + (\langle b, q_0 \rangle - \frac{1}{2} \langle a, q_0 \rangle \, (Tq_1)^2) a \\ &= \langle a, x \rangle \, b + \langle a, x \rangle \, Tq_1 + (\langle b, x \rangle + \frac{1}{2 \langle a, x \rangle} (Px)^2 - \frac{1}{2} \langle a, x \rangle \, (Tq_1)^2) a \\ &= \langle a, x \rangle \, b + \langle a, x \rangle \, Tq_1 + (\langle b, x \rangle - (\langle Bp, v \rangle + \frac{1}{2} \langle a, x \rangle \, v^2)) a \\ &= \langle b, x \rangle \, a + \langle a, x \rangle \, b + Bp + \langle a, x \rangle \, v - (\langle Bp, v \rangle + \frac{1}{2} \langle a, x \rangle \, v^2)) a \\ &= \tilde{p} + Bp + \langle a, x \rangle \, v - (\langle Bp, v \rangle + \frac{1}{2} \langle a, x \rangle \, v^2)) a \end{split}$$

Hence if p := Px and $\tilde{p} := (I - P)x$ then

$$\hat{T}x = \tilde{p} + Bp + \langle a, x \rangle v - (\langle Bp, v \rangle + \frac{1}{2} \langle a, x \rangle v^2))a$$

Thus since \hat{T} is a linear isometry of $\mathbb{E}_{\nu+1}^{n+2}$ it follows that $\hat{T} \in O(\mathbb{E}_{\nu+1}^{n+2})$. Also \hat{T} clearly fixes a so $\hat{T} \in I(\mathbb{P}_{\nu}^{n})$.

Let the map ϕ be as in the hypothesis. Note that $\phi(B, v) = \hat{T}$. ϕ is a Lie group homomorphism, since

$$(TS)x = \psi(x_0, TSx_1)$$

= $\psi((\hat{S}x)_0, T((\hat{S}x)_1))$
= $\hat{T}\hat{S}x$

By definition of \hat{T} it follows that ϕ is injective.

To show that ϕ is surjective, fix $T \in I(\mathbb{P}^n_{\nu})$. Consider the decomposition:

$$p = \langle a, p \rangle b + \langle b, p \rangle a + Pp, \quad p \in \mathbb{E}_{\nu}^{n}$$

Using the fact that $(Tp)^2 = p^2$ with the above decomposition we obtain the following equations:

$$p^{2} = (Tp)^{2} = 2 \langle a, Tp \rangle \langle b, Tp \rangle + (PTp)^{2}$$
$$p \in V \Rightarrow p^{2} = (PTp)^{2}$$
$$p = b \Rightarrow 0 = b^{2} = 2 \langle b, Tb \rangle + (PTb)^{2}$$
$$p = \psi(b, \tilde{p}) \Rightarrow 0 = p^{2} = 2 \langle b, T\tilde{p} \rangle + \langle PTb, PT\tilde{p} \rangle$$

The second equation implies that $PT \in O(V)$. We claim that $\phi(PT, PTb) = T$. This can be seen by decomposing the action of T with respect to the above decomposition and then using the last three equations and the fact that $T \in I(\mathbb{P}^n_{\nu})$.

Hence ϕ is a Lie group isomorphism.

We also note that if $\psi : \mathbb{E}_{\nu}^{n} \to \mathbb{P}_{\nu}^{n}$ is the standard embedding from Eq. (8.3.1), then ψ is equivariant, i.e. in the notation of the proof $\psi \circ T(p) = \hat{T} \circ \psi(p)$.

D.6 Warped Product decompositions of Spherical submanifolds of Pseudo-Euclidean space

D.6.1 Spherical submanifolds of $\mathbb{E}_{\nu}^{n}(\kappa)$

In this section we will classify the spherical submanifolds of $\mathbb{E}_{\nu}^{n}(\kappa)$. In particular we will show that they all have the form $\mathbb{E}_{\nu}^{n}(\kappa) \cap (\bar{p} + W)$ for some $\bar{p} \in \mathbb{E}_{\nu}^{n}$ and some subspace W. Although not all spherical submanifolds will have this form since we are only considering the case of pseudo-Riemannian manifolds. We will see that all spherical submanifolds of $\mathbb{E}_{\nu}^{n}(\kappa)$ arise as restrictions of spherical submanifolds of \mathbb{E}_{ν}^{n} .

The following lemma concerns a submanifold N of $\mathbb{E}_{\nu}^{n}(\kappa)$. We denote by H' the mean curvature normal of N in $\mathbb{E}_{\nu}^{n}(\kappa)$ and H the mean curvature normal of N in \mathbb{E}_{ν}^{n} . Similar definitions hold for the second fundamental forms h' and h. As usual r denotes the dilatational vector field.

Lemma D.6.1

If N is a submanifold of $\mathbb{E}^n_{\nu}(\kappa)$ then the following equations hold:

$$h(X,Y) = h'(X,Y) - \langle X,Y \rangle \frac{r}{r^2}$$

$$H = H' - \frac{r}{r^2}$$
(D.6.1)

In particular, N is an umbilical submanifold of $\mathbb{E}_{\nu}^{n}(\kappa)$ iff it is an umbilical submanifold of \mathbb{E}_{ν}^{n} . In fact, N is a spherical submanifold of $\mathbb{E}_{\nu}^{n}(\kappa)$ iff it is a spherical submanifold of \mathbb{E}_{ν}^{n} .

PROOF These formulas follow from lemma 3.5 and corollary 3.1 in [Che11].

Now we consider the problem of finding the sphere in $\mathbb{E}_{\nu}^{n}(\kappa)$ passing through a point \overline{p} with tangent space V and mean curvature normal z at \overline{p} . We make this precise as follows.

Let $\overline{p} \in \mathbb{E}_{\nu}^{n}(\kappa)$ be arbitrary, $V \subset T_{\overline{p}}\mathbb{E}_{\nu}^{n}(\kappa)$ a non-degenerate subspace with $m := \dim V \ge 1$, $\mu := \operatorname{ind} V$ and $z \in V^{\perp} \cap T_{\overline{p}}\mathbb{E}_{\nu}^{n}(\kappa)$.

Now let $a := \kappa \overline{p} - z$. Then assuming this data defines a submanifold of $\mathbb{E}_{\nu}^{n}(\kappa)$, we use Eq. (D.6.1) to obtain the mean curvature normal in \mathbb{E}_{ν}^{n} at \overline{p} , which is given as follows:

$$z - \kappa \overline{p} = -a$$

Then this determines a sphere in \mathbb{E}_{ν}^{n} with initial data (\bar{p}, V, a) by Theorem D.4.1. Note that $a \neq 0$. In the following theorem we will show that this sphere in \mathbb{E}_{ν}^{n} is in fact the sphere in $\mathbb{E}_{\nu}^{n}(\kappa)$ determined by (\bar{p}, V, a) . First let $W := \mathbb{R}a \oplus V$ and $\tilde{\kappa} := a^{2}$.

Theorem D.6.2 (Spherical submanifolds of $\mathbb{E}^n_{\nu}(\kappa)$)

There is exactly one m-dimensional connected and geodesically complete spherical submanifold \tilde{N} of $\mathbb{E}_{\nu}^{n}(\kappa)$ with $\bar{p} \in \tilde{N}$, $T_{\bar{p}}\tilde{N} = V$ and having mean curvature vector at \bar{p} , z. \tilde{N} is an open submanifold of N; $N = \mathbb{E}_{\nu}^{n}(\kappa) \cap (\bar{p} + W)$ is the spherical submanifold determined by (\bar{p}, V, a) in $\mathbb{E}_{\nu}^{n}(\kappa)$ and \mathbb{E}_{ν}^{n} . In fact, N can be given explicitly as follows (where \simeq means isometric to):

- (a) a is timelike, then $\mu \leq \nu 1$ and $N \simeq H^m_{\mu}(\tilde{\kappa})$
- (b) a is spacelike, then $N \simeq S^m_{\mu}(\tilde{\kappa})$ For cases (b) and (c), let $c = \overline{p} - \frac{a}{\tilde{\kappa}}$ be the center of N, then N is given as follows:

$$N = c + \{ p \in W \mid p^2 = \frac{1}{\tilde{\kappa}} \}$$

(c) a is lightlike, then $\mu \leq \nu - 1$ and $N \simeq \mathbb{E}_{\mu}^{m}$

$$N = \overline{p} + \{ p - \frac{1}{2}p^2 a \mid p \in V \}$$

Remark D.6.3

The relationship between \tilde{N} and N follows from Remark D.4.2, since the above spheres are the spheres in \mathbb{E}^n_{ν} determined by (\bar{p}, V, a) in Theorem D.4.1.

Remark D.6.4

N is a geodesic submanifold of $\mathbb{E}^n_{\nu}(\kappa)$ iff z = 0 iff W intersects the origin.

PROOF First we note that it suffices to show that there exists a single connected and geodesically complete sphere satisfying the initial conditions. By Lemma 3.2.5, it must be unique.

The three definitions of N given above follow directly from Theorem D.4.1 with initial data (\overline{p}, V, a) . Hence the relevant intrinsic properties of N follow from Theorem D.4.1. For the remainder of the proof we will assume N is given by those definitions, and we will prove the following.

Claim D.6.4.1 $N = \mathbb{E}_{\nu}^{n}(\kappa) \cap (\bar{p} + W)$

PROOF Note that the following equations are satisfied: $\langle \bar{p}, \bar{p} \rangle = \frac{1}{\kappa}, \langle a, \bar{p} \rangle = 1$

First we consider the case of Items (a) and (b). We can always write $p = c + \tilde{p}$ where $\tilde{p} \in W$. Also note that the following holds:

$$\begin{aligned} \langle c, c \rangle &= \langle \bar{p}, \bar{p} \rangle - 2 \left\langle \bar{p}, \frac{a}{\tilde{\kappa}} \right\rangle + \frac{1}{\tilde{\kappa}^2} \left\langle a, a \right\rangle \\ &= \frac{1}{\kappa} - 2\frac{1}{\tilde{\kappa}} + \frac{1}{\tilde{\kappa}} \\ &= \frac{1}{\kappa} - \frac{1}{\tilde{\kappa}} \end{aligned}$$

Then since $\langle c, a \rangle = 0$, we have

$$\begin{split} \langle p,p
angle &= c^2 - 2 \left< c, \tilde{p} \right> + \tilde{p}^2 \ &= rac{1}{\kappa} - rac{1}{\tilde{\kappa}} + \tilde{p}^2 \end{split}$$

The above equation shows that $p \in \mathbb{E}^n_{\nu}(\kappa)$ iff $\tilde{p} \in W(\tilde{\kappa})$, which proves the result. Now for Item (c). We can always write $p = \bar{p} + v + wa$ where $v \in V$ and $w \in \mathbb{R}$. Hence

$$\langle p, p \rangle = \frac{1}{\kappa} + v^2 + 2u$$

The above equation shows that $p \in \mathbb{E}^n_{\nu}(\kappa)$ iff $w = -\frac{1}{2}v^2$, which proves the result.

Thus we have shown that N is a spherical submanifold of \mathbb{E}_{ν}^{n} contained in $\mathbb{E}_{\nu}^{n}(\kappa)$. It then follows from Lemma D.6.1 that N is a spherical submanifold of $\mathbb{E}_{\nu}^{n}(\kappa)$ with mean curvature normal z at \bar{p} . Furthermore by Proposition D.1.2 (b), this sphere is of constant curvature $\kappa + z^{2} = a^{2} = \tilde{\kappa}$.

Now we mention when we can restrict a sphere in \mathbb{E}_{ν}^{n} to one in $\mathbb{E}_{\nu}^{n}(\kappa)$. Suppose $(\bar{p}, V, -z)$ determines a sphere in \mathbb{E}_{ν}^{n} with $\bar{p} \in \mathbb{E}_{\nu}^{n}(\kappa)$ and $V \subset T_{\bar{p}}\mathbb{E}_{\nu}^{n}(\kappa)$. Then define z' as

$$z' := z + \kappa \overline{p} \in V^{\perp}$$

We know that $\bar{p} \in V^{\perp}$ and $z \in V^{\perp}$ by hypothesis. In order for $\langle z', \bar{p} \rangle = 0$, we must additionally assume $\langle z, \bar{p} \rangle = -1$. In this case, $(\bar{p}, V, -z)$ define initial data for a sphere in $\mathbb{E}^{n}_{\nu}(\kappa)$. It follows from the above theorem that this sphere is simultaneously the sphere in $\mathbb{E}^{n}_{\nu}(\kappa)$ determined by $(\bar{p}, V, -z)$.

D.6.2 Warped Product decompositions of Spherical submanifolds of Pseudo-Euclidean space

Suppose $\psi : N_0 \times_{\rho_1} N_1 \times \cdots \times_{\rho_k} N_k \to \mathbb{E}^n_{\nu}(\kappa)$ is a warped product decomposition of $\mathbb{E}^n_{\nu}(\kappa)$ associated with initial data $(\bar{p}; \bigoplus_{i=0}^k V_i; a_1, ..., a_k)$ where each $a_i = \kappa \bar{p} - z_i$. By Eq. (D.3.1), the mean curvature vectors at \bar{p} satisfy the following equation:

$$\langle z_i, z_j \rangle = -\kappa \quad i \neq j$$

By Theorem D.6.2, $L_i(\bar{p})$ is a spherical submanifold of \mathbb{E}^n_{ν} determined by (\bar{p}, V_i, a_i) . Note that $a_i \neq 0$. Furthermore the above equation implies that

$$\langle a_i, a_j \rangle = 0 \quad i \neq j$$

Also recall that by assumption, the a_i are linearly independent. Thus the initial data $(\bar{p}; (\mathbb{R}\bar{p} \oplus V_0) \oplus V_1 \oplus \cdots \oplus V_k; a_1, \ldots, a_k)$ determines a proper warped product decomposition of the ambient space \mathbb{E}_{ν}^n . Furthermore, we note that this warped product decomposition is in canonical form; the canonical form was specifically designed to have this property. We now consider the converse problem of restricting a warped product decomposition in \mathbb{E}_{ν}^n to $\mathbb{E}_{\nu}^n(\kappa)$. The following theorem shows that this is always possible when the warped product in \mathbb{E}_{ν}^n is proper and in canonical form:

Theorem D.6.5 (Restricting Warped products to $\mathbb{E}^n_{\nu}(\kappa)$)

Let ψ be a proper warped product decomposition of \mathbb{E}_{ν}^{n} associated with $(\bar{p}; \bigoplus_{i=0}^{\kappa} V_{i}; a_{1}, ..., a_{k})$ which is in canonical form. Suppose $\kappa^{-1} := \bar{p}^{2} \neq 0$ and let $N' := N_{0}(\kappa) \times_{\rho_{1}} N_{1} \times \cdots \times_{\rho_{k}} N_{k}$. Note that $N_{0}(\kappa)$ is an open subset of the sphere in $\mathbb{E}_{\nu}^{n}(\kappa)$ determined by $(\bar{p}, (\bar{p}^{\perp} \cap V_{0}), 0)$. Then $\phi : N' \to \mathbb{E}_{\nu}^{n}(\kappa)$ defined by $\phi := \psi|_{N'}$ is a warped product decomposition of $\mathbb{E}_{\nu}^{n}(\kappa)$ determined by $(\bar{p}; (\bar{p}^{\perp} \cap V_{0}) \bigoplus_{i=1}^{k} V_{i}; a_{1}, ..., a_{k})$.

Furthermore for any point $p \in \text{Im}(\psi)$ with $p^2 \neq 0$, the leaf of the foliation induced by N_i , $L_i(p)$, is simultaneously a sphere in \mathbb{E}^n_{ν} and $\mathbb{E}^n_{\nu}(\frac{1}{p^2})$. Also ψ is in canonical form at every $p \in \text{Im}(\psi)$.

PROOF By Eq. (D.4.9) in Corollary D.4.10 it follows that ϕ is a diffeomorphism onto $\phi(N') \subseteq \mathbb{E}_{\nu}^{n}(\kappa)$. Clearly the restriction of the metric on N to N' is still a warped product metric. Hence it follows that ϕ is a warped product decomposition of $\mathbb{E}_{\nu}^{n}(\kappa)$, i.e. an isometry from a warped product. Furthermore by Theorem D.6.2 it follows that for each $i > 0, N_i$ is also the sphere in $\mathbb{E}_{\nu}^{n}(\kappa)$ determined by (\bar{p}, V_i, z_i) .

Now for the last point, fix $p \in \text{Im}(\psi)$ with $p^2 \neq 0$. Let \tilde{r} be the dilatational vector field in N_0 and $r := \psi_* \tilde{r}$. Can show that r is also the dilatational vector field in \mathbb{E}^n_{ν} (e.g.

see Eq. (D.4.1)). Now if $\rho_i = \langle \tilde{r}, a_i \rangle$, then it follows from Proposition 3.5.9 (2) that the mean curvature vector H_i is:

$$H_i = -\frac{a_i}{\rho_i}$$

Hence $\langle \tilde{r}, -H_i \rangle = 1$. Thus at p, by making the identification r = p, we see that TN_i is orthogonal to $p = \psi_* \tilde{p}$ and $\langle \tilde{p}, -H_i \rangle = 1$. It follows from the discussion following Theorem D.6.2 that $L_i(p)$ is also a sphere in $\mathbb{E}^n_{\nu}(\frac{1}{p^2})$.

Remark D.6.6 (Connectedness)

Remark D.4.7 gives the appropriate modifications of $\operatorname{Im}(\psi)$ when each N_i for i > 0 are required to be connected. When $N_0(\kappa) \simeq H_0^m(\tilde{\kappa})$ or $N_0(\kappa) \simeq S_m^m(\tilde{\kappa})$, $N_0(\kappa)$ is disconnected [O'N83, Section 4.6] and so we modify $N_0(\kappa)$ as follows: By Theorem D.6.2 it follows that $N_0(\kappa)$ is an open subset of the sphere in \mathbb{E}_{ν}^n determined by $(\bar{p}, (\bar{p}^{\perp} \cap V_0), \kappa \bar{p})$. Thus to enforce connectedness, it follows by Remark D.4.2 that we must replace $N_0(\kappa)$ with

$$N_0(\kappa) \cap \{ p \in V_0 \mid \langle \kappa \bar{p}, p \rangle > 0 \}$$

Proof $a = \kappa \bar{p}, \ \tilde{\kappa} = a^2 = \kappa$

$$c = \bar{p} - \frac{a}{\tilde{\kappa}}$$
$$= 0$$

Now we show the effect of this on $\phi(N')$ when $\nu = 1$.

Case 1 a_i is time-like for some i

 $N_0(\kappa)$ is automatically connected since $N_0(\kappa) \subset \{p \in V_0 \mid \langle a_i, p \rangle > 0 \text{ for each i}\}$, then $N_0(\kappa) \subset \{p \in V_0 \mid \langle \kappa \bar{p}, p \rangle > 0\}$ since $\langle a_i, \kappa \bar{p} \rangle = \kappa < 0$ (see [O'N83, P. 143] and Nolker's proof of the hyperbolic case).

Case 2 null case, $a := a_1$ is light-like

 $N_0(\kappa)$ is connected here as well. First observe that it follows from the equation for ψ in Corollary D.4.10 that

$$\langle a, \psi(p_0, p_1) \rangle = \langle a, p_0 \rangle > 0$$

Thus it follows that $N_0(\kappa)$ and $\phi(N')$ are in the time cone opposite to a (see remarks preceding Nolker's proof of the hyperbolic case). Thus it follows that $N_0(\kappa) \subset \{p \in V_0 \mid \langle \kappa \bar{p}, p \rangle > 0\}$, so $N_0(\kappa)$ and hence $\phi(N')$ are connected.

In this case $\phi(N')$ is the maximal connected component of $\mathbb{E}^n_{\nu}(\kappa)$ passing through \bar{p} .

Case 3 a_i is space-like for each i

First observe that it follows from the proof of Corollary D.4.10 that $c = P_0 c \in W_0$ and $p_i - c_i \in W_i$ for i > 0, hence

$$\langle c, \psi(p_0, \dots, p_k) \rangle = \left\langle c, P_0 p_0 + \sum_{i=1}^k \langle a_i, p_0 \rangle (p_i - c_i) \right\rangle$$
$$= \langle c, P_0 p_0 \rangle$$
$$= \langle c, p_0 \rangle$$

Also since since $\langle c, a_i \rangle = 0$, we have that

$$\begin{split} \langle c, c \rangle &= \langle c, \bar{p} \rangle \\ &= \left\langle \bar{p} - \sum_{i=1}^{k} \frac{a_i}{\kappa_i}, \bar{p} \right\rangle \\ &= \frac{1}{\kappa} - \sum_{i=1}^{k} \frac{1}{\kappa_i} \\ &< 0 \end{split}$$

In other words, c is time-like. Also the above equation shows that $\langle c, \kappa \bar{p} \rangle > 0$, thus c and $\kappa \bar{p}$ are in opposite time cones (see [O'N83, P. 143]). Hence,

$$\{p \in V_0 \mid \langle \kappa \bar{p}, p \rangle > 0\} = \{p \in V_0 \mid \langle \kappa c, p \rangle > 0\}$$

Thus since $\langle c, \psi(p_0, \ldots, p_k) \rangle = \langle c, p_0 \rangle$, we see that $\phi(N')$ becomes

$$\phi(N') \cap \{ p \in \mathbb{E}^n_{\nu} \mid \langle \kappa c, p \rangle > 0 \}$$

In the following corollary we show how to obtain any warped product decomposition of $\mathbb{E}_{\nu}^{n}(\kappa)$ by restricting an appropriate warped product decomposition of \mathbb{E}_{ν}^{n} . The "appropriate" warped product product decomposition of \mathbb{E}_{ν}^{n} to restrict follows from the discussion preceding the above theorem. Thus together with the above theorem, we have the following corollary:

Corollary D.6.7 (Warped product decompositions of $\mathbb{E}_{\nu}^{n}(\kappa)$) Suppose $(\bar{p}; \bigoplus_{i=0}^{n} V_{i}; a_{1}, ..., a_{k})$ define initial data for a warped product decomposition of $\mathbb{E}_{\nu}^{n}(\kappa)$. Let ϕ be the warped product decomposition of $\mathbb{E}^n_{\nu}(\kappa)$ given in the above theorem by re-

stricting the warped product decomposition of \mathbb{E}_{ν}^{n} with initial data $(\bar{p}; (\mathbb{R}\bar{p} \oplus V_{0}) \bigoplus_{i=1}^{k} V_{i}; a_{1}, \dots, a_{k})$. Then ϕ is a warped product decomposition of $\mathbb{E}_{\nu}^{n}(\kappa)$ determined by $(\bar{p}; \bigoplus_{i=0}^{k} V_{i}; a_{1}, \dots, a_{k})$.

We now mention which warped product decompositions are possible in $\mathbb{E}^n_{\mu}(\kappa)$. We do this by finding out when it's possible to restrict a warped product on the ambient space. Given a warped product $(V_0 \oplus V_1 \oplus \cdots \oplus V_k; a_1, \ldots, a_k)$ passing through an arbitrary point in \mathbb{E}^n_{ν} , in order to restrict it to $\mathbb{E}^n_{\nu}(\kappa)$, we need it to pass through a point $\bar{p} \in V_0$ with $\bar{p}^2 = \kappa$ satisfying $\langle \bar{p}, a_i \rangle = 1$. So for a fixed $\kappa \neq 0$, we enumerate the distinct warped products in \mathbb{E}^n_{ν} , expand $\bar{p} \in V_0$ so that $\langle \bar{p}, a_i \rangle = 1$ and determine if it's possible for $\bar{p}^2 = \kappa$. By making use of Theorem D.4.5, we have the following results:

Theorem D.6.8 (Warped products in Spherical submanifolds of \mathbb{E}^n and M^n) Suppose $N = N_0 \times_{\rho_1} N_1 \times \cdots \times_{\rho_k} N_k$ is a warped product decomposition of an open subset of a spherical submanifold of \mathbb{E}^n or M^n . This warped product is necessarily proper. If at most one of the N_i are intrinsically flat, then N is isometric to one of the following warped products:

In S^n :

 $S^m \times_{\rho_1} S^{n_1} \times \cdots \times_{\rho_s} S^{n_s}$

In dS^n :

$$dS^{m} \times_{\lambda_{1}} \mathbb{E}^{n_{1}} \times_{\rho_{2}} S^{n_{2}} \times \cdots \times_{\rho_{s}} S^{n_{s}}$$
$$dS^{m} \times_{\tau_{1}} H^{n_{1}} \times_{\rho_{2}} S^{n_{2}} \times \cdots \times_{\rho_{s}} S^{n_{s}}$$
$$S^{m} \times_{\rho_{1}} dS^{n_{1}} \times_{\rho_{2}} S^{n_{2}} \times \cdots \times_{\rho_{s}} S^{n_{s}}$$
$$dS^{m} \times_{\rho_{1}} S^{n_{1}} \times_{\rho_{2}} S^{n_{2}} \times \cdots \times_{\rho_{s}} S^{n_{s}}$$

In H^n :

$$H^{m} \times_{\lambda_{1}} \mathbb{E}^{n_{1}} \times_{\rho_{2}} S^{n_{2}} \times \cdots \times_{\rho_{s}} S^{n_{s}}$$
$$H^{m} \times_{\tau_{1}} H^{n_{1}} \times_{\rho_{2}} S^{n_{2}} \times \cdots \times_{\rho_{s}} S^{n_{s}}$$
$$H^{m} \times_{\rho_{1}} S^{n_{1}} \times_{\rho_{2}} S^{n_{2}} \times \cdots \times_{\rho_{s}} S^{n_{s}}$$

where $\nabla \rho_i, \nabla \tau_i, \nabla \lambda_i$ is a spacelike, timelike, lightlike vector field respectively. **PROOF** For the proof that the warped products are proper, see Lemma D.3.2.

Appendix E

Lexicographic ordering of complex numbers

Complex numbers can be given a natural lexicographic ordering (as in dictionaries) by using their Cartesian product structure:

Definition E.0.9

Suppose $\lambda = a + ib$ and $\omega = c + id$ are complex numbers. We write $\lambda < \omega$ if: b < d or (b = d and a < c)

In the following we use "xor" to mean exclusive or and "or" has its standard meaning. Suppose $\lambda, \omega, \nu \in \mathbb{C}$ and $a \in \mathbb{R}^+$, one can check that this ordering has the following properties:

trichotomy: $\lambda = \omega \operatorname{xor} \lambda < \omega \operatorname{xor} \omega < \lambda$ transitivity: If $\lambda < \omega$ and $\omega < \nu$ then $\lambda < \nu$ translation invariance: If $\lambda < \omega$ then $\lambda + \nu < \omega + \nu$ dilatation invariance: If $\lambda < \omega$ then $a\lambda < a\omega$ skew symmetry: If $\lambda < \omega$ then $-\omega < -\lambda$

Furthermore we note that if $\lambda, \omega \in \mathbb{R}$ then this ordering reduces to the natural ordering of real numbers.

References

- [Ami03] A. V. Aminova. "Projective Transformations of Pseudo-Riemannian Manifolds". In: Journal of Mathematical Sciences 113.3 (2003), pp. 367–470 (cit. on pp. 30, 100).
- [AMR01] R Abraham, J. E. A. Marsden, and T. S. A. Ratiu. Manifolds, Tensor Analysis, and Applications. 3rd ed. Appplied mathematical sciences v. 75. Springer-Verlag, 2001, p. 631 (cit. on p. 7).
- [ANY90] N Abe, Y Nakanishi, and S Yamaguchi. "Circles and spheres in pseudo-Riemannian geometry". English. In: aequationes mathematicae 39.2-3 (1990), pp. 134–145 (cit. on pp. xv, 30, 32, 33).
- [Arn89] V. I. Arnol'd. *Mathematical Methods of Classical Mechanics*. Graduate Texts in Mathematics. Springer, 1989, p. 508 (cit. on pp. 43, 62, 66, 67).
- [BCR00] S Benenti, C Chanu, and G. Rastelli. "The super-separability of the three-body inverse-square Calogero system". In: Journal of Mathematical Physics 41.7 (2000), p. 4654 (cit. on p. 221).
- [BCR02] S Benenti, C Chanu, and G. Rastelli. "Remarks on the connection between the additive separation of the HamiltonJacobi equation and the multiplicative separation of the Schrodinger equation. I. The completeness and Robertson conditions". In: Journal of Mathematical Physics 43.11 (2002), p. 5183 (cit. on p. 57).
- [Ben02] S Benenti. "An outline of the geometrical theory of the separation of variables in the Hamilton-Jacobi and Schrödinger equations". In: SPT 2002: Symmetry and perturbation theory (Cala ... (2002), pp. 2–7 (cit. on p. 106).
- [Ben04] S Benenti. "Separability in Riemannian manifolds". In: *Phil. Trans. Roy. Soc. A, to appear* d (2004) (cit. on pp. xvi, 9, 72, 107, 108, 113, 151, 218).
- [Ben05] S Benenti. "Special Symmetric Two-tensors, Equivalent Dynamical Systems, Cofactor and Bi-cofactor Systems". In: Acta Applicandae Mathematicae 87.1-3 (May 2005), pp. 33–91 (cit. on pp. xvi, 3, 12, 87–89, 91, 93, 94, 97, 98, 107, 112, 113, 151, 159, 214).

[Ben08]	S Benenti. "Algebraic Construction of the Quadratic First Integrals for a Special Class of Orthogonal Separable Systems". In: <i>Symmetries and Overdetermined</i> <i>Systems of Partial Differential Equations</i> . Ed. by M. Eastwood and W. Miller. Vol. 144. The IMA Volumes in Mathematics and its Applications. Springer New York, 2008, pp. 277–304 (cit. on p. 229).
[Ben89]	S Benenti. <i>Stäckel systems and Killing tensors</i> . Jan. 1989 (cit. on pp. xv, 43, 46, 62, 63, 66, 69, 70, 83).
[Ben91]	S Benenti. Separation of variables in the geodesic Hamilton-Jacobi equation. Progress in mathematics. Birkh{ä}user, 1991. Chap. 1, pp. 1–36 (cit. on pp. 66, 68, 73, 75).
[Ben92a]	S Benenti. "Inertia tensors and Stackel systems in the Euclidean spaces". In: <i>Rend. Semin. Mat. Univ. Polit. Torino</i> 50 (1992), pp. 315–341 (cit. on pp. 3, 4, 13, 87, 106, 137, 151).
[Ben92b]	S Benenti. "Orthogonal separation of variables on manifolds with constant curvature". In: <i>Differential Geometry and its Applications</i> 2.4 (Dec. 1992), pp. 351–367 (cit. on pp. 122, 233).
[Ben93]	S Benenti. "Orthogonal separable dynamical systems". In: <i>Differential geometry</i> and its applications (Opava, 1992 (1993), pp. 163–184 (cit. on pp. 3, 9, 56, 151, 227).
[Ben97]	S Benenti. "Intrinsic characterization of the variable separation in the HamiltonJacobi equation". In: <i>Journal of Mathematical Physics</i> 38.12 (1997), p. 6578 (cit. on pp. xv, xvi, 3, 10, 11, 70, 71, 78, 79, 83, 84, 116).
[Bla03]	M. Blaszak. "Separable bi-Hamiltonian systems with quadratic in momenta first integrals". In: (Dec. 2003), p. 53. arXiv: 0312025 [nlin] (cit. on pp. xvii, 220).
[Bla05]	M. Blaszak. "Separable systems with quadratic in momenta first integrals". In: <i>Journal of Physics A: Mathematical and General</i> 38.8 (Feb. 2005), pp. 1667–1685 (cit. on pp. xvii, 219, 220).
[BM03]	A. Bolsinov and V. S. Matveev. "Geometrical interpretation of Benenti systems". In: <i>Journal of Geometry and Physics</i> 44.4 (Jan. 2003), pp. 489–506 (cit. on pp. 97, 98).
[BM13]	A. Bolsinov and V. S. Matveev. "Local normal forms for geodesically equivalent pseudo-Riemannian metrics". In: (Jan. 2013), p. 29. arXiv: 1301.2492 (cit. on p. 117).

[Bog06]	O. I. Bogoyavlenskij. "Algebraic identities for the Nijenhuis tensors". In: Differential Geometry and its Applications 24.5 (Sept. 2006), pp. 447–457 (cit. on p. 239).
[Cal08]	F Calogero. "Calogero-Moser system". In: <i>Scholarpedia</i> 3.8 (Aug. 2008), p. 7216 (cit. on p. 221).
[Cal69]	F Calogero. "Solution of a Three-Body Problem in One Dimension". In: <i>Journal of Mathematical Physics</i> 10.12 (1969), pp. 2191–2196 (cit. on pp. 115, 221).
[Car68]	B. Carter. "Global Structure of the Kerr Family of Gravitational Fields". In: <i>Phys. Rev.</i> 174.5 (Oct. 1968), pp. 1559–1571 (cit. on pp. 3, 84).
[CFS06]	B Coll, J. J. Ferrando, and J. A. Saez. "On the geometry of Killing and conformal tensors". In: <i>Journal of Mathematical Physics</i> 47.6 (2006), p. 062503 (cit. on pp. xv, 25, 50, 51).
[Che11]	B. Y. Chen. <i>Pseudo-Riemannian Geometry, Delta-Invariants and Applications</i> . World Scientific, 2011 (cit. on pp. 29, 259, 260, 264, 277).
[Cra03]	M Crampin. "Conformal Killing tensors with vanishing torsion and the separation of variables in the HamiltonJacobi equation". In: <i>Differential Geometry and its Applications</i> 18.1 (Jan. 2003), pp. 87–102 (cit. on pp. xvi, xvii, 3, 12, 91–94, 98, 99, 106, 107, 123, 151, 154, 172, 180, 185).
[Cra05]	M Crampin. "On the orthogonal separation of variables in the Hamilton-Jacobi equation for geodesics in a Riemannian manifold". In: <i>Differential Geometry and its Applications</i> . Ed. by J Bures, J and Kowalski, O and Krupka, D and Slovak. Matfyzpress, 2005, pp. 453–466 (cit. on p. 106).
[CRA07]	C Chanu, G. Rastelli, and C. Alberto. "Eigenvalues of Killing Tensors and Separable Webs on Riemannian and Pseudo-Riemannian Manifolds". In: <i>Sym-</i> <i>metry, Integrability and Geometry : Methods and Applications</i> 3 (2007), p. 21 (cit. on p. 112).
[Cra07]	M Crampin. "Concircular vector fields and special conformal Killing tensors". In: Differential Geometric Methods in Mechanics and Field Theory Academia Press, Gent (2007), pp. 57–70 (cit. on pp. xvi, xvii, 3, 86, 91, 99–101, 155, 159, 160).
[Cra08]	M Crampin. "Structural equations for a special class of conformal killing tensors of arbitrary valence". In: <i>Reports on Mathematical Physics</i> 62.2 (Oct. 2008), pp. 241–254 (cit. on pp. xvi, 85, 86).
[Die08]	J Dieudonne. <i>Foundations of Modern Analysis</i> . Pure and Applied Mathematics. Academic Press, 2008 (cit. on p. 8).

- [DR07] L Degiovanni and G. Rastelli. "Complex variables for separation of the Hamilton-Jacobi equation on real pseudo-Riemannian manifolds". In: Journal of Mathematical Physics 48.7 (Oct. 2007), p. 073519. arXiv: 0610012 [nlin] (cit. on pp. xix, 68, 74, 75, 84, 233).
- [DU05] F. Dobarro and B. Unal. "Curvature of multiply warped products". In: *Journal of Geometry and Physics* 55.1 (Sept. 2005), pp. 75–106 (cit. on pp. 37, 259).
- [Eis28] L. P. Eisenhart. "Dynamical Trajectories and Geodesics". In: Annals of Mathematics. Second Series 30.1/4 (Jan. 1928), pp. 591–606 (cit. on p. 78).
- [Eis34] L. P. Eisenhart. "Separable systems of Stackel". In: Annals of Mathematics 35.2 (1934), pp. 749–752 (cit. on pp. xv, 2, 10, 11, 70, 71, 75, 83, 84, 106, 113, 119, 120, 123, 124, 130).
- [FIS03] S. H. Friedberg, A. J. Insel, and L. E. Spence. *Linear Algebra*. Pearson Education, 2003 (cit. on pp. 205, 246, 248).
- [FN56] A. Frölicher and A. Nijenhuis. "Theory of vector valued differential forms. Part I". In: *Indagationes Mathematicae* 18 (1956), pp. 338–359 (cit. on pp. xvii, 240–242).
- [GL05] I. Gohberg and P. Lancaster. Indefinite linear algebra and applications. Basel: Birkhauser, 2005, p. 357 (cit. on p. 245).
- [Gre75] W. H. Greub. *Linear Algebra*. 3rd ed. Springer, 1975, p. 476 (cit. on p. 145).
- [Gro11] L. D. Groote. "The role of special conformal Killing tensors in general relativity". In: Journal of Physics: Conference Series 314.1 (2011), p. 12023 (cit. on pp. 99, 233).
- [GVY08] V. S. Gerdjikov, G Vilasi, and A. B. Yanovski. "Vector-Valued Differential Forms". In: *Integrable Hamiltonian Hierarchies*. Ed. by V. S. Gerdjikov, G Vilasi, and A. B. Yanovski. Vol. 748. Lecture Notes in Physics. Springer Berlin Heidelberg, 2008, pp. 459–471 (cit. on pp. 239, 240, 242).
- [Haa55] J Haantjes. "On Xm-forming sets of eigenvectors". In: Indagationes Mathematicae 17 (1955), pp. 158–162 (cit. on pp. xvii, 241).
- [Hal+96] G. S. Hall et al. "On the algebraic structure of second order symmetric tensors in 5-dimensional space-times". In: *General Relativity and Gravitation* 28.9 (1996), pp. 1107–1113 (cit. on p. 145).
- [Hin98] F Hinterleitner. "Global properties of orthogonal separable coordinates for the Klein-Gordon equation in 2+ 1-dimensional flat space-time". In: Sitzungsber. Abt. II 5425 (1998), pp. 133–171 (cit. on p. 214).

- [HM08] J. T. Horwood and R. G. McLenaghan. "Orthogonal separation of variables for the Hamilton-Jacobi and wave equations in three-dimensional Minkowski space". In: Journal of Mathematical Physics 49.2 (2008), p. 23501 (cit. on pp. 11, 214).
- [HMS05] J. T. Horwood, R. G. McLenaghan, and R. G. Smirnov. "Invariant Classification of Orthogonally Separable Hamiltonian Systems in Euclidean Space". In: *Communications in Mathematical Physics* 259.3 (July 2005), pp. 679–709. arXiv: 0605023v1 [arXiv:math-ph] (cit. on pp. 11, 221).
- [HMS09] J. T. Horwood, R. G. McLenaghan, and R. G. Smirnov. "HamiltonJacobi theory in three-dimensional Minkowski space via Cartan geometry". In: *Journal* of Mathematical Physics 50.5 (2009), p. 053507 (cit. on pp. 11, 78).
- [Ika85] T. Ikawa. "On curves and submanifolds in an indefinite-Riemannian manifold".
 In: Tsukuba journal of mathematics 9.2 (1985). Ed. by T. Ikawa, pp. 353–371 (cit. on p. 32).
- [Jac12] N Jacobson. *Basic Algebra I.* Dover books on mathematics. Dover Publications, Incorporated, 2012 (cit. on p. 248).
- [Jel00] W. Jelonek. "Killing tensors and warped product". In: Annales Polonici Mathematici 75.1 (2000), pp. 15–33 (cit. on p. 60).
- [JF94] J.-B. Jun and S. Fueki. "On Some Circles In Pseudo-Riemannian Manifolds".
 In: Nihonkai Mathematical Journal 5.2 (1994), pp. 131–137 (cit. on p. 33).
- [Kal75] E. G. Kalnins. "On the Separation of Variables for the Laplace Equation \Delta \psi + K² \psi = 0 in Two- and Three-Dimensional Minkowski Space". In: SIAM Journal on Mathematical Analysis 6.2 (Apr. 1975), pp. 340–374 (cit. on p. 214).
- [Kal86] E. G. Kalnins. Separation of variables for Riemannian spaces of constant curvature. 1st ed. Longman Scientific & Technical, 1986 (cit. on pp. 2–4, 10, 12, 19, 37, 68, 75, 78, 104, 106, 116, 119, 121–124, 127, 209, 214).
- [Kas10] O. Kassabov. "A characterization of the extrinsic spheres in a Riemannian manifold". In: (Oct. 2010). arXiv: 1010.2863 (cit. on pp. xix, 32, 33).
- [Kaz98] J. Kazdan. Matrices A(t) depending on a Parameter t. 1998 (cit. on p. 8).
- [KM76] E. G. Kalnins and W. Miller. "Lie theory and separation of variables. 9.
 Orthogonal R-separable coordinate systems for the wave equation psi [sub tt] Delta [sub 2] psi =0". In: Journal of Mathematical Physics 17.3 (1976), pp. 331–355 (cit. on p. 214).

[KM78]	E. G. Kalnins and W. Miller. "Lie theory and the wave equation in space- time. 4. The Klein-Gordon equation and the Poincare group". In: <i>Journal of</i> <i>Mathematical Physics</i> 19.6 (1978), pp. 1233–1246 (cit. on p. 214).
[KM79]	E. G. Kalnins and W. Miller. "Separable components for three-dimensional complex riemannian spaces". EN. In: <i>Journal of Differential Geometry</i> 14.2 (1979), pp. 221–236 (cit. on p. 233).
[KM82]	E. G. Kalnins and W. Miller. Separation of variables on n dimensional rieman- nian manifolds. 2. The n-dimensional hyperboloid Hn, Volume 2. University of Waikato, 1982, p. 46 (cit. on pp. 3, 106, 214).
[KM86]	E. G. Kalnins and W. Miller. "Separation of variables on n-dimensional Riemannian manifolds. I. The n-sphere S[sub n] and Euclidean n-space R[sub n]". In: <i>Journal of Mathematical Physics</i> 27.7 (1986), pp. 1721–1736 (cit. on pp. 3, 106, 113, 214).
[KMJ80]	E. G. Kalnins, W. Miller, and Jr. "Killing tensors and variable separation for Hamilton-Jacobi and Helmholtz equations". In: <i>SIAM Journal on Mathematical Analysis</i> 11.6 (1980), pp. 1011–1026 (cit. on pp. xv, xvi, 57, 74).
[KMR84]	E. G. Kalnins, W. Miller, and G. J. Reid. "Separation of Variables for Complex Riemannian Spaces of Constant Curvature. I. Orthogonal Separable Coordinates for Snc and Enc". In: <i>Proceedings of the Royal Society of London. Series</i> A, Mathematical and Physical Sciences 394.1806 (July 1984), pp. 183–206 (cit. on pp. xvii, 181–183, 190, 214).
[Lax07]	P. D. Lax. <i>Linear Algebra and Its Applications</i> . Linear algebra and its applications v. 10. Wiley, 2007 (cit. on p. 8).
[LC04]	T Levi-Civita. "Sulla integrazione della equazione di Hamilton-Jacobi per separazione di variabili". Italian. In: <i>Mathematische Annalen</i> 59.3 (1904), pp. 383–397 (cit. on pp. xv, 2, 68).
[Lee12]	J. M. Lee. <i>Introduction to Smooth Manifolds</i> . 2nd ed. Springer, 2012, p. 723 (cit. on pp. 2, 4, 7, 43, 46, 66).
[Lee97]	J. M. Lee. <i>Riemannian Manifolds: An Introduction to Curvature</i> . Springer, 1997, p. 224 (cit. on pp. 25, 29).
[LL76]	L. D. Landau and E. M. Lifshitz. <i>Mechanics</i> . Butterworth-Heinemann, 1976, p. 170 (cit. on pp. 44, 62, 67).
[LN71]	 D. S. Leung and K. Nomizu. "The axiom of spheres in Riemannian geometry". In: Journal of Differential Geometry 5.3-4 (1971), pp. 487–489 (cit. on pp. xv, 34).

- [LR05] P. Lancaster and L. Rodman. "Canonical Forms for Hermitian Matrix Pairs under Strict Equivalence and Congruence". In: SIAM Review 47.3 (Jan. 2005), pp. 407–443 (cit. on p. 141).
- [Lun01] H. Lundmark. "A New Class of Integrable Newton Systems". In: Journal of Nonlinear Mathematical Physics 8.sup1 (Jan. 2001), pp. 195–199 (cit. on p. 219).
- [Lun03] H. Lundmark. "Higher-Dimensional Integrable Newton Systems with Quadratic Integrals of Motion". In: *Studies in Applied Mathematics* 110.3 (Apr. 2003), pp. 257–296 (cit. on pp. 98, 214, 219).
- [Mar97] C.-M. Marle. "The Schouten-Nijenhuis bracket and interior products". In: Journal of Geometry and Physics 23.34 (Nov. 1997), pp. 350–359 (cit. on pp. 45–47).
- [Mat05] V. S. Matveev. "The eigenvalues of the Sinyukov mapping for geodesically equivalent metrics are globally ordered". In: *Mathematical Notes* 77.3-4 (2005), pp. 380–390 (cit. on p. 98).
- [Mos11] J Moser. "Various Aspects of Integrable Hamiltonian Systems". In: Dynamical Systems. Ed. by C Marchioro. Vol. 78. C.I.M.E. Summer Schools. Springer Berlin Heidelberg, 2011, pp. 137–195 (cit. on p. 191).
- [MRS99] M Meumertzheim, H Reckziegel, and M Schaaf. "Decomposition of twisted and warped product nets". In: *Results in Mathematics* 36.3-4 (1999), 297–312 LA –English (cit. on pp. xv, 25, 26, 35–42, 50, 54).
- [Nak88] Y Nakanishi. "On helices and pseudo-Riemannian submanifolds". In: *Tsukuba journal of mathematics* 12.2 (1988), pp. 469–476 (cit. on p. 33).
- [Nij51] A. Nijenhuis. "Xn- 1-forming sets of eigenvectors". In: Indagationes Mathematicae 13.2 (1951), pp. 200–212 (cit. on pp. xvii, 239, 242).
- [Nij55] A. Nijenhuis. "Jacobi-type identities for bilinear differential concomitants of certain tensor fields I". In: *Indagationes Mathematicae* 17 (1955), pp. 390–397 (cit. on pp. 46, 47).
- [Nol96] S. Nolker. "Isometric immersions of warped products". In: Differential Geometry and its Applications 6.1 (Mar. 1996), pp. 1–30 (cit. on pp. xvii, xviii, 146–148, 160, 161, 259, 263, 266, 267, 274).
- [Nom73] K Nomizu. "Generalized central spheres and the notion of spheres in Riemannian geometry". In: *Tohoku Mathematical Journal* (1973) (cit. on p. 29).

- [NY74] K Nomizu and K Yano. "On circles and spheres in Riemannian geometry". English. In: *Mathematische Annalen* 210.2 (1974), pp. 163–170 (cit. on pp. 30, 32).
- [O'N83] B. O'Neil. Semi-Riemannian Geometry: With Applications to Relativity. 3rd ed.
 Pure and Applied Mathematics Series v. 103. Acad. Press, 1983, p. 468 (cit. on pp. xvi, xviii, 2, 4, 5, 8, 25, 29, 30, 32–34, 37, 40, 138, 140–142, 146, 174, 245, 248, 253, 259–261, 264, 274, 281, 282).
- [Par65] L. A. Pars. A treatise on analytical dynamics. Heinemann, 1965, p. 641 (cit. on p. 75).
- [Pet06] P. Petersen. *Riemannian Geometry*. Vol. 2006. Springer, 2006, p. 424 (cit. on p. 104).
- [Raj14a] K. Rajaratnam. "Canonical forms of Self-Adjoint Operators in Minkowski Space-time". In: (2014). arXiv: 1404.1867 (cit. on pp. 2, 245).
- [Raj14b] K. Rajaratnam. "Warped products and Spaces of Constant Curvature". In: (2014). arXiv: 1404.2547 (cit. on pp. 2, 259).
- [RM14a] K. Rajaratnam and R. G. McLenaghan. "Classification of Hamilton-Jacobi separation in orthogonal coordinates with diagonal curvature". In: (2014). arXiv: 1404.2565 (cit. on pp. 2, 119, 137).
- [RM14b] K. Rajaratnam and R. G. McLenaghan. "Killing tensors, warped products and the orthogonal separation of the Hamilton-Jacobi equation". In: *Journal* of Mathematical Physics 55.1 (Jan. 2014), p. 013505 (cit. on pp. 2, 3, 84, 91, 137).
- [RM14c] K. Rajaratnam and R. G. McLenaghan. "Concircular tensors in Spaces of Constant Curvature: With Applications to Orthogonal Separation of The Hamilton-Jacobi Equation". In: (Apr. 2014). arXiv: 1404.2847 (cit. on pp. 2, 3, 152).
- [Rov98] V. Y. Rovenskii. Foliations on Riemannian Manifolds and Submanifolds.
 Birkh{ä}user Boston, 1998, p. 286 (cit. on pp. 25, 27, 50).
- [RS99] H Reckziegel and M Schaaf. "De Rham decomposition of netted manifolds". In: Results in Mathematics 35.1-2 (1999), 175–191 LA –German (cit. on pp. xv, 36).
- [Sch53] J. Schouten. "On the differential operators of first order in tensor calculus". In: Stichting Mathematisch Centrum. Zuivere Wiskunde 4 (1953), pp. 15–18 (cit. on pp. xv, 45, 47).

[Sha00]	I. G. Shandra. "On the geodesic mobility of Riemannian spaces". In: <i>Mathematical Notes</i> 68.4 (Oct. 2000), pp. 528–532 (cit. on pp. xvi, 97, 99).
[Sta93]	P. Stackel. "Ueber die Bewegung eines Punktes in einern-fachen Mannig- faltigkeit". German. In: <i>Mathematische Annalen</i> 42.4 (1893), pp. 537–563 (cit. on pp. 10, 70, 75, 84).
[TCS05]	G Thompson, M Crampin, and W Sarlet. "Structural equations for a special class of conformal Killing tensors". In: (2005) (cit. on pp. xvii, 3, 86, 91, 155, 174).
[Toj04]	R. Tojeiro. "Conformal de Rham decomposition of Riemannian manifolds". In: (Apr. 2004), p. 17. arXiv: 0404434 [math] (cit. on p. 41).
[Toj07]	R. Tojeiro. "Conformal immersions of warped products". In: <i>Geometriae Dedicata</i> 128.1 (2007), 17–31 LA –English (cit. on p. 146).
[Ton88]	P Tondeur. Foliations on Riemannian manifolds. Universitext (1979). Springer- Verlag, 1988, p. 247 (cit. on pp. 25, 27).
[Win10]	S Winitzki. <i>Linear Algebra via Exterior Products</i> . Lulu Enterprises Incorporated, 2010, p. 285 (cit. on pp. 235–238).
[Woo75]	N. M. J. Woodhouse. "Killing tensors and the separation of the Hamilton-Jacobi equation". In: <i>Communications in Mathematical Physics</i> 44.1 (Feb. 1975), pp. 9–38 (cit. on pp. 43, 47, 62, 63, 65, 83).
[WW03]	C. Waksjo and S. Wojciechowski. "How to Find Separation Coordinates for the Hamilton-Jacobi Equation: A Criterion of Separability for Natural Hamiltonian Systems". In: <i>Mathematical Physics, Analysis and Geometry</i> 6.4 (Apr. 2003), pp. 301–348 (cit. on pp. 3, 12, 113, 116, 137, 214, 221, 227).
[WW05]	S. Wojciechowski and C. Waksjo. "What an Effective Criterion of Separa- bility says about the Calogero Type Systems". In: <i>Journal of Non-linear</i> <i>Mathematical Physics</i> 12.Supplement 1 (2005), p. 535 (cit. on pp. 221, 227).
[Yan40]	K Yano. "Concircular geometry I. Concircular transformations". In: <i>Proceedings</i> of the Japan Academy, Series A, Mathematical Sciences 16.6 (1940), pp. 195– 200 (cit. on p. 100).
[Zeg11]	 A. Zeghib. "Geometry of Warped Products". In: (2011), pp. 1–25. arXiv: 1107.0411 (cit. on pp. 27, 37, 42, 274).

Index

BEKM separation algorithm, 111 Benenti tensor, 85, 94 commute Poisson, 43 Schouten, 46 concircular transformation, 98 vector, 98concircular tensor, 83, 89 irreducible, 91 non-trivial, 90 orthogonal, 90 reducible, 91, 152 conformal Killing tensor, 48 gradient-type, 48orthogonal, 49 torsionless orthogonal, 85 trace-type, 48 diagonal curvature, 102 dilitational vector field, 98 distribution, 6Killing, 25 dKdV equation, 74 first integral, 43 fundamental problems, 1 geodesic Hamiltonian, 42 geodesically equivalent metrics, 95 geometrically equivalent, 152 Haantjes tensor, 238

Hamilton-Jacobi equation, 61 Hamiltonian, 42 natural, 42involution, 43separable, 68iso-canonical form, 164, 171 KEM coordinates, 102separation theorem, 119 Killing Bertrand-Darboux equation, 112tensor, 92Killing tensor, 46 characteristic, 70 Killing-Stackel space, 72 L-sequence, 105 Levi-Civita equations, 67 mean curvature normal, 25 metric-Jordan canonical form complex, 139real, 141 net, 34integrable, 35orthogonal, 35product, 36 pseudo-Riemannian product, 40 twisted product, 40warped product, 40Nijenhuis tensor, 236

orthogonal tensor, 5orthogonally integrable, 35 Poisson bracket, 43Robertson condition, 104 scalar product, 136Schouten bracket, 44 second fundamental form, 25 self-adjoint operator, 5 separable, 67orthogonal, 67 separable chain, 216 skew-normal sequence, 136 space of constant curvature, 7 submanifold geodesic, 28 Lagrangian, 65 spherical, 28umbilical, 28 symplectic form, 43 torsionless, 236 twisted product, 36vertically independent, 64 warped product, 36 decomposition, 258 web, 35KEM, 101 reducible separable, 76 separable, 67